Investigating the relationship between alcohol consumption and type 2 diabetes

A longitudinal analysis of the Whitehall II cohort, 1985-2013

Craig Knott

Thesis submitted for the degree of Doctor of Philosophy

University College London

Declaration

I, Craig Knott, confirm that the work presented in this thesis is my own. Where information has been derived from other sources, I confirm that this has been indicated in the thesis.

Abstract

Background: Although previous studies have reported a J-shaped association between the volume of alcohol consumption and incidence of type 2 diabetes (T2DM), a number of limitations weaken the validity of such findings. This thesis aims to systematically explore the effect of key methodological shortcomings.

Methods: Analyses were undertaken using Whitehall II data from 1985-2013. To examine the degree to which conventional survival analyses might be subject to misclassification error due to the use of a single baseline measure of drinking status, mixed effects models were used to plot the trajectory of alcohol intake according to baseline categories of consumption. Mixed effects models were also stratified by diagnosis status to shed light upon whether increases or decreases in risk are likely to accrue gradually over the life course or occur as a consequence of differences in intake specific to periods of heightened biological sensitivity. Finally, given changes in alcohol consumption across the life course, increasingly complex survival models were used to explore the relationship between different dimensions of the longitudinal trajectory and T2DM risk.

Results: Alcohol consumption within categories of baseline drinking converged over the adult life course toward moderate volumes, with moderate drinkers increasingly contaminated by participants defined at baseline as heavy or infrequent drinkers. Men who developed T2DM were found to increase their consumption up to their date of diagnosis, while drinking among women remained relatively stable up to diagnosis. Marked decreases in consumption were evident among both sexes following diagnosis. Reductions in the risk of T2DM were specific to or most pronounced among female current drinkers in middle age, with drinking in later life associated with an increased risk regardless of sex, after adjustment for prior consumption.

Conclusions: Variations in alcohol consumption across the adult life course highlight the importance of considering drinking histories when defining alcohol consumption categories, with the risk of misclassification error appearing to increase with age. Although reductions in the risk of T2DM were most pronounced among middle-aged women, evidence concerning the determinants of such a sex-specific disparity is lacking. That risks are heightened in older age suggest that any benefits from drinking earlier in the life course may be countered by age-related deteriorations to the alcohol metabolism.

Acknowledgements

My thanks go first and foremost to Annie Britton and Steven Bell, whose input as supervisors has been invaluable. I am hugely grateful for their time, encouragement and expertise during a challenging but rewarding three years.

I am indebted also to my better half for her endless patience and understanding. While other couples have long since started lives together, she has made do with little more than fleeting visits from an irritable and distracted partner. Nothing could ever repay her for such a sacrifice, though I suspect something sparkly might be appreciated.

Special thanks go to my father for his unwavering support and provision of a transport service of far greater reliability than our local privatised rail company. Despite his struggles, he has always found time to help me with all manner of small tasks, for which I am immensely grateful.

My thanks go also to my partner's family who, while not entirely certain what I've been doing for three years, have nonetheless obliged me with a plentiful and welcome supply of tea and cake.

I am grateful to Professor Sarah Wild who travelled from Edinburgh University to examine my MPhil to PhD transfer. Her advice was instrumental in helping to shape and focus my research for the remaining two years.

My final thank you goes the participants of the Whitehall II study and the staff at UCL and elsewhere who make it possible for researchers to utilise such a valuable wealth of health data. And Sam Bassett, who wanted an honourable mention.

Contents

1	Introduction	21
2	Background	25
	2.1 Diabetes mellitus	25
	2.1.1 Costs	25
	2.1.2 Prevalence and incidence	27
	2.1.3 Risk factors	29
	2.2 Alcohol consumption	37
	2.2.1 Trends	37
	2.2.2 Putative biological mechanisms	40
	2.3 Reference groups and confounding	44
3	An updated and revised meta-analysis	49
	3.1 Introduction	49
	3.2 Methods	49
	3.2.1 Selection criteria	49
	3.2.2 Search methods	50
	3.2.3 Data extraction and analysis	51
	3.2.4 Data synthesis	54
	3.3 Results	61
	3.3.1 Excluded studies	61
	3.3.2 Included studies	65
	3.3.3 All data	83
	3.3.4 Sex	84
	3.3.5 Reference group	
	3.3.6 Confounder adjustment	88
	3.3.7 Sub-group analyses	
	3.3.8 Sensitivity analyses	92
	3.3.9 Small-study effects	95
	3.3.10 Pattern of consumption	
	3.4 Strengths	98
	3.5 Limitations	99
	3.5.1 Heterogeneity	
	3.5.2 Quality assessment	
	3.5.3 Stability of consumption	
	3.6 Discussion	102

4	A summary of current evidence	113
5	Research aims	117
6	Data selection and structure	121
	6.1 Introduction	121
	6.2 Whitehall II cohort profile	121
	6.3 Defining alcohol consumption	123
	6.4 Defining T2DM	124
	6.5 Confounding factors	125
	6.5.1 Adiposity	125
	6.5.2 Diet	126
	6.5.3 Ethnicity	127
	6.5.4 Family history of T2DM	127
	6.5.5 Physical activity	127
	6.5.6 Smoking status	128
	6.5.7 Socio-economic status	128
	6.5.8 Variables not included	129
	6.6 Missing data	136
	6.6.1 Mortality	136
	6.6.2 Unit non-response	141
	6.6.3 Item non-response	144
	6.6.4 Multiple imputation	148
	6.7 Statistical power	156
	6.8 Summary of the dataset	157
7	A preliminary conventional survival analysis	164
	7.1 Introduction	164
	7.2 Objective	164
	7.3 Hypotheses	165
	7.4 Methods	165
	7.4.1 Sample	165
	7.4.2 Variables	165
	7.4.3 Statistical analyses	167
	7.5 Results	171
	7.5.1 Descriptive statistics	171
	7.5.2 Age-adjusted models	180
	7.5.3 Multivariable-adjusted models	180
	7.5.4 Frequency interaction	184

	7.5.5 Proportionality assumption	185
	7.5.6 Drink type	186
	7.6 Summary of results	186
	7.7 Limitations	187
	7.8 Discussion	189
8	Trajectories of alcohol consumption	194
	8.1 Introduction	194
	8.2 Objectives	194
	8.3 Hypotheses	195
	8.4 Methods	198
	8.4.1 Sample	198
	8.4.2 Variables	198
	8.4.3 Statistical analysis	199
	8.5 Results	203
	8.6 Summary of results	225
	8.7 Limitations	225
	8.8 Discussion	227
9	Longitudinal alcohol consumption and the risk of T2DM	232
	9.1 Introduction	232
	9.2 Objectives	232
	9.3 Hypotheses	233
	9.3.1 The primary dose-response relationship	233
	9.3.2 Sick quitter effects	233
	9.4 Methods	234
	9.4.1 Sample	234
	9.4.2 Variables	235
	9.4.3 Statistical analysis	237
	9.5 Results	243
	9.5.1 Conventional survival model	243
	9.5.2 Age-varying covariates survival model	245
	9.5.3 Two-stage survival model	247
	9.5.4 Shared random effects model	249
	9.6 Summary of findings	252
	9.7 Limitations	253
	9.8 Discussion	256

10 Discussion	262
10.1 Research questions	262
10.2 Summary of findings	262
10.2.1 Misclassification error	262
10.2.2 Differences in alcohol consumption according to the diagnosis of T2DM $_$	264
10.2.3 Importance of different dimensions of the longitudinal trajectory	265
10.2.4 Sick quitter effects	269
10.3 Limitations	270
10.4 Strengths	272
10.5 Further research	273
10.6 Policy implications	274
11 Appendices	278
11.1 Appendices for Chapter 3	278
11.2 Appendices for Chapter 6	289
11.3 Appendices for Chapter 7	297
11.4 Appendices for Chapter 8	314
11.5 Appendices for Chapter 9	349
12 Reference list	357

Figures

Figure 2.1 Age-specific prevalence of male self-reported doctor-diagnosed DM. Health Survey
for England 1994-2014 27
Figure 2.2 Age-specific prevalence of female self-reported doctor-diagnosed DM. Health Survey
for England 1994-2014 28
Figure 2.3 Crude annual incidence of T2DM per 1,000 person-years at risk. THIN 2000-2013. 29
Figure 2.4 Mean volume of alcohol consumption among male current drinkers, stratified by age
group and country. Health Survey for England and Scottish Health Survey, 2014 38
Figure 2.5 Mean volume of alcohol consumption among female current drinkers, stratified by
age group and country. Health Survey for England and Scottish Health Survey, 2014 38
Figure 2.6 Proportion of male current drinkers who reported episodic heavy consumption.
Opinions and Lifestyle Survey, 2005-2013 39
Figure 2.7 Proportion of female current drinkers who reported episodic heavy consumption.
Opinions and Lifestyle Survey, 2005-2013 39
Figure 3.1 Study flow diagram64
Figure 3.2 Dose-response relationship between the average volume of alcohol consumption and
T2DM risk, utilising all data combined83
Figure 3.3 Scatter diagram of extracted risk estimates, stratified by sex84
Figure 3.4 Dose-response relationship between average volume of alcohol consumption and
T2DM risk, stratified by sex85
Figure 3.5 Sex-adjusted dose-response relationship between average volume of alcohol
consumption and T2DM risk, stratified by reference group86
Figure 3.6 Scatter diagram of extracted risk estimates calculated relative to never drinkers,
stratified by sex and inverse weighted by SE87
Figure 3.7 Dose-response relationship between average volume of alcohol consumption and
T2DM risk, relative to never drinkers, stratified by sex87
Figure 3.8 Dose-response relationship between average volume of alcohol consumption and
T2DM risk, adjusted for sex and reference group, stratified by confounder adjustment88

Figure 3.9 Dose-response relationship between average volume of alcohol consur	mption and
T2DM risk, adjusted for sex and reference group, stratified by method of case ascert	ainment 89
Figure 3.10 Dose-response relationship between average volume of alcohol consul	mption and
T2DM risk, adjusted for sex and reference group, stratified by population region	90
Figure 3.11 Dose-response relationship between average volume of alcohol consul	mption and
T2DM risk, adjusted for sex and reference group, stratified by population type	91
Figure 3.12 Dose-response relationship between average volume of alcohol consul	mption and
T2DM risk, adjusted for sex and reference group, stratified by quality assessment sco	ore92
Figure 3.13 Dose-response relationship between average volume of alcohol consul	mption and
T2DM risk, adjusted for sex and reference group, stratified according to select	tion in the
preceding meta-analysis	93
Figure 3.14 Dose-response relationship between average daily alcohol consumption	and T2DM
risk: sex-specific data, stratified by Jee <i>et al</i> ²³⁵	94
Figure 3.15 Funnel plot of current drinking versus non-drinking, stratified by sex	95
Figure 3.16 Funnel plot of current drinking versus non-drinking, stratified by sex an	d excluding
Jee <i>et al</i> . ²³⁵	96
Figure 3.17 Common latent trajectories of volume of alcohol consumption: the 'c	at's cradle'
	100
Figure 3.18 Predicted mean alcohol consumption trajectories (in units/week) acr	oss the life
course in UK cohort studies	101
Figure 6.1 Directed acyclic graph illustrating direct overadjustment bias	129
Figure 6.2 Directed acyclic graph illustrating indirect overadiustment bias	130
Figure C.2 Decessor de yone graph material and har each test of differences in our incention	
Figure 6.3 Power versus sample size for a log-rank test of differences in survivor fun	ctions_157
Figure 8.1 Hypothesised differences in average alcohol consumption trajectory a	ccording to
T2DM diagnosis	197
Figure 8.2 An illustration of random intercept and random slopes models	201
Figure 8.3 Crude sex-specific linear trajectories of mean weekly volume of alcohol co	onsumption
between the ages of 34-84 years. Observed data	206
Figure 8.4 Crude sex-specific non-linear trajectories of mean weekly volume	of alcohol
consumption between the ages of 34-84 years. Observed data.	207

 Figure 8.5 Crude sex-specific linear trajectories of mean weekly volume of alcohol consumption

 between the ages of 34-84 years, stratified by baseline alcohol consumption category. Observed

 data.
 209

Figure 8.6 Crude sex-specific linear trajectory of mean weekly volume of alcohol consur	nption,
stratified by T2DM diagnosis. Observed data	216
Figure 8.7 Crude sex-specific linear trajectory of mean weekly volume of alcohol consur	nption,
stratified by T2DM diagnosis. Observed data	218
Figure 8.8 Crude trajectories of mean weekly volume of alcohol consumption up to and	beyond
the date of T2DM diagnosis, stratified by sex	220
Figure 8.9 Crude male trajectories of mean weekly volume of alcohol consumption, strat	ified by
baseline alcohol consumption category and T2DM diagnosis. Observed data.	223
Figure 8.10 Crude female trajectories of mean weekly volume of alcohol consumption, st	ratified
by baseline alcohol consumption category and T2DM diagnosis. Observed data.	224
Figure 9.1 Illustration of the longitudinal trajectory and hazard function within age-	varying
covariate and two-stage survival models	239
Figure 9.2 Illustration of the longitudinal trajectory and hazard function within a shared r	andom
effects survival model	241
Figure 9.3 Crude, sex-specific and best-fitting trajectories of the mean weekly volume of	alcohol
consumption (log ₂) between the ages of 40-84 years. Observed data	247

Tables

Table 3.1 Characteristics of selected studies		
Table 3.2 Measures of alcohol consumption, confounder adjustment and effect estimate		
reported by selected studies	73	
Table 6.1 Variable availability in Whitehall II by study wave	122	
Table 6.2 Descriptive summary of Whitehall II data, stratified by wave and sex	131	
Table 6.3 Descriptive summary of Whitehall II data at baseline, stratified by sev	and survival	
	138	
Table 6.4 Proportion of surviving baseline participants with unit non-response	141	
Table 6.5 Descriptive summary of Whitehall II data at baseline, stratified by sex a	and degree of	
unit non-response	142	
Table 6.6 Degree of item non-response by wave, stratified by sex	145	
Table 6.7 Descriptive summary of Whitehall II data at baseline (wave one), stratifi	ed by sex and	
item non-response	146	
Table 6.8 Descriptive summary of imputed Whitehall II data, stratified by wave an	d sex159	
Table 7.1 Baseline characteristics of participants free of T2DM at wave three a	nd with valid	
follow-up data, stratified by sex. Observed data	172	
Table 7.2 Baseline characteristics of participants free of T2DM at wave three a	nd with valid	
follow-up data, stratified by sex and categories of average weekly volum	e of alcohol	
consumption. Observed data	175	
Table 7.3 Age and multivariable-adjusted dose-response relationship between	categories of	
average weekly volume of alcohol consumption and T2DM, stratified by sex. Ol	bserved data.	
	182	
Table 7.4 Iteratively-adjusted dose-response relationship between categories of av	verage weekly	
volume of alcohol consumption and T2DM, stratified by sex. Observed data.	183	
Table 7.5 Multivariable-adjusted interaction between a continuous measure of av	verage weekly	
volume of alcohol consumption and drinking frequency, stratified by sex and ex	xcluding non-	
current and infrequent drinkers. Observed data.	185	

Table 7.6 Multivariable-adjusted dose-response relationship between categories of average

 weekly volume of alcohol consumption, drink type and T2DM, stratified by sex. Observed data.

186

Table 8.1 Crude sex-specific linear trajectory of mean weekly volume of alcohol consur	nption
between the ages of 34-84 years: goodness of fit statistics. Observed data	204
Table 8.2 Crude sex-specific linear and non-linear trajectories of mean weekly volume of a	alcohol
consumption between the ages of 34-84 years. Observed data	205
Table 8.3 Crude sex-specific interaction between the trajectory of mean weekly volu	ime of
alcohol consumption and baseline alcohol consumption category. Observed data.	210
Table 8.4 Baseline characteristics of T2DM-free participants, stratified by T2DM diag	gnosis.
Observed data	212
Table 8.5 Sex-specific interaction between the linear trajectory of mean weekly volu	ime of
alcohol consumption and T2DM diagnosis: goodness of fit statistics. Observed data	214
Table 8.6 Crude sex-specific interaction between the linear trajectory of mean weekly v	volume
of alcohol consumption and T2DM diagnosis. Observed data.	215
Table 8.7 Crude, sex-specific and best-fitting trajectory of mean weekly volume of a	alcohol
consumption, stratified by T2DM diagnosis. Observed data.	217
Table 8.8 Crude trajectories of mean weekly volume of alcohol consumption up to and b	beyond
the date of diagnosis, stratified by sex	219
Table 8.9 Crude sex-specific interaction between the trajectory of mean weekly volu	ime of
alcohol consumption and baseline category, stratified by T2DM diagnosis. Observed data	n 222
Table 9.1 Relationship between the average weekly volume of alcohol consumption and t	he risk
of T2DM, stratified by sex. Shared random effects model, goodness of fit statistics, ob	served
data	243

Table 9.2 Relationship between the average weekly volume of alcohol consumption and the riskof T2DM, stratified by sex. Conventional survival analysis, observed data.244

Table 9.3 Relationship between the average weekly volume of alcohol consumption and the risk

 of T2DM, stratified by sex. Age-varying covariate survival analysis, observed data.
 246

Table 9.4 Relationship between the average weekly volume of alcohol consumption and the riskof T2DM, stratified by sex. Two-stage survival analysis, observed data.248

Table 9.5Multivariable-adjusted relationship between intercept and current valueparameterisations of average weekly volume of alcohol consumption and the risk of T2DM,stratified by sex. Shared random effects survival analysis, observed data.249

Table 9.6 Multivariable-adjusted associations between conditional intercept and current valueparameterisations of average weekly volume of alcohol consumption and the risk of T2DM,stratified by sex. Shared random effects survival analysis, observed data.250

Table 9.7Multivariable-adjusted relationship between intercept and current valueparameterisations of average weekly volume of alcohol consumption and the risk of T2DM,stratified by sex and without adjustment for alcohol consumption category. Shared randomeffects survival analysis, observed data.251

 Table 9.8 Multivariable-adjusted relationship between the rate of change in the average weekly

 volume of alcohol consumption and the risk of T2DM, stratified by sex. Shared random effects

 survival analysis, observed data.
 252

Appendices

Appendix 3.1 Published results from the revised and updated dose-response meta-analysis 278
Appendix 3.2 The Newcastle-Ottawa quality assessment checklist 287
Appendix 6.1 Effect of dietary factors upon the alcohol-T2DM relationship 289
Appendix 6.2 Trace plots illustrating convergence on key variables 291
Appendix 7.1 Crude Nelson-Aalen cumulative hazard estimate, stratified by sex. Observed data. 297
Appendix 7.2 Baseline characteristics of participants free of T2DM at wave three and with valid
follow-up data, stratified by sex. Imputed data 298
Appendix 7.3 Crude Nelson-Aalen cumulative hazard estimate, stratified by sex and category of average weekly volume of alcohol consumption. Observed data 301
Appendix 7.4 Crude Nelson-Aalen cumulative hazard estimate, stratified by sex and frequency of alcohol consumption. Observed data. 302
follow-up data, stratified by sex and categories of average weekly volume of alcohol consumption. Imputed data303
Appendix 7.6 Multivariable-adjusted dose response relationship between categories of average
weekly volume of alcohol consumption and T2DM, stratified by sex. Imputed data 308
Appendix 7.8 Multivariable-adjusted interaction between a continuous measure of average weekly volume of alcohol consumption and drinking frequency, stratified by sex and including non-current and infrequent drinkers. Observed data310
Appendix 7.9 Multivariable-adjusted interaction between a continuous measure of average
weekly volume of alcohol consumption and drinking frequency, stratified by sex and including non-current and infrequent drinkers. Imputed data 311
Appendix 7.10 Multivariable-adjusted dose response relationship between categories of average weekly volume of alcohol consumption and T2DM, stratified by sex. Hazards were permitted to vary as a function of linear time. Observed data 312
Appendix 7.11 Multivariable-adjusted dose-response relationship between categories of average weekly volume of alcohol consumption, drink type and T2DM, stratified by sex. Imputed data313

Appendix 8.1 Crude sex-specific linear trajectory of mean weekly volume of alcohol consumption between the ages of 34-84 years: goodness of fit statistics. Imputed data. ____314

Appendix 8.2 Crude sex-specific linear and non-linear trajectories of mean weekly volume of alcohol consumption between the ages of 34-84 years: results. Imputed data.______315

Appendix 8.3 Crude sex specific trajectories of mean weekly volume of alcohol consumption between the ages of 34-84 years: figure. Imputed data._____316

Appendix 8.4 Crude sex-specific linear and non-linear trajectories of mean weekly volume of alcohol consumption between the ages of 34-84 years: fit statistics. Observed data. _____317

Appendix 8.5 Crude sex-specific linear and non-linear trajectories of mean weekly volume of alcohol consumption between the ages of 34-84 years: fit statistics. Imputed data._____319

Appendix 8.6 Crude sex-specific non-linear trajectories of mean weekly volume of alcohol consumption between the ages of 34-84 years: figure. Imputed data.______323

Appendix 8.8 Crude sex-specific linear trajectories of mean weekly volume of alcohol consumption between the ages of 34-84 years, stratified by baseline alcohol consumption category: figure. Imputed data. _______325

Appendix 8.9 Baseline characteristics of T2DM-free participants, stratified by T2DM diagnosis. Imputed data. 326

Appendix 8.10 Sex-specific interaction between the linear trajectory of mean weekly volume of alcohol consumption and T2DM diagnosis: goodness of fit statistics. Imputed data 329

Appendix 8.11 Crude sex-specific interaction between the linear trajectory of mean weekly volume of alcohol consumption and T2DM diagnosis: results. Imputed data. ______330

 Appendix 8.12
 Crude sex-specific linear trajectory of mean weekly volume of alcohol

 consumption, stratified by T2DM diagnosis: figure. Imputed data.
 331

Appendix 8.13 Crude sex-specific linear and non-linear trajectories of mean weekly alcohol consumption, stratified by T2DM diagnosis: goodness of fit statistics. Observed data. _____332

Appendix 8.14 Crude sex-specific linear and non-linear trajectories of mean weekly alcohol consumption, stratified by T2DM diagnosis: goodness of fit statistics. Imputed data._____334

 Appendix 8.15 Crude, sex-specific and best-fitting trajectories of mean weekly volume of alcohol

 consumption, stratified by T2DM diagnosis: results. Imputed data.

 337

Appendix 8.16 Crude sex-specific best-fitting trajectories of mean weekly volume of alcohol consumption, stratified by T2DM diagnosis: figure. Imputed data. ______ 338

Appendix 8.17 Crude linear and non-linear trajectories of alcohol consumption up to and beyond the date of diagnosis, stratified by sex: goodness of fit statistics. Observed data. ____ 339

Appendix 8.18 Crude linear and non-linear trajectories of alcohol consumption up to and beyond the date of diagnosis, stratified by sex: goodness of fit statistics. Imputed data. _____ 340

Appendix 8.19 Crude trajectories of mean weekly volume of alcohol consumption up to and beyond the date of diagnosis, stratified by sex. Imputed data. ______ 343

Appendix 8.20 Crude sex-specific interaction between the trajectory of mean weekly volume of alcohol consumption and baseline alcohol consumption category, stratified by T2DM diagnosis: results. Imputed data. ______ 344

Appendix 8.21 Crude male trajectories of mean weekly volume of alcohol consumption, stratified by baseline alcohol consumption category and T2DM diagnosis: figure. Imputed data. 345

Appendix 8.22 Crude female trajectories of mean weekly volume of alcohol, stratified by baseline alcohol consumption category and T2DM diagnosis: figure. Imputed data. ______ 346

Appendix 8.23 Crude sex-specific best-fitting trajectory of mean weekly volume of alcohol consumption, stratified by T2DM diagnosis and excluding non-drinkers. Observed data.____ 347

Appendix 8.24 Crude sex-specific and best-fitting trajectory of mean weekly volume of alcohol consumption, stratified by T2DM diagnosis: a competing risk sensitivity analysis. Observed data. 348

Appendix 9.1 Alternative distributional functions of the cumulative baseline hazard______349Appendix 9.2 Relationship between the average weekly volume of alcohol consumption and therisk of T2DM, stratified by sex. Conventional survival analysis, imputed data. ______350

Appendix 9.3 Relationship between the average weekly volume of alcohol consumption and the risk of T2DM, stratified by sex. Age-varying covariate survival analysis, imputed data._____ 351

Appendix 9.4 Relationship between the average weekly volume of alcohol consumption and the risk of T2DM, stratified by sex. Two-stage survival analysis, imputed data. ______ 352

Appendix 9.5 Multivariable-adjusted relationship between intercept and current value parameterisations of average weekly volume of alcohol consumption and the risk of T2DM,

stratified by sex. Shared random effects survival analysis, time-to-event timescale, observed data. ______353

Appendix 9.6 Multivariable-adjusted associations between conditional intercept and current value parameterisations of average weekly volume of alcohol consumption and the risk of T2DM, stratified by sex. Shared random effects survival analysis, time-to-event timescale, observed data. ______354

Appendix 9.7 Multivariable-adjusted relationship between the rate of change in the average weekly volume of alcohol consumption and the risk of T2DM, stratified by sex and adjusted for intake at the intercept. Shared random effects survival analysis, observed data. _____355

Glossary of units and abbreviations

ABV	Alcohol by volume
BIC	Bayesian information criterion
BMI	Body mass index
CHD	Coronary heart disease
CI	Confidence interval
CRP	C-reactive protein
CVD	Cardiovascular disease
DM	Diabetes mellitus
FFQ	Food frequency questionnaire
FP	Fractional polynomial
FPG	Fasting plasma glucose
g	Grams
GL	Glycaemic load
GWAS	Genome-wide association study
HbA1c	Glycated haemoglobin A1c
HDL	High density lipoprotein
HOMA	Homeostasis model assessment
HR	Hazard ratio
MAR	Missing at random
MCAR	Missing completely at random
MICE	Multiple imputation chained equations
MID	Multiple imputation, then deletion
ml	Millilitre
mmol/L	Millimoles per litre
NHS	National Health Service
nmol/L	Nanomoles per litre
OGTT	Oral glucose tolerance test
OR	Odds ratio
RR	Relative risk
SHBG	Sex hormone-binding globulin
SNP	Single nucleotide polymorphism
SMD	Standardised mean difference
T1DM	Type 1 diabetes mellitus
T2DM	Type 2 diabetes mellitus
THIN	The Health Improvement Network
ΤΝΓ-α	Tumour necrosis factor alpha
UK	United Kingdom
US	United States
WHO	World Health Organisation
WW2	World War Two

Chapter 1 Introduction

1 Introduction

Diabetes mellitus (DM) is a condition implicated in the development of vascular dysfunction and thereby a heightened risk of numerous vascular conditions, including coronary heart disease (CHD) and ischaemic stoke,^{1,2} as well as blindness, kidney failure and limb loss.³ Beyond its human costs, the financial burden of the condition is sizeable, at up to £13.8bn of National Health Service (NHS) expenditure in 2010/11 and an estimated annual loss to the United Kingdom (UK) economy of £15.4bn through absenteeism and early mortality.⁴

Such factors are made all the more alarming by trend data indicating a rise in the prevalence of DM over recent decades, particularly within older age groups.⁵ By 2014, DM was estimated to affect 6.4% of adults living in the United Kingdom, or around 2,913,538 people – an increase of 67% relative to 1994.⁶ Thus, although the rate of newly diagnosed cases appears to have plateaued in recent years,⁷ DM continues to pose considerable human and economic costs that represent a substantial public health challenge.

Fortunately, around 85% of known cases are of a type caused largely by exposure to deleterious yet modifiable lifestyle factors,⁸ indicating that many future cases may be preventable. With this in mind, Public Health England published in 2015 a systematic review into the effectiveness of preventative lifestyle intervention programmes, of which the majority appeared to target improvements to physical activity and diet, including the regulation of fat and fibre consumption.⁹

In contrast to these programmes, a growing body of evidence has suggested that alcohol consumption may play a role in reducing the risk of type 2 diabetes mellitus (T2DM), with one meta-analysis from 2009 reporting that men and women who consumed ~22-24 g/day (154-168 g/week) of ethanol may be subject to a one-third reduction in the risk of T2DM, relative to never drinkers,¹⁰ or a volume of intake equivalent to around 1.2 pints of 4% alcohol by volume (ABV) lager per day.¹¹

With alcohol consumption appearing to represent a modifiable risk factor to which the majority of UK adults are exposed, Chapter 2 provides a more detailed overview of the evidence concerning alcohol consumption and T2DM, as well as a number of biological mechanisms by which alcohol consumption has been hypothesised to modify a person's risk of T2DM. The chapter also includes a summary of recent trends in alcohol consumption behaviour, and outlines a number of methodological limitations that may have led to an overestimation of reductions in T2DM risk at moderate volumes of alcohol intake. Acknowledging these

Chapter 1: Introduction

shortcomings, Chapter 3 reports results from an updated and revised meta-analysis that set out to explore the effect of such factors upon the observed dose-response relationship, with Chapter 4 providing a summary of the current literature, including gaps in the evidence base, such as a failure within conventional literature to consider the effect of longitudinal changes to drinking behaviour. Chapter 5 sets out a number of aims by which the longitudinal relationship between alcohol consumption and T2DM risk can be better understood, with Chapter 6 describing the selection and structure of a dataset for such an exploration and Chapter 7 establishing the suitability of the chosen cohort for quantifying the association between alcohol consumption and T2DM. Chapter 8 reports changes in alcohol consumption over the adult life course and investigates whether these trajectories differ according to T2DM diagnosis, while Chapter 9 utilises increasingly complex survival analyses to formally explore the relationship between different dimensions of the longitudinal trajectory and T2DM risk. Chapter 10 summarises findings from the analyses undertaken and documents a series of policy implications and areas for future research. Chapter 1: Introduction

Chapter 2 Background

2 Background

2.1 Diabetes mellitus

DM is a chronic metabolic disorder characterised by the body's inability to effectively regulate the metabolism of carbohydrates and fats, leading to chronically elevated levels in the bloodstream. Such a homeostatic impairment is currently understood to take two primary forms:

- Type I diabetes (T1DM): an auto-immune disease in which an abnormal immune response leads to the destruction of insulin-producing pancreatic β-cells. Given the joint role of insulin in promoting the absorption of glucose from the blood into muscle and fat tissues, and the inhibition of glucose release from the liver, absolute insulin deficiency as associated with T1DM results in a build-up of carbohydrates and fats in the bloodstream.¹² T1DM has been identified as most common among younger individuals and estimated to represent around 5-10% of known cases worldwide, or around 15% of known cases in England.⁸
- Type 2 diabetes (T2DM): T2DM is characterised by relative insulin deficiency (reduced production) and/or the presence of insulin resistance (impaired response), which develop progressively over time and commonly following prolonged exposure to detrimental lifestyle factors, such as obesity.⁸ Accounting for around 85-90% of known cases, T2DM represents both the most common and preventable component of DM.

Regardless of type, these metabolic abnormalities have been linked to an increased risk of numerous vascular complications and corollary costs to health services and the economy.

2.1.1 Costs

2.1.1.1 Morbidity and mortality

Although the precise nature of the relationship between DM and vascular pathogenesis appears complex and multifactorial, with biological mechanisms currently posited rather than conclusively established,^{13,14,15} DM has been associated with the onset of numerous deleterious vascular conditions.³

Chief among these, macrovascular conditions appear to represent the leading cause of morbidity and mortality among diabetic patients worldwide, with cardiovascular disease (CVD) accounting for around half of all documented deaths.¹⁶ This elevated cardiovascular risk profile

25

Chapter 2: Background

among people with DM was first identified in 1979, when data from the Framingham study indicated that participants with the condition were twice as likely to develop CVD as those without, even after adjustment for age, sex and common cardiovascular risk factors such as smoking.¹⁷ Subsequent studies have consistently identified an elevated risk of macrovascular health complications among people with DM.¹ The latest macrovascular risk estimates are provided by an analysis of data obtained via the CALIBER programme, which links data from four English electronic health databases.² Based on a cohort of over 1.9 million participants, T2DM was positively associated with more than a 50% increase in the risk of heart failure (HR 1.56, 95% CI 1.45-1.69) and myocardial infarction (HR 1.54, 95% CI 1.42-1.67), as well as a 72% increase in the risk of ischaemic stroke (HR 1.72, 95% CI 1.52-1.95). Elsewhere, studies indicate that recovery from such conditions may also be impaired among persons with T2DM, with such individuals having exhibiting a heightened risk of stroke recurrence and post-stroke mortality,^{18,19,20,21} physical disability²² and dementia.^{23,24}

Aside from macrovascular conditions and their negative consequences for health, diabetesinduced damage to the microvasculature represents a further serious concern, increasing the risk of conditions ranging from kidney failure and incontinence to limb loss and blindness.³

2.1.1.2 Primary care expenditure

Through its direct impact upon population health, DM necessitates investment in a wide range of health services. Applying estimates of disease prevalence and population size to inpatient and outpatient data available from Hospital Episode Statistics and other sources, two recent studies have attempted to calculate expenditures for healthcare resources including diagnostic tests, primary care consultations, prescription drugs and treatment for diabetes-attributable vascular conditions. Depending on the methods and data utilised, these direct costs have been estimated at between £9.8bn²⁵ and £13.8bn⁴ in 2010/11, or around one-tenth of total NHS expenditure over the period.²⁶ These figures represent the latest published estimates of DM-related primary care expenditure.

2.1.1.3 Indirect costs

Costs associated with DM are not limited to health services. As a chronic and sometimes debilitating condition, DM can also introduce indirect costs to the economy through premature mortality, early retirement, absenteeism and an increased dependence on social welfare and care services.

Few sources of UK-specific data are available for determining indirect costs attributable to diabetes.^{4,25} One approximation was undertaken by The Economist Intelligence Unit in 2007 using data from the United States (US),²⁷ which estimated foregone earnings attributable to absenteeism and mortality at £1.7bn/annum in the UK. Elsewhere, attempting to include indirect costs due to presenteeism, informal care and welfare, estimates of between £13.9bn²⁵ and £15.4bn⁴ were estimated for 2010/11.

Whatever the precise figures, direct and indirect costs look to be sizeable – an observation notable given stagnated or reduced NHS, public health and social care budgets,^{28,29,30} plus ongoing increases in the number of cases requiring treatment.

2.1.2 Prevalence and incidence

According to data published at the end of 2015,⁵ the prevalence of self-reported DM cases among non-institutionalised adults in the England rose from 2.4% to 6.2% between 1994 and 2014 (Figures 2.1 and 2.2), reaching 2,707,850 cases by the end of the period.³¹ Taking a more inclusive measurement, capturing both institutionalised and non-institutionalised residents from across the UK as a whole, objective data collected as part of the Quality and Outcomes Framework (QOF) indicated a prevalence of 6.4%, or 2,913,538 cases in 2014/15.⁶ Temporal increases in the prevalence of T2DM appear most pronounced among older age groups, with prevalence consistently greatest among men.



Figure 2.1 Age-specific prevalence of male self-reported doctor-diagnosed DM. Health Survey for England 1994-2014.



Figure 2.2 Age-specific prevalence of female self-reported doctor-diagnosed DM. Health Survey for England 1994-2014.

However, a marked rise in the prevalence of T2DM does not necessarily translate to a DM epidemic; an increase in prevalence may occur instead as a result of successful public health initiatives, such as improved disease management and consequent decreases in premature mortality.³² Looking instead to the number of *new* cases diagnosed each year, trend data from The Health Improvement Network (THIN)^{33,34} indicate that, while the incidence of T1DM has been low and constant over time, the age and sex standardised rate of T2DM rose by 63% between 1996 and 2005, to 4.31 cases per 1,000 person-years at risk.³⁵ Although alarming, more recent figures indicate that these earlier increases may now have stabilised, falling to 3.99 and 3.73 cases per 1,000 person-years at risk among men and women in 2013 (Figure 2.3).⁷

These figures indicate that increases in the prevalence of DM are most likely a consequence of improvements to the treatment and management of the condition, and resultant reductions in premature mortality. However, although an average 15,665 new T2DM cases were diagnosed annually between 2000 and 2013,⁷ the human and economic costs of DM are not inexorable.



Figure 2.3 Crude annual incidence of T2DM per 1,000 person-years at risk. THIN 2000-2013.

2.1.3 Risk factors

A number of risk factors have been identified for the development of T2DM, ranging from adiposity and smoking, to alcohol consumption and physical activity.^{36,37} While not all factors are modifiable, current epidemiologic evidence indicates that the risk of T2DM may be attenuated in large part through simple lifestyle changes. Several risk factors are discussed below, beginning with adiposity and ending with alcohol consumption.

2.1.3.1 Adiposity

Some of the rise in the incidence of T2DM observed between 1996 and 2005 was posited by researchers as attributable to rising obesity over the period, with the proportion of newly diagnosed patients presenting as obese having risen by 22% over the period, reaching 56% by 2005.³⁵ This was better substantiated by results from the British Regional Heart Study, which indicated that 25.9% (95% CI 16.5-38.3%) of the rise in T2DM incidence among men between 1984-1992 and 1999-2007 was attributable to an age-adjusted increase in average body mass index (BMI) of 1.42kg/m² (95% CI 1.10-1.74kg/m²).³⁸

An aetiological association between obesity and T2DM has been supported by multiple longitudinal studies, including a meta-analysis of 15 cohorts, which identified a standard deviation increase in BMI raised the relative risk (RR) of T2DM of by 55% (RR 1.55, 95% CI 1.43-1.69), with similar increases identified across alternative measures of adiposity, including waist circumference (RR 1.63, 95% CI 1.49-1.79) and waist-hip ratio (RR 1.52, 95% CI 1.40-1.66).³⁹ Although waist circumference was found to be a stronger predictor of T2DM risk than BMI and waist-hip ratio in primarily white study populations, this relationship differed according to a

diverse range of participant characteristics. Regardless of the measure adopted, the importance of adiposity as a predictor of T2DM risk is supported by studies that indicate decreased insulinmediated glucose transport and metabolism within adipose tissues relative to skeletal muscle, contributing to the development of insulin resistance and hyperinsulinaemia.⁴⁰

2.1.3.2 Diet

Given links between adiposity and T2DM risk, a number of dietary exposures are likely to be important modifiable risk factors for T2DM risk, including fat, carbohydrate and fibre consumption.

Fibre has been hypothesised to reduce T2DM risk by hastening intestinal transit, decreasing the amount of carbohydrates and fats digested and dispersed into the bloodstream.^{41,42} Data pooled from 17 studies indicated a 9% reduction in risk per 10 g/day increase in total fibre consumption (RR 0.91, 95% CI 0.87-0.96).

Through their effect on blood sugar and obesity, carbohydrate-rich foods have been found to be linearly associated with the risk of T2DM. With foods ranked according to their glycaemic load (GL), with one unit equal to the effect on blood sugar of consuming 1 g of glucose, figures collated from 18 cohort studies indicated a 3% increase in T2DM risk for every 20 unit increase in GL per day (RR 1.03, 95% CI 1.00-1.05).⁴³

By contrast, results reported by a recent meta-analysis of eight longitudinal studies indicated no dose-response association between saturated fat consumption and the risk of T2DM, with elevated risks observed only among crudely adjusted models.⁴⁴ Similarly, despite polyunsaturated fats commonly found in fish having properties thought advantageous toward a lowering of T2DM risk, including reductions in serum lipids, platelet aggregation and blood pressure,⁴⁵ data pooled from 18 cohorts identified no association between T2DM and fish consumption (RR per 100 g/day 1.12, 95% CI 0.94-1.34) or polyunsaturated fatty acids (RR per 250 mg/day 1.04, 95% CI 0.97-1.10).⁴⁶

Reporting an array of positive, inverse and null associations, studies sampled as part of the saturated and polyunsaturated fat meta-analyses were notable in their considerable degree of heterogeneity, drawing into question the validity of reporting single summary measures. It was possible that real dose-response effects were lost due to noise resulting from marked differences in methods, populations and quality. While the proportion of variation in risk explained by between-study variance was low to moderate among fibre (I^2 =29%) and GL studies (I^2 =54%),⁴³ heterogeneity was considerable among those that investigated saturated fats

 $(l^2=91\%)^{44}$ and polyunsaturated fats (fish consumption, $l^2=81\%$; polyunsaturated fat consumption, $l^2=78\%$).⁴⁵ Additionally, the majority of sampled studies had adjusted for some measure of adiposity. With obesity likely positioned on the causal pathway between fat consumption and T2DM risk, it was possible that risks associated with dietary exposures had been biased toward the null by having controlled for an important intermediate factor.^{47,48}

Taken collectively, fibre and carbohydrate appear to have an important influence upon T2DM risk beyond any effects they might have upon adiposity, while the effects of saturated and polyunsaturated fats remain inconclusive owing to substantial heterogeneity between studies.

2.1.3.3 Heritability

Although not modifiable, a predisposition to T2DM appears to be transferred vertically between generations via genetic inheritance. Compared with age-specific rates of T2DM as found in the US general population, individuals with a parental history of T2DM were found to have 2.3 times the risk of T2DM, while those with both a parent and a grandparent exhibited 3.5 times the risk of having developed the condition.⁴⁹

Looking to the incidence of T2DM as documented by the largest twin study to have explored the heritability of T2DM, the degree of concordance was reported to be 34% (95% Cl 27-41%) among monozygotic and 16% (95% Cl 12-20%) among dizygotic twin pairs, with levels of concordance higher among female twins pairs⁵⁰ Indeed, the proportion of total variance calculated as attributable to genetic factors was highest among women, at 69% of male and 79% of female variance, suggesting that women may be more likely to inherit a genetic predisposition to the disease.⁵⁰ Other studies have reported higher levels of concordance, though sample sizes were substantially smaller and therefore subject to greater variance.^{51,52}

Beyond twin studies, new data are becoming available from genome-wide association studies (GWASs), which sequence or impute the entire human genome and quantify the degree of association between SNPs and disease. To date, the number of SNPs associated with T2DM risk has risen in just a few years from 38⁵³ and 39⁵⁴ to more than 70,^{55,56} with almost two dozen variants implicated in dysfunctional metabolic processes, including abnormalities in insulin secretion and insulin response.⁵³

Despite such findings, SNPs that reach genome-wide statistical significance currently account for only around 6%⁵⁷ to 10%⁵³ of total variance in T2DM risk attributed to heritability, suggesting that more genes have yet to be discovered, have too small an effect on their own to be selected at the GWAS significance threshold,⁵⁸ or that heritability estimates derived from twin studies

may have been overestimated or unreliable, such as through their small sample sizes.⁵⁹ Until these issues are overcome, adjustment for family history of T2DM may present a pragmatic approach by which heritability may be accounted for in observational studies.

2.1.3.4 Physical activity

Physical activity is also a significant risk factor for T2DM, with longitudinal studies from a range of populations indicating an inverse relationship.^{60,61} Specifically, results from the latest available meta-analysis indicates a 35% reduction in T2DM risk among study participants in the highest exposure category relative to those in the lowest category (RR 0.65, 0.59-0.71), with heterogeneity low across the 14 applicable studies (I²=18%).⁶¹ Reductions in risk were also evident among individuals who switched from low to moderate or higher categories (RR 0.64, 95% CI 0.50-0.70, n=7, I²=0%), suggesting that any negative effects conferred by low physical activity may be reversible.

Although reductions in T2DM risk associated with physical activity may be mediated via an effect upon adiposity,⁶² both factors has been identified as independent. This is supported by the results from physical activity meta-analyses, which identified large measures of effect despite adjustment for BMI across the majority of constituent studies. However, obesity appears to be the stronger determinant of risk.⁶³

Accordingly, various additional mechanisms have been proposed beyond the direct effect of physical activity upon adiposity. These include exercise-induced increases to (a) the capillarisation of muscle tissue,^{64,65} (b) the production of glucose transporter proteins,^{66,67,68,69} and (c) muscle glycogen synthase activity, an enzyme involved in the conversion of glucose into glycogen for energy storage.^{70,71} Together, these three processes have been posited to increase the rate at which glucose can be metabolised and transferred out of the bloodstream and into skeletal muscle tissue, resulting in improved insulin sensitivity as evident in the form of lower post-load blood glucose measurements among individuals who exercise for longer durations and with greater intensity.⁷²

2.1.3.5 Smoking

The latest available meta-analysis of 84 longitudinal cohort studies and indicated that current smokers exhibit a 37% greater risk of T2DM relative to non-smokers (RR 1.37, 95% CI 1.33-1.42).⁷³ This relationship was also found to be dose-dependent, with risks largest among those classified as heavy smokers (RR 1.57, 95% CI 1.47-1.66). A stratification of non-smokers revealed an elevated risk among former smokers relative to those that reported never smoking (RR 1.14,

95% CI 1.10-1.18), suggesting that complete life-ling abstention may be optimal for reducing T2DM risk, but that elevated risks conferred by current smoking may still be attenuated by cessation. This latter inference is supported by the inverse relationship between the duration of cessation and T2DM risk.⁷⁴ Were a causal relationship present between smoking and the development of T2DM, the authors estimated that around 12% of male and 24% of female T2DM cases worldwide were attributable to active smoking.

Paradoxically, however, smoking has repeatedly been associated with advantageous changes in adiposity,⁷⁵ with BMI lowest among current smokers and smoking cessation linked to increases in BMI even after adjustment for baseline adiposity.⁷⁶ Putative mechanisms for such an inverse relationship range from smoking being a behavioural alternative to eating⁷⁵ to a direct genetic link between the two factors.⁷⁷ Whatever the underlying process, smoking appears associated with marked reductions in one of the leading risk factors for T2DM. Additionally, although smokers tend to exhibit a clustering of obesogenic health behaviours,⁷⁸ the dose-response nature of the association between smoking and T2DM risk and the magnitude of observed effect sizes were such that the relationship appeared unlikely to be attributable to residual confounding alone.

Any causal mechanism between smoke exposure and T2DM remains unclear, though there is some suggestion from experimental studies that smoking may increase insulin resistance,^{79,80,81} with improvements to insulin sensitivity observed following cessation.⁸² Such a relationship may be mediated by nicotine through its apparent effect upon insulin signalling proteins.⁸³

2.1.3.6 Alcohol consumption

2.1.3.6.1 Average volume of alcohol consumed

An increasing body of evidence has highlighted the role that the average volume of alcohol consumed may play in modifying the risk of T2DM. According to a recent meta-analysis, which comprised 20 longitudinal studies, a J-shaped dose-response relationship exists between the volume of alcohol consumption and T2DM risk.¹⁰ Relative to never drinkers, reductions in risk were reported to be greatest at 22 g/day (RR 0.87, 0.76-1.00) among men and 24 g/day (RR 0.60, 0.52-0.69) among women, with risks rising incrementally thereafter. A newer meta-analysis published after the analyses undertaken in Chapter 3 also reported a non-linear dose-response relationship, with risks lowest among adults that consumed 91-168 g/week (men: RR 0.80, 95% CI 0.72-0.89; women: RR 0.57, 95% CI 0.48-0.67), relative to the lowest consumption category

reported among constituent studies. Peak reductions in risk were evident at around 22 g/day (RR 0.57, 95% CI 0.51-0.62) among men and 30 g/day (RR 0.70, 0.64-0.76) among women.⁸⁴

These findings have been countered by a Mendelian randomisation meta-analysis of 48 international cohorts.⁸⁵ Rather using a self-reported measure of alcohol intake, adults were instead categorised according to a genetic proxy for alcohol consumption. Because genotypes tend to be assigned at random during meiosis (assuming that the choice of partner was not associated with the genotype (panmixia)), groups of drinkers defined according to a genetic variant should benefit from a more even distribution of confounding factors akin to a randomised controlled trial, strengthening any inferences drawn.⁸⁶ However, despite being less prone to reverse causality and confounding, it should be noted that results derived using genetic data are subject to a unique set of assumptions. Aside from the requirement that a genetic polymorphism be associated with the exposure of interest, explaining a substantial proportion of variance in the volume of alcohol consumption consumed, it is assumed that a selected marker be associated with the outcome of interest solely through its effect on alcohol consumption, e.g. is unrelated to other genetic traits that may influence the outcome differentially (antagonistic pleiotropy).⁸⁶

In this particular analysis, rs1229984 was selected as a single-nucleotide polymorphism (SNP) in the alcohol dehydrogenase 1B gene (*ADH1B*), responsible for encoding an enzyme implicated in the metabolism of alcohol.⁸⁷ Compared to the more common G-allele variant of the SNP (rs1229984(G)), carriers of an A-allele (rs1229984(A)) exhibit an impaired clearance rate of alcohol metabolites, leading to increased flushing, palpitations and drowsiness through a resultant accumulation of acetaldehyde, a metabolic by-product of alcohol.⁸⁸ Due to the adverse biological response it elicits following alcohol consumption, rs1229984(A) has been associated with lower volumes of daily alcohol consumption and a higher prevalence of non-drinking,⁸⁸ as well as a lower odds of alcohol dependence.⁸⁹

Results from this Mendelian randomisation meta-analysis were in concordance with these negative effects. Although mean intake was not reported, carriers of the rs1229984(A) polymorphism had a 17% (95% CI 15.6-18.9%, I²=64%) lower weekly volume of alcohol consumption and a 27% higher odds of being a non-drinker (OR 1.27, 95% 1.21-1.34, I²=73%), relative to non-carriers (rs1229984(G)).⁸⁵ Notably, an analysis of all available individual participant data showed no difference in the odds of T2DM by genetic variant (p=0.627, n=145,063). Similarly, when data were restricted solely to those who defined themselves as current drinkers, no decrease in odds was detected among rs1229984(A) carriers (OR 0.97, 95%

CI 0.86-1.09, n=111,140) despite a lower volume of alcohol consumption. Such findings suggest that poor adjustment for confounding factors may have explained at least some of the apparent reduction in risk observed at moderate volumes of alcohol consumption present in observational studies, with around one-third of such studies having provided crude or age-adjusted estimates only.¹⁰ Weak confounder adjustment among current observations studies is an issue discussed in more detail in Section 2.3.

A number of alternative explanations for a null result were possible. With consideration to the biological effect of the A-allele polymorphism, the absence of any reduction in risk among lighter drinkers with a genetically impaired alcohol metabolism might be attributable to higher concentrations of inflammatory alcohol metabolites, which may have offset any reduction in risk otherwise conferred by a more moderate volume of alcohol consumption. This was argued by authors of a paper which analysed participants from the Nurses' Health Study and Health Professionals Follow-Up Study.⁹⁰ While reductions in risk were observed among self-reported moderate drinkers in accordance with the 2009 meta-analysis, the presence of an alcohol dehydrogenase polymorphism was found to significantly attenuate the reductions in risk reported among moderate drinkers (p=0.02). Such a finding led the authors to conclude that, were there a causal relationship between the volume of alcohol intake and T2DM, the risk of T2DM may be mediated by an impaired clearance of inflammatory metabolites such as acetaldehyde^{91,92,93} (see also Section 2.2.2.3).

Aside from this, a null result may also be a consequence of numerous statistical limitations.⁹⁴ Firstly, the Mendelian randomisation meta-analysis does not rule out a relationship between some other dimension of alcohol consumption and the risk of T2DM, such as the effect of drinking pattern. As highlighted in the following section, there are indications from some observational studies that the frequency of drinking or propensity for binge drinking may represent important determinants of T2DM risk beyond a simple measure of average volume consumed. Secondly, only a relatively small proportion of the variation in alcohol consumption was explained by the *ADH1B* genotypes. Were the mean volume of alcohol intake equal to a value of 50 g/week, for example, then the average difference in consumption between carriers and non-carriers would equate to just 8.5 g/week, or barely one unit. In this sense it was possible that the disparity in consumption between the two groups was too small to detect any effect. Thirdly, the prevalence of rs1229984(A) carriers within the pooled dataset was low, at just 7% of the sample, meaning that statistical power may have been insufficient to detect any effect, if present. Accordingly, though Mendelian randomisation studies can offer advantages over
observational studies, evidence from cohort and case-control studies still have an important role to play in developing a better understanding of the relationship between alcohol consumption and T2DM risk.

2.1.3.6.2 Drinking pattern

Beyond the volume of alcohol consumed on an average day or week, there is some suggestion that the pattern of alcohol consumption may represent a second and important component of T2DM risk. Of the 20 studies selected as part of the 2009 meta-analysis, just three were documented as having investigated some measure of consumption patterning.¹⁰ The first, which utilised data from a US occupational cohort, found that the frequency of consumption over an average week was significantly associated with T2DM risk, even after adjustment for the average volume of alcohol consumption (p=<0.001).⁹⁵ Specifically, each additional drinking day per week was associated with an independent and multivariable adjusted reduction in risk of 7% (RR 0.93, 95% CI 0.90-0.97).

The second publication, a prospective cohort of monozygotic twins, looked instead at episodic heavy drinking. Although they provided no detailed dose-response analysis of the interaction between volume and pattern, they did document that female participants who reported consuming >179 g on any one occasion at least one a month during the preceding year had double the risk of T2DM relative to those that did not report binge drinking (RR 2.1, 95% CI 1.0-4.4).⁹⁶ No such association was observed among men, though the average volume of alcohol consumption among binge drinkers was not reported.

Finally, the third publication, based on a male multi-ethnic prospective cohort study, stratified volumes of weekly alcohol intake by the number of drinking days during the week.⁹⁷ Although sub-group sample sizes were small and confidence intervals consequently wide, point estimates revealed a notable interaction between weekly volume and frequency. At the lowest volume of alcohol consumption (<70 g/week), little difference in risk was evident when stratified by frequency of drinking over the week, suggesting that such a volume of intake may have been too low to detect an interaction effect in such a small sample. However, at higher volumes of consumption, risks were lower among those who distributed their consumption over a greater number of days. For instance, relative to non-drinkers, those who consumed >210 g/day over 1-3 days/week were five times more likely to have developed T2DM (RR 5.21, 95% Cl 1.79-15.2), while those that distributed the same volume over a full seven days exhibited the lowest risk of T2DM (RR 0.68, 95% Cl 0.39-1.20). Consumption within this group was equivalent to around 30

g/day, or a level almost equal to that associated with the peak reduction in risk reported by the 2009 meta-analysis.¹⁰

Although studies that investigated joint associations between the two dimensions of drinking and T2DM risk were few in number and explored the relationship in different ways, both the volume and frequency of consumption appeared to represent important risk factors, with T2DM risk appearing lowest among regular moderate drinkers, and any benefits potentially offset by concentrated periods of episodic heavy consumption.

2.2 Alcohol consumption

Representing a potentially modifiable risk factor for T2DM and appearing from observational studies to reduce risks at regular moderate volumes of alcohol consumption, trends in the volume and pattern of UK alcohol consumption are described below. With a peak reduction in risk appearing to be conferred at an volume of around 22-24 g/day, according to estimates from longitudinal observational studies,¹⁰ trend data gave an indication as to the likely risk profile of the UK general population and thereby the degree to which the burden of T2DM could potentially be attenuated through population-level changes in alcohol consumption behaviour.

2.2.1 Trends

2.2.1.1 Volume of alcohol consumption

Current national alcohol guidelines advise that the total volume of consumption exceed no more than 21 units/week among men and 14 units/week among women.⁹⁸ A recent review of the available evidence recommended that these thresholds be revised to 14 units/week for both sexes.⁹⁹ This newly proposed threshold equates to 16 g/day, or a volume close to the 24 g/day nadir in T2DM risk reported by observational studies.¹⁰

In aggregate, data from the Health Survey for England 2014⁵ and the Scottish Health Survey 2014¹⁰⁰ both indicate that consumption among men and women is close to or within current national guidelines, measuring an average 19.0 g/day among English men and 9.9 g/day among English women,⁵ with values slightly lower in Scotland. Even within age groups where consumption is highest, volumes are not of a level associated with an increased risk of T2DM. For instance, among English men aged 65-74 years, consumption measured 23.0 g/day on average (Figure 2.4), with 11.9 g/day consumed by English women aged 16-24 years (Figure 2.5). However, it was unlikely that the volume of drinking is distributed uniformly across all seven days. As discussed in the next section, alcohol intake is marked by instances of episodic heavy consumption potentially associated with an increased risk of T2DM.^{96,97}



Figure 2.4 Mean volume of alcohol consumption among male current drinkers, stratified by age group and country. Health Survey for England and Scottish Health Survey, 2014.





2.2.1.2 Pattern of alcohol consumption

To get some idea as to the prevalence of episodic heavy consumption in the general population, data were obtained from the annual Opinions and Lifestyle Survey, which reported alcohol consumption from a random probability sample of UK residents.¹⁰¹ Based on the data available, episodic heavy consumption was defined as any current drinker who reported consuming more than twice the recommended daily limit on their heaviest drinking day in the week prior to

interview, equating to >63 g among men and >47 g among women. The degree of episodic heavy consumption was lowest among women in all age groups (Figures 2.6 and 2.7). Although gradually declining over time, episodic heavy consumption is most common within younger age groups regardless of sex. By 2013, 20.9% of male and 15.8% of female current drinkers aged 16-24 years reported consuming more than twice the recommended limit on at least one day in the week.



Figure 2.6 Proportion of male current drinkers who reported episodic heavy consumption. Opinions and Lifestyle Survey, 2005-2013.



Figure 2.7 Proportion of female current drinkers who reported episodic heavy consumption. Opinions and Lifestyle Survey, 2005-2013.

If regular moderate alcohol consumption is causally associated with marked reductions in T2DM risk, relative to never drinkers, then the presence of episodic heavy drinking among around onefifth of younger drinkers suggests that UK drinking behaviours may not be optimal for the attenuation of T2DM risk. Such trend data also suggest that an increase in the burden of T2DM may become apparent over coming decades as younger episodic heavy drinkers grow older.

2.2.2 Putative biological mechanisms

There are multiple proposed mechanisms by which alcohol consumption might influence the risk of T2DM. Three of the most commonly discussed hypothesised pathways are discussed below.

2.2.2.1 Insulin sensitivity

One pathway concerns the effect of alcohol upon insulin sensitivity, which refers to the rate of metabolic response following insulin exposure, including the metabolism, transport and storage of carbohydrates and fats.^{102,103} High insulin sensitivity is marked by the prompt metabolism of carbohydrates and fats in the presence of nominal insulin concentrations. Here, elevations in blood sugar, such as those following meals, are acute and easily regulated without the need for dietary or medicinal control. Conversely, low insulin sensitivity is indicated by an impaired insulin response, resulting in chronically elevated levels of blood glucose. Often referred to as insulin resistance, these chronic elevations in blood glucose trigger a compensatory feedback loop that can result in a state of hyperinsulinaemia.

To date, analyses exploring the relationship between alcohol and insulin sensitivity have been predominantly cross-sectional,¹⁰⁴ with just three out of 16 non-interventional studies identified as part of a recent review having been longitudinal in design. Of all the observational data identified, both U-shaped and inverse linear associations with insulin sensitivity were found, as measured variously via fasting insulin, fasting glucose or homeostasis model assessment (HOMA).

Among studies that identified a U-shaped relationship, insulin sensitivity was greatest at volumes of alcohol consumption at which reductions in risk were reported by Baliunas *et al*,¹⁰ ranging from 9 g/day¹⁰⁵ and 20 g/day¹⁰⁶ to anything below 40 g/day.¹⁰⁷ By contrast, upper limits in studies that identified a linear relationship ranged from >23 g/day¹⁰⁸ to >100 g/day.¹⁰⁹ Interestingly, although a cross-sectional analysis of data obtained from the French DESIR cohort identified a strong and sizeable linear relationship with average volume of alcohol consumption

among both sexes (p=<0.001),¹¹⁰ no such association was identified in a three-year longitudinal analysis later undertaken on the same cohort bar a linear increase in glucose among men.¹¹¹

A separate review and meta-analysis looked specifically at results from interventional studies that spanned at least a two-week period.¹¹² A total 14 studies were selected, with participant samples ranging in size from 17 to 51. Although no difference in fasting glucose concentration was identified among either male (p=0.48, study n=5) or female (p=0.94, n=6) drinkers, relative to controls, a significant sex interaction was reported between alcohol consumption and insulin sensitivity (p=0.018), which tended to be higher among female (SMD 0.16, 95% CI ⁻0.04-0.37, p=0.12, study n=5) but not male drinkers (SMD –0.30, 95% CI ⁻1.23-0.64, p=0.54, study n=5). However, heterogeneity among the studies was particularly high (I²=95%). When a study largely responsible for such heterogeneity was removed, no significant difference in insulin sensitivity between men and women remained (p=0.180). With regard to the effect of alcohol upon fasting insulin, significant reductions in concentration were observed among women (p=0.02, study n=6). While no such effect was reported among men, analyses were based on estimates extracted from just two small interventional studies of low precision (men: p=0.59, study n=2). Data from the two studies indicated a similarly advantageous direction of effect.

Although a meta-analysis of alcohol consumption and T2DM risk found greater reductions in risk among moderate female drinkers than their male equivalents,¹⁰ observational and interventional research were insufficient to conclude that this difference in dose-response may have occurred due to a sex-specific disparity in alcohol-induced insulin response. In addition to the majority of observational research thus far being cross-sectional, negating any conclusions concerning the direction of effect, results from interventional studies should also be viewed with caution. Such studies were small in size, few in number and reported heterogeneous levels of alcohol intake. Collectively, such limitations constrain any attempt to generalise interventional results to the general population.

2.2.2.2 High-density lipoprotein

Besides its putative effect upon markers of insulin sensitivity, alcohol may confer some reduction in the risk of T2DM through a causal relationship with the production of high-density lipoprotein (HDL). HDL has been linked to a number of positive metabolic processes, including the reverse cholesterol transport process, which transports lipids out of the bloodstream for storage in the liver.^{113,114} Elsewhere, HDL has appeared to play a role in the transfer of glucose away from the bloodstream and into adipose tissue for storage, as well as the promotion of

glucose metabolism by skeletal muscle tissue.^{115,116} Finally, there has been some suggestion that HDL may help promote insulin secretion by β -cells and protect them against apoptosis.¹¹⁶

Investigating alcohol-related changes to HDL concentration, data were abstracted from 25 interventional studies.¹¹⁷ The researchers identified a linear dose-response relationship, whereby HDL concentrations increased by 0.133mg/dl per gram of daily alcohol consumption, relative to controls. At the average volume of alcohol consumption (30 g/day), which was close to the nadir in risk reported by the 2009 T2DM meta-analysis,¹⁰ HDL concentration increased by 8.3% over a period of four weeks compared with controls (3.99 mg/dl, 95% Cl 3.25-4.73).¹¹⁷ This linear relationship was supported by a more recent meta-analysis of 33 studies (p=0.013),¹¹⁸ with similarly positive findings reported by a recent trial of 224 participants followed over two years (2.0 mg/dL, 95% Cl 1.6-2.2 mg/dL).¹¹⁹ The earlier of the two meta-analyses reported the dose-response relationship by sex, and found a stronger association among men than women (men: β =0.134 mg/dl per 1 g/day alcohol; women β =0.095 mg/dl per 1 g/day alcohol), though differences between sexes were not significant (p=0.930).¹¹⁷ However, the absence of a statistically significant interaction may have been due to low statistical power as opposed to the absence of a sex-specific dose-response, with only three small studies reporting female data.

Taking a different approach, a recent Mendelian randomisation meta-analysis found no overall difference in the concentration HDL (p=0.259) when data from 46 studies were stratified according to a genetic polymorphism predictive of variance in alcohol consumption,⁸⁵ though this null finding may have been a result of limitations outlined elsewhere (see Section 2.1.3.6.1).

That observed relationships in prospective cohort studies were linear as opposed to J-shaped suggested that any beneficial increase in HDL concentration following increased alcohol consumption may be offset by metabolic harms at higher intakes, such as a triggering of pancreatic β-cell apoptosis and signalling dysfunction.¹²⁰ Moreover, even if alcohol consumption is causally associated with increases in HDL concentration, there is an indication that HDL may not itself be associated with the onset of T2DM despite its apparent role in effective glucose transport and storage. For instance, in a recent Mendelian randomisation study based on 47,627 participants sampled by the Copenhagen City Heart Study and the Copenhagen General Population Study, no association was found between SNPs linked to HDL concentration and T2DM risk (p=0.550); significant dose-response relationships were specific to models that utilised observational data as opposed to a genetic indicator of HDL variance.¹²¹

2.2.2.3 Inflammation

A further hypothesis concerns the effect that alcohol may have upon inflammatory response,^{122,123} a mechanism triggered following damage to human tissue or cellular function. Specifically, inflammation and a consequent production of pro-inflammatory proteins has been associated with the disruption of endothelial and pancreatic β -cell functioning, impairing the breakdown and transport of glucose, increasing blood glucose concentrations and thereby implicated in the pathogenesis of T2DM.^{93,124,125}

As with studies investigating alcohol-related changes in HDL concentration, observational data concerning inflammatory markers is scarce and almost exclusively cross-sectional.^{122,126} Focussing on results from interventional studies, a recent meta-analysis identified publications that reported dose-response relationships between alcohol consumption and a range of inflammatory biomarkers.¹¹⁸ These included C-reactive protein (CRP), understood to increase in concentration during periods of inflammatory response but with an unknown function beyond its prognostic value,¹²⁷ and tumour necrosis factor alpha (TNF- α), a pro-inflammatory protein hypothesised to play a role in the development of insulin resistance through its suppression of insulin signalling.^{128,129,130} The concentrations of such markers were not found to be influenced by the volume of alcohol consumption, though the number of constituent studies was small (e.g. CRP, n=5; TNF- α , n=3).¹¹⁸ By contrast, a Mendelian randomisation meta-analysis of 42 longitudinal studies found CRP concentrations to be 3.4% (95% Cl 1.1-5.7%) lower among Aallele carriers whose average volume of alcohol consumption was 17% lower than among noncarriers (p=<0.001).⁸⁵ Moreover, when a non-linear dose-response relationship was estimated based on data reported by 22 Mendelian randomisation studies, CRP concentrations were predicted to be elevated only at volumes of alcohol intake greater than around 166 g/week, relative to zero alcohol consumption.¹³¹

Data from these Mendelian randomisation studies therefore indicate an advantageous reduction in the concentration of CRP at relatively moderate volumes of weekly consumption, consistent with the J-shaped association observed between alcohol intake and T2DM risk.¹⁰ However, in the absence of supporting evidence from interventional or longitudinal studies, the plausibility of such a pathway remains in question. This was especially so given a lack of robust evidence linking inflammatory intermediates to T2DM. For instance, while a study of the Whitehall II and Northwick Park Heart Study II datasets reported significant associations between baseline concentrations of CRP and the risk of insulin resistance (HOMA), hyperglycaemia and T2DM, null associations were present when data were stratified according

to SNPs significantly associated with variance in CRP concentration but independent of confounding factors such as obesity, blood pressure and socioeconomic position.¹³² Elsewhere, consideration alternative inflammatory markers, a recent Mendelian randomisation metaanalysis reported null associations between genetic variants linked to concentrations of interleukin 1 receptor antagonist, a regulator of inflammatory response mechanisms, and T2DM risk (OR 0.99, 95% CI 0.97-1.01).¹³³ Although meta-analyses of Mendelian randomisation studies have linked SNPs associated with variance in the concentration of interleukin 6 (a cell signalling molecule implicated in the suppression of inflammatory response) to a slightly lower but statistically significant risk of CHD of between three¹³⁴ and five per cent,¹³⁵ no studies appear to have thus far explored associations between SNPs of interleukin 6 and T2DM risk.

2.2.2.4 Summary

Longitudinal observational studies that investigated putative mechanisms by which alcohol consumption might attenuate the risk of T2DM have tended to be lacking, with the majority of studies being cross-sectional in design and thus inappropriate for developing an understanding of cause and effect. Interventional studies have provided a better source of data, but their generalisability has been limited by heterogeneous designs, short durations and singular volumes of consumption, obfuscating any understanding of dose-response effects following chronic alcohol intake at the population level. Finally, with large-scale Mendelian randomisation studies having identified no association between either HDL or inflammatory markers and T2DM, the likelihood of such pathways being the route by which alcohol may confer differences in T2DM risk is low.

2.3 Reference groups and confounding

Although observational and experimental data have hinted at potential biological mechanisms by which alcohol may induce a reduction in T2DM risk, the validity of such studies was questionable. As an alternative to a direct biological effect, it is possible that observed reductions in T2DM risk among moderate drinkers may instead have been a statistical artefact attributable factors such as the selection of a reference group potentially predisposed to the condition under study, and a failure to account for any such predisposition.

For instance, of the 20 studies selected as part of the 2009 alcohol-diabetes meta-analysis, 18 calculated risks associated with current drinking relative to pooled non-drinkers.¹⁰ Criticism has been levelled at this choice of abstention group on the grounds that is may capture former drinkers whose lifestyle behaviours and health status may place them at increased risk of negative health events, including T2DM.^{136,137,138} Such criticism is supported by studies that

report a higher prevalence of T2DM risk factors among non-drinkers than moderate consumers of alcohol following adjustment for differences in age and sex,^{139,140} including being of Asian ethnicity, having a high BMI and low physical activity. Elsewhere, studies also report a higher prevalence of poor self-reported health and limiting longstanding illness among non-drinkers than any other drinking category,^{141,142,143} with the onset of ill-health associated with a subsequent cessation of alcohol consumption.^{144,145,146} In at least one instance, the prevalence of limiting longstanding illness has been found to exhibit a J-shaped relationship, with the proportion of illness highest among non-drinkers and lowest among moderate consumers of alcohol.¹⁴¹ Such data suggest that pooled non-drinking categories may indeed comprise a group of former drinking 'sick quitters' who cease alcohol consumption due to ill-health. Such individuals were likely to have health profiles that placed them at greater risk of T2DM than moderate drinkers, potentially confounding the dose-response relationship between drinking and T2DM risk in poorly adjusted studies.

Of the studies selected as part of the 2009 meta-analysis, risk estimates abstracted from 13 of the 20 selected studies had been adjusted for just three or fewer confounding factors, with one-third of studies providing only crude or age-adjusted estimates. There was therefore a possibility that estimated reductions in the risk of T2DM among healthier moderate drinkers may be overestimated. Unfortunately, the meta-analysis provides no sensitivity analysis examining the effect of confounder adjustment or abstention reference category upon the shape and magnitude of the dose-response relationship. Similarly, a new meta-analysis published after the analyses undertaken for Chapter 3 provided no adjustment or stratification according to the choice of reference category by consistent studies, opting only to use the lowest category of consumption in each case, however so defined.⁸⁴ As such, the impact of former drinkers and poor confounder adjustment upon reported dose-response relationships is currently unknown.

In the absence of such an analysis, the possibility that reductions in T2DM risk may have been overestimated is instead supported by a recent analysis of studies that explored the relationship between average volume of alcohol consumption and all-cause mortality.¹³⁶ Of the 21 all-cause mortality studies to have included former drinkers within a pooled non-drinking reference group, all reported a J-shaped association consistent with the 2009 T2DM meta-analysis. However, of a further seven studies that had explicitly excluded former drinkers, no difference in risk was reported between non-drinkers and moderate drinkers. Similarly, in a meta-analysis of 34 longitudinal studies,¹⁴⁷ reductions in the risk of all-cause mortality were lower and present across a narrower range of alcohol intake among studies that utilised a robust never drinking

45

reference category (n=27) rather than a group of pooled non-drinkers (n=21). These analyses and others¹⁴⁸ all support the assertion that former drinkers represent a less healthy drinking category predisposed to or already suffering from chronic health conditions.

Although these findings suggest that former drinkers should be excluded from among pooled non-drinkers so as to leave a category of never drinkers, it should be noted that even the use of a never drinking reference category has drawn criticism. This has included the impact of such a reference group upon statistical power given their relatively small sample size in populations where alcohol consumption is otherwise normative,^{149,150} and evidence suggesting that morbidities in early adulthood may predict never drinking, potentially subjecting never drinkers to the same limitations as former drinkers.¹⁵¹ These limitations have led some to recommend the use of an occasional drinking category as a reference group.^{138,152} However, without knowing their average volume of consumption at each drinking occasion, such a category could capture either episodic heavy drinkers who appear to be at the greatest risk of T2DM, or episodic moderate drinkers that may drink infrequently due to the same morbidities as former drinkers. It is unclear why such a heterogeneous group would prove a better choice of reference category than robustly defined never drinkers, especially in a well-adjusted study of sufficient size.

Looking back to the 2009 meta-analysis, the authors attempted to work around the bias that may have arisen from selected studies having utilised a pooled non-drinking reference category confounded by less healthy former drinkers. Specifically, Baliunas *et al*¹⁰ weighted risk estimates reported by such studies according to the sex-specific proportion of former drinkers documented by studies where non-drinkers had been explicitly bifurcated into groups of never and former drinkers.

The validity of such an approach is questionable. In addition to having calculated weights from just five studies, closer inspection revealed that two of the five studies had not utilised a robust never drinking category. One defined never drinkers as those that had never consumed more than 11 alcoholic drinks in any year,⁹⁷ while the second defined never drinkers as those that reported no alcohol consumption during the five years preceding baseline and consumption no greater than four drinks per day prior to those five years.¹⁵³ In the latter instance, never drinkers may actually have been former drinkers who had once routinely consumed 72.7 g/day, assuming that each drink was equal to one average (1750 ml) glass of 13% ABV wine.¹¹

As well as estimating sex-specific proportions of former drinkers based on studies that had not defined abstention categories in a robust manner, it was unclear how reliably the baseline

prevalence of former drinkers averaged from five studies could be applied to 15 studies of disparate populations and demographics, with alcohol consumption differing according to factors beyond sex alone, including age, economic wealth, smoking and education,^{154,155} plus both per capita consumption and the prevalence of non-drinking differing markedly between countries.¹⁵⁴

In summary, an updated and revised meta-analysis was required to explore the impact of confounder adjustment and reference category selection upon the dose-response relationship between the volume of alcohol consumption and T2DM risk. Such analyses would help highlight the degree to which previous research may have overestimated any reductions in risk through a failure to properly account for a potential predisposition of former drinkers toward the development of T2DM.

Chapter 3

An updated and revised meta-analysis of the dose-response relationship between alcohol consumption and T2DM

3 An updated and revised meta-analysis

3.1 Introduction

Observational studies that utilised a self-reported measure of alcohol consumption appear to indicate a peak reduction in the risk of T2DM at levels of consumption below the equivalent of 8.5 pints of 4% ABV lager per week (22 g/day).¹⁰ However, such studies predominantly calculated risks relative to pooled non-drinkers, a group likely to comprise former drinkers whose poor health and clustered negative lifestyle behaviours potentially predispose them to the development of T2DM.¹³⁶ In addition, two-thirds of selected studies adjusted for three or fewer confounding factors, with one-third providing only crude or age-adjusted estimates. As a consequence, an unequal distribution of risk factors for T2DM may have been largely unaccounted for, leaving observed reductions in risk less a consequence of a direct biological effect and more a statistical artefact. This possibility is hinted at by studies of all-cause mortality, which report an attenuation or nullification of reductions in risk following the removal of less healthy former drinkers from the abstention reference category,^{136,147} as well as a Mendelian randomisation meta-analysis which found no relationship between drinking intensity and T2DM.⁸⁵

To gauge the degree to which observed reductions in risk may have been overestimated, this chapter details an updated and revised meta-analysis of longitudinal observational studies, including sensitivity analyses chosen explicitly to explore the effect of confounder adjustment and choice of abstention reference category upon the dose-response relationship. The revised analyses also discuss new studies that report joint associations between the volume of alcohol consumption and drinking pattern. It was hypothesised that any reductions in risk would be shallower among studies that used a more robust abstention category or adjusted for a broader range of confounding factors. Results from the meta-analysis were later published.¹⁵⁶ A copy of the publication is included in Appendix 3.1.

3.2 Methods

3.2.1 Selection criteria

3.2.1.1 Types of study

Studies were limited to those of cohort and case-control design, and both community and occupational datasets were considered. Although occupational datasets may have captured a narrower spectrum of drinking and lifestyle behaviours,^{157,158} data would still have captured

moderate levels of consumption – a range of particular interest given apparent reductions in T2DM risk at lower levels of intake.¹⁰

3.2.1.2 Types of participant

All adults aged ≥16 years were considered eligible regardless of sex, ethnicity or setting.

3.2.1.3 Types of exposure

Given apparent differences in dose-response between men and women,¹⁰ sex-specific selfreported alcohol consumption was selected as the exposure of interest. With a non-linear doseresponse relationship having previously been identified between alcohol consumption and T2DM risk,¹⁰ studies were only considered if risks were reported across \geq 3 categories denoting the average volume of alcohol consumption, inclusive of reference category. Studies were also excluded if the volume of alcohol consumption could not be converted into g/day, and if any abstention category was contaminated with participants other than non-drinkers (i.e. included light or occasional drinkers of some definition).

3.2.1.4 Types of outcome

T2DM was selected as the outcome of interest. Diagnostic tests and their respective thresholds have varied over time, with initial World Health Organisation (WHO) recommendations from 1965¹⁵⁹ having been revised in 1980,¹⁶⁰ 1985¹⁶¹ and 1998,¹⁶² with glycated haemoglobin A1C (HbA1c) added as a new diagnostic indicator in 2011.¹⁶³

Restricting selection to publications that defined T2DM according to current recommendations would have unnecessarily excluded publications prior to 2011, which were likely to have adopted the gold standard diagnostic criteria at the time they were published. Such an approach would also have excluded publications that utilised self-reported measures of T2DM. An inclusive range of diagnostic measures were thus considered in order to maximise the number of studies selected, including: all objective diagnoses, all self- or physician-reported diagnoses, any reported hypoglycaemic drug treatment, or linkage to clinical registry data. Where not explicitly defined, cases of DM among adults were assumed to be T2DM given the tendency for T1DM to occur early in life.⁸

3.2.2 Search methods

PubMed (MEDLINE), Embase, The Cumulative Index to Nursing and Allied Health Literature (CINAHL) and the Alcohol and Alcohol Problems Science (ETOH) databases were each searched for relevant studies.

Where possible, searches of each database identified publications with titles or abstracts containing the terms 'alcohol', 'ethanol' or 'drink*', plus 'diabet*', 'NIDDM' or 'T2D*', plus 'cohort', 'inciden*', 'prospective', 'longitudinal', 'case' or 'retrospective'. During each electronic search, no limits were placed upon the language or date of publication. Searches were undertaken on 18/02/2014.

Of publications selected for inclusion in the final meta-analysis, referenced and referencing publications were searched for additional literature not captured by initial electronic searches. Grey literature was excluded owing to its lack of peer review. Although technical reports and conference papers may reflect the latest research in a given field, results can be preliminary and not subject to peer review.¹⁶⁴

3.2.3 Data extraction and analysis

3.2.3.1 Study selection

3.2.3.1.1 Screening

Duplicate publications were omitted by Craig Knott (CK), who then screened the titles, abstracts and full texts of remaining publications to exclude any that failed to report a longitudinal association between volume of alcohol consumption and T2DM among persons aged \geq 16 years.

3.2.3.1.2 Shortlisting

Screened publications were then shortlisted by CK, Annie Britton (AB) and Steven Bell (SB) according to *a priori* selection criteria. These criteria were as follows:

- The full text was available, allowing the identification of population characteristics and methods used;
- Data were reported separately for men and women, enabling the analysis of differences in dose-response according to sex;
- The volume of alcohol consumption was defined across ≥3 categories so as to permit a dose-response analysis;
- The abstention category free of contamination by current drinkers, i.e. the inclusion of light current drinkers;
- Reported data were sufficient for determining both the average volume of alcohol intake and the degree of T2DM risk for each consumption category.

During shortlisting, CK reviewed all screened entries while AB and SB each acted as second reviewer for two-thirds of the screened entries. This ensured all studies were reviewed

independently by at least two individuals, with one-third reviewed by all three reviewers. In the presence of any differences of opinion, a publication was reviewed by all three reviewers and the majority decision upheld.

Agreement between pairs of reviewers was quantified via Cohen's kappa statistic (κ), while agreement between all three reviewers was determined via the calculation of a Fleiss' kappa coefficient. The Fleiss kappa coefficient served as an extension of κ and permitted the calculation of agreement across more than two reviewers.¹⁶⁵ Inter-rater agreement could take any value from –1 to +1, standardised such that a value of +1 was indicative of perfect agreement between reviewers and -1 a complete absence of agreement across all shortlisted publications.¹⁶⁶

3.2.3.2 Data requests

If a publication was agreed not to meet shortlisting criteria, contact was made with authors in an effort to obtain revised data, such as risks calculated relative to an uncontaminated reference category. Similarly, authors of publications that met shortlisting criteria were contacted if published data were found to be insufficient for the extraction or calculation of risk estimates, or sex-specific data were not reported.

In each instance, authors were given one month in which to respond. If no additional data were received, the publication was excluded (Figure 3.1). Where revised data were provided, this was noted in tables describing the characteristics of selected studies (Tables 3.1 and 3.2).

3.2.3.3 Duplicate studies

Publications that analysed the same dataset were identified and duplicates omitted with consideration to their respective degree of confounder adjustment, sample size and length of follow-up. Decisions were reached qualitatively by consensus, as detailed in Section 3.3.1.

3.2.3.4 Exposure data

A variety of methods were used for deriving volumes of alcohol consumption from categories of intake. Where the central tendency of each category was explicitly reported, the median or mean value was used, with preference given to the former measure owing to the skewness of continuous alcohol consumption.

If studies reported no central tendency for a category of alcohol consumption, intake was estimated according to the median of the lower and upper limit. However, in some circumstances the upper limit of the heaviest drinking category was not reported. Looking to meta-analyses undertaken elsewhere, methods applied for estimating an unknown upper limit were found to be arbitrary and inconsistent. While some authors operated on the assumption

that the range of a category with an undefined upper limit was equal to that which preceded it,^{167,168} or three-quarters the width of the preceding category,¹⁰ others multiplied the category's lower limit by a value of 1.2.¹⁶⁹ With no reason to conclude that any method was better than the other, unspecified upper limits were defined as 1.2 times the value of the lower limit. Where no lower bound was defined for the bottom category of volume of alcohol consumption, a value of 0.1 g was assumed.

In circumstances where alcohol consumption was reported only as the total number of drinks consumed over a given period of time, intake in g/day was estimated assuming country-specific standard drinks¹⁷⁰ unless otherwise defined by the study authors.

Finally, the volume of alcohol consumption as categorised according to periods longer than a day were converted into daily estimates assuming an even distribution of consumption over the period. For instance, a weekly volume of alcohol consumption was divided by seven.

3.2.3.5 Risk estimates

Extracted risk estimates took a number of forms. The pooling of ORs alongside other effect estimates risked being inappropriate given indications from simulated cohorts that ORs tend to overestimate equivalent RRs the greater the incidence of an event.¹⁷¹ In an effort to bring ORs and RRs into greater concordance, ORs and their respective CIs were adjusted according to the Zhang and Yu method shown in Formula 3.1.

RR=OR/(1-p)+(p*OR)

Formula 3.1 Approximation of relative risks according to reported odds ratios

Here, p is a value equal to the proportion of T2DM among unexposed participants. Although not perfect, such an approach improves the suitability of pooling extracted ORs alongside other risk estimates for a high-incidence event such as T2DM.

Of shortlisted studies, some reported risk estimates according to a reference category other than abstention. These risk estimates were recalculated such that all risk estimates were relative to an equivalent category of non-drinking. This was undertaken iteratively using the Hamling method.¹⁷² In short, the method uses the total number of exposed and unexposed participants in conjunction with the original risk estimates reported for each consumption category (k) to derive an approximated kx2 table of either cases and controls (for case-control studies) or cases and at-risk participants (for cohort studies) at each level of alcohol consumption. This table can then be used to calculate adjusted risk estimates for any consumption category relative to a reference category of choice. The calculation of corresponding CIs incorporated the covariance

between the original risk estimates from which the kx2 table was constructed, ensuring that the degree of variance for each new estimate was not underestimated.¹⁷²

In all cases, estimates were extracted from models that incorporated the maximum number of confounding variables without adjustment for putative mediators – i.e. markers of insulin sensitivity, HDL concentration or inflammation. Such models would likely have been subject to an overadjustment bias, effectively controlling for the very mechanisms under investigation and thereby biasing risk estimates toward the null to a degree proportionate to the magnitude of dose-response effect explained by the mediating factor.⁴⁷

3.2.4 Data synthesis

3.2.4.1 Model selection

Although meta-analyses have explored dose-response relationships according to a series of predefined alcohol consumption categories,¹⁷³ step functions provide a crude operationalisation of the relationship under study. Although easily interpretable by lay audiences, categories simplify the dose-response association, modelling risk as constant across a category's entire range and changing only at the exact threshold of each category.¹⁷⁴ In so doing, dose-response relationships become conditional upon both the number and width of exposure categories chosen, with the potential to inadvertently conceal nuanced changes in risk when categories are broad and few in number.

Fractional polynomial (FP) regression models provided a more flexible means of analysing nonlinear dose-response relationships, utilising alcohol consumption data in its continuous form.¹⁷⁵ An extension of standard linear regression, FP regression models use transformations of alcohol consumption according to a restricted range of fractional powers (x^{-2} , x^{-1} , $x^{-0.5}$, $\ln(x)$, $x^{0.5}$, x^1 , x^2 and x^3).¹⁷⁶ Added as a single term, these transformations are capable of modelling monotonic dose-response relationships, while as a pair can also form non-monotonic functions with a single turning point, such as the J-shaped dose-response association previously observed between volume of alcohol consumption and T2DM risk. For example, a FP regression that comprises two polynomial terms is expressed as per Formula 3.2, with x^{p1} and x^{p2} equal to any pre-defined transformation of alcohol consumption from the restricted range of fractional powers, $\beta_{1,...,n}$ equal to the change in risk for every unit change in $x^{p1,...,n}$, and β_0 equal to the constant.

$$\mathsf{RR} = \beta_0 + \beta_1 \mathbf{x}^{\mathsf{p}1} + \beta_2 \mathbf{x}^{\mathsf{p}2} + \varepsilon$$

Formula 3.2 Derivation of regression coefficients according to a polynomial exposure

To identify the best-fitting transformation, all possible FP combinations were compared against a null model containing only a constant parameter. Fit was determined according to the deviance statistic (D), equivalent to the sum of squared residuals under OLS regression, or the degree to which observed data differed from the fitted function. Accordingly, the best-fitting model was that which reported a D value closest to zero.

FP regression models were constructed using the -fp- command in Stata 13.¹⁷⁷ As the -fpcommand calculates dose-response relationships using weighted least squares regression, any resulting coefficients are under the assumption that data are independent. Such an assumption is invalid in this instance given that groups of risk estimates are nested within selected studies and calculated relative to a common reference category. When applied to non-independent data, least squares regression produces estimates with too great a degree of precision.¹⁷⁸ Accordingly, using an approach similar to the Hamling method,¹⁷² covariance present between groups of extracted risk estimates was used to inflate standard errors reported by the bestfitting FP regression models. This was undertaken using the -glst- command.¹⁷⁹

3.2.4.2 Heterogeneity

An assumption commonly underlying the pooling of data for meta-analysis is that, for a given level of volume alcohol consumption, corresponding effect estimates extracted from each study (y_i) were approximations of the same underlying effect (θ), with the extracted estimates each distributed around θ according to random sampling error.¹⁸⁰ Under this assumption, an inverse variance weight is typically applied (Formula 3.3), providing less prominence to smaller studies on the grounds that more precise information about the same underlying effect would be provided by larger selected studies.

$w=1/SE_i^2$

Formula 3.3 Inverse variance weight

However, given that no two selected studies were likely to have been methodologically identical, some degree of between-study heterogeneity was expected beyond simple random error,¹⁸¹ with each y_i thereby estimating an underlying effect specific to the characteristics of its design and population (θ_i). In such a circumstance, the use of an inverse variance weight would be inappropriate, giving prominence to a sub-sample of selected studies that may have estimated an underlying effect different to those given lower prominence. Accordingly, random effects models were used throughout. These assumed that each effect estimate (y_i) for a given level of

exposure approximated its own specific underlying effect (θ_i), and that each underlying effect (θ_i) was itself distributed around a 'grand' value for all possible effects (μ) at the given level of exposure. The introduction of random effects thereby added an additional variance component (τ) equal to the estimated deviation between each θ_i and μ , thereby attenuating the inverse variance weight proportionate to the estimated deviation of each effect estimate's underlying effect from the grand effect of interest (Formula 3.4). Results from the random effects models were more conservative, with wider confidence intervals in the presence of between-study heterogeneity. Where no between-study heterogeneity was present (τ =0), however, and the random effects models would naturally report figures analogous to a fixed effect model.¹⁸⁰

$w=1/SE_i^2+\tau$

Formula 3.4 Inverse variance weight plus a random effects variance component

The degree of between-study heterogeneity was quantified using the I² index.¹⁸¹ This estimated the proportion of total variance in risk as attributable to factors other than random error, and was calculated for each fitted polynomial term according to Formula 3.1:

$I^{2}=((Q-df)/Q)*100\%$

Formula 3.5 Calculation of the I² statistic

Here, df was equal to the number of constituent studies minus one, and the Cochran Q statistic equal to the sum of squared differences between each abstracted data point and the fitted polynomial function, with the contribution of each data point equal to its weight in the dose-response meta-analysis. An I² of 0% would thus indicate complete homogeneity between studies, with no variability between risk estimates other than that which would be expected through random error. Beyond this, the following guidelines were used for interpreting the I² index: 0-40%, low heterogeneity; 30-60%, moderate heterogeneity; 50-90%, substantial heterogeneity; 75% to 100%: considerable heterogeneity.¹⁸²

3.2.4.3 Publication bias

There was a possibility that studies that investigated the relationship between volume of alcohol consumption and T2DM risk were more likely to have been published if they reported a significant and beneficial dose-response association. For instance, in an analysis of publication rates among 29,729 biomedical abstracts presented at scientific conferences, half were eventually published in full and the probability of publication associated with positive directions of effect (RR 1.30, 95% CI 1.14-1.47) and statistically significant results (RR 1.17, 95% CI 1.02-1.35).¹⁸³ Similar findings have been identified elsewhere.¹⁸⁴ If such publication bias were present

in the field of alcohol research, published studies selected for meta-analysis would have underrepresented smaller studies or those that reported contrary results and thereby overestimated the apparently advantageous relationship between moderate alcohol consumption and T2DM risk.

Funnel plots have commonly been used as a means of visualising the degree of publication bias present across sampled studies, plotting effect estimates along the x-axis and an indicator of their precision rising along the y-axis. In the absence of bias, less precise studies would be distributed evenly around the base of the plot with more precise studies clustered near the top, collectively resembling an inverted and symmetrical funnel. Accordingly, an asymmetrical distribution of risk estimates would indicate that studies of a particular statistical significance or direction of effect may have been less likely to reach publication.

Funnel plots are conventionally constructed by including a single risk estimate from each selected study, with all risk estimates relating to a consistently defined binary exposure (i.e. the risk of exposure versus non-exposure) such that they are assumed to be distributed around the same overall effect in a manner akin to a fixed effect meta-analysis.^{185,186} Given that dose-response meta-analyses required the abstraction of multiple risk estimates from each selected study, with estimates pertaining to differing volumes of alcohol intake, such an assumption would have been invalidated were all extracted data included in a single plot.

Accordingly, new risk estimates were recalculated using the Hamling method so as to represent the risk of current drinking relative to non-drinking in each study.¹⁷² These new estimates were then plotted against their respective standard errors. The contribution of each risk estimate to the calculation of an overall risk of current drinking was then estimated according to an inverse variance weighted linear regression equivalent to a fixed effects meta-analysis.¹⁸⁸ Random effects were not used when calculating the summary estimate and doing so risked shifting the summary estimate such that the small study effects being investigated may become masked.¹⁸⁸ Risk estimates were plot on a logarithmic scale to ensure that effects in opposite directions but of the same magnitude were all equidistant from a risk of 1.0.¹⁸² A formal test of funnel plot asymmetry was also undertaken using the Egger's test, which tested the null hypothesis of no difference in effect according to sample size by way of an inverse variance weighted linear regression of standardised risk estimates against their SEs.¹⁸⁷

Although a customary to view funnel plots as indicators of publication bias, such an interpretation should be made with consideration to factors other than publication bias that

may have produced an asymmetrical distribution. Alternative explanations include selection bias (selection criteria may have inadvertently excluded studies distributed around specific areas of the funnel plot, such as if negative associations between an exposure and event were most common among occupational cohorts but such populations had been excluded) and heterogeneity among selected studies (that, free of selection bias, selected studies still differed by characteristics associated with the relationship of interest, such as primarily representing a population sub-group in whom a particular direction of effect was more common). Accordingly, any funnel plots should only be used as a rough guide and best interpreted as a graphical representation of small-study effects – i.e. the tendency for studies of lower power to show lower levels of precision.¹⁸⁸

3.2.4.4 Quality assessment

The quality of selected studies was assessed using the Newcastle-Ottawa quality assessment scale (Appendix 3.2).¹⁸⁹ It comprised eight questions ranging from the representativeness of the sample, the method of case ascertainment and the length of follow-up. A single point could be awarded for each question apart from one concerning confounder adjustment for which up to two points could be awarded. One point was awarded for adjustment for any confounding factors, with an additional point awarded if a study adjusted for 'the most important factor'. This was defined as adiposity, based on a study by Hu *et al*, which identified excess body fat as the strongest predictor of T2DM in a cohort of female nurses.¹⁹⁰ Study quality was thus ranked on a scale from 0-9 points, with larger scores indicative of higher quality.

Besides confounder adjustment, a number of other questions had to be subjectively defined. For instance, an 'adequate' period of aggregate follow-up was defined as at least six years in total. This threshold was decided upon with consideration to research by Tabak *et al*,¹⁹¹ which examined 13-year trajectories of fasting and post-load glucose prior to the development of T2DM. They identified a gradual increase in glucose concentration over time, followed by a rapid elevation in concentration around three (fasting glucose) and six (post-load glucose) years prior to T2DM diagnosis. Assuming that study participants were free of T2DM at baseline with normative blood glucose readings, it was thus expected that a period of at least six years would have been required for blood glucose levels to reach a concentration sufficient for the formal diagnosis of T2DM.

3.2.4.5 Sub-group analyses

In addition to a summary dose-response analysis of all extracted data combined, a series of secondary analyses were undertaken to explore whether the relationship differed according to pre-specified factors.

3.2.4.5.1 Sex interaction

Given a higher incidence of T2DM³⁵ and a slower metabolism of alcohol among men,^{192,193,194} it was possible that the relationship between the volume of alcohol consumption and T2DM risk may have differed according to sex. Such a hypothesis was supported by data from the last available meta-analysis,¹⁰ which indicated that female drinkers experienced greater reductions in T2DM risk across a wider range of alcohol consumption than men, relative to quasi-never drinkers. Accordingly, in addition to the primary analysis of all data combined, consideration was given to a possible sex interaction. In addition to formally testing a sex interaction, the dose-response relationship was reported separately for men and women.

3.2.4.5.2 Reference group

To explore the effect of reference group selection upon the dose-response relationship, abstracted data were adjusted for any differences in dose-response according to sex and then an interaction term tested according to whether risk estimates had been calculated relative to never or pooled non-drinkers. Data were then stratified and dose-response relationships reported separately according to reference category. It was hypothesised that the magnitude of any reductions in T2DM risk at moderate volumes of alcohol consumption would be smaller among studies that excluded former drinkers from their reference category.

3.2.4.5.3 Confounder adjustment

As noted previously, it was hypothesised that confounder adjustment was likely to have a marked impact upon whether reductions in T2DM risk were observed at moderate levels of alcohol intake, with the magnitude of any reduction posited to be smaller among studies with more comprehensive adjustment for confounding factors.

To test this, dose-response data were adjusted for sex and reference group, with an interaction term then tested according to each study's degree of confounder adjustment. Confounder adjustment was defined according to a binary variable denoting whether extracted risk estimates were either crude or age-adjusted only, or multivariable-adjusted. In addition to formally testing whether the dose-response relationship differed according to the degree of

confounder adjustment, results were reported separately for each category of the binary variable.

3.2.4.5.4 Other sub-group analyses

Aside from sex, reference category and confounder adjustment, a number of other factors were posited to be likely modifiers of the observed dose-response relationship and thereby sources of heterogeneity between studies. These included the baseline age of participants, method of case ascertainment, population type, population region¹⁹⁵ and overall study quality assessment score. The reasoning behind the selection of such factors was as follows:

- Baseline age: It was hypothesised that any reduction in risk would be lower among older participants owing to age-related decreases in liver function and body water, increasing and prolonging the concentration of blood alcohol and alcohol metabolite beyond those seen in younger individuals at equivalent volumes of alcohol consumption.^{196,197,198,199} Accordingly, any reductions in risk at moderate intakes might therefore be smaller within older populations as a consequence of prolonged accumulations of alcohol metabolites and their impact upon oxidative stress and a triggering of inflammatory response.^{200,201}
- Case ascertainment: Subjective diagnoses of T2DM may have been subject to recall or self-reporting biases, leading to cases of T2DM being underreported. Depending on how any underreporting was distributed according to the volume of alcohol consumption, studies that utilised a subjective measure of T2DM diagnosis may have either overestimated or underestimated the magnitude of any dose-response relationship. Although an analysis investigating the validity of self-reported T2DM in epidemiologic studies found subjective measures to be reliable.²⁰²
- Follow-up duration: It was possible that studies with shorter periods of follow-up may have captured few new cases of T2DM and thereby have both insufficient power for the detection of dose-response effects.
- Population region: Given a genetic susceptibility among Asian populations to impaired alcohol metabolism²⁰³ and increased T2DM risk,²⁰⁴ it was hypothesised that any reductions in risk among moderate drinkers would be less pronounced among Asian studies in much the same way as older drinkers.
- Population type: Termed the 'healthy worker effect', studies have found participants in occupational cohorts to be healthier than those sampled in general population studies.²⁰⁵ Accordingly, were a reduction in risk observed among moderate drinkers, any

such reduction in risk may have been shallower in studies of general populations whose participants were less healthy and potentially at greater risk of T2DM beyond any adjusted confounding factors.

• Quality assessment: With studies scoring low on the Newcastle-Ottawa assessment scale¹⁸⁹ potentially being less well adjusted for confounding factors, utilising self-reported outcome data, short follow-up periods and unrepresentative samples, it was posited that dose-response relationships reported by publications of lower quality would differ markedly in some fashion from those of higher quality.

Unfortunately, in the absence of individual participant data, the effect of differences in baseline age and follow-up duration were not explored due to the risk of aggregation bias, which would have inappropriately treated all study participants as having shared the same average age or length of observation.²⁰⁶ For all remaining factors, interaction terms were formally tested in a mutually exclusive manner following adjustment for sex, reference category and confounder adjustment, if found to be significant modifiers of the dose-response relationship. Data were then stratified according to the value of each factor.

To explore the role of case ascertainment as a modifier of the dose-response relationship, a categorical variable was created denoting whether case ascertainment was subjective, objective or some combination of the two. A binary variable was created to investigate population type, indicating whether studies sampled occupational groups or the general population. Asian and non-Asian studies were identified and coded into a binary variable, studies also coded according to whether they fell below the median quality assessment score.

3.2.4.6 Sensitivity analyses

Having identified a study that contributed a substantial proportion of sampled data, an *a posteriori* sensitivity analysis was undertaken to explore the effect of including such a study upon the pooled analyses. An additional sensitivity analysis was also carried out to compare the dose-response relationship according to whether selected studies had been sampled by the preceding meta-analysis. Such an analysis would highlight whether the results of any newly analysed studies differed markedly to those previously examined.

3.3 Results

3.3.1 Excluded studies

As shown in the study flow diagram (Figure 3.1), 2,357 unique publications were identified by the initial search procedure, with preliminary screening by CK leading to the exclusion of 2,255

publications that reported no temporal association between volume of alcohol consumption and T2DM risk. The remaining 102 publications were then shortlisted by three reviewers against *a priori* selection criteria.

CK assessed all 102 publications, while AB shortlisted the first two-thirds (n=68) and SB shortlisted the last two-thirds (n=72), providing an overlap of one-third (n=38) for which shortlisting was undertaken by all three reviewers. Agreement between reviewers was high, with a κ of 0.86 between CK and SB, 0.91 between CK and AB, and 0.82 between all three reviewers.

Of the 102 publications that reported a temporal association between volume of alcohol consumption and T2DM risk, reviewers agreed that the full text of one was unobtainable, 45 reported <3 levels of alcohol intake, and seven provided data insufficient for estimating drinking in g/day. A further seven publications provided no sex-specific risk estimates, one reported information insufficient for calculating risk estimates for each consumption category (i.e. a figure without numbered axes or estimates), and one reported an abstention category contaminated by current drinkers.

With 40 shortlisted publications agreed to have m*et* all selection criteria, a further six suitable publications were identified following an examination of their referenced and referencing material. The resulting 46 publications were then checked to ensure that cohorts were each sampled only once. Where the same dataset was analysed by multiple publications, risk estimates were included from only the most robust analysis. Agreed upon by all three reviewers, the qualitative rationale in each instance was as follows:

- Four of the 46 shortlisted publications analysed the US Nurses' Health Study. Two of these were excluded for providing substantially shorter periods of follow-up and smaller sample sizes than the chosen publication.^{207,208} Although the third publication provided almost twice the follow-up duration of any other paper, it was excluded due to its much smaller sample size and stratification of risk estimates for each consumption category according to a genetic polymorphism as opposed to the volume of alcohol consumption.⁹⁰
- Three studies analysed the US Health Professionals' Follow-up Study. The first was excluded for providing data on 46,000 fewer participants than the chosen publication.⁹⁰
 Although the second paper included adjustment for two additional confounding factors

than the chosen publication,²⁰⁹ it sampled over 5,000 fewer participants and utilised half the period of follow-up than the third publication.

- Three studies also analysed the multi-centre European Prospective Investigation into Cancer and Nutrition (EPIC) study, each with roughly equivalent levels of confounder adjustment. The first sampled over 10,000 fewer participants than the chosen publication and was thus excluded.²¹⁰ While the second publication sampled 8,800 more participants than the chosen study,²¹¹ it was excluded in preference of a paper that sampled a much wider range of European countries (Netherlands versus Denmark, France, Germany, Italy, Netherlands, Spain, Sweden and the United Kingdom).
- Two papers were published in 2010 by the same primary author, each sampling the same Japanese Gifu Prefectural Center for Health Check and Health Promotion cohort. Reporting similar levels of confounder adjustment and follow-up duration, the publication with the smaller sample size was excluded.²¹²
- Two other shortlisted publications were also published by an identical primary author, with each sampling employees from the same Japanese gas company with equivalent levels of confounder adjustment and follow-up. Here, the study with the smaller sample size was excluded.²¹³

The removal of these eight duplicate studies left a final sample of 38 unique studies.



Figure 3.1 Study flow diagram

3.3.2 Included studies

As per the study flow diagram in Figure 3.1, 38 unique studies met *a priori* selection criteria out of an initial 2,704 search results. Of these, nine had provided additional, updated or recalculated risk estimates via personal correspondence.

Data extracted from the 38 studies represented 1,082,639 male and 819,966 female participants, among whom 79,633 and 46,293 cases of T2DM were reported. Crude or age-adjusted estimates were provided by 15 (39.5%) studies. Of the 23 remaining studies, measures of adiposity were universally accounted for, with smoking controlled by 16 (69.6%) studies, physical activity by 15 studies (65.2%), and familial heritability by 10 studies (43.4%). Other confounding factors included education (n=9, 39.1%), dietary variables (n=6, 26.1%), blood pressure (n=5, 21.7%), ethnicity (n=3, 13.0%) and some marker of social status such as occupational grade (n=3, 13.0%). More detailed information concerning the characteristics of the selected studies are reported in Table 3.1, while Table 3.2 lists the risk estimates for each level of alcohol consumption.

Despite capturing an additional 18 studies relative to the preceding alcohol-T2DM metaanalysis, only five studies were identified as having used a never drinking category strictly defined as complete abstention across the life course.^{96,219,225,230,244} These included Kao *et al*,²¹⁹ who defined never drinkers as those who answered in the negative to the two following questions: "Do you presently drink alcoholic beverages?" and "Have you ever consumed alcoholic beverages?" Similarly, Carlsson *et al*,⁹⁶ defined never drinkers as baseline non-drinkers who also reported no change in alcohol consumption at any time prior, while Wannamethee *et al*²²⁵ defined never drinkers as those who reported no consumption at all periods of follow-up (ages 15-17, 18-22, 23-30, and 31-40 years of age). Where a study's definition of never drinking was deemed inadequate, such as through an inclusion of infrequent⁹⁷ or former drinkers,^{153,234} such studies were classified as having use a pooled non-drinking reference category.

Table 3.1 Characteristics of selected studies

First author	Year	Country	Dataset	Design	Population	Age	Duration	Exposure ascertainment	Case ascertainment
						(years)	(years)		
Holbrook ²¹⁴	1990	US	Rancho	Cohort	Community	40-79	14	Self-reported average weekly	A positive FPG or OGTT result, or a
			Bernardo					consumption of any alcoholic drinks.	self-reported doctor-diagnosed
									diabetes.
Kawakami ²¹⁵	1997	Japan	Large electrical	Cohort	Occupational	18-53	7.9	Self-reported mean consumption of	A positive FPG result among
			company					alcoholic drinks usually consumed per	participants identified following a
								week during the past year.	urine sample as having glycosuria.
Tsumura ²¹⁶	1999	Japan	Osaka Health	Cohort	Occupational	35-61	9.7	Self-reported usual daily volume	A positive FPG or OGTT result.
			Survey					consumption.	
Ajani ²¹⁷	2000	US	Physicians'	Cohort	Occupational	40-85	12.1	Self-reported frequency of consumption.	Self-reported doctor diagnosis.
			Health Study					Responses interpreted as number of	
								standard drinks consumed.	
Wei ²¹⁸	2000	US	Cooper Clinic	Cohort	Community	30-79	6.1	Self-reported average weekly volume	A positive FPG result. Of those
			Study					consumption of beer, wine or hard	without a positive result, a self-
								liquor. Alcohol content estimated as 1.1	reported history of diabetes or use
								g/1 oz beer, 2.7 g/1 oz wine, and 15.1 g	of hypoglycaemic medication.
								/1 oz liquor.	
Conigrave ⁹⁵	2001	US	Health	Cohort	Occupational	40-75	10.9	Self-reported average frequency of	A positive FPG, non-FPG or OGTT
			Professionals'					consumption, with average daily volume	result, or an elevated plasma glucose
			Follow-up					estimated assuming 12.8g/beer,	reading on two different occasions,
			Study					11.0g/glass of wine, 14.0g/glass of	or self-reported hypoglycaemic
								liquor.	treatment.

Hu ¹⁹⁰	2001	US	Nurses'	Cohort	Occupational	30-55	15.3	Self-reported average frequency	Self-reported cases confirmed
			Health Study					of beer, wine and liquor	following a positive FPG result, or
								consumption. Daily volume	an elevated randomly measured
								estimated according to assumed	plasma glucose concentration.
								nutrient content by drink type.	
Kao ²¹⁹	2001	US	ARIC	Cohort	Community	45-64	Men: 5.3	Self-reported average number of	A positive FPG, or the self-reported
							Women: 5.4	drinks consumed per week.	use of hypoglycaemic medication,
								Assumed 12.0 g/drink.	or a self-report of doctor diagnosis.
Meisinger ²²⁰	2002	Germany	MONICA	Cohort	Community	35-74	Men: 7.5	Self-reported volume of beer,	Self-reported doctor diagnosis or
							Women: 7.6	wine and spirit consumption on	self-reported use of hypoglycaemic
								the previous workday and the	medication.
								previous weekend.	
Wannamethee ²²¹	2002	UK	British	Cohort	Community	40-59	16.8	Self-reported frequency, quantity,	Self-reported T2DM, confirmed via
			Regional					and type of alcohol consumption.	primary care records.
			Heart Study						
Carlsson ⁹⁶	2003	Finland	Finnish Twin	Cohort	Community	≥18	28	Self-reported volume of alcohol	Death certificates, the National
			Cohort					consumed during an average	Hospital Discharge Register and the
								week (beer, wine) or month	Medication Register of the Social
								(spirits).	Insurance Institution.
100222	2002	Korea	Steel	Cohort	Occupational	25-55	Λ	Self-reported data. No further	A positive EPG result
Lee	2003	когеа	company	Conort	Occupational	25-55	4	detail published.	A positive FPG result.

Nakanishi ²²³	2003	Japan	Building contractor	Cohort	Occupational	35-59	6.1	Self-reported frequency of alcohol consumption per week, and usual amount consumed per day in units of "go" (23 g ethanol).	A positive FPG result, or self- reported use of hypoglycaemic medication.
Sawada ²²⁴	2003	Japan	Gas company	Cohort	Occupational	20-40	13.6	Self-administered questionnaire. No further detail published.	A positive FPG or OGTT result, or the self-reported prescription of hypoglycaemic medication.
Wannamethee ²²⁵	2003	US	Nurses' Health Study II	Cohort	Occupational	25-42	8.1	Self-reported beverage-specific frequency of consumption during the past year. Volume consumption estimated according to assumed drink-specific ABVs.	A positive FPG or OGTT result, or elevated plasma glucose levels on two different occasions, or self- reported hypoglycaemic treatment.
Lee ²²⁶	2004	US	lowa Women's Health Study	Cohort	Community	55-69	9.3	Self-administered questionnaire. No further detail published.	Self-reported doctor diagnosis.
Waki ²²⁷	2005	Japan	JPHC Study	Cohort	Community	40-59	10	Self-reported type and frequency of alcohol consumption per week, and the usual amount of alcohol consumed per day.	Self-reported doctor diagnosis.
Hodge ⁹⁷	2006	Australia	Melbourne Collaborative Cohort Study	Cohort	Community	40-69	4	Self-reported beverage-specific average frequency and quantity of consumption, plus volume via a seven-day diary.	Self-reported doctor diagnosis.

Hu ²²⁸	2006	Finland	FINMONICA	Cohort	Community	35-74	Men: 13.0 Women: 13.8	Self-administered questionnaire. No further detail published.	National Hospital Discharge Register or National Social Insurance Drug Register, confirmed via a positive FPG or OGTT result, or self-reported hypoglycaemic drug treatment.
Strodl ²²⁹	2006	Australia	Australian Women's Health Survey	Cohort	Community	70-74	3	Self-administered questionnaire. No further detail published.	Self-reported doctor-diagnosis.
Burke ²³⁰	2007	Australia	Kimberley Aborigines	Cohort	Community	15-88	12.9	A contextualised diary of the last two 48-hour drinking periods.	Linkage to hospital admission and mortality records.
Djoussé ¹⁵³	2007	US	Cardiovascular Health Study	Cohort	Community	≥65	6.3	Self-reported usual frequency of beer, wine and liquor consumption, and quantity consumed on an average occasion.	A positive FPG result or self-reported hypoglycaemic drug treatment.
Maty ²³¹	2008	US	Alameda County Study	Cohort	Community	17-94	34	Self-reported frequency of consumption, and quantity consumed on an average occasion.	Self-reported doctor-diagnosed diabetes.
Onat ²³²	2009	Turkey	Turkish Adult Risk Factor Study	Cohort	Community	≥18	7.4	Self-administered questionnaire. No further detail published.	A positive FPG result or self-reported hypoglycaemic drug treatment.
Roh ²³³	2009	Korea	Annual health evaluation	Cohort	Community	Not reported	4	Self-reported frequency of consumption, and quantity consumed on an average occasion.	A positive FPG result.

								Self-reported average frequency of	
		US	Black			21-69	9.4	beer, wine, and liquor consumption	Self-reported doctor diagnosis,
Boggs ²³⁴	2010		Women's	Cohort	Community			during the previous year. Volume	excluding cases among participants
			Health Study					consumption estimated assuming 12.0	aged <30 years.
								g/drink.	
			Korean Cancer			30-95		Self-administered questionnaire. No	Outpatient treatment for diabetes
Jee ²³⁵	2010	Korea	Prevention	Cohort	Community		14	further detail nublished	(at least three visits for diabetes
			Study						care per 365 days).
		Japan	Gifu Center					Self-administered questionnaire	A positive EPG result or self-
Nagava ²³⁶	2010		for Health	Cohort	Community	30-59	Men:8.2 Women: 7.7	econfirmed by a personal interview	reported hypoglycaemic drug
Nuguyu	2010		Promotion					with a public health nurse. No further	treatment
								detail published.	
		France	DESIR	Cohort	Community	30-65	9	Self-reported usual daily consumption	
								of wine, beer, cider and spirits. The	A positive FPG or HbA1c result, or
Balkau ²³⁷	2011							following volumes were assumed: 10.0	self-reported treatment for
								g/125 ml wine or 250 ml of beer/cider,	diabetes.
								and 7.0 g/20 ml of spirits.	
		Denmark,						Self-reported number and frequency of	Self-reported doctor-diagnosed
		France,						glasses of beer, cider, wine, sweet	diabetes or self-reported
		Germany,		Nested				liquor, distilled spirits or fortified wines	treatment for diabetes, or linkage to primary or secondary care registers or diabetes and pharmaceutical registers.
Beulens ²³⁸	2012	Italy,	EPIC-InterAct	case-	Community	35-70	9.9	consumed during the previous 12	
		Netherlands,		cohort				months. Volume estimated according	
		Spain,						to assumed glass volume and ethanol	
		Sweden, UK						content for each drink type.	

Cullmann ²³⁹	2012	Sweden	Stockholm Diabetes Prevention Program	Cohort	Community	35-56	8-10	Self-reported frequency and quantity of medium and strong beer, wine, dessert wine and spirits.	A positive FPG or OGTT reading.
Sato ²⁴⁰	2012	Japan	Kansai Healthcare Study	Cohort	Occupational	40-55	3.5	Self-reported weekly frequency of alcohol consumption and quantity consumed per average drinking day. Volume estimated according to a Japanese standard drink equivalent to 23.0 g/180 ml sake.	A positive FPG result or self- reported hypoglycaemic drug treatment.
Stringhini ²⁴¹	2012	UK	Whitehall II	Cohort	Occupational	35-55	14.2	Self-reported quantity of drinks consumed in the previous week, converted to UK units. No further detail published.	A positive FPG or OGTT result, or self-reported doctor-diagnosed diabetes, or self-reported hypoglycaemic drug treatment.
Teratani ²⁴²	2012	Japan	Steel company	Cohort	Occupational	Not reported	4.4	Self-reported quantity of daily alcohol consumption. Volume estimated assuming 22.0 g/180 ml sake, 500 ml beer, 60 ml whiskey, 180 ml wine, or 110 ml shochu.	Self-reported doctor diagnosis, or a positive HbAlc result, or ≥6.1% or self-reported hypoglycaemic drug treatment.
Abbasi ²⁴³	2013	Netherlands	PREVEND	Cohort	Community	28-75	8	Self-administered questionnaire. No further detail published.	A positive FPG or random plasma glucose result, or a self- reported doctor diagnosis, or objective hypoglycaemic drug treatment.
Heianza ²⁴⁴	2013	Japan	TOPICS 11	Cohort	Community	26-80	10.2	Self-reported frequency of consumption and average quantity of alcohol consumed per occasion.	Positive FPG or HbA1c result, or self- reported doctor-diagnosed diabetes.
------------------------	------	--------	---------------------------------	--------	-----------	-------	------	--	--
Rasouli ²⁴⁵	2013	Norway	Nord-Trøndelag Health Survey	Cohort	Community	≥20	11	Self-reported, beverage-specific average quantity of beer, wine or spirits consumed over a two-week period. Assumed 16g/beer, 12g/glass of wine or spirits.	Of those with self-reported diabetes, T1DM was tested using a marker of pancreatic autoimmune damage. Those with a negative result were classified as having T2DM.
Shi ²⁴⁶	2013	China	Shanghai Men's Health Study	Cohort	Community	40-74	5.4	Self-reported type, frequency, and usual quantity of alcohol consumed (wine, beer, and liquor).	Self-reported cases confirmed via a positive FPG or OGTT result or use of hypoglycaemic medication.

	_	Alcohol consumption				Risk of T2D	N	_	
First author	Sex	Exposure categories ^a	Estimated g/day ^b	Cases (n)	Non-cases (n)	Measure	Effect estimates	Confounder adjustment	Quality
		Non-drinkers	Non-drinkers	6	31		1.00 (reference)		
	Mon	0.1-84.3 g/week	6.0	7	53		0.72 (95% CI 0.26-1.98)		
	IVIEII	84.4-176.0 g/week	18.6	6	55		0.61 (95% CI 0.21-1.74)		
Holbrook ²¹⁴						risk		Age	7
		Non-drinkers	Non-drinkers	16	68	1 ISK	1.00 (reference)		
	Women	0.1-41.3 g/week	6.0	7	67		0.50 (95% CI 0.22-1.14)		
	women	41.4-117.4 g/week	18.6	12	60		0.88 (95% CI 0.44-1.73)		
		117.5-750 g/week	66.2	12	61		0.86 (95% CI 0.44-1.70)		
		0 ml/week	0.0	11 ^c	590°		1.00 (reference)	Age: BMI: education: family history	
Kawakami ²¹⁵	Men	<300 ml/week	16.9	23¢	1,595°	Hazard	1.04 (95% CI 0.47-2.32)	of T2DM; occupation; physical	7
Kawakami Wien	≥300 ml/week	40.6	12¢	533¢	ratio	1.09 (95% Cl 0.44-2.67)	activity; smoking; shift pattern		
		Non-drinkers	Non-drinkers	76	1,058		1.00 (reference)		
		0.1-19.0 ml/day	7.5	95	1,226		0.98 (95% CI 0.73-1.33)		
Tsumura ²¹⁶	Men	19.1-29.0 ml/day	19.0	120	1,386	Relative	1.08 (95% CI 0.81-1.44)	Age	6
		29.1-50.0 ml/day	31.2	60	1,057	LISK	0.80 (95% Cl 0.57-1.12)		
		≥50.1 ml/day	47.4	105	1,179		1.40 (95% Cl 1.04-1.88)		
		Rarely/Never drinkers	Rarely/Never drinkers	145	2,900		1.00 (reference)		
		1-3 drinks/month	0.9	111	2,189		1.03 (95% Cl 0.80-1.33)		
A :=:::217	Man	1 drinks/week	2.0	122	2,806	Relative	0.89 (95% Cl 0.70-1.14)	Age; BMI; physical activity;	C
Ajani	wien	2-4 drinks/week	6.0	157	4,614	risk	0.74 (95% Cl 0.59-0.93)	smoking; treatment group	0
		5-6 drinks/week	11.0	80	2,613		0.67 (95% Cl 0.51-0.88)		
	≥1 drink/day	16.8	151	5,063		0.57 (95% CI 0.45-0.73)			

Table 3.2 Measures of alcohol consumption, confounder adjustment and effect estimates reported by selected studies

		Non-drinkers	Non-drinkers	36	1,811		1.00 (reference)		
W/oi ²¹⁸		1-61.8 g/week	4.8	21	1,675		0.78 (95% CI 0.44-1.37)		
Wei	Men	61.9-122.7 g/week	13.1	16	1,682	Relative risk ^{d,e}	0.56 (95% CI 0.31-1.00)	Age; family history of 12DM;	6
		122.8-276.6 g/week	26.6	35	1,655		1.22 (95% CI 0.75-1.98)	years of follow-up	
		≥276.6 g/week	83.5	41	1,661		1.32 (95% CI 0.83-2.11)		
		0 g/day	0.0	416	10,656		1.00 (reference)		
		0.1-4.9 g/day	2.3	450	11,356		1.09 (95% CI 0.95–1.24)		
		5.0-9.9 g/day	7.3	214	6,941		0.88 (95% CI 0.74–1.04)	Age; BMI	
Conigrave ⁹⁵	Men	10.0-14.9 g/day	12.3	163	6,050	Relative risk	0.77 (95% CI 0.64–0.92)		5
		15.0-29.9 g/day	19.7	174	6,321		0.80 (95% CI 0.67–0.96)		
		30.0-49.9 g/day	38.1	116	4,419		0.72 (95% CI 0.58–0.88)		
		≥50.0 g/day	70.1	38	1,419		0.64 (95% CI 0.46–0.89)		
		0 g/day	0.0	1,715	27,165		1.00 (reference)	Age; family history of T2DM;	
11190	Maman	0.1-5 g/day	2.6	1,034	26,997	Dolotivo riek	0.78 (95% CI 0.72–0.84)	menopausal status; time;	c
nu ¹⁹⁰	women	5.1-10.0 g/day	7.6	189	9,155	Relative risk	0.56 (95% CI 0.48–0.65)	use of postmenopausal	0
		>10 g/day	12.0	358	17,480		0.59 (95% CI 0.52–0.66)	hormone therapy	
		Lifetime abstainers	Lifetime abstainers	69	600		1.00 (reference)		
		Former drinkers	Former drinkers	118	978		0.93 (95% CI 0.70-1.24)		
		≤1 drink/week	0.1	74	741		0.88 (95% CI 0.64-1.23)		
	Men	1.1-7 drinks/week	6.1	139	1,227		0.98 (95% CI 0.74-1.30)		
		7.1-14 drinks/week	17.7	55	670		0.72 (95% CI 0.50-1.02)	Age: BMI: education:	
		14.1-21 drinks/week	29.3	32	281		0.94 (95% CI 0.62-1.41)	ethnicity; family history of	
		>21 drinks/week	57.4	60	379		1.75 (95% CI 1.26-2.44)	T2DM; history of	
Kao ²¹⁹						Relative risk ^{d,e}		hypertension; physical	7
		Lifetime abstainers	Lifetime abstainers	236	1,987		1.00 (reference)	activity; smoking; total	
		Former drinkers	Former drinkers	108	872		1.00 (95% CI 0.75-1.34)	energy intake; waist-hip	
		≤1 drink/week	0.1	110	1,626		0.92 (95% CI 0.72-1.17)	ratio	
	Women	1.1-7 drinks/week	5.7	90	1,226		0.99 (95% CI 0.74-1.33)		
		7.1-14 drinks/week	16.7	18	378		0.75 (95% CI 0.45-1.25))	
		14.1-21 drinks/week	28.9	5	125		0.60 (95% CI 0.24-1.47)		
		>21 drinks/week	49.6	2	56		0.39 (95% CI 0.10-1.55)		

		0 g/day	0.0	23	439		1.00 (reference)		
	Men	0.1-39.9 g/day	20.0	46	1,518		0.59 (95% CI 0.36-0.96)		
		≥40 g/day	48.0	58	968		1.14 (95% CI 0.71-1.82)		
Meisinger ²²⁰						Relative risk		None	6
		0 g/day	0.0	48	1,212		1.00 (reference)		
	Women	0.1-19.9 g/day	10.0	26	1,199		0.56 (95% CI 0.35-0.89)		
		≥20 g/day	24.0	12	618		0.50 (95% CI 0.27-0.93)		
		Non-drinkers	Non-drinkers	4	285		1.00 (reference)		
		<1 unit/week	0.6	62	1,150		0.91 (95% CI 0.50-1.65)	Age; BMI; history of CHD;	
Wannamethee ²²¹	Men	1-15 units/week	7.9	99	1,612	Relative risk ^e	0.74 (95% CI 0.45-1.20)	physical activity; smoking;	9
		15-42 units/week	32.7	64	1,361		0.60 (95% CI 0.36-0.99)	social class	
		>42 units/week	63.2	18	566		0.87 (95% CI 0.50-1.51)		
		Lifetime abstainers	Lifetime abstainers	64	1,045		1.00 (reference)		
		Former drinkers	Former drinkers	11	151		0.91 (95% CI 0.46-1.80)		
	Men	<5 g/day	3.1	181	2,525		1.06 (95% CI 0.78-1.42)		
		5–30 g/day	10.7	261	4,480		0.86 (95% CI 0.63-1.16)		
		>30 g/day	42.8	75	1,023		0.90 (95% CI 0.61-1.32)		
Carlsson ^{f,96}						Hazard ratio		Age; BMI; smoking	8
		Lifetime abstainers	Lifetime abstainers	280	2,977		1.00 (reference)		
		Former drinkers	Former drinkers	2	75		0.93 (95% CI 0.23-3.73)		
	Women	<5 g/day	2.3	273	5,655		0.79 (95% CI 0.66-0.95)		
		5–20 g/day	6.9	55	2,173		0.66 (95% CI 0.47-0.91)		
		>20 g/day	25.9	10	303		0.79 (95% CI 0.40-1.55)		
		Non-drinkers	Non-drinkers	23	816		1.00 (reference)		
		≤90 g/week	6.5	33	1,793		0.66 (95% CI 0.39-1.12)		
Lee ²²²	Men	91-180 g/week	19.4	11	733	Relative risk	0.54 (95% CI 0.26-1.10)	None	3
		181-360 g/week	38.6	11	497		0.79 (95% CI 0.39-1.61)		
		>360 g/week	61.7	5	133		1.32 (95% CI 0.51-3.42)		
		0 g/day	0.0	63	358		1.00 (reference)		
		0.1-22.9 g/day	11.5	67	467		0.87 (95% CI 0.60-1.26)	Age; BMI; family history of	
Nakanishi ²²³	Men	23.0-45.9 g/day	34.5	66	632	Relative risk ^e	0.66 (95% CI 0.47-0.93)	T2DM; physical activity;	7
		46.0-68.9 g/day	57.5	107	774	774	0.78 (95% CI 0.56-1.10))) smoking	
		≥69 g/day	82.8	67	352		0.95 (95% Cl 0.65-1.38)		

		Non-drinkers	Non-drinkers	50	1,412		1.00 (reference)	Age; BMI; family history	
Sawada ²²⁴	Men	1-45 g/day	23.5	206	2,814	Relative risk	1.59 (95% CI 1.16-2.17)	of T2DM; fitness; high	7
		≥46 g/day	55.2	24	239		1.68 (95% CI 1.03-2.76)	blood pressure; smoking	
		Lifelong abstainers	Lifelong abstainers	181	14,736		1.00 (reference)		
		Former drinkers	Former drinkers	334	23,791		1.18 (95% CI 0.98-1.41)	Age	
Wannamathaa ²²⁵	Womon	0.1-4.9 g/day	2.5	336	44,048	Polativo rick	0.67 (95% CI 0.56-0.80)		4
wannamethee	women	5.0-14.9 g/day	10.0	70	18,309	Relative risk	0.34 (95% CI 0.25-0.44)	Age	4
		15.0-29.9 g/day	22.5	8	2,308		0.29 (95% CI 0.15-0.60)		
		≥30 g/day	36.0	6	758		0.63 (95% CI 0.28-1.42)		
		Non-drinkers Non-drinkers		1,168	15,829		1.00 (reference)		
Lee ²²⁶	Women	1-14 g/day	8.0	675	15,592	Rate ratio	0.60 (95% CI 0.55-0.66)	None	4
		≥15 g/day	18.0	78	2,356		0.47 (95% CI 0.37-0.59)		
		Non/infrequent drinkers	Non/infrequent drinkers	196	3,834		1.00 (reference)		
	Mon	≤23.0 g/day	11.55	169	3,162		1.08 (95% CI 0.88-1.32)		
	IVIEII	23.1-46.0 g/day	34.55	174	2,735		1.24 (95% CI 1.02-1.52)		
		>46.0 g/day	55.32	164	2,479		1.23 (95% CI 1.00-1.52)	Age; BMI; family history	
Waki ²²⁷						Relative risk ^d		of T2DM; hypertension;	6
		Non/infrequent drinkers	Non/infrequent drinkers	436	13,919		1.00 (reference)	physical activity; smoking	
	Women	≤4.9 g/day	2.5	15	465		1.14 (95% CI 0.69-1.90)		
Sawada ²²⁴ Men Sawada ²²⁴ Men 4-45 g/day ≥ 46 g/day ≥ 46 g/day ≥ 46 g/day ≥ 46 g/day Former drinkers 0.1-4.9 g/day 5.0-14.9 g/day ≥ 30 g/day ≥ 30 g/day ≥ 30 g/day ≥ 15 g/day ≥ 15 g/day ≥ 15 g/day $\geq 3.1-46.0$ g/day $\geq 3.1-46.0$ g/day $\geq 3.1-46.0$ g/day $\geq 3.1-46.0$ g/day ≥ 46.0 g/day $\geq 3.1-46.0$ g/day $\leq 3.1-46.0$ g/day $\geq 3.1-46.0$ g/day $\leq 3.1-46.0$ g/day $\geq 3.1-46$	women	5.0-11.5 g/day	8.25	16	636		0.81 (95% CI 0.49-1.34)		
	>11.5 g/day	13.92	13	481		0.79 (95% CI 0.45-1.38)			

		Lifetime abstainer	Lifetime abstainer	25	1,795		1.00 (reference)		
		Former drinkers	Former drinkers	17	500		2.44 (95% CI 1.29-4.52)		
		<10 g/day	4.3	56	3,031		1.55 (95% CI 0.95-2.50)		
	wen	10-19.9 g/day	15.0	30	2,247		1.21 (95% CI 0.69-2.07)		
		20-29.9 g/day	24.2	13	1,333		0.80 (95% CI 0.40-1.59)	Age: BMI: country of birth:	
Hodge ⁹⁷		≥30 g/day	45.0	38	3,129	Relative risk ^d	0.86 (95% Cl 0.50-1.57)	dietary glycaemic index; dietary energy intake; waist-	8
		Lifetime abstainers	Lifetime abstainers	114	7,729		1.00 (reference)	hip ratio	
		Ex-drinkers	Ex-drinkers	9	589		1.12 (95% CI 0.55-2.24)		
	Women	<10 g/day	3.5	32	5 <i>,</i> 659		0.66 (95% CI 0.44-1.00)		
		10-19.9 g/day	15.0	18	2,838		0.82 (95% CI 0.49-1.37)		
		≥20 g/day	30.2	10	2,210		0.60 (95% CI 0.30-1.17)		
		Non-drinkers	Non-drinkers	223	3,608		1.00 (reference)		
	Men	1-100 g/week	7.2	190	3,661		0.91 (95% CI 0.75-1.11)	Age; BMI; food consumption	
		>100 g/week	17.1	104	2,402		0.74 (95% CI 0.58-0.95)	(bread; coffee, fruit, tea,	
Hu ²²⁸						Hazard ratio		sausage, vegetable);	8
		Non-drinkers	Non-drinkers	357	6,350		1.00 (reference)	smoking: study year: systolic	
	Women	1-100 g/week	7.2	87	3,877		0.74 (95% CI 0.57-0.94)	blood pressure	
		>100 g/week	17.1	3	523		0.23 (95% CI 0.07-0.73)	·	
		Non-drinkers	Non-drinkers	87	2,698		1.00 (reference)		
Strodl ²²⁹	Womon	Rarely drinkers	Rarely drinkers			Polativo ricka	1.00 (95% CI 0.74-1.35)	Nono	2
Strour	women	1-2 drinks/day	15.0	54	2,922	Relative HSK	0.58 (95% CI 0.42-0.82)	None	5
		≥3 drinks/day	36.0	12	306		1.21 (95% CI 0.67-2.17)		
		Life-long abstainers	Life-long abstainers	7	14		1.00 (reference)		
	Mon	Ex-drinkers	Ex-drinkers	14	40		0.78 (95% CI 0.37-1.65)		
	WIEIT	<150 g/day	88.0	12	86		0.37 (95% CI 0.16-0.82)		
		≥150 g/day	209.0	8	48		0.43 (95% CI 0.18-1.04)		
Burke ²³⁰						Relative risk		None	6
		Life-long abstainers	Life-long abstainers	25	66		1.00 (reference)		
	Women	Ex-drinkers	Ex-drinkers	11	38		0.82 (95% CI 0.44-1.52)		
	vonen	<100 g/day	57.0	10	48		0.63 (95% CI 0.33-1.21)		
		≥100 g/day	136.0	9	18		1.21 (95% CI 0.65-2.28)		

		Never drinkers	Never drinkers	37	476		1.00 (reference)		
		Former drinkers	Former drinkers	10	183		0.7 (95% Cl 0.3-1.4)		
	Men	<1 drink/week	0.4	13	326		0.5 (95% CI 0.3-0.9)		
		1-6 drinks/week	4.0	24	421		0.6 (95% Cl 0.4-1.1)		
Dioussá ¹⁵³		≥7drinks/week	30.0	25	384		0.8 (95% Cl 0.4-1.3)	Age; BMI;	
Djousse						Relative risk		education;	8
		Never drinkers	Never drinkers	74	1,221		1.00 (reference)	smoking	
		Former drinkers	Former drinkers	10	143		1.2 (95% CI 0.6-2.3)		
	Women	<1 drink/week	0.4	23	582		0.7 (95% Cl 0.4-1.1)		
		1-6 drinks/week	4.0	13	400		0.6 (95% Cl 0.3- 1.1)		
		≥7 drinks/week	30.0	5	285		0.4 (95% Cl 0.2-1.0)		
		0 drinks/month	0.0	21	373		1.00 (reference)		
	Men	1-45 drinks/month	10.6	85	1,652		0.92 (95% CI 0.58-1.46)		
		≥46 drinks/month	25.4	34	591		1.02 (95% Cl 0.60-1.73)		
Maty ²³¹						Relative risk		None	4
		0 drinks/month	0.0	42	771		1.00 (reference)		
	Women	1-45 drinks/month	10.6	116	1,969		1.08 (95% Cl 0.76-1.52)		
		≥46 drinks/month	25.4	9	250		0.67 (95% Cl 0.33-1.36)		
		Non-drinkers	Non-drinkers	102	936		1.00 (reference)		
	Men	<3 drinks/day	16.0	46	434		1.23 (95% Cl 0.88-1.73)		
		>3 drinks/day	38.4	14	71		1.91 (95% Cl 1.06-3.45)		
Onat ^{f,232}						Relative risk		Age; physical	6
		Non-drinkers	Non-drinkers	157	1,384		1.00 (reference)	activity, smoking	
	Women	<3 drinks/day	16.0	2	63		0.38 (95% CI 0.11-1.23)		
		>3 drinks/day	38.4	0	4		Too few data		
		Non-drinkers	Non-drinkers	150	276		1.00 (reference)		
Pob ²³³	Mon	1-14 g/day	8.0	251	412	Rolativo rick	1.08 (95% Cl 0.91-1.26)	Nono	F
NUII	Roh ²³³ Men	15-29 g/day	22.5	166	200	Relative HSK	1.29 (95% Cl 1.09-1.53)	NOTE	J
Roh ²³³ Men	≥30 g/day	36.0	123	139		1.33 (95% Cl 1.11-1.60)			

		Never drinkers	Never drinkers	1,669	20,457		1.00 (reference)		
		Former drinkers	Former drinkers	1,159	9,966		1.22 (95% Cl 1.13-1.31)		
D = = = = 234		1-3 drinks/week	4.0	552	7,658	Deletive viel:	0.84 (95% CI 0.76-0.93)	Age; questionnaire	-
BOBB224	women	4-6 drinks/week	10.0	132	2,530	Relative risk	0.60 (95% CI 0.51-0.72)	cycle; energy intake	5
		7-13 drinks/week	20.0	97	1,484		0.70 (95% CI 0.57-0.86)		
		≥14 drinks/week	33.6	43	654		0.71 (95% CI 0.52-0.96)		
		0 g/day	0.0	14,407	172,786		1.00 (reference)		
		1-24 g/day	13.0	33,332	418,536		0.95 (95% CI 0.93–0.97)		
	Men	25-4 g/day	37.5	7,588	80,680		0.99 (95% CI 0.96–1.02)		
		50-99 g/day	75.0	4,188	41,104		1.05 (95% CI 1.01-1.08)		
Jee ^{f,235}		≥100 g/day	120.0	1,440	13,703	Hazard ratio	1.04 (95% CI 0.99–1.10)	Age; age ² ; BMI; physical activity	8
		0 g/day	0.0	24,860	359,916		1.00 (reference)		
		1-24 g/day	13.0	3,596	60,024		0.90 (95% CI 0.87–0.93)		
	women	25-49 g/day	37.5	6	210		1.85 (95% CI 0.77–4.43)		
		≥50 g/day	60.0	2	46		1.03 (95% Cl 1.00-1.06)		
		0 g/day	0.0	212	3,940		1.00 (reference)		
	Mon	<25 g/day	12.5	198	4,035		0.92 (95% CI 0.76-1.11)		
	WIEIT	25-40 g/day	32.5	223	4,071		1.02 (95% CI 0.85-1.22)		
Nagaya ²³⁶		≥40 g/day	48.0	236	3,913	Relative risk	1.11 (95% Cl 0.93-1.33)	None	6
		0 g/day	0.0	188	6,434		1.00 (reference)		
	Women	<25 g/day	12.5	30	1,413		0.73 (95% Cl 0.50-1.07)		
		≥25 g/day	30.0	6	297		0.70 (95% Cl 0.31-1.56)		
		0 g/day	0.0	18	206		1.00 (reference)		
	Men	<20 g/day	2.0	27	411		0.77 (95% CI 0.42-1.40)		
	Wien	20-39 g/day	23.0	79	844		0.84 (95% CI 0.49-1.40)		
Balkau ^{f,237}		≥40 g/day	67.0	47	244	Relative risk ^d	1.27 (95% Cl 0.73-2.16)	Education; physical activity; smoking	5
		0 g/day	0.0	35	206		1.00 (reference)		
	Women	<20 g/day	1.0	35	411		0.95 (95% CI 0.59-1.48)		
		≥20 g/day	21.0	22	1088		0.87 (95% CI 0.51-1.43)		

		0 g/day	0.0	485	452		1.00 (reference)		
		0.1-6.0 g/day	3.1	1,303	1,262		1.03 (95% CI 0.86-1.24)		
		6.1-12.0 g/day	9.1	890	891		0.93 (95% CI 0.79-1.09)		
	Men	12.1-24.0 g/day	18.1	1,116	1,166		0.97 (95% CI 0.83-1.13)		
		24.1-60.0 g/day	42.1	1,448	1,555		0.89 (95% CI 0.77-1.02)	Age; BMI; education; food	
		60.1-96.0 g/day	78.1	393	363		0.80 (95% CI 0.65-0.99)	consumption (coffee, fruit,	
Beulens ²³⁸		>96.0 g/day	115.2	126	85	Hazard ratio ^e	1.10 (95% Cl 0.79-1.54)	meat, total energy, vegetable); physical	8
		0 g/day	0.0	1,601	2,013		1.00 (reference)	activity; smoking;	
		0.1-6.0 g/day	3.1	2,429	3,828		0.91 (95% CI 0.86-0.96)		
	Women	6.1-12.0 g/day	9.1	743	1,483		0.75 (95% CI 0.66-0.84)		
		12.1-24.0 g/day	18.1	623	1,322		0.79 (95% CI 0.70-0.90)		
		>24 g/day	28.8	402	838		0.81 (95% CI 0.69-0.95)		
		Non-drinkers	Non-drinkers	10	62		1.00 (reference)		
		0.01-6.79 g/day	3.4	46	501		0.62 (95% CI 0.32-1.19)		
Beulens ²³⁸ Cullmann ²³⁹ Sato ²⁴⁰	Men	6.80-13.01 g/day	9.9	28	488		0.41 (95% CI 0.23-0.73)		
		13.02-22.13 g/day	17.6	41	505		0.56 (95% CI 0.33-0.96)		
		≥22.14 g/day	g/day 0.0 485 452 1.00 (reference) 1-60 g/day 3.1 1,303 1,262 1.03 (95% Cl 0.86-1.24) 1.12.0 g/day 9.1 890 891 0.93 (95% Cl 0.37-1.09) 2.1-24.0 g/day 18.1 1,116 1,166 0.97 (95% Cl 0.33-1.13) 4.1-60.0 g/day 78.1 393 363 0.80 (95% Cl 0.77-1.02) Age; BMI; education; food 96.0 g/day 115.2 126 85 Hazard ratio* 1.10 (95% Cl 0.79-1.54) meat, total energy, wegetable; physical g/day 0.0 1,601 2,013 1.00 (reference) activity; smoking; 1.12.0 g/day 9.1 743 1,433 0.75 (95% Cl 0.36-0.94) activity; smoking; 2.1-24.0 g/day 18.1 623 1,322 0.79 (95% Cl 0.70-0.90) activity; smoking; 2.1-24.0 g/day 2.8.8 402 838 0.41 (95% Cl 0.32-0.73) activity; smoking; 0.16-6.79 g/day 3.4 46 501 0.62 (95% Cl 0.32-0.17) activity; smoking; 1.01-6.79 g/day						
Cullmann ²³⁹						Relative risk ^{d,e}		family history of T2DM;	8
		Non-drinkers	Non-drinkers	6	94		1.00 (reference)	physical activity; smoking	
		0.01-1.49 g/day	0.8	34	724		0.92 (95% CI 0.37-2.26)		
	Women	1.50-4.71 g/day	3.1	14	766		0.39 (95% CI 0.18-0.83)		
		4.72-8.75 g/day	6.7	20	739		0.69 (95% CI 0.34-1.41)		
		≥8.76 g/day	10.5	24	755		0.87 (95% CI 0.43-1.75)		
		Non-drinkers	Non-drinkers	142	1,479		1.00 (reference)		
Sato ²⁴⁰	Men	0.1-2.0 drinks/day	14.7	350	4,055	Hazard ratio	0.94 (95% CI 0.78-1.15)	Δσο	5
5410	WICH	2.1-4.0 drinks/day	42.7	268	3,093		0.94 (95% CI 0.77-1.15)	ABC	5
		≥4.1 drinks/day	68.9	118	1,126		1.16 (95% CI 0.91-1.48)		
		Non-drinkers	Non-drinkers	131	2,287		1.00 (reference)		
		1-76 g/week	6.3	71	1,677		0.81 (95% CI 0.61, 1.08)		
Teratani ^{f,242}	Men	77-153 g/week	15.7	73	1,243	Hazard ratio	0.94 (95% CI 0.70, 1.26)	None	5
		154-307 g/week	22.0	85	1,469		0.95 (95% CI 0.72, 1.25)		
		≥308 g/week	44.0	104	1,283		1.14 (95% CI 0.88, 1.49)		

		0 units/week	0.0	85	623		1.00 (reference)		
	Men	1-21 units/week	12.4	369	3,037		0.96 (95% CI 0.75-1.22)		
		≥21 units/week	25.2	102	825		1.04 (95% CI 0.77-1.39)		
Stringhini ^{f,241}						Hazard ratio		Age; ethnicity	4
		0 units/week	0.0	111	540		1.00 (reference)		
	Women	1-14 units/week	8.5	139	1,198		0.73 (95% CI 0.56-0.94)		
		≥14 units/week	16.8	13	195		0.51 (95% CI 0.28-0.92)		
		No/Almost never	No/Almost never	47	496		1.00 (reference)		
		1-4 drinks/month	0.8	32	379		0.90 (95% CI 0.58-1.38)		
	Men	2-7 drinks/week	6.3	76	1,121		0.73 (95% CI 0.52-1.04)		
		1-3 drinks/day	19.8	53	768		0.75 (95% CI 0.51-1.09)		
Abbasi ^{f,243}		≥4 drinks/day	47.5	18	257		0.76 (95% CI 0.45-1.28)		
						Relative risk		None	6
		No/Almost never	No/Almost never	70	1,106		1.00 (reference)		
		1-4 drinks/month	0.8	39	655		0.94 (95% Cl 0.65-1.38)		
	Women	2-7 drinks/week	6.3	34	1,084		0.51 (95% CI 0.34-0.76)		
		1-3 drinks/day	19.8	22	491		0.72 (95% CI 0.45-1.15)		
		≥4 drinks/day	47.5	3	69		0.70 (95% CI 0.23-2.17)		
		Lifetime abstainers	Lifetime abstainers	15	138		1.00 (reference)		
		Former drinkers	Former drinkers	10	30		2.83 (95% Cl 1.27-6.31)		
		8-54 g/week	2.9	35	199		1.74 (95% Cl 0.95-3.19)		
Hojanzaf.244	Mon	55-98 g/week	10.9	31	214	Polativo rick	1.54 (95% CI 0.83-2.86)	٨٥٥	7
Tieldil2d ^{,,}	WIEII	99-160 g/week	17.6	23	221	Relative HSK	0.94 (95% CI 0.49-1.80)	Age	,
		161-229 g/week	24.7	30	230		1.43 (95% CI 0.76-2.66)		
		230-287 g/week	32.9	37	236		1.61 (95% CI 0.88-2.93)		
		288-748 g/week	66.3	35	166		2.38 (95% CI 1.29-4.38)		

		Abstainers	0.0	44	1,513		1.00 (reference)		
		0.01-4.9 g/day	1.7	324	11,343		0.94 (95% Cl 0.66-1.35)		
	Men	5.0-9.9 g/day	6.9	96	3,855		0.81 (95% CI 0.54-1.22)		
		10.0-14.9 g/day	11.7	18	1,387		0.46 (95% CI 0.25-0.85)	Age; BMI; education; family history of T2DM; physical activity; smoking	
Rasouli ^{f,244}		≥15 g/day	19.7	16	807	Hazard ratio	0.79 (95% Cl 0.42-1.46)		7
		Abstainers	0.0	74	3,342		1.00 (reference)		
	Womon	0.01-4.9 g/day	1.1	330	15,774		1.34 (95% CI 0.99-1.83)		
	women	5.0-9.9 g/day	6.6	33	2,220		1.37 (95% CI 0.86-2.20)		
		≥10 g/day	12.0	5	504		1.12 (95% CI 0.44-2.85)		
		Non-drinker	Non drinker	894	33,415		1.00 (reference)	Age; BMI; education; energy intake;	
chif.246	Mon	<1 drink/day	9.6	74	3,115	Hazard ratio	0.88 (95% CI 0.70-1.12)	family history of T2DM; hypertension; income; occupation; physical activity;	6
51117	IVICII	1-2.9 drinks/day	26.0	169	8,349	Hazard ratio	0.80 (95% CI 0.67-0.94)		0
		≥3 drinks/day	53.6	101	3,973		0.91 (95% CI 0.74-1.13)	smoking; wait-hip ratio	

^a The upper limit of the highest exposure category was conservatively defined as 1.2 times the value of the lower bound, unless explicitly defined within each publication.

^b Conversions into g/day were undertaken according to the median volume of alcohol consumption in each category. Means were used if medians were unreported. If neither indicator of central tendency were reported for each category, the median of the upper and lower bounds was used.

^c Figures were obtained via personal correspondence and reflected the crude number of cases and non-cases in each exposure category. These figures therefore differed slightly from those contained within the analytical sample of the original study. Figures reported via personal correspondence were used only for the estimation of covariance between reported coefficients. Total of cases and non-cases in the analytical sample were 41 and 2,271 respectively.

^d RRs were estimated based on reported ORs using the Zhang and Yu formula.¹⁷¹

^e The Hamling method was used to recalculate risk estimates according to a reference category other than that originally reported.¹⁷²

^f Additional, updated or recalculated risk estimates were provided via personal correspondence and may thus have differed from those reported by the original publication.

3.3.3 All data

Data pooled from all 38 selected studies are presented in Figure 3.2. Constituent studies each contributed \geq 3 point estimates, inclusive of reference group. In the figure, each risk estimate is plotted as a bubble of size proportionate to the inverse of its standard error such that larger bubbles represent more precise estimates.

Visual inspection of plotted risk estimates suggested sizable heterogeneity between data points at equivalent volumes of alcohol consumption. This was corroborated by the calculation of an I^2 index for each fitted polynomial, which revealed an I^2 of 75% (95% CI 67-80%) along the first polynomial, and 50% (95% CI 31-63%) along the second.

The greatest reduction in risk is evident at an alcohol intake of 12 g/day, equating to an 18% decrease in risk relative to pooled non-drinkers (RR 0.82, 95% 0.79-0.85). Risks increased incrementally above this volume, with significant reductions in risk evident up to 64 g/day. The best-fitting FP model provided a substantially better parameterisation of the dose-response relationship than a standard linear regression (p=<0.001).



Figure 3.2 Dose-response relationship between the average volume of alcohol consumption and T2DM risk, utilising all data combined

3.3.4 Sex

A scatter diagram of risk estimates stratified by men and women revealed that the doseresponse association differed by sex, with a reduction in risk at moderate volumes of alcohol consumption appearing most pronounced among women (Figure 3.3).



Figure 3.3 Scatter diagram of extracted risk estimates, stratified by sex

A sex-interaction was thus introduced against each polynomial term and a likelihood ratio test run to compare the basic FP model to one inclusive of a sex interaction. The interaction was significant overall (Wald test, p=<0.001) and improved the fit of the basic model (p=<0.001).

Sex-stratified results are reported in Figure 3.4. The dose-response relationships shown in Figure 3.4 were restricted to levels <140 g/day due to a dearth of risk estimates concerning higher volumes of alcohol intake. The results shown in Figure 3.4 appear to indicate that any reduction in risk relative to pooled non-drinkers may have been specific to women. While men show no reduction in risk at any volume, statistically significant reductions in the risk of T2DM are observed among women who reported average volume of alcohol consumption of <71 g/day. Reductions in risk were greatest at 35 g/day (RR 0.66, 95% CI 0.55-0.78), or close to two pints of 4% ABV lager.¹¹



Figure 3.4 Dose-response relationship between average volume of alcohol consumption and T2DM risk, stratified by sex

3.3.5 Reference group

Only five studies were deemed to have utilised a strictly-defined never-drinking category, with four studies having reported risk estimates for men, and four for women. Having identified a significant interaction by sex, an additional interaction was included according to whether constituent studies had calculated risks relative to never or pooled non-drinkers. The dose-response relationship differed significantly by both sex (p=<0.001) and reference group (p=<0.001), with the addition of a reference group interaction improving the fit of the dose-response relative to one containing an interaction only for sex (p=0.009).

To visualise differences in dose-response according to abstention category, risk estimates were stratified by reference group and adjusted for sex through an inclusion in each model of a sex interaction term. Sex-adjusted differences in dose-response are reported in Figure 3.5, with the level of consumption restricted to <100 g/day due to the small number of risk estimates available above that level of consumption among studies that utilised a never drinking reference category. Reductions in risk appear most pronounced among studies that calculated risks relative to pooled non-drinkers, evident up to 82 g/day and largest at 12 g/day (RR 0.80, 95% 0.83-0.86). Although a U-shaped dose-response relationship is also evident among studies that

used a strictly-defined never drinking reference category, decreases in risk are smaller in magnitude and not statistically significant, peaking at 41 g/day (RR 0.94, 95% CI 0.72-1.20).



Figure 3.5 Sex-adjusted dose-response relationship between average volume of alcohol consumption and T2DM risk, stratified by reference group

The dose-response relationship was then calculated based solely on studies that reported a never-drinking abstention category. As shown in Figure 3.6, risk estimates were sparse particularly at higher volumes of alcohol consumption. As such, the plotted results shown in Figure 3.7 are restricted to values of alcohol consumption <140 g/day due to the wide confidence intervals reported by the model. Given the small number of constituent studies, Figure 3.7 should be interpreted with caution. Nevertheless, the results indicated that potential reductions in risk at moderate volumes of alcohol consumption remained specific to women when studies excluded former drinkers from the reference category.

Chapter 3: An updated and revised meta-analysis of the dose-response relationship between alcohol consumption and T2DM



Figure 3.6 Scatter diagram of extracted risk estimates calculated relative to never drinkers, stratified by sex and inverse weighted by SE



Figure 3.7 Dose-response relationship between average volume of alcohol consumption and T2DM risk, relative to never drinkers, stratified by sex

3.3.6 Confounder adjustment

A total 15 studies reported crude or age-adjusted estimates (n=15), with the rest providing more comprehensively adjusted data (n=23). Although sex (p=<0.001) and reference category (p=<0.001) remained statistically significant modifiers of the dose-response relationship, little statistical difference was evident according to whether or not risk estimates had been extracted from crude or age adjusted studies or those with more comprehensive confounder adjustment (p=0.165). The addition of a confounder adjustment interaction provided no marked improvement in model specification of a sex-adjusted model (p=0.168).

When data were stratified by confounder adjustment and adjusted for differences in doseresponse attributable to sex and reference group, multivariable-adjusted estimates appear to show shallower reductions in risk at moderate levels of alcohol consumption (Figure 3.8). This relationship was little changed when using an alternative confounding variable that defined studies according to whether their degree of adjustment was above or below the mean of four confounding factors.



Figure 3.8 Dose-response relationship between average volume of alcohol consumption and T2DM risk, adjusted for sex and reference group, stratified by confounder adjustment

3.3.7 Sub-group analyses

Beyond analyses stratified according to factors of primary interest, a number of further subgroup analyses were undertaken, investigating the effect of putative modifiers of the doseresponse relationship.

3.3.7.1 Case ascertainment

Cases of T2DM were defined according to a variety of methods, summarised as participant selfreports (n=11), objective ascertainment (n=21), or a combination thereof (n=6). Given the small number of studies to have employed mixed methods of case ascertainment, attention was focussed upon the subset of studies that used either a subjective or objective methods. Accordingly, data were restricted to the 32 applicable studies. The addition of an interaction according to method of case ascertainment provided little improvement over a dose-response model adjusted for differences by sex and reference category. While sex (p=<0.001) and reference category (p=<0.001) remained statistically significant modifiers, this was not the case for method of case ascertainment overall (p=0.166).



Figure 3.9 Dose-response relationship between average volume of alcohol consumption and T2DM risk, adjusted for sex and reference group, stratified by method of case ascertainment

Looking to the stratified dose-response relationship reported in Figure 3.9, studies that utilised an objective measure of case ascertainment reported reductions in risk of smaller magnitude and across a narrower range of consumption than studies with cases defined according to subjective self-reports. Among the latter, a peak 35% reduction in T2DM risk was observed at

20 g/day (RR 0.65, 95% CI 0.55-0.76) versus a 14% reduction in risk at 4 g/day (RR 0.86, 95% CI 0.82-0.91) among studies that defined cases using an objective measure, relative to pooled nondrinkers after adjustment for sex and reference group.

3.3.7.2 Population region

A total 13 studies sampled participants from Asian regions, with the remaining 25 studies having sampled participants from non-Asian countries. Population region represented a statistically significant modifier of the dose-response relationship (p=<0.001), even after accounting for the effect of sex (p=<0.001) and reference category (p=0.029). Inclusion of an interaction according to population region improved specification over a model that comprised interactions only for sex and reference category (p=<0.001).

Looking to the stratified dose-response relationship shown in Figure 3.10, risk estimates extracted from Asian studies reported no reduction in T2DM risk at any level of average volume of alcohol consumption. By comparison, estimates from non-Asian countries exhibited a J-shaped association even after adjustment for sex and reference category, with an intake of an average 26.5 g/day associated with a 27% reduction in risk relative to adjusted non-drinkers (RR 0.73, 95% CI 0.67-0.78).



Figure 3.10 Dose-response relationship between average volume of alcohol consumption and T2DM risk, adjusted for sex and reference group, stratified by population region

3.3.7.3 Population type

Another factor posited as a modifier of the relationship between volume of alcohol consumption and T2DM was population type. Of selected studies, 12 sampled occupational cohorts and 26 analysed participants from the general population. Alongside the addition of sex (p=<0.001) and reference group (p=0.008) interaction terms, the dose-response relationship did not differ significantly according to population type (p=0.407). The inclusion of an interaction by population type did not improve a model already adjusted for differences according to sex and reference category (p=0.410).

Little difference in dose-response was observed when data were stratified according to population type, though reductions in risk among general population studies may have been less pronounced than those reported by occupational samples (Figure 3.11).





3.3.7.4 Quality assessment

The quality of selected studies ranged from three to nine points out of nine. One study received the maximum score of nin e,²²⁵ while two studies received a score of just three^{222,229} for limitations including the use of a self-reported measure of T2DM, no information concerning the derivation of alcohol consumption volume, short periods of follow-up and no adjustment for confounding factors. The median score was six, indicating a moderate level of quality on average.

There was no significant difference in overall dose-response according to whether studies fell below a median quality assessment score of six points (p=0.837) after adjustment for differences according to sex (p=<0.001) and reference category (p=0.01). Although reductions in risk appeared greater among studies of below median quality, the difference was small and not significant (Figure 3.12).



Figure 3.12 Dose-response relationship between average volume of alcohol consumption and T2DM risk, adjusted for sex and reference group, stratified by quality assessment score

3.3.8 Sensitivity analyses

3.3.8.1 Newly analysed studies

Of the 20 publications analysed by the preceding meta-analysis of volume of alcohol consumption and T2DM, 17 had been selected for analysis as part of the revised meta-analysis. Of those not selected, two were superseded by newer analyses of the same studies that also benefitted from larger samples and periods of follow-up,^{207,210} whilst the remaining study did not report sex-specific estimates.²⁴⁷ This left 21 publications with analyses that were either updated since the last meta-analysis, or else missed by the previous authors.

A statistically significant difference in dose-response was identified according to whether studies had been previously sampled by Baliunas *et al*¹⁰ (p=<0.001) after accounting for any differences attributable to sex (p=<0.001) or choice of reference category (p=0.357). Stratifying the data revealed that newly analysed studies appeared to report a shallower reduction in risk at

moderate volumes of alcohol intake after adjustment for sex and reference category (Figure 3.13). Of studies sampled in the previous meta-analysis, a peak 29% reduction in risk was observed at 20.5 g/day (RR 0.71, 95% Cl 0.61-0.82) versus a 13% reduction at 14 g/day (RR 0.87, 95% Cl 0.83-0.91) among newly analysed studies, relative to adjusted non-drinkers.

Potentially explaining at least some of this discrepancy, sex-specific logistic regression analyses comprising putative sources of heterogeneity reported that previously analysed studies of men were more likely to have used subjective case ascertainment (p=<0.001) and sampled non-Asian populations (p=<0.001), while studies reporting female data adjusted for a greater number of confounders (p=0.029), less likely to have analysed occupational samples (p=0.028) and were more likely to have utilised subjective measures of case ascertainment (p=0.020) and a strictly-defined never drinking category (p=0.003).





3.3.8.2 Sensitivity of dose-response to a high precision study

Having extracted data from all selected studies, 63% of constituent participants were found to have been sampled by a single, high quality study.²³⁵ Questioning the generalisability of any results where more than half of participants had been drawn from a single publication, a sensitivity analysis was undertaken to compare the impact of the study upon the observed dose-

response relationship. As shown in Figure 3.14, the inclusion of the large study had little overall effect upon the dose-response relationship beyond contributing to a reduction in variance. This was posited to be attributable to the calculation of dose-response relationships using random effects, which applies a lower weight to larger studies. On this basis, as well as its high quality assessment score, the study was considered appropriate for inclusion in all reported analyses. As per the sex-specific analyses reported in Figure 3.4, plots were restricted to volumes of consumption below 140 g/day due to a dearth of risk estimates at higher levels.



Figure 3.14 Dose-response relationship between average daily alcohol consumption and T2DM risk: sexspecific data, stratified by Jee *et al*²³⁵

3.3.9 Small-study effects

Recalculated estimates denoting the risk of current drinking versus non-drinking were plotted against the standard error of their log (Figure 3.15). The resulting funnel plots indicated that, while male risk estimates appeared largely symmetrically dispersed around a summary estimate denoting the overall level of risk reported for current drinkers (p=0.181), female estimates may have been asymmetrical (p=0.063), with selected risk estimates appearing more likely to report a reduction in T2DM than might have been expected in the absence of small-study effects.



Figure 3.15 Funnel plot of current drinking versus non-drinking, stratified by sex

Given the summary estimates were calculated using a simple inverse variance weight,¹⁸⁸ asymmetry was hypothesised to be a consequence of a high precision study that reported results contrary to the majority of other studies. A prime suspect for such an effect was a large, high quality Korean study, which represented 65% of sampled participants and reported a smaller reduction in risk than many other selected studies.²³⁵ The impact of the Korean study upon estimated small study effects is reported in Figure 3.16. After its exclusion, observed asymmetry among female risk estimates was indeed attenuated, with no significant different in risk estimates according to the degree of study precision among either men (p=0.610) or women (p=0.508). Small study effects and the risk of any potential publication bias was low.



Figure 3.16 Funnel plot of current drinking versus non-drinking, stratified by sex and excluding Jee *et al*.²³⁵

3.3.10 Pattern of consumption

Of the 21 newly analysed studies, only three of their associated publications included a joint analysis of both the volume of alcohol consumption and drinking pattern. The first publication reported that middle-aged male binge drinkers (>1 bottle of wine on any one occasion \geq 1/month, or around 95.6 g assuming a 750 ml bottle of 13% ABV wine¹¹) had a 67% (OR 1.67, 95% CI 1.11-1.20) greater odds of T2DM relative to light drinkers (0.01-6.79 g/day).²³⁹ This finding was in keeping with a previously discussed study, which reported double the risk of T2DM among women that consumed >179 g on any one occasion at least once a month during the preceding year (RR 2.1, 95% CI 1.0-4.4).⁹⁶ In both cases, however, reporting of the joint association between volume and pattern was limited, providing no indication as to the average volume of alcohol consumption among binge drinkers. It was possible their elevated risk of T2DM may have been attributable to a high average intake as opposed to infrequent periods of episodic heavy consumption.

A more comprehensive analysis was reported by Sato *et al*,²⁴⁰ which explicitly set out to explore the joint association between the frequency and volume of alcohol consumption in a cohort of Japanese men. While no difference in the risk of T2DM was observed among men who consumed low volumes of alcohol (\leq 28 g/day) infrequently throughout the week (1-3 days), relative to pooled non-drinkers (RR 0.93, 95% CI 0.72-1.20), reductions in risk were apparent when the same low volume was consumed on at least four days in an average week (RR 0.74, 95% CI 0.58-0.95). Such a result indicated that moderate volumes of alcohol consumption may only confer a reduction in risk when drinking is regular.

The third and final study to have investigated the two dimensions of alcohol consumption was authored by Heianza *et al.*²⁴⁴ Sampling 1,605 Japanese men, risks associated with the average volume of alcohol consumption per drinking occasion were stratified according to four weekly frequencies. Consistent with the preceding analysis, risks were lowest among participants who frequently (\geq 6 days/week) consumed moderate volumes of alcohol (23 g/day). Relative to this group, those who consumed the same moderate volume of alcohol across just 2-3 days/week reported a 48% greater risk of T2DM (RR 1.48, 0.65-3.37). Notably, risks were consistently greatest at all frequencies within a light drinking category that also included pooled non-drinkers, providing further indications as to the unsuitability of pooled non-drinkers as a reference category.

3.4 Strengths

In revising the 2009 meta-analysis, these most recent results benefitted from the addition of new studies published after the last search in 2008, plus the inclusion of two historic papers missed or discounted during previous meta-analyses. Thus, while Baliunas *et al*¹⁰ sampled 20 studies, 477,200 participants and 12,556 cases, revised analyses were based on data obtained from 38 studies, 1,902,605 participants and 125,926 cases. Although the majority of these additional observations were extracted from a single Korean study,²³⁵ a sensitivity analysis that compared results calculated with and without the large study showed negligible difference in the dose-response relationship beyond an expected change in precision (Figure 3.14).

Some alcohol researchers have argued that never drinkers may present a more suitable abstention category than pooled non-drinkers.¹³⁶ Although the preceding meta-analysis reportedly adopted a never drinking reference category when calculating the risk of T2DM among current drinkers, the never drinking category was only approximated. Specifically, effect estimates reported by 15 studies that used a non-drinking reference category were weighted according to the average sex-specific proportion of former drinkers reported by five studies that included a never drinking abstention category. This approach assumed that the average prevalence of former drinkers reported by the five studies was equal to that unreported by the remaining 15 studies and their varying population characteristics. In addition to the possibility that the study-specific proportion of former drinkers was determined by factors other than sex alone, two of the five studies from which an average proportion of former drinkers had been estimated did not define never and former drinking in a robust manner. For instance, one study defined never drinkers as participants who described themselves as non-drinkers at baseline and who also reported (a) not having changed consumption during the five years prior to baseline, and (b) never having regularly consumed ≥ 5 drinks/day.¹⁵³ Here, never drinkers were likely to comprise a number of former or infrequent moderate drinkers. Accordingly, it was possible that the previous meta-analysis may not have reliably estimated the proportion of never drinkers contained among pooled non-drinking categories. Contrary to their method, the updated meta-analysis opted instead to test for an interaction in dose-response according to type of reference category defined by selected studies. Such an approach avoided the potential for having weighted data under inappropriate assumptions. Beyond this, the revised metaanalysis also tested the effect of hypothesised modifiers of the dose-response relationship, providing insight into their effect upon summary dose-response estimates.

3.5 Limitations

3.5.1 Heterogeneity

Heterogeneity between sampled studies was high, with the breadth of effect estimates at equivalent volumes of consumption complicating interpretation of the overall dose-response relationship. In order to better understand the effect of such heterogeneity upon the observed dose-response relationship, potential explanatory factors were posited *a priori* and, where appropriate, explored individually through the inclusion of a corresponding interaction term along each fitted polynomial, with results then reported separately at each level of the a priori factor. The use of meta-regression to jointly test differences in dose-response according to all putative sources of heterogeneity was avoided owing to the potential for low statistical power, even when effect sizes and the number of studies are large.^{248,249,250} Though suggested that statistical power may have been sufficient in instances where the number of factors did not exceed a ratio of one to every 10 studies,²⁵¹ simulations have suggested that power may remain low in circumstances where heterogeneity is high.²⁵²

3.5.2 Quality assessment

Although the quality of selected studies was assessed using the Newcastle-Ottawa assessment scale,¹⁸⁹ with studies found to be of moderate quality on average (a score of 6/9), quality assessment tools are subject to notable limitations. Although a wide range of instruments have thus far been devised for the assessment of quality among non-randomised studies, each comprised assessment criteria that were disparate in both number and nature.²⁵³ In addition to the use of different rating scales or summary scores that risked weighing the importance of component items in ways disproportionate to their impact upon the validity of a given study, their differing designs were such that the choice of tool was likely to have had a sizeable bearing upon the assessment of study quality.^{254,255} Alongside such general limitations, the Newcastle-Ottawa tool used to assess the quality of studies selected as part of the revised meta-analysis had criticisms ranging from the tool's focus upon the generalisability of results as opposed to a study's internal validity,²⁵⁶ to weak inter-rater reliability on some questions.^{257,258} With these limitations in mind, results from the Newcastle-Ottawa quality assessment tool should be considered only as a rough guide as opposed to a definitive measure of study quality.

3.5.3 Stability of consumption

In defining alcohol consumption using only one baseline measure, almost all studies sampled as part of the meta-analysis had modelled risk under an assumption that alcohol consumption was stable over the course of the study. However, having investigated datasets for which repeated

measures of alcohol consumption were available, researchers identified numerous disparate trajectories of alcohol consumption behaviour as a function of time.²⁵⁹ As such, the cross-sectional categorisation of alcohol consumption may have been an inappropriate approach for defining drinking, with baseline categories potentially subject to misclassification error.¹⁵⁰

Defined as latent classes, numerous discrete trajectories have been identified across the adult life course.^{260,261,262,263} For example, looking to analyses that captured alcohol consumption data between the ages of 16 and 42 years, five distinct classes were observed in a Finnish cohort.²⁶⁴ Here, in addition to stable categories of light (22%), moderate (35%) and heavy consumption (23%), 9% of participants decreased and 11% increased their consumption between the ages of 16 and 42.²⁶⁴ Elsewhere, in a study of Canadians aged \geq 50 years, nine different classes were identified over an average six years.²⁶⁵ As well as stable light (7%), moderate (11%), heavy (2%) never (6%) and former drinkers (8%), plus those who increased (16%) or decreased (31%) their consumption, this older cohort also captured U-shaped (10%) and inverted U-shaped (9%) trajectories.²⁶⁵



Figure 3.17 Common latent trajectories of volume of alcohol consumption: the 'cat's cradle'

As evident from just two examples, exact trajectories identified within cohorts will differ according to their baseline age, duration of observation and frequency of repeated measures. Despite this, four trajectories appear robust to such factors, having been consistently observed across cohorts: stable non-drinking and current drinking, increasing consumption and decreasing consumption – a so-called 'cat's cradle' of latent classes (Figure 3.9).²⁵⁹ Of these four classes, increasing consumption may be most common during early adulthood, with younger participants typified by never drinking and therefore exhibiting little opportunity for alternative trajectories of consumption. Conversely, in later life, decreased consumption may become the

dominant trajectory following factors including age-related deteriorations to health status. This disparity was hinted at by the examples above,^{264,265} and supported by findings from an analysis of mean volume consumption calculated from nine UK-based cohort studies (Figure 3.10).²⁶⁶



Figure 3.18 Predicted mean alcohol consumption trajectories (in units/week) across the life course in UK cohort studies. Reproduced from Britton *et al*²⁶⁶ in BMC Medicine under a Creative Commons Attribution License (4.0).

Despite such evidence highlighting the variability in the volume of alcohol consumption as a function of time, the vast majority of alcohol-T2DM studies modelled the dose-response relationship according to a single baseline measurement, with just one out of the 38 selected publications having utilised repeated measures of alcohol consumption. The authors of the

paper modelled the dose-response relationship using an unconditional logistic regression model equivalent to an extended Cox survival model with time-varying covariates updated at six-month intervals over a period of 10 years.¹⁵³ Relative to never drinkers, results showed attenuated and non-significant reductions in risk across all levels of current drinking after accounting for changes to alcohol consumption over time.¹⁵³

A second publication, which was excluded from the meta-analysis, reported analyses of male participants in the Health Professionals Follow-Up Study.²⁶⁷ Compared with stable none or light drinkers (0-4.9 g/day), those who increased their consumption to moderate levels (5.0-29.9 g/day) in any four-year period exhibited a statistically significant 25% reduction in hazard (HR 0.75, 95% CI 0.62-0.90), while those who became heavy drinkers (\geq 30.0 g/day) showed a sizeable but not significant 65% reduction in hazard (HR 0.35 95% CI 0.11–1.10). Of moderate drinkers that reduced their consumption to none or light drinking, a 9% increase in hazards was observed (HR 1.09, 95% CI 0.92-1.30). Such participants may have reduced their intake due to poor health, though no such increase in risk was evident among heavy drinkers who switched to none or light consumption (HR 0.78, 95% CI 0.44-1.38). Such inconsistent results may have been a consequence of particularly small sample size in a number of sub-groups, with estimates subject to sampling error.

The variability of alcohol consumption across the life course suggests that the dose-response relationship may be more complex than modelled by conventional analytical methods. With the validity of existing research potentially limited by a failure to account for the effect of such variability upon T2DM risk, epidemiological research needs to explicitly adopt analytical approaches capable of giving consideration to changes in drinking over time.

3.6 Discussion

Although few additional studies were found to have reported a strictly-defined never drinking abstention category, sex-adjusted analyses confirmed that the choice of reference category had a marked and statistically significant impact upon the observed dose-response relationship, with reductions in risk at moderate volumes of alcohol intake appearing specific to studies that excluded less healthy former drinkers from their abstention category (Figure 3.5). Results from Figure 3.5 thus mirror findings from meta-analyses of other health conditions, which reported attenuations in risk reduction when former drinkers were explicitly excluded from an abstention reference category.^{136,147} Such a difference in dose-response provides support to arguments made by some academics against the use of a pooled non-drinking category, ^{136,137} and suggested

that reductions in risk at moderate volumes of alcohol consumption may thus have been overestimated by close to 95% of observational studies.

Beyond the impact of reference category, little independent difference in the overall doseresponse relationship was observed according to whether or not risk estimates were multivariable-adjusted. Such a finding suggests either that confounder adjustment had little additional impact after the exclusion of former drinkers, or that the type as opposed to number of confounding factors may have been a more important determinant of heterogeneity in observed dose-response.

Reductions in risk were previously identified as smallest among men.¹⁰ These were rendered completely absent in revised analyses following the addition of new data (Figures 3.4 and 3.6). These new data were more likely to have been obtained from Asian countries (53% versus 21%) and have utilised an objective measure of ascertainment (89% versus 71%) – two factors associated with less pronounced reductions in risk even after accounting for differences attributable to sex and reference category (Figures 3.10 and 3.9 respectively). Among women, peak reductions in risk were broadly comparable between the two meta-analyses: a 40% peak reduction was reported at 24 g/day among women (RR 0.60, 95% CI 0.52-0.69) sampled in the earlier meta-analysis,¹⁰ and a 34% peak reduction at 35 g/day (RR 0.66, 95% CI 0.55-0.78) in the revised meta-analysis.

It was unclear why the dose-response relationship between the volume of alcohol consumption and T2DM risk operated differently according to sex, though five conceivable hypotheses were proposed: systematic differences in study characteristics; sex-specific differences in drinking pattern; sex-specific differences in the effect of alcohol upon putative biological pathways; sex hormones; and drink type. These were explored separately and in detail below:

1. It was possible that sex-specific disparities in dose response may have been in some part a consequence of systematic differences in study characteristics. Following a stratification of study characteristics by sex, female data was found more likely to have been extracted from studies of non-Asian regions (61% versus 87%) or that utilised a subjective measure of case ascertainment (35% versus 18%). Both factors were associated with more pronounced reductions in risk than in studies of Asian regions (Figure 3.10) and that used an objective measure of defining T2DM (Figure 3.9). Although studies of women were also more likely to have sampled the general

population (87% versus 70%), population type was observed to have very little independent effect upon dose-response (Figure 3.11).

An *a posteriori* analysis was thus undertaken to explore the significance of case ascertainment and population region interactions solely among female data. Both case ascertainment (p=0.002) and population region (p=0.010) were found to be statistically significant modifiers of the female dose-response relationship, with plotted results similar to those reported in Figures 3.9 and 3.10 when both sexes were combined.

Although no sizeable difference was found in the proportion of studies that reported multivariable-adjusted risk estimates (men: 64%; women: 61%), it was possible that the degree of confounder adjustment was less important than the type of confounders for which dose-response relationships were adjusted. For instance, male studies were more often adjusted for a number of key risk factors for T2DM, including adiposity (45% versus 39%), family history of T2DM (24% versus 17%) and smoking (39% versus 30%).

Given the association between these risk factors and the risk of T2DM, as well as the possibility that they appear most prevalent among non-drinkers than moderate drinkers,^{136,139} it is plausible that discrepancies in the type of confounder adjustment alongside other differences in study characteristics may have explained at least some of the difference in dose-response between men and women. Such a conclusion is supported in part by the absence of any alcohol-T2DM relationship in a recent Mendelian randomisation meta-analysis, in which confounding factors should have been more randomly distributed between drinking groups.⁸⁵

2. With the risk of T2DM modelled exclusively according to the volume of alcohol consumption, no account was given to drinking pattern upon T2DM. The importance of such a consideration is illustrated by a recent meta-analysis of seven observational studies that reported data on both dimensions of alcohol consumption investigating the risk of ischaemic heart disease (IHD).²⁶⁸ Here, relative to never drinkers, a 36% (RR 0.64, 95% CI 0.53, 0.71) reduction in risk was documented among those that reported a moderate level of average volume of alcohol consumption (<30 g/day) and no episodic heavy drinking. By contrast, no reduction in risk was evident among those who reported the same moderate level of average volume consumption but additionally indicated episodic heavy drinking (RR 1.12, 95% CI 0.91, 1.37). Of studies selected as part of the revised meta-analysis, few had considered the effect of consumption pattern, but</p>

largely mirrored findings reported by Roerecke *et al* when researching IHD:²⁶⁸ that regular moderate alcohol consumption was associated with the lowest incidence of T2DM,^{95,97,240,244} with risks most pronounced among infrequent heavy drinkers.^{96,239,244}

Such results indicate a possibility that any putative health benefit conferred by regular exposure to moderate volumes of alcohol may be countered by infrequent episodes of heavy consumption. Unfortunately, data for IHD were not stratified by sex, with the majority of T2DM studies having only analysed men or women separately. Accordingly, it was unclear whether a volume-pattern interaction operated similarly among both men and women. Nevertheless, assuming a comparable effect in both sexes, sex-specific differences in the relationship between the volume of alcohol consumption and T2DM risk may have been attributable to sex-specific disparities in the prevalence of episodic heavy drinking.

The possibility of a difference in consumption pattern between men and women is supported by data collated from 7,193 attendees at European 172 general practices.²⁶⁹ Among participants, the odds of episodic heavy drinking (\geq 6 drinks on any one occasion at least once every month) among non-hazardous male drinkers (an AUDIT score of <8) was four times that of women following adjustment for country, age, employment status, mental and physical health (OR 4.38, 95% CI 3.27, 5.85). Similarly, of data reported by the Opinions and Lifestyle Survey, episodic heavy drinking was found to be greatest among men.¹⁰¹

Although more data are required, current evidence indicates that both the volume and pattern of alcohol consumption are likely to be important modifiers of T2DM risk. Differences in dose-response between men and women may therefore have been a consequence of differences in consumption pattern. Specifically, a greater prevalence of episodic heavy drinking among men.

3. Observed differences in dose-response between men and women may have been attributable to sex-specific disparities in the effect of alcohol upon putative biological pathways. Evidence in support of such a hypothesis was limited, however.

Looking to research concerning a possible effect of alcohol upon insulin sensitivity, a meta-analysis of 10 interventional studies identified a significant sex interaction between average volume of alcohol consumption and insulin sensitivity (p=0.018).¹¹² Specifically, insulin sensitivity was found to be higher among female but not male

drinkers, relative to controls. However, heterogeneity among the studies was high, and when a study largely responsible for such heterogeneity was removed, no significant difference in insulin sensitivity between men and women remained (p=0.180). Although women were also found to exhibit significant reductions in fasting insulin concentration, the absence of any detectable effect among men was due to a dearth of data; men were represented by just two small interventional studies that, despite low precision, each reported similar reductions in fasting insulin following alcohol consumption.

A second potential pathway concerned alcohol-mediated increases in HDL concentration. Although a linear dose-response relationship was identified across 33 interventional studies,¹¹⁸ a sex interaction was not statistically significant. Such a result was mirrored by analyses reported in a Mendelian randomisation meta-analysis, which found no alcohol-mediated change to HDL concentration and no significant difference in mediation between men and women.⁸⁵

Finally, regarding the effect of alcohol consumption upon inflammatory response, a meta-analysis of interventional studies to have researched various inflammatory biomarkers found no difference in concentration by alcohol consumption,¹¹⁸ and while a Mendelian randomisation meta-analysis of 42 longitudinal studies found concentrations of CRP to be lower among A-allele carriers whose average volume of alcohol consumption was 17% lower than non-carriers, the magnitude of the difference was small and not stratified by sex.⁸⁵

Aside from the methodological weaknesses that underpinned such studies, evidence indicating a difference in alcohol-mediated biological response between men and women was lacking.

4. A fourth possibility was that sex hormones may have played a role in altering the effect of alcohol upon T2DM risk. Associated with levels of insulin resistance and glucose concentrations independent of adiposity,^{270,271,272} at least two hormonal biomarkers have been implicated as modifiers of T2DM risk, including estradiol, a female sex hormone, and sex hormone-binding globulin (SHBG), a protein responsible for the transport of sex hormones. Of nine case-control studies to have reported concentrations of estradiol according to T2DM status, levels were significantly higher among those who developed T2DM (SMD 12.8 nmol/L, 95% CI 3.44, 22.2 nmol/L), though with no apparent interaction by sex (p=0.870).²⁷³ Of 20 case-control studies to

have reported on SHBG, concentrations were significantly lower among female cases than controls (SMD -16.23 nmol/L, 95% CI -20.24, -12.22 nmol/L), with little difference observed among men (SMD -5.07, 95% CI -11.92, 1.77).²⁷³ Effect modification of SHBG concentrations by sex were statistically significant (p=<0.001). Lower levels of circulating SHBG were also identified in cohort studies, where the risk of T2DM was substantially lower in participants with higher concentrations of SHBG, particularly among women.²⁷³

In addition a lack of understanding concerning the precise biological mechanism by which factors such as SHBG may modify T2DM risk,²⁷³ research investigating the effect of alcohol upon circulating concentrations of sex hormones is scarce, particularly how moderate volumes of alcohol consumption might specifically elicit advantageous changes in female sex hormone concentration. One cross-sectional analysis data from Women's Health Study stratified alcohol consumption according to whether or not participants consumed in excess of 25 g/day. Estradiol concentrations were elevated only within the higher consumption category, relative to pooled infrequent and nondrinkers (β =0.17, 95% CI 0.05, 0.29).²⁷⁴ While concentrations of SHBG also appeared to be higher among regular current drinkers, differences in SHBG were not statistically significant (β =0.02, 95% CI -0.11, 0.15). Elsewhere, a cross-sectional study of 202 premenopausal women reported that those who consumed ≥10 g/day had an 18% higher mean salivary estradiol concentrations than those who consumed <10 g/day (p=0.034).²⁷⁵ Similar findings have been reported by cross-sectional studies elsewhere,²⁷⁶ as well as linear increases in estradiol concentrations across categories of weekly alcohol consumption,²⁷⁷ and acute elevations in small placebo-controlled studies of female participants following dosages of 0.5 g/kg (or around 27.5 g in a 55 kg woman)²⁷⁸ and 0.7 g/kg (or around 38.5 g in a 55 kg woman).²⁷⁹

Such studies were limited by only having sampled a single sex, meaning that side-byside comparisons of alcohol-related changes to estradiol concentration were not possible. Similarly, with most studies having reported concentrations according to only a one or two volumes of alcohol consumption, a more complete understanding of the dose-response relationship was unclear. However, based on the few small studies available, there was an indication that the association between alcohol consumption and estradiol concentration may sit contrary to what is expected, with concentrations appearing elevated at volumes of alcohol consumption otherwise associated with a peak reduction in T2DM risk among women.¹⁰ The validity of beneficial alcohol-induced
changes to endogenous sex hormone concentrations having served as a pathway by which women might experience the greatest reductions in T2DM risk therefore seemed weak.

5. Finally, there was a possibility that sex-specific differences in dose-response may have been a consequence of men and women having consumed different types of alcoholic drink; that reductions in risk were less to do with the total volume of ethanol so much as some other drink-specific component. Such a possibility was plausible given a purported role of flavonoids and stilbenes commonly found in fruit-based drinks as anti-oxidants capable of impairing inflammatory response, and promoting both good blood flow and endothelial function for the effective transport of blood glucose.^{280,281}

Of the studies selected as part of the revised meta-analysis, seven reported multivariable-adjusted risk estimates according to drink type. However, in many cases, the stratification of risk estimates by drink type and categories of volume consumption led to small sub-group samples and high imprecision, weakening inferences drawn concerning any effect modification by drink type.

In one study, risks stratified by drink type were only reported for men owing to the low volume of consumption among women. Among the 5,423 male US participants, multivariable-adjusted odds of T2DM showed no independent association for any drink type.²¹⁹ Although point estimates indicate greater reductions in risk among men who drank wine at higher frequencies (e.g. OR 0.63 at >25 g/day, 95% CI 0.08-4.95), relative to non-drinkers, the estimates were subject to substantial imprecision given the small number of male wine drinkers. In a male cohort that benefitted from a much larger sample size, reductions in risk were statistically significant among moderate wine drinkers (HR 0.80 at 25 g/day, 95% CI 0.65-0.97), relative to pooled non-drinkers, with no reduction evident among beer or spirit drinkers.²⁴⁶ By contrast, a final male cohort reported independent linear reductions in risk per 15 g/day increase in consumption for beer (RR 0.80, 95% 0.70-0.91), white wine (RR 0.77, 95% Cl 0.65-0.90), and spirits (RR 0.85, 95% CI 0.76-0.94), plus some indication of a reduction in risk among red wine drinkers (RR 0.91, 95% CI 0.78-1.08).⁹⁵ However, in this instance confounder adjustment was much lower, accounting only for age and BMI. Nevertheless, in another model with greater covariate adjustment, but with additional adjustment for potential mediators of the alcohol-T2DM relationship, significant associations were unchanged.

In a US cohort study, participants were categorised according to their preferred drink type.²⁴⁵ Unfortunately, to boost sample size, data were adjusted for sex as opposed to stratified. All current drinkers showed reductions in risk relative to non-drinkers, but with no difference in risk according to whether participants consumed beer (RR 0.7, 95% CI 0.4-1.1), wine (RR 0.6, 95% CI 0.4-0.9) or spirits (RR 0.6, 95% CI 0.4-0.9). Conversely, in another cohort where sex-specific data were combined, reductions in risk were evident among wine drinkers only (HR 0.93 per g/day wine, 95% CI 0.87-0.99), with point estimates no different from the null for all other drink types.¹⁵³

Relative to never drinkers, reductions in risk among Australian female drinkers were specific to those who reported wine consumption.⁹⁷ However, none of the relationships were statistically significant due to high imprecision owing to a small number of cases within sub-groups. Imprecision was also high among men, though with reductions in risk indicated across all drink types. In particular, a statistically significant linear association trend was observed between increased wine consumption and decreased T2DM risk (p=0.037). Benefiting from a much larger number of cases per sub-group, a European multicentre prospective case–cohort study also reported significant effect modification among male wine drinkers (p=0.003), with hazards lowest at 24.1-60.0 g/day (HR 0.82, 95% CI 0.71-0.95) relative to very light drinkers (0.1-6.0 g/day).²³⁸ Little variation from the null was evident across categories of beer (p=0.660) or spirit (p=0.440) consumption in men. Among women, significant effect modification was documented for wine (p=0.048) and fortified wine (p=0.023) drinkers, with hazards lowest in each case among those who reported consuming 24.1-60.0 g/day (wine: HR 0.89, 95% CI 0.72-1.11; fortified wine: HR 0.64, 95% CI 0.24-1.70).

On balance, reductions in risk appeared most consistently reported among male and female participants who reported consuming wine, particularly in studies of greater precision.^{238,246} Although in keeping with research indicating that chemical compounds in fruit-based drinks such as red wine may have anti-inflammatory effects,^{280,281} with sexspecific reductions in T2DM thereby a result of compounds other than ethanol, a recent probabilistic study casts doubt on such a conclusion. Using dose-response data from animal models, researchers estimated the comparative risks and benefits of alcohol consumption according to the putative carcinogenicity of ethanol and the anti-carcinogenicity of the stilbene resveratrol.²⁸² Based on abstracted data, the authors estimated that a person would have to consume at least 111 glasses of wine before

reaching a therapeutic dose of resveratrol at which the risk of cancer would begin to decrease. Having an effect over 100,000 times more potent than resveratrol, the increased risk of cancer from ethanol consumption would far outweigh any anti-carcinogenicity from resveratrol.

Although equivalent analyses have yet to be published for T2DM risk, such a finding suggests that reductions in risk among wine drinkers may instead be confounded by differences in health and lifestyle according to drink preference. Although the few papers that did stratify T2DM risk by drink type were generally included a number of primary T2DM risk factors such as age, adiposity, physical activity and smoking, residual confounding was a possibility, with marked differences in the characteristics wine, beer and spirit drinkers according a wide range of socio-economic variables.^{283,284} In the absence of high precision studies able to isolate the independent effect of drink-specific compounds, a clear conclusion concerning the clinical effect of different drinks at moderate levels of alcohol consumption remains in question.

Although sex-specific differences were observed in the dose-response relationship between volume of alcohol consumption and T2DM risk, it was not possible to say with certainty which factor or combination of factors contributed to reductions in risk at moderate levels of consumption being specific to women. Based upon the available literature, the most credible determinants included a statistical artefact arising from differences in study characteristics, or else the result of biological mechanisms modified by sex-specific variations in drinking behaviour, such as consumption frequency or drink preference.

Chapter 4

A summary of current evidence

4 A summary of current evidence

DM has been linked to the development of multiple vascular morbidities,^{1,3} the management and treatment of which was estimated to have accounted for around one-tenth of the NHS budget in 2010/11, or £13.8bn.⁴ The effects of DM are not restricted to health and healthcare systems, however, with deteriorations in health having sizeable impacts on patients' ability to maintain jobs and care for themselves independently.⁴ Fortunately, of all known cases, T2DM represents by far the largest proportion, at around 85-90% of UK diagnoses.⁸ As such, the burden of DM may be largely attributable to modifiable environmental exposures, such as obesity and smoking, with the potential to attenuate disease risk in the general population through modest lifestyle changes.

A growing literature has indicated that alcohol consumption may represent one such change. Results from an updated and revised meta-analysis indicated a J-shaped dose-response relationship relative to pooled non-drinkers (Figure 3.2), with reductions in risk most pronounced among women (Figure 3.4). Although few in number, studies have also indicated that the frequency of consumption may represent an important modifier of T2DM risk, with T2DM lowest among regular moderate drinkers, and any reductions in risk potentially offset by concentrated periods of episodic heavy consumption.

To date, a number of biological mechanisms have been proposed to explain apparent reductions in T2DM risk at moderate levels of alcohol consumption, including alcohol-related improvements to insulin sensitivity, increases in the concentration of the lipid transport protein HDL, and a reduction in the expression of inflammatory factors implicated in the disruption of endothelial and pancreatic β -cell functioning. Research investigating such pathways has tended to be limited, with most existing observational studies having been cross-sectional in nature, preventing causal inference. Where studies were interventional, these were commonly constrained by small sample sizes, acute durations and single levels of exposure, undermining a determination of dose-response within each sample.

Where results from interventional studies have been reported, acute improvements in insulin sensitivity and fasting insulin have been observed,¹¹² as well as linear increases in HDL cholesterol concentrations.^{117,119} However, a beneficial relationship between alcohol consumption and HDL concentration has not been supported by Mendelian randomisation studies,⁸⁵ with such studies also reporting a null relationship between SNPs of HDL variance and T2DM risk.¹²¹ In a reversal of this apparent disparity in the findings of interventional and

Mendelian randomisation research, acute interventional trials currently report no association between alcohol consumption and inflammatory markers (though the number of such studies is small),¹¹⁸ while results from Mendelian randomisation studies indicate a J-shaped association between alcohol and CRP consistent with the dose-response between alcohol consumption and T2DM risk.¹³¹ And although a longitudinal cohort study has found associations between the concentration of CRP and T2DK risk, such an association was not identified when T2DM risk was stratified according to SNPs of variance in CRP concentration.¹³² Elsewhere, Mendelian randomisation studies have reported conflicting evidence concerning the relationship between inflammatory markers and vascular conditions.^{133,134,135} Thus, although conceivable that alcohol may indeed act upon T2DM risk through such pathways, results from existing studies are inconsistent and limited by small sample sizes.

Studies included within the updated and revised meta-analyses were subject to their own shortcomings, undermining the inference that moderate volumes of alcohol consumption may confer a reduction in T2DM risk. In all but two of the cohorts analysed by Baliunas *et al*, risk estimates among drinkers had been calculated relative to a group of pooled non-drinkers.¹⁰ Although conventional in epidemiological studies to measure the effect of an exposure relative to those who were not exposed, the use of a pooled non-drinking category has drawn criticism through the manner in which it captures a number of former drinkers whose health and lifestyle may predispose them to the development of T2DM.^{136,137} As such, apparent reductions in T2DM risk at moderate levels of consumption may have been overestimated, particularly among studies with poor adjustment for determinants of T2DM risk, such as physical activity and diet.

To explore this further, the updated and revised meta-analysis included sensitivity analyses that explicitly documented the effect of reference group and confounder adjustment upon reductions in risk observed in aggregate data. These analyses supported a significant, sex-adjusted interaction according to the choice of reference group, with reductions in risk particular to studies that calculated risks relative to pooled non-drinkers (Figure 3.5). Furthermore, after accounting for the effect of sex and reference category upon the dose-response relationship, a weak interaction was observed according to the degree of confounder adjustment, with multivariable-adjusted reductions in risk at moderate volumes of alcohol consumption appearing less pronounced than those derived from crude or age-adjusted estimates, though present across a broader range of consumption (Figure 3.8). Alongside these two factors, population region also appeared to have a sizeable effect upon dose-response, with smaller reductions in risk reported by studies that sampled participants from Asian regions (Figure 3.10).

Aside from the issue of reference group selection and confounder adjustment, a further limitation of the current literature concerned the tendency of longitudinal studies to define alcohol consumption according to a single baseline measure. Despite growing evidence that drinking may vary markedly across the life course,²⁵⁹ only one selected study was found to have explored the effect of changes in alcohol consumption upon the risk of T2DM.¹⁵³ Of studies not captured by the revised and updated meta-analysis, only one additional publication was found to give any consideration to the variability of alcohol intake over time.²⁶⁷

Results from the two available studies hint at a dose-response relationship more complex than captured by conventional analytical approaches. The first reported attenuated and non-significant reductions in T2DM risk after accounting for longitudinal changes to the volume of alcohol consumption among sampled participanst,¹⁵³ while the second reported significant differences in T2DM according to the longitudinal trajectory of alcohol consumption.²⁶⁷ For instance, of moderate drinkers (5.0-29.9 g/day) that reduced their consumption to none or light drinking (0-4.9 g/day), a 9% increase in hazards was observed (HR 1.09, 95% CI 0.92-1.30) on average over each four-year period of follow-up, indicating a possible sick quitter effect. Although limited, such analyses suggest that the categorisation of participants according to their consumption at baseline risked introducing some degree of misclassification error.¹⁵⁰ Accordingly, future research should apply analytical approaches capable of describing and modelling longitudinal changes to alcohol consumption and their association with T2DM risk.

Chapter 5

Research aims

5 Research aims

Although results from the updated and revised meta-analysis suggest the presence of reductions in T2DM risk among women at moderate volumes of alcohol intake, with T2DM risk elevated even at low volumes among men, effect estimates abstracted from constituent studies were calculated according to a single baseline measure. Despite apparent changes in alcohol consumption behaviour across the adult life course,²⁵⁹ very few studies appear to have investigated the impact of such variation upon a person's risk of T2DM. In the few instances where T2DM risk has been modelled as a function of longitudinal changes to alcohol consumption, results indicate a relationship more complex than captured by conventional analytical approaches,^{153,267} supporting recent calls for alcohol studies to give a better account of variations in alcohol consumption across the life course.^{266,285,286,287} With this in mind, a series of analyses will be undertaken to help better understand the longitudinal dynamics of the dose-response relationship, beginning with a preliminary survival analysis designed to confirm the suitability of the chosen dataset for the analysis of dose-response relationships between alcohol consumption and T2DM.

Aim 1: Establish whether baseline alcohol consumption is associated with T2DM risk (Chapter 7)

- Undertake a conventional multivariable-adjusted survival analysis, reporting the doseresponse relationship between baseline categories of average weekly volume of alcohol consumption and T2DM risk.
- Investigate whether the volume of alcohol consumption alone was sufficient for modelling T2DM risk by testing an interaction between the volume and frequency of alcohol consumption.
- Compare results from the conventional survival analyses against those summarised as part of the revised meta-analysis in Chapter 3.

Aim 2: Determine the degree to which baseline categories of alcohol consumption risked being subject to misclassification error as a result of longitudinal changes in alcohol consumption (Chapter 8)

• Establish whether sex-specific trajectories of the mean weekly volume of alcohol consumption were constant over the captured life course.

• Establish the longitudinal stability of alcohol intake within categories defined by baseline consumption.

Aim 3: Describe differences in the mean volume of alcohol consumption over the life course according to T2DM diagnosis (Chapter 8)

- Determine whether trajectories of alcohol consumption differed according to whether or not participants developed T2DM.
- Describe the nature of any differences in the mean weekly volume of alcohol consumption over the period leading up to the time of censoring or T2DM diagnosis.

Aim 4: Formally explore the utility of more advanced survival models in developing a better understanding of the relationship between alcohol consumption and T2DM (Chapter 9)

- Conduct a series of increasingly advanced survival models to estimate the sex-specific risk of T2DM as a function of age-varying alcohol consumption.
- Compare dose-response relationships reported according to different parameterisations of the longitudinal trajectory.
- Investigate the effect of adjustment for heterogeneous non-drinking groups upon the sex-specific dose-response relationship.
- Establish whether declining trajectories represent a group of drinkers at elevated risk of T2DM.

With these aims established, Chapter 6 details the selection and structure of a dataset considered suitable for exploring longitudinal trajectories of alcohol consumption and their relationship with T2DM.

6 Data selection and structure

6.1 Introduction

To investigate the sensitivity of conventional survival analyses to longitudinal changes in average volume of alcohol consumption and drinking patterns, it was necessary to identify a dataset with a number of fundamental characteristics.

Given that alcohol consumption behaviours appeared to change across all periods of adulthood,^{264,265} it was important first of all to analyse a dataset that spanned as broad a period of the adult life course as possible, with repeated measures of alcohol consumption that were both comparable between waves and of sufficient frequency as to capture acute fluctuations in drinking over time. However, while many longitudinal datasets were found to include repeated measures of alcohol consumption, none detailed the full adult life course, each having documented a restricted life period, such as early adulthood²⁸⁸ or older age.²⁸⁹

Of the numerous longitudinal cohorts available, Whitehall II provided the best compromise between the period of adulthood captured, frequency of repeated alcohol measures, sample size and availability of variables, including objectively defined measures of T2DM. For instance, while the English Longitudinal Study of Ageing benefits from regular repeated measures and a sample of similar size to Whitehall II, its coverage of the adult lifecourse is skewed toward later life and comprises just 14 years of follow-up relative to the three decades of data afforded by Whitehall II.²⁹⁰ Elsewhere, in terms of coverage, the MRC National Survey of Health and Development 1946 is perhaps one of the closest to Whitehall II, with measures of alcohol consumption available between the ages of 36 and 64 years. However, it is limited by self-reported T2DM data, fewer repeated measures and a sample one third the size of that available in Whitehall II.²⁹¹ Other cohorts, such as the Twenty-07 Study²⁹² and the Caerphilly Prospective Study,²⁹³ benefit from a similar number of repeated measures to Whitehall II but with very small sample sizes.

6.2 Whitehall II cohort profile

The Whitehall II cohort was established in 1985 as a longitudinal occupational study and comprised 10,308 (6,895 male and 3,413 female) civil servants aged 35-55 years who worked in the offices of 20 Whitehall departments.²⁹⁴ All eligible employees were invited to participate via letter, of whom 73% consented.²⁹⁵ Baseline measurements were obtained between 1985 and 1988 by way of a self-administered questionnaire and a clinical examination. Participants were

followed up at regular intervals of between a median 1.6 and 5.5 years, with self-administered questionnaires completed at each wave and a clinical examination undertaken once every two waves (Table 6.1).

Although Whitehall II represents a geographically-concentrated and occupationally-narrow cohort, a recent paper confirmed that aetiological associations identified within the cohort are consistent with those reported from studies of general population samples, implying that aetiological analyses based on Whitehall II data are likely to offer sound external validity.²⁹⁶

Age range	34-56	37-60	40-64	42-66	45-69	48-71	50-74	54-77	55-80	60-84
Wave number	1	2	3	4	5	6	7	8	9	11
Alcohol consumption										
Frequency										
Volume										
'Always non-drinker?'										
<u>T2DM</u>										
FPG ^a										
HbA1c ^b										
OGTT ^c										
Self-reported										
<u>Covariates</u>										
Age										
BMI ^d										
Diet ^e										
Education										
Ethnicity										
Employment status										
Family history of T2DM										
Government office region										
Household income										
Index of Multiple Deprivation										
Occupational grade										
Physical activity										
Sex										
Smoking status										

Table 6.1 Variable availabilit	y in Whitehall II by study wave
--------------------------------	---------------------------------

^aFasting plasma glucose; ^bGlycated hemoglobin A1c; ^cOral glucose tolerance test; ^dBody mass index; ^eDietary variables, including fibre and saturated fat.

The main aim of wave 10 was to validate self-completed measures of psychiatric morbidity in older people. As only 337 participants were sampled for the validation exercise, data for wave 10 are not listed.

6.3 Defining alcohol consumption

As shown in Table 6.1, estimates of average weekly alcohol consumption were available in waves one, two, three, five, seven, nine and 11. Participants were asked to report the number of alcoholic drinks they had consumed in the preceding week according to "measures" of spirits, "glasses" of wine, or "pints" of beer or cider. A standard measure of spirits and a glass of wine were each assumed to contain one unit of alcohol, and a pint of beer or cider two units of alcohol. Based on data provided by Drinkaware,¹¹ these volumes were roughly equivalent to a small (125 ml) glass of 10% ABV white wine (1.2 units) and a 4% ABV pint of lager (2.3 units).

Given recent increases to both the size of wine glasses (175 ml and 250 ml) and the average strength of wines (11.5-13.5% ABV),²⁹⁷ it was likely that alcohol consumption from wine is conservatively estimated in Whitehall II.²⁹⁸ This was supported by a number of small convenience studies that asked drinkers to pour self-defined usual glasses of various drink types. Among these, volumes of alcohol contained within a usual glass of wine were consistently greater than those assumed by Whitehall II, with mean alcohol content measuring 15 g of alcohol among 283 drinkers from six locations across South-East England,²⁹⁹ or around double the volume assumed in Whitehall II. Elsewhere, almost half of participants from a Scottish convenience study poured in excess of 16 g alcohol.³⁰⁰

While wine consumption data could have been inflated retrospectively, any such alteration risked being arbitrary and simplistic given variations in underestimation between convenience studies according to factors such as age, sex and setting.²⁹⁹ In a recent study that investigated the impact of increasing the assumed alcohol content of wine from 8 g/glass to 16 g/glass, doubling the value of wine consumption had the effect of reducing the risk of all-cause, CVD and cancer mortality among very heavy drinkers (men: \geq 408 g/week, women: \geq 288 g/week), relative to moderate drinkers (men: 1-168 g/week; women: 1-112 g/week).³⁰⁰ Risks among former and pooled non-drinkers were attenuated, though not to a degree as to be statistically significant. Accordingly, any underestimation of wine consumption would not have influenced the direction of the dose-response association under study, but may lead to an overestimation of risk at higher volumes. With this in mind, the volume of alcohol consumption was left unchanged with the caveat that risks among very heavy, former and non-drinkers were likely to have been biased upward among participants who predominantly consumed wine.

The total volume of alcohol consumption was reported within the Whitehall II dataset as average units/week. These data were transformed into grams/week assuming a UK unit equal to 7.9 g ethanol, consistent with conversions undertaken as part of the revised meta-analysis. At wave

one, median reported alcohol consumption was 63.2 g/week (IQR 23.7-134.3 g/week, n=6,840) and 23.7 g/week (IQR 0.0-55.3 g/week, n=3,374) among men and women respectively (Table 6.2). Median weekly alcohol consumption varied over the course of the study, peaking among men and women at wave five (ages 45-69).

Whitehall II also included data concerning the pattern of alcohol consumption, operationalised as the frequency of alcohol consumption over the last 12 months. In wave 11, frequencies were defined as: 'never', '≤1 occasion/month', '2-4 occasions/month', '2-3 occasions/week' and '>3 occasions/week'. In all other waves, frequencies were recorded as: 'none', 'special occasions', '>1/month', '>1/week', 'daily' and 'almost daily'. To maintain consistency across waves in the classification of consumption frequency, new variables were derived for each wave of data and categories defined as: 'none', '<1 occasion/week', '1-3 occasions/week', 'daily or almost daily'. These derivations were in keeping with a prior publication that used alcohol consumption frequency data from Whitehall II.²⁶⁶

6.4 Defining T2DM

Self-reported T2DM was documented at all waves, defined as any self-reported doctor-diagnosis or self-reported prescription of anti-diabetic medication (Table 6.1). Given that close to onethird of T2DM cases may be missed by subjective measures of T2DM,³⁰¹ self-reports were supplemented by objective data. These were obtained using measures of blood glucose drawn during clinical examinations at waves three (1991-93, ages 39-64), five (1997-99, ages 45-69), seven (2003-04, ages 50-74), nine (2008-09, ages 55-79) and 11 (2012-2013, ages 61-85) following a minimum five-hour fast. At each clinical examination, cases of T2DM were defined according to the 1998 WHO criteria: a fasting plasma glucose (FPG) reading \geq 7.0mmol/L.³⁰² Data following an oral glucose tolerance test (OGTT) were also available, for which participants consumed 389 ml of Lucozade (75 g anhydrous glucose) over a five-minute period, with a blood sample taken two hours later. However, as shown in Table 6.1, this measure was unavailable in wave 11. Similarly, while HbA1c concentrations were also documented as an alternative objective diagnostic indicator, such data were only available from wave seven onwards. To maintain a consistent definition of T2DM over time, FPG was thus selected as the objective measure of choice alongside self-reported data.

To ensure a baseline free of T2DM, known cases between waves one through three were excluded from any analyses involving T2DM. After wave three, a total 916 new cases were observed (men: 620; women: 296), with a median follow-up 20.1 years (IQR 14.8-20.6 years) among men and 19.8 years (IQR 12.8-20.4 years) among women (Table 6.2).

6.5 Confounding factors

All confounding factors identified during the literature review were available in Whitehall II, including adiposity, dietary variables (fats, carbohydrates and fibre), ethnicity, physical activity, smoking status and family history of T2DM (Table 6.1).

6.5.1 Adiposity

Despite its strong association with T2DM risk,³⁹ there was a possibility that adiposity may have operated as a mediator of the relationship between volume of alcohol consumption and T2DM risk, with evidence indicating that calories consumed via alcoholic drinks tended to be additional to those derived from other dietary sources such as evening meals, leading to an acute overconsumption of calories relative to non-drinkers.³⁰³ However, research linking average volume of alcohol consumption to increases in adiposity itself is conflicting, with longitudinal studies having reported a variety of null, inverse and positive associations.^{303,304} Notably, where positive associations have been identified, absolute differences in adiposity across alcohol consumption categories were modest. For example, over a period of three-years, each 14 g increase in alcohol consumption per drinking occasion was associated with an increase in BMI of 0.03 kg/m² (0.02, 0.04 kg/m²) among men, or a rise of 0.03 kg/m² (95% CI -0.01, 0.07 kg/m²) for each additional day of weekly consumption.³⁰⁵ Coefficients among women were even smaller and nonsignificant. Elsewhere, in an analysis of three large occupational cohorts, each 14 g increase in the volume of consumption per drinking occasion was associated with an increase in weight of 0.19 kg (95% CI 0.10, 0.27 kg) over a period of four years, or just 0.05 kg/annum.³⁰⁶ Similarly small increases in weight have been identified elsewhere.³⁰⁷

On balance, given the body of available evidence, it was decided that while calories derived from alcoholic drinks may have some direct effect upon adiposity, the magnitude of such an association appears diminutive relative to other risk factors for adiposity, such as physical inactivity and poor diet. Accordingly, adiposity was included as a confounding factor in all multivariable-adjusted models, with negligible risk of overadjustment.

Multiple measures of adiposity are available in Whitehall II, including BMI and waist circumference. Each has been strongly associated with an increased risk of T2DM, with risk estimates of comparable magnitudes and the predictive utility of each measure differing according to a range of participant characteristics.³⁹ BMI was selected as a well-known indicator of adiposity. Data on participants' height and weight were captured at each clinical examination. Mean BMI measured 24.6 kg/m² (95% CI 23.3, 25.8 kg/m², n=6,883) among men and 24.8 kg/m²

(95% CI 23.1, 26.5 kg/m², n=3,441) among women at wave one, with mean BMI appearing to increase gradually at each successive wave (Table 6.2).

6.5.2 Diet

A number of dietary factors were considered in accordance with the literature review, including carbohydrates, fibre and dietary fats (e.g. polyunsaturated and trans fats). The consumption of such factors was estimated based on responses to an anglicised food frequency questionnaire (FFQ) based on that used in the US Nurses' Health study.³⁰⁸ Foods commonly eaten in the UK were added to the FFQ as per the UK arm of the EPIC cohort study.³⁰⁹ A common portion size was assigned to each food, and participants then asked how often, on average, they consumed such a portion during the previous year. Frequencies ranged from 'never or less than once per month' to 'six or more times per day'. The reported frequency for each food item was then converted to an estimated daily intake, with the consumption of constituent nutrients computed by multiplying the daily frequency of consumption by the estimated nutrient content of each portion.³¹⁰

Unfortunately, dietary variables were only available in waves three, five and seven of the Whitehall II cohort. In order to explore the merit of truncating the number of repeated measures of alcohol consumption in favour of greater covariate adjustment, the contribution of dietary factors upon the alcohol-T2DM relationship was explored alongside other covariates of interest by way of multiple sex-stratified Cox survival models. The method adopted was consistent with studies selected as part of the revised meta-analysis, with all covariates modelled only according to their values at baseline. The sample was restricted to T2DM-free participants at wave three, with T2DM defined according to a positive FPG, self-reported doctor diagnosis or use of hypoglycaemic medication. Dietary variables were treated as continuous variables that denoted their proportion of total energy intake (g/100g). With the exception of fibre, all dietary variables were subjected to a logarithmic transformation owing to skewness. In keeping with conventional survival models, categorical average weekly volume of alcohol consumption was selected as the exposure of interest.

The reference model was adjusted for all *a priori* covariates (age, BMI, ethnicity, family history of T2DM, physical activity, sex, smoking and socio-economic status), excluding any dietary variables. Dietary variables were then added in a mutually exclusive manner, with a final model adjusted for all covariates and dietary variables. Results from these analyses identified a statistically significant inverse association between fibre consumption and T2DM risk among men, independent of adiposity (p=0.018), while all other dietary variables were far from

significant in both sexes (Appendix 6.1). In all models, dietary variables appeared to have little impact upon the relationship between alcohol consumption and T2DM risk. The relationship was similarly unchanged when alcohol consumption was modelled as a continuous variable, or when dietary variables were coded as tertiles.

Such a series of results indicated that other covariates, particularly BMI, may have captured much of the effect of diet upon T2DM risk. Dietary factors were thus omitted as covariates from any subsequent multivariable-adjusted model.

6.5.3 Ethnicity

The ethnic group of participants was self-reported during screening at waves one and five. The ethnicity of participants was coded as 'white', 'South Asian' or 'other', which isolated into a separate category any Asian participants who were likely to have a heightened genetic sensitivity to alcohol²⁰³ and susceptibility to the development of T2DM.²⁰⁴ At baseline, 92.0% (95% CI 91.3-92.6) of participants were of a white ethnic background (Table 6.2).

6.5.4 Family history of T2DM

Family history of diabetes was self-reported by participants in waves one and two. Data were coded as a dichotomous variable according to whether a participant reported a parent or sibling as having developed T2DM. The variable was thus limited by failing to capture familial cases that developed after the first two waves of study. Among those who participated at baseline, 10.4% (95% CI 9.7-11.2%, n=709) of men and 14.1% (95% CI 12.9-15.3%, n=470) of women reported a family history of T2DM (Table 6.2).

6.5.5 Physical activity

Information regarding physical activity was ascertained via a 20-item questionnaire that included questions on the frequency and duration of participation in activities including walking, cycling, housework, gardening and sports in the preceding four weeks. These responses were used to derive the average number of hours spent doing each activity per week. Activities were classified as either mildly (e.g. gardening and housework), moderately (e.g. walking), or vigorously (e.g. running and swimming) energetic, relative to laying at rest.³¹¹ Time spent in mildly energetic physical activity was discounted as it predominantly captured women, potentially leading to a small sub-group in any adjusted analyses stratified by sex. This was consistent with a previous analysis using Whitehall II physical activity data.³¹²

A categorical physical activity variable was created, with participants classified according to WHO physical activity recommendations:³¹³ meeting guidelines (≥150 minutes of moderate-

intensity exercise per week, or \geq 75 minutes of vigorous-intensity activity); inactive (<60 minutes of moderate physical activity and <60 minutes of vigorous physical activity; below guidelines (anyone not inactive or meeting the WHO guidelines).

The proportion of participants who reported meeting WHO physical activity guidelines was greatest among men and women at wave five (ages 45-69), at 91.0% (95% CI 90.1-91.8%, n=4,234) and 79.6% (95% CI 77.8-81.4%, n=1,576) respectively (Table 6.2). This proportion then fell with increasing age up to the final wave of data, at 54.3% (95% CI 52.8-55.8%, n=2,389) among men and 36.6% (95% CI 34.4-38.8%, n=661) among women by wave 11.

6.5.6 Smoking status

Smoking data were collected at each wave and participants categorised according to whether they reported being a current, former or never smoker. The proportion of current smokers was greatest at wave one, at 15.9% (95% Cl 15.1-16.8%, n=1,091) of men and 23.4% (95% Cl 22.0-24.9%, n=795) of women (Table 6.2). The proportion of current smokers declined as the cohort grew older, falling to 3.4% (95% Cl 2.9-3.9%, n=141) of men and 4.8% (95% Cl 3.8-5.9%, n=80) of women by wave 11.

6.5.7 Socio-economic status

Given the long history of research highlighting a social gradient in health inequalities, socioeconomic variables were considered as a means of capturing some degree of confounding not captured by other *a priori* selected variables. Such a decision was supported by research that indicated a social gradient in abstention and heavy drinking,^{314,315} which were disproportionally concentrated among those of low education or in unskilled occupations. Elsewhere, heavy drinking has also been associated with being unemployed or materially deprived,^{316,317} or living in deprived neighbourhoods.³¹⁸ Similarly, the risk of T2DM also appears greatest among study participants of low income,^{319,320} education,³²⁰ socio-economic status,³²¹ or area deprivation.³²²

Although income, education, government office region (GOR) and Index of Multiple Deprivation (IMD) score were all available as potential socio-economic variables, data were present on each for only a sub-set of the baseline sample: GOR and IMD data were only available for 33.0% (n=3,398) of participants at wave one, income for 69.5% (n=7,161), and education for 74.5% (n=7,681). In complete-case analyses, such a high proportion of missing data would have led to a substantial reduction in sample size.

Two alternative surrogate indicators of socio-economic status were thus selected: last known hierarchical occupational grade, based on salary and work role (administrative,

professional/executive, clerical/support), and employment status (employed, retired, redundant/sacked/sick/other). Where individuals lacked an occupational grade in waves after baseline measurement, such as due to having left the civil service through retirement, redundancy or long-term sickness, the last known occupational grade was used. Over 90.7% of male participants at wave one were among the top two tertiles of occupational grade, compared with 50.3% of female participants. The proportion of retired individuals rose at each successive wave to 77.7% (95% CI 76.5-78%, n=5,018) of men and 84.2% (95% CI 82.4-85.8%, n=1,554) of women by wave 11 (Table 6.2). To assess whether the inclusion of education, income, GOR and IMD would have helped capture additional confounding by socio-economic status, correlations were calculated between these factors and both employment status and occupational grade at wave three. The correlation between occupational grade and employment status was low (r=-0.02), indicating that each was likely to describe different dimensions of socio-economic status. Weak to moderate correlations were found between occupational grade and income, IMD and education, with Pearson's r ranging from 0.29-0.53, suggesting that such factors may have had some moderate additional predictive value in adjusted analyses. The weakest correlation was present between occupational grade and GOR (r=-0.08), indicating that area of residence may be an important socio-economic covariate despite the geographically concentrated nature of the cohort. Some degree of residual confounding within adjusted analyses was thus plausible.

6.5.8 Variables not included

6.5.8.1 Direct mediators

Some selected studies reported models that also adjusted for factors such as serum insulin, glucose or triglyceride concentration. However, these factors were likely to exist on the causal pathway (see Section 2.2.2). By adjusting for a factor on the causal pathway between alcohol consumption and T2DM risk, the primary dose-response relationship under investigation will be biased toward the null.⁴⁷ This is illustrated in Figure 6.1, where E indicates the exposure of interest (weekly volume of alcohol consumed), M the mediating mechanisms (insulin sensitivity, cholesterol, inflammation), and D the dependent event of interest (T2DM).

 $E \longrightarrow M \longrightarrow D$

Figure 6.1 Directed acyclic graph illustrating direct overadjustment bias

6.5.8.2 Indirect mediators

Although self-rated general health was considered as a variable that may have helped adjust for confounding through an uneven distribution of poor health across alcohol consumption categories, and thereby potential differences in susceptibility to the development of T2DM, its inclusion was ruled out due to the possibility that it may also have served as a surrogate for mediating factors through which alcohol was hypothesised to modify T2DM risk. For instance, were moderate drinkers more likely to report good self-rated general health relative to non-drinkers, this may have been attributable in part to (a) advantageous differences in alcohol-induced insulin sensitivity, HDL cholesterol concentration and inflammation, and (b) a correspondingly lower burden of negative health conditions as a consequence of such a favourable metabolic profile, such as lower rates of stroke^{323,324,325} and CHD.^{326,327,328} The hypothesised role of self-rated general health as a surrogate mediator is illustrated in Figure 6.2, where U represents omitted mediators (inflammation, cholesterol and insulin sensitivity) and M the observed surrogate mediator (self-rated general health).



Figure 6.2 Directed acyclic graph illustrating indirect overadjustment bias

The exclusion of self-reported health as a covariate was supported by its absence as a confounding variable in models reported by selected studies, with preference given to lifestyle and demographic variables. Unfortunately, the degree to which a general health variable would have served as either a confounder or mediator could not be investigated statistically, each distinguishable only on a conceptual basis.³²⁹

I			Wave of	study		
l	1	m	S	7	6	11
	% (95% CI)					
Variable	c	L	c	c	r	L
<u>Men</u> Participation (%)						
Participated	100.0	87.8	79.3	71.0	0.09	64.7
	6,895	6,057	5,473	4,893	4,759	4,459
Non-response		11.0	17.7	23.4	22.0	22.0
		757	1,218	1,613	1,515	1,521
Died		1.2	3.0	5.6	9.0	13.3
		81	204	389	621	915
Age						
Mean years	44.5 (44.4-44.7) ^a	$49.9 (49.8-50.1)^{a}$	55.7 (55.5-55.9) ^a	$61.1 (60.9-61.2)^{a}$	65.8 (65.7-66.0) ^a	69.7 (69.5-69.9) ^a
	6,895	6,057	5,473	4,893	4,759	4,459
Alcohol consumption frequency (%)						
None in past year	3.2 (2.8-3.6)	3.6 (3.2-4.2)	3.4 (2.9-3.9)	3.6 (3.1-4.1)	4.6 (4.0-5.2)	2.0 (1.6-2.5)
	220	209	168	171	213	83
<1/week	20.5 (19.5-21.4)	17.0 (16.1-18.0)	13.7 (12.8-14.7)	13.5 (12.6-14.5)	13.9 (12.9-14.9)	25.9 (24.6-27.3)
	1,407	277	682	645	644	1,083
1-3 times/week	42.5 (41.4-43.7)	39.4 (38.1-40.6)	33.0 (31.7-34.3)	30.5 (29.2-31.8)	30.1 (28.8-31.5)	26.7 (25.4-28.1)
	2,923	2,259	1,640	1,458	1,396	1,117
Daily or almost daily	33.8 (32.7-34.9)	40.0 (38.7-41.3)	49.9 (48.6-51.3)	52.4 (51.0-53.8)	51.4 (49.9-52.8)	45.4 (43.9-46.9)
	2,325	2,295	2,485	2,506	2,379	1,898
Alcohol consumption volume						
Median g/week	63.2 (23.7-134.3) ^b	63.2 (23.7-142.2) ^b	94.8 (31.6-181.7) ^b	79.0 (31.6-158.0) ^b	71.1 (23.7-142.2) ^b	63.2 (23.7-126.4) ^b
	6,840	5,734	4,984	4,765	4,619	4,400
BMI						
Mean kg/m²	24.6 (24.5-24.6) ^a 6,883	25.1 (25.1-25.2) ^a 5,591	26.0 (25.9-26.1) ^a 3,996	26.6 (26.5-26.7) ^a 4,569	26.6 (26.5-26.7) ^a 4,424	26.6 (26.4-26.7) ^a 4,039

Table 6.2 Descriptive summary of Whitehall II data, stratified by wave and sex

Employment status (%)						
Employed	100.0	90.7 (90.0-91.4)	67.3 (66.0-68.5)	52.9 (51.5-54.3)	33.1 (31.8-34.5)	19.4 (18.2-20.6)
	6,895	7,981	5,001	3,413	2,039	1,101
Retired		7.4 (6.7-8.0)	26.4 (25.3-27.6)	42.1 (40.8-43.5)	63.0 (61.7-64.4)	77.7 (76.5-78.9)
		598	2,214	3,123	4,416	5,018
Other ^c		1.9 (1.6-2.3)	6.3 (5.7-7.0)	5.0 (4.4-5.6)	3.8 (3.3-4.4)	2.9 (2.5-3.5)
		236	606	403	295	185
Ethnicty (%)						
White	92.0 (91.3-92.6)	92.8 (92.1-93.4)	93.5 (92.8-94.1)	94.0 (93.3-94.6)	94.2 (93.5-94.8)	94.6 (93.9-95.2)
	6,306	5,602	5,116	4,592	4,472	4,211
South Asian	5.3 (4.8-5.9)	4.9 (4.4-5.5)	4.4 (3.9-5.0)	4.2 (3.7-4.8)	4.0 (3.5-4.6)	3.7 (3.2-4.3)
	366	297	243	207	191	166
Other ^d	2.7 (2.3-3.1)	2.3 (2.0-2.7)	2.0 (1.7-2.5)	1.8 (1.4-2.2)	1.8 (1.5-2.2)	1.7 (1.4-2.1)
	183	140	112	87	86	76
Family history of T2DM (%)						
Yes	10.4 (9.7-11.2)	10.5 (9.8-11.4)	10.3 (9.5-11.2)	10.1 (9.3-11.0)	9.9 (9.1-10.8)	9.8 (9.0-10.7)
	209	630	557	486	464	432
No	89.6 (88.8-90.3)	89.5 (88.6-90.2)	89.7 (88.8-90.5)	(20.0-0.68) (20.7)	90.1 (89.2-90.9)	90.2 (89.2-91.0)
	6,078	5,342	4,836	4,336	4,232	3,972
Incident diabetes ^c (%)						
Yes	0.9 (0.7-1.2)	1.8 (1.4-2.1)	2.5 (2.1-3.0)	3.8 (3.3-4.4)	6.1 (5.4-6.9)	0.9 (0.7-1.2)
	63	108	129	177	276	38
No	99.1 (98.8-99.3)	98.2 (97.9-98.6)	97.5 (97.1-97.9)	96.2 (95.6-96.7)	93.9 (93.2-94.6)	99.1 (98.8-99.3)
	6,401	5,894	5,081	4,499	4,246	4,205
Occupational grade (%)						
Administrative (top)	38.4 (37.2-39.5)	48.0 (46.8-49.3)	50.9 (49.6-52.2)	53.5 (52.1-54.9)	54.4 (53.0-55.9)	13.9 (12.9-14.9)
	2,647	2,909	2,785	2,617	2,591	619
Professional (middle)	52.3 (51.1-53.5)	45.1 (43.9-46.4)	43.0 (41.7-44.3)	41.7 (40.3-43.0)	40.7 (39.4-42.1)	48.4 (47.0-49.9)
	3,607	2,733	2,352	2,038	1,939	2,160
Clerical (bottom)	9.3 (8.6-10.0)	6.9 (6.2-7.5)	6.1 (5.5-6.8)	4.9 (4.3-5.5)	4.8 (4.2-5.5)	37.7 (36.3-39.1)
	641	415	336	238	229	1.680

Physical activity ^e (%)						
Inactive	9.5 (8.8-10.2)	14.4 (13.5-15.3)	3.8 (3.3-4.4)	5.6 (5.0-6.3)	23.7 (22.5-25.0)	28.7 (27.4-30.1)
	645	826	177	267	1,110	1,264
Below guidelines	37.3 (36.1-38.4)	36.4 (35.1-27.6)	5.2 (4.6-5.9)	7.5 (6.8-8.3)	18.3 (17.2-19.4)	17.0 (15.9-18.1)
	2,544	2,087	243	357	854	746
Met guidelines	53.3 (52.1-54.5)	49.2 (47.9-50.5)	91.0 (90.1-91.8)	86.9 (85.9-87.8)	58.0 (56.6-59.4)	54.3 (52.8-55.8)
	3,636	2,826	4,234	4,138	2,715	2,389
Smoking (%)						
Never	47.7 (46.5-48.9)	44.7 (43.3-46.0)	46.8 (45.5-48.2)	45.3 (43.9-46.7)	44.0 (42.6-45.5)	42.5 (41.0-44.0)
	3,264	2,418	2,369	2,192	1,972	1,783
Former	36.3 (35.2-37.5)	42.1 (40.8-43.5)	43.6 (42.2-45.0)	47.1 (45.7-48.5)	49.9 (48.4-51.4)	54.2 (52.7-55.7)
	2,486	2,282	2,204	2,278	2,235	2,274
Current	15.9 (15.1-16.8)	13.2 (12.3-14.1)	9.6 (8.8-10.4)	7.6 (6.9-8.4)	6.1 (5.4-6.8)	3.4 (2.9-3.9)
	1,091	715	484	370	273	141
Women						
Participation (%)						
Participated	100.0	80.8	70.2	60.8	58.7	54.2
	3,413	2,758	2,397	2,074	2,002	1,849
Non-response	ı	17.9	26.8	33.5	31.6	31.2
		611	914	1,144	1,078	1,065
Died		1.3	3.0	5.7	9.8	14.6
		44	102	195	333	499
Age						
Mean years	45.8 (45.6-46.0) ^a	51.1 (50.8-51.3) ^a	56.6 (56.3-56.8) ^a	61.7 (61.4-61.9) ^a	66.4 (66.2-66.7) ^a	70.1 (69.9-70.4) ^a
	3,413	2,758	2,397	2,074	2,002	1,849
Alcohol consumption frequency (%)						
None in past year	6.4 (5.6-7.2)	7.3 (6.4-8.4)	6.8 (5.8-7.9)	9.2 (8.0-10.5)	11.8 (10.4-13.3)	7.6 (6.4-9.0)
	217	189	141	183	224	124
<1/week	39.5 (37.9-41.2)	35.9 (34.0-37.7)	31.0 (29.0-33.0)	32.4 (30.4-34.5)	31.4 (29.3-33.5)	43.4 (41.0-45.8)
	1,345	925	644	648	598	708
1-3 times/week	34.3 (32.7-35.9)	34.8 (33.0-36.6)	30.6 (28.7-32.6)	27.9 (26.0-29.9)	27.0 (25.1-29.1)	21.7 (19.8-23.8)
	1,167	897	637	557	515	354
Daily or almost daily	19.8 (18.5-21.2)	22.0 (20.5-23.7)	31.6 (29.7-33.7)	30.5 (28.5-32.6)	29.9 (27.8-31.9)	27.3 (25.2-29.5)
	674	568	658	609	569	446

Alcohol consumption volume						
Median g/week	23.7 (0.0-55.3) ^b 3,374	23.7 (0.0-55.3) ^b 2,578	31.6 (0.0-79.0) ^b 2,098	25.7 (0.0-79.0) ^b 1,986	23.7 (0.0-63.2) ^b 1,902	15.8 (0.0-55.3) ^b 1,827
BMI						
Mean kg/m ²	24.8 (24.6-24.9) ^a	25.7 (25.5-25.9) ^a	26.4 (26.2-26.7) ^a	27.2 (27.0-27.5) ^a	27.3 (27.1-27.6) ^a	27.2 (26.9-27.5) ^a
	3,411	2,483	1,685	1,881	1,769	1,576
Employment status (%)						
Employed	100.0	90.1 (88.9-91.2)	56.3 (54.3-58.3)	40.5 (38.4-42.6)	23.3 (21.5-25.2)	12.8 (11.4-14.4)
	6,895	2,485	1,342	836	465	237
Retired	ı	5.5 (4.7-6.4)	32.6 (30.8-34.6)	51.8 (49.6-53.9)	71.1 (69.0-73.0)	84.2 (82.4-85.8)
		152	778	1,069	1,420	1,554
Other ^c	ı	4.4 (3.7-5.2)	11.0 (9.8-12.4)	7.7 (6.7-9.0)	5.7 (4.7-6.8)	3.0 (2.3-3.9)
		121	263	160	113	55
Ethnicty (%)						
White	85.5 (84.3-86.7)	86.0 (84.6-87.3)	86.6 (85.2-87.9)	87.1 (85.6-88.5)	87.5 (85.9-88.9)	87.9 (86.3-89.3)
	2,875	2,353	2,070	1,801	1,746	1,622
South Asian	6.6 (5.8-7.5)	6.1 (5.2-7.0)	6.2 (5.3-7.2)	6.1 (5.2-7.3)	6.0 (5.0-7.1)	5.7 (4.8-6.9)
	222	166	147	127	120	106
Other ^d	7.9 (7.0-8.8)	7.9 (7.0-9.0)	7.2 (6.3-8.3)	6.8 (5.8-7.9)	6.5 (5.5-7.7)	6.3 (5.3-7.5)
	264	217	173	140	130	117
Family history of T2DM (%)						
Yes	14.1 (12.9-15.3)	13.9 (12.6-15.2)	13.7 (12.3-15.1)	13.8 (12.3-15.3)	13.6 (12.2-15.2)	13.3 (11.8-14.9)
	470	375	322	1,756	1,697	241
No	85.9 (84.7-87.1)	86.1 (84.8-87.4)	86.3 (84.9-87.7)	86.2 (84.7-87.7)	86.4 (84.8-87.8)	86.7 (85.1-88.2)
	2,874	2,329	2,032	280	268	1,573
lncident diabetes ^c (%)						
Yes	1.1 (0.8-1.5)	1.5 (1.1-2.0)	3.0 (2.4-3.8)	4.2 (3.4-5.2)	7.1 (6.0-8.3)	0.6 (0.3-1.1)
	37	42	68	84	134	10
No	98.9 (98.5-99.2)	98.5 (98.0-98.9)	97.0 (96.2-97.6)	95.8 (94.8-96.6)	92.9 (91.7-94.0)	99.4 (99.0-99.7)
	3,033	2,689	2,203	1,896	1,758	1,739

Occupational grade (%)						
Administrative (top)	11.2 (10.1-12.3)	15.4 (14.1-16.8)	18.6 (17.1-20.2)	21.2 (19.5-23.0)	22.1 (20.3-23.9)	6.5 (5.5-7.7)
	381	425	466	439	442	120
Professional (middle)	39.1 (37.5-40.8)	44.1 (42.2-46.0)	44.8 (42.8-46.8)	47.5 (45.3-49.6)	46.8 (44.6-48.9)	24.0 (22.1-26.0)
	1,336	1,216	1,074	985	936	443
Clerical (bottom)	49.7 (48.0-51.4)	40.5 (38.7-42.3)	36.6 (34.7-38.5)	31.3 (29.4-33.4)	31.2 (29.2-33.2)	69.6 (67.4-71.6)
	1,696	1,117	877	650	624	1,286
Physical activity ^e (%)						
Inactive	25.7 (24.3-27.3)	35.6 (33.8-37.5)	11.2 (9.9-12.7)	12.4 (11.0-13.9)	31.7 (29.7-33.8)	43.5 (41.3-45.8)
	856	919	222	245	613	787
Below guidelines	38.8 (37.1-40.4)	32.7 (30.9-34.5)	9.1 (8.0-10.5)	9.6 (8.4-11.0)	23.1 (21.3-25.1)	19.9 (18.1-21.8)
	1,289	844	181	190	447	360
Met guidelines	35.5 (33.9-37.1)	31.7 (29.9-33.5)	79.6 (77.8-81.4)	78.1 (76.2-79.8)	45.1 (42.9-47.4)	36.6 (34.4-38.8)
	1,180	817	1,576	1,548	872	661
Smoking (%)						
Never	53.2 (51.5-54.8)	51.2 (49.2-53.2)	54.1 (52.0-56.2)	53.8 (51.7-56.0)	53.6 (51.3-55.8)	52.3 (49.9-54.6)
	1,805	1,229	1,158	1,098	966	879
Former	23.4 (22.0-24.9)	30.9 (29.1-32.8)	32.7 (30.7-34.7)	36.1 (34.1-38.3)	40.3 (38.1-42.6)	43.0 (40.6-45.4)
	795	742	200	737	750	723
Current	23.4 (22.0-24.9)	17.9 (16.5-19.5)	13.2 (11.8-14.7)	10.0 (8.8-11.4)	6.1 (5.1-7.3)	4.8 (3.8-5.9)
	795	431	283	204	113	80
^a Mean and 95% confidence interva	l; ^b Median and inte	r-quartile range; ^c Re	dundant, student, lo	ng-term sick, long-te	rm carer; ^d e.g. black	Caribbean,
African and Arabic; ^e Meeting guide minutes of moderate and <60 minu	elines (≥150 minutes utes of vigorous acti	s of moderate-intens vity; below guidelin	sity or ≥75 minutes o es (anyone not inact	f vigorous-intensity a ive or meeting guide	ctivity per week); ina lines)	active (<60

6.6 Missing data

Evident in Table 6.2 were three sources of missing data: unit non-response, representing attrition by individuals who opted not to participate at all in a given wave; item non-response, characterised as individuals who participated but opted not to answer one or more questions in a given wave; and mortality.

Were data missing completely at random (MCAR), no systematic differences would exist between participants with and without missing values.³³⁰ Under such a circumstance, missing data would be ignorable, with the only resultant limitation being a reduction in statistical power. In practice, however, data are rarely MCAR.³³¹

The following sections explored whether individuals with missing data differed markedly from those that provided valid responses. If differences were present on characteristics associated with T2DM, missing data would be considered informative and non-ignorable due to its potential to introduce attrition or survivorship biases to any complete-case analyses of the alcohol-T2DM relationship.^{332,333} For example, were individuals who opted not to answer certain questions found to have exhibited characteristics that placed them at greater risk of T2DM than those with complete covariate data, such as current smoking and physical inactivity, then complete-case analyses would have risked selecting a disproportionately healthy sub-sample and T2DM risk potentially underestimated as a consequence.

6.6.1 Mortality

Information concerning the mortality of study participants is available via linkage to the NHS Central Registry.²⁹⁴ By wave 11, a total 13.3% (n=915) of men and 14.6% (n=499) of women were documented as having died (Table 6.2).

As with any other form of attrition, the death of participants risked producing results that were more optimistic than would otherwise be the case, with many of the risk factors for T2DM also being risk factors for mortality. If participants at high risk of T2DM had died prior to the development of the condition, the resulting dataset would have been biased, with the remaining cohort representing disproportionately healthy, low-risk participants relative to the original sample. To investigate the degree to participant characteristics differed according to mortality status, Table 6.3 summarises participant characteristics at wave one, stratified according to whether or not participants died during the course of the study.

Men and women who died were found to exhibit a higher proportion of negative risk factors for T2DM than those that survived, including older age and higher BMI, as well as lower levels of

physical activity and a higher prevalence of current smoking. Men who died were also more likely to be lighter, occasional drinkers than those that survived, though lower volumes of alcohol consumption and less frequent drinking may have been attributable to their poor health status at baseline. That participants lost to mortality appeared of poorer health and at greater risk of T2DM at baseline, it was possible that mortality produced a healthier analytical subsample in which T2DM risk was underestimated. In such an instance, mortality would be informative and represent a competing risk event, with death altering a participant's probability of experiencing T2DM.³³⁴

Unfortunately, no agreement was found concerning how best to deal with the informative effect of a competing risk such as mortality.³³² Approaches have included deriving an outcome measure that included both mortality and the primary outcome of interest,³³⁵ or assigning those who died with the worst outcome value (e.g. coding all deceased individuals as having developed T2DM).³³⁶ In each case, the modification of an outcome variable in such a manner rendered difficult the interpretation of results within the context of the primary relationship of interest.³³⁷ With a lack of agreement over the most appropriate method, and existing methods potentially complicating any inferences concerning the alcohol-T2DM relationship, no attempt was made to account for the effect of mortality except to note that analyses potentially underestimated the risk of event.

		Men			Women	
	Survived	Died	Difference	Survived	Died	Difference
Variables (wave 1)	% (95% CI) n	% (95% CI) n		% (95% CI) n	% (95% CI) n	
Age						
Mean years	44.0 (42.3-45.7) ^b 5.980	48.2 (46.5-49.8) ^b 915	<0.001	45.3 (43.5-47.0) ^b 2.914	48.8 (47.2-50.4) ^b 499	<0.001
Alcohol consumption frequer	ncy			·		
None in past year	2.9 (2.5-3.4) 173	5.2 (3.9-6.8) 47	<0.001	6.3 (5.4-7.2) 182	7.1 (5.1-9.7) 35	0.227
<1/week	19.8 (18.8-20.8) 1,181	24.9 (22.2-27.8) 226		38.9 (37.1-40.7) 1,131	43.1 (38.8-47.6) 214	
1-3 times/week	43.5 (42.3-44.8) 2,596	36.0 (32.9-39.2) 327		34.7 (33.0-36.5) 1,010	31.7 (27.7-35.9) 157	
Daily or almost daily	33.8 (32.6-35.0) 2,016	34.0 (31.0-37.1) 309		20.1 (18.7-21.6) 584	18.1 (15.0-21.8) 90	
Alcohol consumption volume	8					
Median g/week	63.2 (23.7-134.3) ^c 5,931	55.3 (15.8-150.1) ^c 909	0.040	23.7 (0.0-55.3) ^c 2,882	23.7 (0.0-55.3) ^c 492	0.167
BMI						
Mean kg/m²	24.4 (23.3-25.6) ^b 5,971	25.4 (24.0-26.9) ^b 912	<0.001	24.6 (22.9-26.3) ^b 2,914	25.8 (23.9-27.7) ⁵ 497	<0.001
Ethnicity						
White	92.3 (91.6-92.9) 5,493	90.2 (88.1-92.0) 813	0.093	85.3 (84.0-86.6) 2,455	87.0 (83.6-89.7) 420	0.206
South Asian	5.1 (4.6-5.7) 305	6.8 (5.3-8.6) 61		6.9 (6.0-7.9) 199	4.8 (3.2-7.1) 23	
Other ^d	2.6 (2.2-3.1) 156	3.0 (2.1-4.3) 27		7.8 (6.9-8.8) 224	8.3 (6.1-11.1) 40	

Table 6.3 Descriptive summary of Whitehall II data at baseline, stratified by sex and survival

Family history of T2DM						
No	89.9 (89.1-90.6) 5,298	87.3 (85.0-89.4) 780	0.021	86.3 (85.0-87.5) 2,465	83.8 (80.3-86.8) 409	0.142
Yes	10.1 (9.4-10.9) 596	12.7 (10.6-15.0) 113		13.7 (12.5-15.0) 391	16.2 (13.2-19.7) 79	
Incident diabetes from wave 3	~					
No	89.4 (88.5-90.2) 4 583	87.5 (84.6-90.0) 520	0.178	88.5 (87.1-89.7) 2 016	88.7 (84.4-91.8) 258	0.920
Yes	10.6 (9.8-11.5) 657	12.5 (10.0-15.4) 74		11.5 (10.3-12.9) 263	11.3 (8.2-15.6) 33	
Occupational grade						
Administrative (top)	12.7 (11.9-13.6) 760	12.3 (10.4-14.6) 113	0.016	5.0 (4.3-5.9) 146	3.6 (2.3-5.7) 18	<0.001
Professional (middle)	46.6 (45.4-47.9) 2.788	42.1 (38.9-45.3) 385		19.4 (18.0-20.8) 564	11.4 (8.9-14.5) 57	
Clerical (bottom)	40.7 (39.4-41.9) 2,432	45.6 (42.4-48.8) 417		75.6 (74.0-77.2) 2,204	85.0 (81.5-87.9) 424	
Physical activity ^e						
Inactive	9.0 (8.3-9.8) 536	12.1 (10.1-14.4) 109	0.001	24.4 (22.8-26.0) 692	33.6 (29.5-37.9) 164	<0.001
Below guidelines	36.9 (35.7-38.1) 2,186	39.7 (36.5-42.9) 358		39.5 (37.8-41.4) 1,122	34.2 (30.1-38.6) 167	
Met guidelines	54.0 (52.8-55.3) 3,201	48.2 (45.0-51.5) 435		36.1 (34.3-37.8) 1,023	32.2 (28.2-36.5) 157	
Smoking						
Never	49.3 (48.0-50.6) 2,924	37.4 (34.3-40.6) 340	<0.001	55.9 (54.1-57.7) 1,620	37.1 (33.0-41.5) 185	<0.001
Former	36.6 (35.4-37.8) 2,170	34.8 (31.7-37.9) 316		24.0 (22.4-25.5) 694	20.3 (17.0-24.1) 101	
Current	14.1 (13.3-15.0) 838	27.8 (25.0-30.8) 253		20.1 (18.7-21.6) 583	42.6 (38.3-47.0) 212	

The number of participants listed for each variable differed according to item non-response on each given variable at baseline.
[*] To explore differences between non-response groups, one-way ANOVA was used on continuous data, and the chi ² test on
categorical data (where continuous data exhibited a non-normal distribution, data were log-transformed prior to testing; ^b Mean
and 95% confidence interval; ^c Median and inter-quartile range; ^d e.g. black Caribbean, African and Arabic; ^c Meeting guidelines (≥150
minutes of moderate-intensity or 275 minutes of vigorous-intensity activity per week); inactive (<60 minutes of moderate and <60
minutes of vigorous activity; below guidelines (not inactive or meeting guidelines).

6.6.2 Unit non-response

Some participant data were also missing due to unit non-response. Of participants who survived the course of the study, only 65.5% of male and 53.4% female baseline participants were found to have taken part at all waves (Table 6.4). Such a loss of data was likely to have a substantial impact upon statistical power and thereby inflate the risk of type 2 error in any complete-case analyses. To elucidate whether unit non-response may have introduced some degree of attrition bias, baseline characteristics of surviving participants were stratified according to the number of waves for which unit non-response was observed (Table 6.5).

	Men	Women
Number of ways with unit pop response	%	%
Number of waves with unit non-response	n	n
0	65.5	53.4
	3,919	1,555
1	9.0	10.4
	537	303
2	5.4	6.1
	331	178
3	6.4	7.9
	383	229
4	6.4	8.9
	381	258
5	7.2	13.4
	429	391

Table 6.4 Proportion of surviving baseline participants with unit non-response

Differences in baseline characteristics are evident according to participants' degree of unit non-response. As shown in Table 6.5, those with greater levels of unit non-response were more likely to be from minority ethnic backgrounds, lower occupational grades, less physically active and have the greatest proportion of current smokers. Those with unit non-response across \geq 4 waves were also found to have a lower volume and frequency of alcohol consumption, potentially due to poor health. In addition to a worse metabolic risk profile, participants with unit non-response had a lower incidence of T2DM, suggesting T2DM may have been underreported due to non-participation at clinical examinations. In summary, complete-case analyses may be subject to some degree of attrition bias, with any analytical sample skewed toward those with a more favourable risk profile.³³⁸

		Unit non-res	sponse	
	0 waves	1-3 waves	≥4 waves	Difference ^a
Variables (wave 1)	% (95% CI)	% (95% CI)	% (95% CI)	
	n	n	n	
Men				
Age				
Mean years	44.2 (44.0, 44.4) ^b 3,919	43.5 (43.2, 43.8) ^b 1,251	43.5 (43.1, 43.9) ^b 810	<0.001
Alcohol consumption frequency				
None in past year	2.2 (1.8, 2.7)	4.1 (3.1, 5.3)	4.3 (3.1, 6.0)	<0.001
	87	51	35	
<1/week	18.8 (17.6, 20.1)	19.1 (17.0, 21.4)	25.6 (22.7, 28.7)	
	736	239	206	
1-3 times/week	44.7 (43.1, 46.2)	42.5 (39.8, 45.3)	39.5 (36.1, 42.9)	
	1,747	531	318	
Daily or almost daily	34.3 (32.8, 35.8)	34.3 (31.7, 36.9)	30.6 (27.6, 33.9)	
	1,341	428	247	
Alcohol consumption volume				
Median g/week	63.2 (31.6, 134.3) ^c	71.1 (31.6, 150.1) ^c	47.4 (15.8, 126.4) ^c	< 0.001
	3,891	1,239	801	
BMI				
Mean kg/m ²	243(242 244) ^b	248(246 249) ^b	246 (24 4 24 8) ^b	<0.001
Mean kg/m	24.3 (24.2, 24.4)	24.0 (24.0, 24.5)	24.0 (24.4, 24.8)	
Ethnicity	5,512	1,200	005	
NA/h:+-				-0.001
white	94.9 (94.1, 95.5)	89.1 (87.3, 90.8)	84.3 (81.5, 86.6)	<0.001
South Acian	3,/10			
South Astan	5.7 (5.1, 4.5) 142	0.0 (5.0, 0.4)	9.7 (7.8, 12.0)	
e i d	1 E (1 1 1 0)	ده ۱۵ (۲ ۱ ۲ ۲)	60 (4 6 7 0)	
Other"	1.5 (1.1, 1.5)	4.0 (3.1, 3.3)	0.0 (4.0, 7.9)	
Family history of T2DM	50	50	-0	
				0 500
NO	9.9 (9.0, 10.9)	10.0 (8.4, 11.8)	11.2 (9.2, 13.6)	0.536
Vez	384	123	89	
res	90.1 (89.1, 91.0)	90.0 (88.2, 91.6)	88.8 (80.4, 90.8)	
	3,485	1,109	704	
Occupational grade				
Administrative (top)	43.1 (41.6, 44.7)	31.7 (29.1, 34.3)	28.3 (25.3, 31.5)	<0.001
	1,689	396	229	
Professional (middle)	52.4 (50.8, 53.9)	55.3 (52.5, 58.1)	52.8 (49.4, 56.3)	
	2,052	692	428	
Clerical (bottom)	4.5 (3.9, 5.2)	13.0 (11.3, 15.0)	18.9 (16.3, 21.7)	
	178	163	153	
Physical activity ^e				
Inactive	7.6 (6.8, 8.5)	10.3 (8.7, 12.1)	14.1 (11.8, 16.7)	<0.001
	297	126	113	
Below guidelines	37.9 (36.4, 39.5)	33.9 (31.3, 36.6)	36.5 (33.3, 39.9)	
	1,476	417	293	
Met guidelines	54.4 (52.9, 56.0)	55.8 (53.0, 58.6)	49.4 (45.9, 52.8)	
	2,119	686	396	

Table 6.5 Descriptive summary of Whitehall II data at baseline, stratified by sex and degree of unit nonresponse

Smoking				
Never	51.0 (49.4, 52.5)	45.9 (43.2 <i>,</i> 48.7)	46.3 (42.9, 49.8)	< 0.001
	1,983	570	371	
Former	37.0 (35.5, 38.5)	37.0 (34.3, 39.7)	33.8 (30.6, 37.2)	
	1,440	459	271	
Current	12.0 (11.0, 13.1)	17.1 (15.1, 19.3)	19.9 (17.2, 22.8)	
	467	212	159	
Women				
Age				
Mean years	44.6 (44.3, 44.9) ^b	45.9 (45.4, 46.3) ^b	46.2 (45.7, 46.6) ^b	< 0.001
	1.555	710	649	
Alcohol consumption frequency				
None in past year	52(1261)	76 (59 98)	71 (5697)	<0.001
None in past year	5.2 (4.2, 0.4)	7.0 (5.5, 5.8)	7.4 (5.0, 9.7)	<0.001
<1/week	35 2 (32 9 37 7)	41 0 (37 4 44 7)	40	
	547	41.0 (37.4, 44.7) 290	43.4 (41.0, 43.2) 294	
1-3 times/week	36.9 (34.6. 39.4)	33.8 (30.4, 37.4)	30.6 (27.1, 34.2)	
	573	239	198	
Daily or almost daily	22.7 (20.7, 24.8)	17.5 (14.9. 20.5)	16.7 (14.0, 19.7)	
	352	124	108	
Alcohol consumption volume				
Median g/week				<0.001
Mediali g/ week	31.6 (7.9, 63.2)	23.7 (0.0, 55.3)	15.8 (0.0, 47.4)	<0.001
	1,534	707	641	
ВМІ				
Mean kg/m ²	24.3 (24.1, 24.5) ^b	24.9 (24.6, 25.3) ^b	24.9 (24.6, 25.2) ^b	<0.001
	1,555	710	649	
Ethnicity				
White	88.8 (87.1, 90.3)	81.6 (78.6 <i>,</i> 84.3)	80.7 (77.4, 83.7)	<0.001
	1,379	573	503	
South Asian	5.4 (4.4, 6.7)	8.3 (6.4, 10.5)	9.1 (7.1, 11.7)	
	84	58	57	
Other ^d	5.8 (4.7, 7.1)	10.1 (8.1, 12.6)	10.1 (8.0, 12.7)	
	90	71	63	
Family history of T2DM				
No	13.3 (11.7, 15.1)	13.3 (11.0, 16.0)	15.0 (12.4, 18.0)	0.539
	203	93	95	
Yes	86.7 (84.9, 88.3)	86.7 (84.0 <i>,</i> 89.0)	85.0 (82.0, 87.6)	
	1,322	606	537	
Occupational grade				
Administrative (top)	16.7 (14.9, 18.6)	7.3 (5.6 <i>,</i> 9.5)	5.4 (3.9 <i>,</i> 7.4)	<0.001
	259	52	35	
Professional (middle)	46.9 (44.5, 49.4)	32.3 (28.9, 35.8)	31.7 (28.3, 35.4)	
	730	229	206	
Clerical (bottom)	36.4 (34.0, 38.8)	60.4 (56.8 <i>,</i> 64.0)	62.9 (59.1, 66.5)	
	566	429	408	
Physical activity ^e				
Inactive	20.3 (18.3, 22.4)	28.5 (25.2, 32.0)	29.9 (26.5, 33.6)	< 0.001
	310	194	188	
Below guidelines	42.5 (40.1, 45.0)	36.1 (32.6, 39.8)	36.0 (32.3, 39.8)	
	650	246	226	
Met guidelines	37.2 (34.8, 39.6)	35.4 (31.9, 39.1)	34.1 (30.5, 37.9)	
	568	241	214	
Smoking				
---------	-------------------	-------------------	-------------------	--------
Never	59.4 (57.0, 61.9)	50.8 (47.1, 54.5)	53.1 (49.2, 56.9)	<0.001
	919	358	343	
Former	25.4 (23.2, 27.6)	24.5 (21.5, 27.9)	20.0 (17.1, 23.2)	
	392	173	129	
Current	15.2 (13.5, 17.1)	24.7 (21.6, 28.0)	26.9 (23.6, 30.5)	
	235	174	174	

Figures excluded participants who died over the course of the study. Figures included participants who had T2DM at waves 1-3. Number of participants listed under each variable differed according to the level of item non-response at baseline (wave one).

^aTo explore differences between non-response groups, one-way ANOVA was used on continuous data, and the chi² test on categorical data (where continuous data exhibited a non-normal distribution, data were log-transformed prior to testing; ^bMean and 95% confidence interval; ^cMedian and 25th and 75th percentiles; ^de.g. black Caribbean, African and Arabic; ^eMeeting guidelines (≥150 minutes of moderateintensity or ≥75 minutes of vigorous-intensity activity per week); inactive (<60 minutes of moderate and <60 minutes of vigorous activity; below guidelines (not inactive or meeting guidelines).

6.6.3 Item non-response

The third source of missing data within the Whitehall II cohort was among individuals who participated in a given wave but provided no answer to a specific question or provided no anthropometric measure or blood draw at clinical examination. Most *a priori* selected covariates were subject to some degree of item non-response, as summarised below in Table 6.6. Item non-response was greatest for the BMI variable at wave five, with measurements absent among 27.0% of male and 29.7% of female participants. This was understood to have been due to administrative problems at centres where clinical examinations were held. As a consequence, excess item non-response on BMI at this wave could be treated as MCAR.

In addition to a cumulative loss of 34.5% of male and 46.5% of female baseline participants through unit non-response in complete-case analyses (Table 6.4), analytic samples risked being attenuated even further by item non-response. To isolate differences in baseline characteristics according to item non-response, the dataset was restricted to the 5,474 individuals who participated at all waves, with descriptive statistics then reported for each variable as stratified according to whether complete-case data were provided at all waves (Table 6.7). Close to half of participants free of unit non-response provided complete-case data (n=2,673). Among both sexes, those with item non-response had a higher BMI and proportion of minority ethnic participants at baseline (wave one), as well as lower occupational grade, physical activity and average volume and frequency of alcohol consumption. As with unit non-response, the loss of data through item non-response appeared to represent a further source of attrition bias, with complete-case analyses inevitably sampling a subset of participants with a more favourable T2DM risk profile.

In order to reduce the impact of attrition bias upon any analyses, values missing due to unit or item non-response were estimated using multiple imputation. This process and its underlying assumptions are outlined in the following section.

-	Wave 1	Wave 3	Wave 5	Wave 7	Wave 9	Wave 11
	%	%	%	%	%	%
Variables with item non-response	n	n	n	n	n	n
Men						
Alcohol consumption frequency	0.3	5.2	9.1	2.3	2.7	6.2
	20	317	498	113	127	278
Alcohol consumption volume	0.8	5.3	8.9	2.6	2.9	1.3
	55	323	489	123	135	58
BMI	0.2	7.7	27.0	6.6	7.0	9.4
	12	466	1,477	324	335	420
Employment status	-	0.0	0.6	0.4	0.2	0.0
		0	34	19	7	1
Ethnicity	0.6	0.3	0.0	0.1	0.2	0.1
	40	18	2	7	10	6
Family history of T2DM	1.6	1.4	1.5	1.5	1.3	1.2
	108	85	80	71	63	55
Incident diabetes from wave 3 ^a	-	-	2.5	2.4	3.2	3.1
			134	115	148	135
Physical activity	1.0	5.2	15.0	2.7	1.7	1.4
	70	318	819	131	80	60
Smoking status	0.8	10.6	7.6	1.1	5.9	5.9
-	54	642	416	53	279	261
Women						
Alcohol consumption frequency	0.3	6.5	13.2	3.7	4.8	11.7
	10	179	317	77	96	217
Alcohol consumption volume	1.1	6.2	12.3	4.1	4.8	1.2
·	39	168	288	83	94	21
BMI	0.1	10.0	29.7	9.3	11.6	14.8
	2	275	712	1.3	233	273
Employment status	-	0.0	0.6	0.4	0.2	0.2
		0	14	9	4	3
Ethnicity	1.5	0.8	0.3	0.3	0.3	0.2
	52	22	7	6	6	4
Family history of T2DM	2.0	2.0	1.8	1.8	1.9	1.9
	69	54	43	38	37	35
Insident disketse from were 2ª	_	_	3 1	2.8	3 9	3.8
incident diabetes from wave 3			73	56	77	69
Physical activity	2.6	6.5	17.4	4.4	3.5	2.2
	88	178	418	91	70	41
Smoking status	0 5	12 9	10 7	17	71	٩ ٥
Shioking status	18	356	256	35	143	167
^a Restricted to diabetes-free particin	ants at way	e 3				

Table 6.6 Degree of item non-response by wave, stratified by sex

-	lt	em response	
-	Complete	Incomplete	Difference ^a
Variables (wave 1)	% (95% CI)	% (95% CI)	
	n	n	
<u>Men</u>			
Age			
Mean years	44.1 (43.8, 44.3) ^b	44.3 (44.1, 44.6) ^b	0.163
	1,976	1,943	
Alcohol consumption frequency			
None in past year	0.8 (0.5, 1.3)	3.7 (3.0, 4.7)	<0.001
	15	72	
<1/week	19.0 (17.4, 20.8)	18.6 (16.9, 20.4)	
	376	360	
1-3 times/week	45.7 (43.6, 48.0)	43.6 (41.4, 45.8)	
	904	843	
Daily or almost daily	34.5 (32.4, 36.6)	34.1 (32.0, 36.3)	
	681	660	
Alcohol consumption			
Median g/week	71.1 (31.6, 142.2) ^c	63.2 (23.7, 126.4) ^c	<0.001
	1.976	1.915	
BMI			
Mean kg/m ²	24 2 (24 1 24 3) ^b	211(212 215) ^b	0.051
Wear Kg/m	1 976	1 936	
Ethnicity	11570	1.550	
White	96.5 (95.6, 97.2)	93.2 (92.0, 94.2)	<0.001
	1,907	1,809	
South Asian	2.3 (1.7, 3.1)	5.0 (4.1, 6.1)	
	46	97	
Other ^e	1.2 (0.8, 1.7)	1.8 (1.3, 2.5)	
	23	35	
Family history of T2DM ^d			
No	91.1 (89.8, 92.3)	89.0 (87.5, 90.3)	0.023
	1,801	1,684	
Yes	8.9 (7.7, 10.2)	11.0 (9.7, 12.5)	
	175	209	
Incident diabetes from wave 3			
No	89.4 (87.9, 90.7)	88.2 (86.7, 89.6)	0.261
	1,766	1,649	
Yes	10.6 (9.3, 12.1)	11.8 (10.4, 13.3)	
	210	220	
Occupational grade			
Administrative (top)	43.3 (41.1, 45.5)	42.9 (40.7, 45.1)	0.006
	856	833	
Professional (middle)	53.2 (51.0, 55.4)	51.5 (49.3, 53.7)	
	1,051	1,001	
Clerical (bottom)	3.5 (2.8, 4.4)	5.6 (4.7, 6.7)	
	69	109	

Table 6.7 Descriptive summary of Whitehall II data at baseline (wave one), stratified by sex and item non-response

Physical activity ^e			
Inactive	5.9 (4.9, 7.0)	9.4 (8.2, 10.8)	<0.001
	116	181	
Below guidelines	40.5 (38.4, 42.7)	35.2 (33.1, 37.4)	
Matguidalinas			
Met guidennes	55.0 (51.4, 55.8) 1 059	55.5 (55.1, 57.5) 1 060	
Smoking	1,000	1,000	
Never	50 1 (47 8 52 3)	519(497542)	0.085
	989	994	0.005
Former	38.6 (36.5, 40.8)	35.4 (33.3, 37.5)	
	763	677	
Current	11.3 (10.0, 12.8)	12.7 (11.3, 14.3)	
	224	243	
<u>Women</u>			
Age			
Mean years	43.7 (43.3, 44.1) ^b 697	45.3 (44.9, 45.7) ^b 858	<0.001
Alcohol consumption frequency			
None in past year	2.6 (1.6, 4.1)	7.3 (5.7, 9.2)	<0.001
	18	62	
<1/week	32.1 (28.8, 35.7)	37.8 (34.6, 41.1)	
	224	323	
1-3 times/week	38.5 (34.9, 42.1)	35.7 (32.5, 39.0)	
	305	305	
Daily or almost daily	26.8 (23.7, 30.3)	19.3 (16.8, 22.1)	
	165	165	
Alcohol consumption			
Median g/week	31.6 (7.9, 71.1) ^c	23.7 (0.0, 55.3) ^c	<0.001
	697	837	
BMI			
Mean kg/m ²	23.9 (23.6, 24.2) ^b	24.7 (24.4, 25.0) ^b	<0.001
	697	858	
Ethnicity			
White	92.8 (90.7, 94.5)	85.5 (83.0, 87.7)	<0.001
	647	732	
South Asian	2.4 (1.5, 3.9)	7.8 (6.2, 9.8)	
	17	67	
Other ^e	4.7 (3.4, 6.6)	6.7 (5.2, 8.5)	
	33	57	
Family history of T2DM ^d			
No	87.9 (85.3, 90.2)	85.6 (83.1, 87.9)	0.184
	613	709	
Yes	12.1 (9.8, 14.7)	14.4 (12.1, 16.9)	
Incident diskets from when 2	84	119	
Incluent diabetes from wave 3			0.000
NO	90.4 (88.0, 92.4)	85.3 (82.8, 87.6)	0.003
Vec	03U 96/76 1201	/10 147/137 173)	
	67	122	

Occupational grade			
Administrative (top)	20.5 (17.7, 23.7)	13.5 (11.4, 16.0)	< 0.001
	143	116	
Professional (middle)	51.4 (47.6, 55.1)	43.4 (40.1, 46.7)	
	358	372	
Clerical (bottom)	28.1 (24.9, 31.6)	43.1 (39.8, 46.5)	
	196	370	
Physical activity ^e			
Inactive	17.8 (15.1, 20.8)	22.4 (19.7, 25.3)	0.007
	124	186	
Below guidelines	46.6 (42.9, 50.4)	39.1 (35.8 <i>,</i> 42.5)	
	325	325	
Met guidelines	35.6 (32.1, 39.2)	38.5 (35.3 <i>,</i> 41.9)	
	248	320	
Smoking			
Never	58.2 (54.5, 61.9)	60.4 (57.1, 63.7)	0.222
	406	513	
Former	27.4 (24.2, 30.8)	23.7 (20.9, 26.7)	
	191	201	
Current	14.3 (11.9, 17.2)	15.9 (13.6, 18.5)	
	100	135	

Securetienel avede

Data were restricted to individuals who participated at all waves. Sample sizes differed according to item non-response at baseline (wave one).

^aTo explore differences between non-response groups, one-way ANOVA was used on continuous data, and the chi² test on categorical data (where continuous data exhibited a non-normal distribution, data were log-transformed prior to testing; ^bMean and 95% confidence interval; ^cMedian and 25th and 75th percentiles; ^de.g. black Caribbean, African and Arabic; ^eMeeting guidelines (≥150 minutes of moderate-intensity or ≥75 minutes of vigorous-intensity activity per week); inactive (<60 minutes of moderate and <60 minutes of vigorous activity; below guidelines (not inactive or meeting guidelines).

6.6.4 Multiple imputation

Given the concomitant issues of reduced sample size and a potential introduction of attrition bias through unit and item non-response, the decision was made to try and populate any values deemed missing due to non-response.

A number of different imputation methods were available for handling missing data. One of the most basic was mean substitution, which would have replaced missing values with the observed variable mean. However, such an approach would have assumed that missing data were MCAR, such that the mean of the observed data provided a sound approximation of the missing values of a given variable, as in the case of excess item non-response on the BMI at wave five due to an administrative error, or a blood sample having been dropped at random in a laboratory. This assumption was not supported by the analyses undertaken in preceding sections. Additionally,

adding no new information to the dataset beyond what was already observed, mean substitution would have resulted in an underestimation of variance.³³⁹ Alternative simple imputation methods, such as hot deck imputation and its replacement of missing data with an observed value selected at random from participants with similar characteristics, were each subject to the same two limitations.³³⁹

Among more complex methods, multiple imputation represented a popular and methodologically superior approach to those described above.³⁴⁰ Unlike simple imputation methods, the most likely value of each missing data point was predicted multiple times via a series of iterative regression models, with each series of predictions forming its own datas*et al*ongside the observed data.³⁴¹ Values from each prediction were then averaged, and associated variances calculated according to two components: the standard variance present within each dataset, and the variance present between datasets. As a consequence, as well as adding new data as opposed to merely duplicating what has already been observed, variances following multiple imputation are larger and better reflect the uncertainty surrounding true values of the missing data. Given these benefits, multiple imputation was selected as the imputation method of choice.

6.6.4.1 Statistical analysis

Given the non-monotone nature of item non-response across waves (Table 6.2), as well as the presence of multiple variable types (e.g. categorical and continuous data), multiple imputation was undertaken using chained equations (MICE), which represented a method suited to handling such data characteristics.^{342,343}

MICE offered a sequential approach to multiple imputation, whereby variables with missing data were imputed iteratively via a sequence of regression models that predicted missing data as conditional on all other observed variables.³⁴² For each variable missing data, the most appropriate regression model was utilised, including logistic regression models for binary data and multinomial logistic regression for non-ordered categorical variables. Given the difficulty with which linear regression models predict missing data on skewed continuous variables,³⁴³ skewed data were log transformed prior to imputation, then converted back to their original scale post-imputation.

The imputation procedure began by populating all missing data with random values so as to permit the inclusion of missing data points within regression models. Then, starting with the variable for which the proportion of missing data was lowest (x_1) , observed data on that variable

were regressed against all other covariates in the model ($x_2,...,x_p$). Missing values on x_1 were then replaced by data drawn from the predicted distribution of values missing on x_1 , given $x_2,...,x_p$.³⁴⁴ Once complete, the imputation procedure moved to the variable with the next lowest proportion of missing data (x_2). Observed values of x_2 were then regressed on all other covariates ($x_1, x_3,..., x_p$), with x_1 including all data predicted at the preceding stage.³⁴⁴ This process continued iteratively until missing values had been imputed for all variables with missing data, representing one complete iteration of the imputation procedure.

At the start of a second iteration, missing data were populated with predictions from the previous iteration as opposed to random noise. The calculation of new iterations continued until the predicted values were consistent between iterations. In such a circumstance, any variation in predicted values between iterations appeared random and the data were said to have converged. At the point of convergence, predicted data were stored in a dataset as one complete imputation.³⁴⁴

In order to reliably capture the degree of uncertainty surrounding the predicted values, 50 imputations were run. This was in keeping with guidance which advises that the total number of imputed datasets should be equal to at least the total proportion of participants without complete-case data (see Tables 6.4 and 6.6).³⁴⁴

The imputation procedure was undertaken while data were stored in their wide format. This modelled each wave of a given variable as a separate covariate in each regression model, allowing for the modelling of missing data as a function of observations at all times that preceded and followed the wave of unit or item non-response. Data were imputed using Stata 13 and the -mi- package. The *augment* option was also applied to overcome any circumstances in which perfect prediction between variables may have occurred, adding random observations with very low weights to the included variables.

6.6.4.2 Diagnostics

Imputed estimates was examined to identify any incongruous values, such as negative alcohol consumption. Where any invalid values were identified, the imputation model was modified accordingly and re-run. In the case of average weekly volume of alcohol consumption, for example, this involved the use of a truncated regression model, which restricted the lower limit of predicted alcohol consumption to a value of zero.

Trace plots were then undertaken to check that predicted values of missing data had converged.³⁴⁵ These plotted the mean predicted value of a variable at each iteration, indicating

visually whether any variation in values between iterations had become random such that additional iterations provided no further improvement to the predictions made. The optimal number of iterations recommended to reach convergence ranged from 10³⁴⁴ to 20³⁴⁶ depending upon the degree of correlation between variables. A total 20 iterations were initially run for each imputation, though convergence was not achieved on a minority of variables. The number of iterations was thus doubled to 40, which was found to be a sufficient number of iterations to reach convergence across all covariates. Examples of the resulting trace plots are reported in Appendix 6.2.

Once predictions had successfully converged and were deemed congruous, descriptive statistics were reported according to whether data were observed or imputed. Due to marked differences in participant characteristics according to the presence of missing data (Tables 6.5 and 6.7), the missingness process was considered informative and results thus documented according to both the imputed and observed datasets where possible within the limitations of Stata.

6.6.4.3 Variable selection

All variables identified *a priori* for use in any subsequent statistical analysis were included in the imputation model. These substantive variables thus comprised the frequency and total weekly volume of alcohol consumption plus an interaction term between these two dimensions, all confounding factors, and T2DM. Specifically, T2DM was represented by three variables: binary variable that denoted the number of new cases of T2DM captured at each wave of measurement, a continuous variable that represented the time to event, and the cumulative baseline hazard function. The inclusion of these outcome variables in the imputation model ensured that missing values on any independent variables each reflected the covariance that existed between observed covariate data and the risk of T2DM. Failure to include such variables risked biasing toward the null any associations subsequently calculated between covariates and T2DM risk when using imputed data.^{347,348}

With missing values predicted conditional upon observed data, the multiple imputation model operated under the assumption that data were missing at random (MAR), i.e. that any systematic differences between participants with and without missing data could be entirely according to observed data included in the imputation model.³⁴⁹ Accordingly, the plausibility of the MAR assumption and the performance of any imputation model was strengthened by an inclusive covariate selection strategy, including a broad range of auxiliary variables likely to covary with the variables upon which missing data were to be predicted.^{342,350} A number of suitable auxiliary variables were theorised and are described in the section below.

151

Unfortunately, despite the inclusion of numerous auxiliary variables, it was possible that the MAR assumption may have been invalid; that some proportion of the systematic difference in participant characteristics evident between those with and without missing data (Tables 6.5 and 6.7) may have been a consequence of unobserved factors not included in the imputation model. ^{340,341} Nevertheless, in the case of both unit and item non-response, the MAR assumption was considered reasonable given the range of auxiliary variables considered.

6.6.4.3.1 Predictors of unit non-response

Although the reason for unit non-response was unknown, potential explanations were hypothesised to include instances were individuals were not solicited at all (e.g. due to inaccurate contact records) or participants who were solicited but unable to participate (e.g. due to physical or mental incapacity, or scheduling problems).

The first group was believed to represent a very small proportion of total unit non-response, given efforts by cohort administrators to maintain accurate contact information and actively trace those lost to postal contact.²⁹⁴ Such individuals were assumed to be MCAR. Of those with unit non-response across just a couple of waves, attrition was most logically explained by scheduling problems such as work commitments or holidays, or the development of acute physical or mental incapacity sufficient to preclude participations. By contrast, of those with unit non-response across the majority of waves, attrition was most likely attributable to the development of chronic physical or mental incapacity that prevented participation over the long term.

In either case, it was thought possible to capture information concerning the probability of unit non-response through the inclusion of variables concerning mental and physical health prior to unit non-response, as well as factors likely to influence time available to participate. In the first instance, mental incapacity was included as an auxiliary variable via two variables. The first was the General Health Questionnaire (GHQ), a 30-item screening instrument administered at all waves and designed to capture self-reported symptoms of anxiety, depression and associated psychosocial dysfunction, with scores that ranged from 0 to 30.³⁵¹ Defined as any score greater than four,³⁵² the tool has been validated on Whitehall II data for the detection of mild psychiatric disorder.³⁵² Mild psychiatric disorder was thus included as a predictor of unit non-response by way of a binary GHQ variable. In addition to psychiatric morbidity, any unit non-response due to impaired mental cognition was captured via results from the Mini Mental State Examination (MMSE), a 30-point screening instrument for the quantitative assessment of global cognitive function.³⁵³ The MMSE comprised a series of 11 questions concerning seven distinct cognitive domains, including orientation to time, attention and recall, and has been validated as sensitive to the detection of moderate-to-severe cognitive impairment.³⁵⁴ The MMSE was first introduced in wave 5 and applied only to persons aged ≥ 60 years. MMSE data were then available at all subsequent waves and delivered to all participants. In an effort to capture participants' physical health, participants' self-rated general health was also included, available at all waves and coded according to three categories: 'very good or excellent', 'good' and 'fair or poor'.

With regard to any unit non-response attributable to scheduling problems, the imputation model included data concerning the average number of hours worked in a week, as captured at wave three, as well as the amount of time participants reported being 'worn out' in the four weeks prior to each wave, defined as 'all of the time', 'most of the time', 'a good bit of the time', 'some of the time', 'a little of the time' or 'none of the time'. Regression models also included a variable that denoted whether family obligations reduced a participant's time available for relaxation. These data were documented at waves three, five and seven, with categories defined as 'not at all', 'to some extent', 'a great deal', or 'not applicable', e.g. no family obligations.

6.6.4.3.2 Predictors of item non-response

Alcohol consumption

Although alcohol consumption appeared lower at baseline (wave one) among those with item non-response at later waves (Table 6.7), it was possible that participants who opted not to disclose their consumption at later waves had experienced an alcohol use disorder.³⁵⁵ Such individuals may not have disclosed their consumption behaviour out of shame or an inability to reliably recall volumes accurately due to inebriation. In case missing alcohol consumption data were associated with problematic levels of drinking, data from the CAGE questionnaire were included as a screening instrument for the detection of problem drinking,³⁵⁶ which was found to correlate well with clinical diagnoses of alcoholism.³⁵⁷ The CAGE questionnaire was first introduced in the Whitehall II study at wave 3 and comprises four questions concerning alcohol consumption behaviour and participants' perceptions thereof, such as whether they ever felt guilty about their drinking behaviour. Alcohol dependence was defined according to a score $\geq 2.^{356}$

With studies also having identified lower levels of alcohol intake among those in poor health,^{144,145,146,141,358} self-rated general health was posited to provide a useful correlate of alcohol consumption. Alongside this, substantive ethnicity and socio-economic variables were thought to help explain any religious or genetic factors that might have influenced alcohol

consumption or the willingness of participants to report it,³⁵⁹ as well as differences in alcohol consumption by socio-economic strata.³⁶⁰

<u>BMI</u>

Beyond substantive variables such as physical activity,³⁶¹ socio-economic factors,^{362,363} and smoking,³⁶¹ potential correlates of BMI were thought to include: mild psychiatric disorder, as captured via the GHQ;³⁶⁴ blood pressure;³⁶⁵ plasma triglycerides and HDL;^{366,367} and dietary factors identified during the literature review, including carbohydrate, polyunsaturated fat, trans fat and fibre.

Blood pressure readings were obtained at each clinical examination, averaged across two readings with continuous systolic and diastolic blood pressure variables documented in millimetres of mercury (mmHg).³⁶⁸ Blood samples were also taken at each clinical examination, with triglyceride and HDL concentrations recorded in mmol/L.³⁶⁹ Data concerning dietary factors were obtained via the FFQ, available at waves three, five and seven only.

It was possible that information on BMI was absent for some individuals due to hairstyles or clothing that would have rendered invalid any attempt to measure height or weight. Missing data in such cases were assumed to be MCAR, or else partially correlated with ethnicity, such as being unable to accurately measure height owing to religious headdresses.

Employment status

It was hypothesised that the probability of retirement would have been correlated with age, while the probability of exiting the labour market for reasons other than retirement was posited to be associated with indicators of mental health and self-reported general health.³⁷⁰

<u>Ethnicity</u>

Item non-response on the ethnicity variable was low (Table 6.6). However, given notable ethnic inequalities in labour participation in the UK,^{371,372} it was hypothesised that any missing ethnicity data could be predicted through a partial correlation with variables that denoted employment status and occupational grade.

Family history of T2DM

Item non-response on this variable was also low, absent for just 1.6% of male and 2.0% of female baseline participants (Table 6.6). It was hypothesised that these data may have been missing because participants simply did not know whether there was a history of T2DM in the family. Such data were thus assumed to be MCAR.

Incidence of T2DM

Possible randomly distributed reasons for not having an objective indicator of T2DM diagnosis were posited to include refusal to have bloods taken, inability by the clinic nurse to obtain a suitable blood sample or poor adherence to the necessary protocol (e.g. had not fasted). Conversely, the lack of a subjective self-reported measure might have been attributable to the inability of a participant to recall whether or not T2DM had been reported by a doctor, or a lack of time with which to attend the clinic or complete all necessary measurements upon attendance. Missing values of T2DM were thus predicted according to indicators of availability and cognition, as described previously, as well as key risk factors for the disease, including age, BMI, ethnicity and both serum triglyceride and HDL concentration.

Physical activity

BMI was considered a key correlate of physical activity level, along with other clinical markers such as serum HDL and triglycerides.³⁷³ Aside from such factors, studies indicate that, among employed individuals, time spent in leisure-time physical activity may be determined by factors that influence the amount of free time available to undertake such activity.³⁷⁴ These factors were included as auxiliary predictors of physical activity, as described for unit non-response. Further correlates were thought to include socio-economic factors and self-rated general health, with different types and amounts of leisure-time physical activity undertaken across social strata.^{375,376,377}

<u>Smoking</u>

As was likely with both BMI and physical activity, it was posited that item non-response may been greater among those participants who perceived their smoking behaviour to be less socially desirable. Given the clustering of negative health behaviours,^{78,378} it was postulated that alcohol consumption, physical activity and BMI would all correlate with smoking, as well as clinical measures such as blood pressure³⁷⁹ and serum HDL concentration.³⁸⁰

6.6.4.4 Results

6.6.4.4.1 Convergence

Trace plots indicated good convergence across all variables of interest, with between-iteration variation in predicted values appearing stable at the point of draw following 40 iterations. These data are reported in Appendix 6.2 and indicated that further iterations would have provided no improvement to the predicted values, with any variation between imputations having been a consequence of random error.

6.6.4.4.2 Summary of imputed data

Wave-specific characteristics of participants within the imputed dataset are reported below in Table 6.8, with observed cohort data reported in Table 6.2. After accounting for unit and item non-response, participants within the imputed dataset reported higher levels of average weekly volume consumption, particularly among women, suggesting that many may not have reported their consumption due to perceived social undesirability. The imputed data also predicted higher levels of BMI among women, relative to the observed dataset, suggesting that women may have systematically declined such a measurement for similar reasons. A higher proportion of imputed individuals were employed, relative to those that were observed, suggesting that many who did not participate may have done so due to scheduling problems and time constraints. Both sexes with missing data were also more likely to have been of minority ethnic origin, lower occupational grade and current smokers. Additionally, while not shown in Tables 6.2 and 6.8, the proportion of participants who reported being of very good or good general health was highest among those who were observed, with any discrepancies between the two datasets largest in later waves. For instance, of those who participated in wave 11, 48.0% of men and 44.2% of women reported being in good or very good general health. By contrast, within the imputed dataset, equivalent figures were just 40.4% and 34.8% respectively. Systematic differences between the observed (Table 6.2) and imputed data (Table 6.8) confirm that missing data were unlikely to have been MCAR, with complete-case analyses subject to attrition bias and the selection of a healthier sub-sample of participants.

6.7 Statistical power

Of the 3,413 women who participated at baseline, just 2,094 (61.4%) were free of prevalent T2DM at wave three and provided complete-case data concerning alcohol consumption, the confounding factors outlined in Section 6.5, and incident T2DM. Among these, only 247 incident cases were documented. There is therefore a possibility that statistical power may be insufficient to detect relatively small dose-response or interaction effects among women.³⁸¹

To establish the power of any subsequent survival model to detect a difference in survivor functions for a one-unit change in consumption among women, the -stpower- package was used.³⁸¹ This assumed a conventional alpha level of 0.05, a sample of 2,094 participants, and survival in the reference group equal to the average observed among all female participants (88.2%). The power to detect a 20% reduction in hazards per unit increase in consumption was estimated at just 0.39, indicating a 61% probability of type II error. Based upon the graph in Figure 6.3, a sample of ~5,900 women would be required to reach a conventional power

threshold of 0.80. These calculations thus indicate that low statistical power may present a particular limitation for dose-response analyses of female Whitehall II participants.





6.8 Summary of the dataset

Although numerous longitudinal cohorts are available, Whitehall II offers a number of benefits relevant to the study of longitudinal consumption trajectories and associations with T2DM. Perhaps of foremost importance, the cohort includes multiple waves of follow-up data for alcohol consumption and a wide range of potential confounders. The regular frequency of these follow-up data is such that acute fluctuations in health behaviours are more likely to be captured, increasing the accuracy of plotted trajectories and attempts to model risk with consideration to changes in participant characteristics across 50 years of the adult life course. Finally, the availability of an objective measure of T2DM reduces the potential for underestimating the incidence of T2DM relative to the use of self-reported data.

However, with a predominantly male sample, statistical power is likely to be low for analyses of women. Additionally, as with any cohort, a number of participants were lost to follow-up or failed to answer all questions included in self-completion questionnaires, with complete-case data available for just 65.5% of men and 53.4% of women (Table 6.4). Systematic differences in baseline characteristics are apparent according to the presence of unit or item non-response, with observed participants having reported a lower metabolic risk profile at the beginning of the

study than those with missing data. Complete-case analyses thus risked being subject to selection bias, sampling a healthier sub-set of the source population in a manner that may impact the validity of any dose-response relationship estimated. Although missing data have been imputed using chained equations and a broad range of ancillary variables, the use of imputed data within analytical models is currently limited. As noted in forthcoming chapters, these limitations included variability in the predicted values of stratification variables between imputations and an inability to use imputed event data within survival models. These restrictions were such that little difference is reported between results based on the observed and imputed datasets. In addition to restrictions concerning the range of statistical tests that can be applied to imputation models, including goodness-of-fit statistics, it was decided to report results based upon the observed data as the primary analyses, with results obtained from the imputed dataset reported in the appendices.

			Wave of	study		
Imputed variables	1	3	5	7	6	11
	% (95% CI)					
Men	n=6,895	n=6,814	n=6,691	n=6,506	n=6,274	n=5,980
Age						
Years	44.5 (44.4-44.7)	49.8 (49.7-50.0)	55.5 (55.4-55.6) ²	60.9 (60.8-61.1) ²	65.8 (65.6-65.9)*	69.6 (69.5-69.8) ²
Alcohol cons. frequency						
None in past year	3.2 (2.8-3.6)	3.3 (2.9-3.8)	3.1 (2.7-3.6)	3.3 (2.8-3.8)	4.1 (3.6-4.7)	2.7 (2.2-3.1)
<1/week	20.4 (19.5-21.4)	15.9 (14.9-16.8)	12.3 (11.4-13.1)	12.4 (11.5-13.3)	12.6 (11.7-13.6)	27.1 (25.6-28.5)
1-3 times/week	42.5 (41.3-43.6)	37.2 (36.0-38.5)	31.7 (30.4-33.0)	28.6 (27.3-29.9)	28.7 (27.4-30.0)	26.6 (25.1-28.0)
Daily or almost daily	33.9 (32.8-35.0)	43.6 (42.3-44.8)	52.9 (51.4-54.4)	55.8 (54.4-57.2)	54.6 (53.0-56.1)	43.7 (41.9-45.5)
Alcohol cons. volume						
g/week	63.2 (23.7-135.2) ^b	63.2 (23.7-138.7) ^b	94.8 (39.5-181.4) ^b	82.8 (37.5-158.0) ^b	71.1 (31.6-134.1) ^b	65.9 (27.0-126.8) ^b
BMI						
kg/m²	24.6 (24.5-24.6) ²	25.2 (25.1-25.3) ²	26.1 (26.0-26.2) ²	26.6 (26.5-26.7) ^a	26.6 (26.5-26.8) ²	26.6 (26.4-26.7)
Employment status						
Employed	100.0	90.1 (89.4-90.9)	67.1 (65.9-68.4)	53.0 (51.6-54.4)	32.8 (31.3-34.2)	19.3 (18.0-20.5)
Retired	•	7.5 (6.8-8.2)	25.9 (24.8-27.1)	41.2 (39.8-42.5)	62.8 (61.4-64.3)	77.6 (76.2-79.0)
Other ^c	•	2.4 (1.9-2.8)	6.9 (6.2-7.7)	5.8 (5.1-6.6)	4.4 (3.6-5.2)	3.1 (2.5-3.7)
Ethnicity						
White	92.0 (91.4-92.6)	92.0 (91.4-92.7)	92.1 (91.4-92.7)	92.2 (91.5-92.8)	92.1 (91.5-92.8)	92.3 (91.6-92.9)
South Asian	5.3 (4.8-5.9)	5.3 (4.8-5.8)	5.2 (4.7-5.7)	5.1 (4.6-5.6)	5.2 (4.6-5.7)	5.1 (4.6-5.7)
Other ^d	2.7 (2.3-3.1)	2.7 (2.3-3.1)	2.7 (2.3-3.1)	2.7 (2.3-3.1)	2.7 (2.3-3.1)	2.6 (2.2-3.0)

Table 6.8 Descriptive summary of imputed Whitehall II data, stratified by wave and sex

Family history of T2DM						
Yes	10.5 (9.7-11.2)	10.5 (9.8-11.2)	10.4 (9.6-11.1)	10.3 (9.6-11.1)	10.2 (9.5-11.0)	10.1 (9.4-10.9)
No	89.5 (88.8-90.3)	89.5 (88.8-90.2)	89.6 (88.9-90.4)	89.7 (88.9-90.4)	89.8 (89.0-90.5)	89.9 (89.1-90.6)
Occupational grade						
Administrative (top)	38.4 (37.2-39.5)	46.0 (44.9-47.2)	48.0 (46.8-49.2)	49.3 (48.1-50.5)	49.8 (48.5-51.0)	12.7 (11.9-13.6)
Professional (middle)	52.3 (51.1-53.5)	45.7 (44.5-46.9)	43.9 (42.7-45.1)	43.0 (41.8-44.2)	42.4 (41.2-43.6)	46.6 (45.4-47.9)
Clerical (bottom)	9.3 (8.6-10.0)	8.2 (7.6-8.9)	8.1 (7.4-8.8)	7.7 (7.0-8.3)	7.8 (7.2-8.5)	40.7 (39.4-41.9)
Physical activity ^e						
Inactive	9.7 (9.0-10.4)	15.1 (14.1-16.0)	4.8 (4.1-5.4)	7.5 (6.4-8.5)	26.0 (24.5-27.5)	18.9 (17.9-19.9)
Below guidelines	37.2 (36.1-38.3)	36.1 (34.8-37.3)	5.8 (5.1-6.5)	8.5 (7.6-9.3)	18.9 (17.8-20.1)	14.7 (13.8-15.6)
Met guidelines	53.1 (51.9-54.3)	48.9 (47.6-50.2)	89.5 (88.5-90.4)	84.0 (82.6-85.4)	55.1 (53.3-56.8)	66.4 (65.2-67.6)
Smoking						
Never	47.8 (46.6-49.0)	46.3 (45.1-47.6)	45.4 (44.1-46.6)	44.1 (42.8-45.4)	44.2 (42.6-45.9)	44.3 (42.7-45.9)
Former	36.3 (35.1-37.4)	39.7 (38.5-41.0)	43.6 (42.2-44.9)	46.4 (45.0-47.8)	48.3 (46.7-49.8)	51.0 (49.4-52.6)
Current	15.9 (15.1-16.8)	13.9 (13.0-14.8)	11.0 (10.1-11.9)	9.5 (8.6-10.4)	7.5 (6.6-8.4)	4.7 (3.8-5.7)
	% (95% CI)	% (95% CI)	% (95% CI)	% (95% CI)	% (95% CI)	% (95% CI)
Women	n=3,413	n=3,369	n=3,311	n=3,218	n=3,080	n=2,914
Age						
Years	45.7 (45.5-45.9) ^a	51.1 (50.9-51.3) ²	56.8 (56.5-57.0)*	62.1 (61.9-62.3) ^a	67.0 (66.8-67.3)*	70.9 (70.6-71.1)*
Alcohol cons. frequency						
None in past year	6.4 (5.6-7.2)	6.6 (5.7-7.5)	6.6 (5.5-7.6)	7.5 (6.5-8.6)	9.6 (8.4-10.8)	7.2 (6.1-8.3)
<1/week	39.5 (37.9-41.1)	32.2 (30.5-33.9)	25.4 (23.6-27.1)	27.5 (25.6-29.4)	26.6 (24.6-28.6)	44.8 (42.2-47.4)
1-3 times/week	34.3 (32.7-35.9)	35.5 (33.7-37.2)	32.5 (30.5-34.4)	30.0 (27.8-32.2)	30.4 (28.4-32.3)	23.4 (21.6-25.2)
Daily or almost daily	19.9 (18.5-21.2)	25.7 (24.1-27.3)	35.6 (33.5-37.7)	35.0 (32.7-37.2)	33.4 (31.3-35.5)	24.6 (22.4-26.8)
Alcohol cons. volume						
g/week	23.7 (0.0-55.3) ^b	23.7 (6.3-61.5) ^b	39.4 (13.3-83.8) ^b	34.4 (11.2-78.8) ^b	28.5 (7.9-63.0) ^b	26.6 (5.5-61.5) ^b

BMI						
kg/m ²	24.8 (24.6-24.9)	25.9 (25.7-26.0) ^a	26.8 (26.6-27.0) ²	27.4 (27.2-27.6)	27.5 (27.2-27.7)	27.3 (27.0-27.5)*
Employment status						
Employed	100.0	89.5 (88.3-90.6)	54.3 (52.2-56.3)	38.3 (36.2-40.4)	21.0 (19.1-22.8)	11.4 (9.9-12.9)
Retired		5.5 (4.7-6.4)	33.5 (31.6-35.4)	52.7 (50.6-54.9)	72.8 (70.8-74.8)	85.5 (83.7-87.2)
Other ^c	•	5.0 (4.1-5.9)	12.2 (10.9-13.6)	8.9 (7.5-10.4)	6.2 (4.9-7.6)	3.1 (2.1-4.1)
Ethnicity						
White	85.5 (84.3-86.7)	85.4 (84.2-86.6)	85.3 (84.1-86.5)	85.2 (84.0-86.5)	85.2 (83.9-86.5)	85.3 (84.0-86.6)
South Asian	6.6 (5.7-7.4)	6.6 (5.8-7.5)	6.7 (5.9-7.6)	6.8 (5.9-7.7)	6.8 (5.9-7.7)	6.9 (6.0-7.8)
Other ^d	7.9 (7.0-8.8)	7.9 (7.0-8.9)	8.0 (7.1-8.9)	7.9 (7.0-8.9)	8.0 (7.1-9.0)	7.8 (6.8-8.8)
Family history of T2DM						
Yes	14.1 (12.9-15.2)	14.1 (12.9-15.3)	14.2 (13.0-15.4)	14.2 (13.0-15.4)	14.0 (12.7-15.2)	13.7 (12.4-15.0)
No	85.9 (84.8-87.1)	85.9 (84.7-87.1)	85.8 (84.6-87.0)	85.8 (84.6-87.0)	86.0 (84.8-87.3)	86.3 (85.0-87.6)
Occupational grade						
Administrative (top)	11.2 (10.1-12.3)	13.8 (12.6-14.9)	15.3 (14.1-16.6)	16.5 (15.2-17.8)	17.1 (15.8-18.4)	5.0 (4.2-5.8)
Professional (middle)	39.1 (37.5-40.8)	42.1 (40.4-43.8)	42.0 (40.3-43.6)	42.5 (40.8-44.2)	42.3 (40.5-44.0)	19.4 (17.9-20.8)
Clerical (bottom)	49.7 (48.0-51.4)	44.1 (42.5-45.8)	42.7 (41.0-44.4)	41.0 (39.3-42.7)	40.6 (38.9-42.4)	75.6 (74.1-77.2)
Physical activity ^e						
Inactive	26.3 (24.8-27.8)	36.4 (34.5-38.2)	13.3 (11.6-14.9)	16.2 (13.8-18.6)	35.0 (32.7-37.3)	35.3 (33.5-37.0)
Below guidelines	38.2 (36.6-39.9)	33.5 (31.7-35.4)	10.2 (8.7-11.6)	11.4 (10.0-12.8)	22.5 (20.7-24.2)	19.3 (17.9-20.8)
Met guidelines	35.5 (33.9-37.1)	30.1 (28.4-31.9)	76.6 (74.5-78.7)	72.4 (69.8-75.1)	42.5 (40.0-45.0)	45.4 (43.6-47.2)
Smoking						
Never	53.2 (51.5-54.9)	51.6 (49.9-53.3)	51.2 (49.4-53.0)	50.7 (48.8-52.6)	52.9 (50.7-55.2)	53.4 (51.1-55.6)
Former	23.4 (22.0-24.8)	28.7 (27.0-30.3)	33.1 (31.2-34.9)	36.3 (34.4-38.2)	39.2 (37.0-41.5)	40.4 (38.1-42.6)
Current	23.4 (22.0-24.8)	19.8 (18.3-21.2)	15.8 (14.2-17.3)	13.0 (11.5-14.5)	7.9 (6.4-9.3)	6.2 (4.8-7.7)

	les could not be used in any statistical model.	artile range; ^c Redundant, student, long-term sick, long-term carer; ^d e.g. black Caribb
Wave-specific figures excluded participants who died.	No event data were reported as imputed dependent variable	^a Mean and 95% confidence interval; ^b Median and inter-quai

oean, African and Arabic; ^{*}Meeting guidelines (2150 minutes of moderate-intensity or 275 minutes of vigorous-intensity activity per week); inactive (<60 minutes of moderate and <60 minutes of vigorous activity; below guidelines (anyone not inactive or meeting guidelines). 2

Chapter 7

A preliminary conventional survival analysis

7 A preliminary conventional survival analysis

7.1 Introduction

Given changes to drinking behaviour across the life course as evidenced by multiple cohort studies,²⁶⁶ it is possible that conventional survival models that define intake according to a single baseline measure may have been simplistic in their approach and subject to misclassification error through a failure to consider the effect of variations in alcohol consumption over time.

Prior to exploring such shortcomings in more detail, this chapter reports preliminary analyses undertaken to confirm the suitability of the Whitehall II dataset for such an analysis. Specifically, results are reported from a series of conventional survival models in order to establish whether observed dose-response relationships within the Whitehall II cohort were consistent with those identified in the revised meta-analysis. Although a conventional survival analysis of T2DM risk has previously been published using Whitehall II data, pooled heterogeneous non-drinking categories were used with adjustment only for age and ethnicity.²⁴¹

Unlike the majority of existing studies, the conventional survival model reported in this chapter will also jointly examine two dimensions of alcohol consumption to ascertain whether drinking frequency represent an additional independent determinant of T2DM risk alongside the volume of consumption. The chapter will also explore the dose-response relationship between different drink types, given evidence discussed in Chapter 3 indicating the possibility of effect modification.

7.2 Objective

The objectives of the chapter were thus to:

- Undertake a conventional survival analysis, reporting the dose-response relationship between baseline categories of average weekly volume of alcohol consumption and T2DM risk.
- Investigate whether any interaction existed between the average volume of alcohol intake and the frequency of consumption.
- Establish whether independent dose-response relationships exist between drink type and T2DM risk.
- Compare results against those summarised as part of the revised meta-analysis in Chapter 3.

7.3 Hypotheses

Given that aetiological associations identified within Whitehall II are accordant with those found in general population cohorts,²⁹⁶ it was hypothesised that results from conventional survival analyses applied to Whitehall II data would be broadly similar to those reported in Chapter 3. Specifically, that:

- Reductions in T2DM risk will be greatest among women who consumed alcohol at moderate volumes;
- Any elevated risks will be most pronounced among former drinkers and those that reported a low frequency of alcohol intake.
- After accounting for consumption of other drink types, reductions in risk will be most pronounced among wine drinkers.

7.4 Methods

7.4.1 Sample

The sample was restricted to T2DM-free individuals who participated at baseline plus at least one other wave from which longitudinal time-to-event data could be determined. The baseline was defined as wave three, the first wave at which cases of T2DM were identified via both subjective and objective measures.

7.4.2 Variables

7.4.2.1 Alcohol consumption

With wave three representing the first period at which data were sufficient for the demarcation of never and former drinkers, participants were defined according to categories of alcohol consumption reported at wave three.

Using the volume and frequency variables described in Section 6.3, abstainers were categorised as either never drinkers or non-current drinkers. Never drinkers were defined as those who reported never having consumed alcohol over the year preceding participation at wave three and as being 'always a non-drinker'. To provide a stricter definition of never drinkers, this category excluded any participant who reported 'always being a non-drinker' at wave three yet also reported non-zero consumption at wave one (n=47), leaving a total 106 men and 111 women categorised as never drinkers. Non-current drinkers were thus defined as those who reported no alcohol consumption in the year preceding participation at wave three, and who reported not having always been a non-drinker. This category also included the 47 participants

who described themselves as never drinkers at wave three but reported non-zero consumption at wave one. The non-current drinkers were considered roughly analogous to a former drinking category and totalled 103 men and 78 women.

Current drinkers were defined according to their reported total volume of alcohol intake in the week preceding baseline. Male current drinkers were categorised as having consumed 0.1-50.0 g/week (n=1,692), 50.1-100.0 g/week (n=1,163), 100.0-150.0 g/week (n=725) or >150.0 g/week (n=1,316). For women, among whom the volume of alcohol consumption was lower, current drinkers were defined as having drunk 0.1-50.0 g/week (n=1,081), 50.1-100.0 g/week (n=387) or >100.0 g/week (n=323). Intervals of 50 g/week were chosen to strike a balance between being narrow enough to detect differences in T2DM risk across relatively small changes in the volume of alcohol consumption, and being broad enough to each capture an ample number of cases. Additionally, the use of regular intervals permitted a direct comparison of dose-response between men and women.

Alongside current drinkers, a separate group of 'infrequent drinkers' was defined, which represented participants who reported having consumed alcohol in the year preceding interview, but that did not drink alcohol in the week prior to measurement. The category of infrequent drinkers thus excluded participants who reported consuming alcohol at some point in the year prior to interview, but did not provide an answer to the questions concerning consumption in the last week; such individuals may have consumed alcohol in the last week, but missing data were such that this could not be determined. In total, 628 men and 597 women were classified as infrequent drinkers, among whom close to 85% reported drinking alcohol less than once a week, monthly or only on special occasions.

To test for an interaction between average weekly volume and the frequency with which alcohol was consumed over the week, a consumption pattern variable was defined according to whether or not participants reported 'daily or almost daily' alcohol consumption at wave three. This binary variable was chosen for consistency with previous studies that reported reductions in risk as greatest among participants that regularly consumed moderate volumes of alcohol throughout the week.^{95,97,240,244}

Finally, continuous variables denoting the weekly volume of consumption of different drink types were derived based on responses to questions concerning the number of "measures" of spirits, "glasses" of wine, or "pints" of beer or cider consumed. A glass of wine and a measure of

spirits were each assumed to contain one unit of alcohol (7.9 g), with two units of alcohol (15.8 g) in and a pint of beer or cider.

7.4.2.2 T2DM

As documented in Section 6.4, cases of T2DM were defined according to any self-reported doctor diagnosis or use of hypoglycaemic medication, or a positive FPG result following clinical examination. Analyses focussed upon incident cases of T2DM and so therefore excluded any cases prevalent at baseline.

As the number of participants predicted to have developed T2DM over the course of the study varied between imputations, analyses applied to the imputed dataset were restricted to participants with observed T2DM diagnosis data such that the number of participants remained static between imputations.³⁴²

7.4.2.3 Covariates

All covariates selected *a priori* were included in multivariable-adjusted models, operationalised as described in Section 6.5. Included covariates were age, BMI, ethnicity, employment status, family history of T2DM, occupational grade, physical activity, and smoking status. As with alcohol consumption, all covariates were defined at baseline, with the exception of ethnicity for which data were obtained from waves one and five.

7.4.3 Statistical analyses

7.4.3.1 Data synthesis

7.4.3.1.1 Cumulative hazard functions

Where cumulative hazard functions were reported, these were calculated using the Nelson-Aalen estimator, which plotted the cumulative hazard as a function of the number of participants at risk and the number of events observed at each point in time. Differences in cumulative hazard functions were tested using the log-rank test of equality.

7.4.3.1.2 Categorical dose-response

In keeping with methods applied by conventional analyses, Cox proportional hazards models were used to investigate the relationship between categories of alcohol consumption and T2DM risk. At its most simple, the Cox survival model is expressed according to Formula 7.1.³⁸²

 $h_i(t)=h_0(t)\exp\{\beta x_i\}$

Formula 7.1 Calculation of a Cox proportional hazard model

Here, the hazard of T2DM was calculated as a function of the baseline hazard of the sample at a given point in time ($h_0(t)$), which represented the hazard for a participant when all covariates were equal to zero, plus the effect of substantive covariate values (x_i) and their corresponding coefficients (β).

Results were reported separately for men and women based on apparent sex-specific differences in dose-response reported by the revised meta-analysis (Chapter 3). For each sex, age-adjusted and multivariable-adjusted dose-response relationships were described. Although few in number, never drinkers were selected as the reference group of choice for consistency with analyses undertaken as part of the revised and updated meta-analysis reported in Chapter 3, permitting a direct comparison of dose-response.

7.4.3.1.3 Frequency interaction

To investigate whether the effect of weekly volume of alcohol consumption differed according to the frequency of intake, a multiplicative interaction between the weekly volume of alcohol consumption and drinking frequency was included in the model. A categorical drinking variable denoting graduated volumes of intake was not used as part of the interaction analyses due to the small number of participants and cases when categories were divided according to values of the frequency variable. Instead, the weekly volume of alcohol consumption was modelled as a continuous variable, with a binary frequency variable denoting whether or not participants consumed alcohol on a daily or almost daily basis.

Non-current and infrequent drinkers were excluded from the interaction analysis so that relationships were calculated relative to never drinkers, consistent with preceding models. However, a sensitivity analysis was also run to examine the effect of their inclusion in a multivariable-adjusted interaction model.

Hazards were reported per 10 g/week increase in consumption and robust Huber-White standard errors were utilised for the calculation of confidence intervals due to the positive skewness of the continuous alcohol consumption variable.^{383,384,385} These standard errors are robust to the heteroscedasticity of residuals when modelling skewed variables, and thereby account for the underestimation of variance that can occur in its presence.

Models that incorporated a continuous measure of alcohol intake assumed a linear doseresponse relationship with T2DM incidence in conflict with the J-shaped associations previously observed. Accordingly, the possibility of a non-linear relationship was explored by raising the continuous measure of average volume of alcohol consumption according to a range of fractional powers, as per the method described in Section 3.2.4.1. The fit of the non-linear doseresponse models was then compared against the original linear model.

7.4.3.1.4 Drink type

To establish the association between different drink types and the risk of T2DM, the weekly volume of alcohol consumption from wine, spirits and beer or cider were each included concurrently within a multivariable-adjusted model as separate continuous variables. Consumption of each drink type was not categorised in an effort to avoid small sub-group sample size and low precision – a limitation inherent to a number of existing studies to have explored the effect of drink type upon T2DM risk.^{97,219,245} No adjustment was made for total consumption to avoid overadjustment in instances where participants only consumed a single type of drink at baseline.

As described above, these models assumed a linear relationship between drink type and T2DM risk. Accordingly, the volume of each drink type was raised to a series of fractional powers as described in Section 3.2.4.1. Results from the best-fitting models were reported.

7.4.3.2 Goodness of fit

To determine which models best fit the underlying data, the Bayesian information criterion (BIC) was calculated in each instance. Developed as a tool to aid model selection, the BIC was calculated as follows:³⁸⁶

$BIC=(-2^{*}ln(\ell))+(ln(n_{T}))(n_{p})$

Formula 7.2 Calculation of the Bayesian information criterion

Here, ℓ refers to the likelihood function, or the probability of having reached the calculated effect estimates given the included parameters. As per the deviance statistic described in Section 3.2.4.1, a perfectly specified model would report a value of zero. The log of the total number of observations (or in the case of nested data, the effective sample size) was expressed as n_T , while n_p denoted the total number of included parameters. Unlike the likelihood function, which always improves when new parameters are introduced, the BIC includes a penalty to models of greater complexity due to their reduced parsimony and increased risk of overfitting, risking coefficients having modelled random error specific to the Whitehall II dataset and thereby reducing the generalisability of findings to other samples.

Likelihood and thereby BIC statistics could not be reported for analyses of imputed data, as the final estimates were not drawn from a single model but from identical models applied to

multiple different datasets.³⁴² At the time of writing, there was no clear solution to the derivation of summary fit statistics for analyses of imputed data.³⁸⁷ Though one group of researchers proposed a method by which likelihood statistics could be averaged across imputations,³⁸⁸ the package was not compatible with the latest -mi- commands and applicable only to a restricted range of regression models (i.e. linear, logistic and ordered logistic).³⁸⁹ Simpler alternative recommendations included documenting the individual fit statistics for one, a sub-sample or all constituent imputation models.³⁸⁷ Due to the time costs involved in calculating and storing 50 fit statistics for each statistical model, the mean and range of log-likelihood and BIC statistics are reported in each case according to just the first three imputations.

7.4.3.3 Proportionality assumption

Standard survival models operate under an assumption of proportional hazards: that relative differences in hazards between categories of volume of alcohol consumption are proportionate over time compared to the hazard of the reference group, such that the relative hazard of T2DM for a given category is approximately equal at all time points and thus suitable for reporting according to a single hazard ratio. This assumption was tested in each sex-specific multivariable-adjusted model through the inclusion of an interaction between categories of weekly alcohol consumption and a linear expression of time. Any violations of proportionality are reported.

7.4.3.4 Statistical package

Analyses of observed data were undertaken using the -st- package of commands available in Stata 13,³⁹⁰ while results from analyses of imputed data were reported using the -mi- suite.³⁴² As Stata 13 did not permit the use of imputed diagnosis data due to differences in the predicted number of cases and duration of follow-up between imputations, analyses of both observed and imputed datasets were both restricted to participants with observed T2DM diagnosis data only. Similarly, due to the small number of never and non-current drinkers, it was not possible to derive imputed alcohol consumption categories at wave three; no imputation model would converge even when the number covariates against which never drinking was predicted were drastically reduced. Accordingly, imputed analyses reported in this chapter accounted only for attrition due to missing covariate data.

7.5 Results

7.5.1 Descriptive statistics

A total 5,456 men and 2,434 women provided valid alcohol consumption and T2DM data, with a median 20.2 and 20.0 years of follow-up respectively. Of those with valid alcohol consumption and T2DM diagnosis data, a total 589 incident cases of T2DM were documented among men and 279 cases among women, or 10.8% and 11.5% of the sample. In multivariable-adjusted analyses, complete-case data were available for 4,869 men and 2,094 women, among whom 527 and 247 cases of T2DM were documented. Women appeared to develop T2DM at a slightly faster rate over time (p=0.078, Appendix 7.1).

Notable differences in metabolic risk profile were evident at baseline between those that did and did not develop T2DM (Table 7.1). Consistent with findings from the literature review reported in Section 2.1.3, those who developed T2DM over the course of the study tended to be older, had higher values of BMI, were more likely to have been of Asian ethnicity, have a family history of T2DM, poorer self-reported general health, lower physical activity or be of lower occupational grade. The same differences were evident within the imputed dataset (Appendix 7.2).

Differences in crude cumulative hazards were present across categories of weekly volume of alcohol consumption, particularly among women (men: p=0.090; women: p=<0.001). Male noncurrent and never drinkers exhibited the greatest incidence of T2DM over the course of the study, with cumulative hazards roughly equivalent among current and infrequent drinkers (Appendix 7.3). By contrast, while female current drinkers also had a lower overall incidence of T2DM than non-current and never drinkers, cumulative hazards were greatest among those defined as infrequent drinkers (Appendix 7.3). For both men and women, the cumulative hazards were greatest among those who reported no consumption or a frequency of less than one occasion a week over the year preceding baseline interview (Appendix 7.4).

As might be expected given their higher incidence of T2DM, non-current and never drinkers exhibited a worse metabolic risk profile at baseline than current drinkers, with male and female abstainers having poorer self-reported general health, lower physical activity, a lower occupational grade, a greater proportion of South Asian participants and a higher prevalence of family history of T2DM than current drinking categories (Table 7.2). Similarly, differences in the baseline characteristics of current drinkers and infrequent drinkers were most pronounced among women. Specifically, female infrequent drinkers tended to be older, have a higher BMI,

lower occupational grade, lower physical activity, poorer self-reported general health, and a greater proportion of South Asian participants and individuals who were out of work for reasons such as long-term illness. In all cases, imputed descriptive data were comparable (Appendix 7.5).

		Men			Women	
	T2DM	Censored D	oifference ^a	T2DM	Censored	Difference ^a
Variables (wave 3)	% (95% CI)	% (95% CI)		% (95% CI)	% (95% CI)	
	c	c		c	c	
Age						
Mean years	50.4 (49.9-50.9) ^b	49.8 (49.6-49.9) ^b	0.017	52.1 (51.3-52.8) ^b	50.7 (50.5-51.0) ^b	<0.001
	620	5.103		296	2.274	
Alcohol consumption frequency						
None in past year	5.1 (3.6-7.2)	3.2 (2.7-3.7)	0.005	10.0 (7.0-14.1)	6.6 (5.6-7.7)	<0.001
	30	155		28	142	
<1/week	19.8 (16.8-23.3)	16.7 (15.6-17.7)		51.8 (45.9-57.6)	33.1 (31.1-35.1)	
	117	812		145	713	
1-3 times/week	34.7 (31.0-38.7)	40.2 (38.8-41.6)		25.7 (20.9-31.2)	36.4 (34.4-38.5)	
	205	1,959		72	785	
Daily or almost daily	40.3 (36.4-44.4)	40.0 (38.6-41.3)		12.5 (9.1-16.9)	23.9 (22.2-25.8)	
	238	1,947		35	516	
Alcohol consumption volume						
Never drinkers ^c	2.9 (1.8-4.6)	1.5 (1.2-1.9)	0.183	5.4 (3.2-8.7)	4.0 (3.2-4.9)	<0.001
	17	75		15	86	
Non-current drinkers ^d	2.2 (1.3-3.8)	1.6 (1.3-2.0)		4.6 (2.7-7.9)	2.6 (2.0-3.4)	
	13	80		13	56	
Infrequent drinkers ^e	10.3 (8.1-13.1)	11.1 (10.2-12.0)		37.1 (31.6-43.0)	20.9 (19.2-22.6)	
	61	540		104	450	
0.1-50.0g/week	27.6 (24.2-31.4)	29.6 (28.4-30.9)		38.9 (33.4-44.8)	42.7 (40.6-44.8)	
	163	1,445		109	920	
50.1-100.0g/week	19.5 (16.5-22.9)	20.5 (19.4-21.7)		8.6 (5.8-12.5)	16.0 (14.6-17.7)	
	115	1,000		24	346	
100.1-150.0g/week	12.7 (10.3-15.7)	12.9 (12.0-13.8)		5.4 (3.2-8.7)	13.9 (12.5-15.4)	
	75	627		15	299	
>150.0g/week ^f	24.7 (21.4-28.4)	22.7 (21.6-23.9)		ı	ı	
ì	146	1,107				

Table 7.1 Baseline characteristics of participants free of T2DM at wave three and with valid follow-up
data, stratified by sex. Observed data.

BMI						
Mean kg/m²	26.9 (26.6-27.2) ^b 620	24.9 (24.8-25.0) ^b 4,729	<0.001	29.5 (28.8-30.2) ^b 280	25.1 (24.9-25.3) ^b 2,053	<0.001
Employment status						
Employed	90.8 (88.3-92.8) 563	90.9 (90.1-91.7) 4,641	0.033	88.5 (84.3-91.7) 262	90.7 (89.4-91.8) 2,062	0.213
Retired	6.3 (4.6-8.5) 39	7.5 (6.8-8.2) 382		5.4 (3.3-8.7) 16	5.4 (4.6-6.4) 123	
Other ^g	2.9 (1.8-4.6) 18	1.6 (1.3-1.9) 80		6.1 (3.9-9.5) 18	3.9 (3.2-4.8) 89	
Ethnicity						
White	84.8 (81.8-87.5) 526	94.6 (93.9-95.2) 4,815	<0.001	71.2 (65.7-76.1) 210	89.1 (87.7-90.3) 2,012	<0.001
South Asian	11.8 (9.5-14.6) 73	3.4 (3.0-4.0) 175		14.6 (11.0-19.1) 43	4.6 (3.8-5.5) 103	
Other	3.4 (2.2-5.1) 21	2.0 (1.6-2.4) 101		14.2 (10.7-18.7) 42	6.4 (5.4-7.5) 144	
Family history of T2DM						
No	81.1 (77.8-84.1) 495	91.2 (90.4-91.9) 4,589	<0.001	68.8 (63.1-73.9) 198	89.2 (87.8-90.4) 1,989	<0.001
Yes	18.9 (15.9-22.2) 115	8.8 (8.1-9.6) 444		31.3 (26.1-36.9) 90	10.8 (9.6-12.2) 242	
Occupational class						
Administrative (top)	42.4 (38.6-46.4) 263	49.9 (48.5-51.3) 2,546	<0.001	5.4 (3.3-8.7) 16	17.5 (16.0-19.1) 398	<0.001
Professional (middle)	48.5 (44.6-52.5) 301	44.3 (43.0-45.7) 2,263		42.2 (36.7-48.0) 125	45.9 (43.8-47.9) 1,043	
Clerical (bottom)	9.0 (7.0-11.6) 56	5.8 (5.2-6.4) 294		52.4 (46.6-58.0) 155	36.6 (34.7-38.6) 833	

Physical activity ^h						
Inactive	17.1 (14.3-20.4) 101	13.4 (12.5-14.4) 655	0.050	41.8 (36.1-47.7) 117	33.9 (32.0-36.0) 732	0.006
Below guidelines	35.6 (31.8-39.6) 210	36.8 (35.4-38.1) 1.791		33.9 (28.6-39.7) 95	33.2 (31.2-35.2) 716	
Met guidelines	47.3 (43.3-51.3) 279	49.8 (48.4-51.2) 2,426		24.3 (19.6-29.7) 68	32.9 (30.9-34.9) 709	
Self-reported general health						
Very good/Excellent	40.5 (36.6-44.5) 239	53.7 (52.3-55.1) 2,616	<0.001	31.1 (25.9-36.8) 87	43.5 (41.4-45.6) 937	<0.001
Good	45.1 (41.1-49.1) 266	37.9 (36.6-39.3) 1,848		44.6 (38.9-50.5) 125	42.8 (40.7-44.9) 922	
Fair/Poor	14.4 (11.8-17.5) 85	8.3 (7.6-9.1) 406		24.3 (19.6-29.7) 68	13.7 (12.3-15.3) 296	
Smoking						
Never	38.5 (34.5-42.6) 215	46.1 (44.6-47.5) 2,123	<0.001	54.0 (47.9-60.0) 141	51.1 (48.9-53.3) 1,029	0.645
Former	42.2 (38.2-46.4) 236	41.9 (40.5-43.3) 1,929		28.7 (23.5-34.6) 75	31.2 (29.2-33.3) 629	
Current	19.3 (16.2-22.8) 108	12.0 (11.1-13.0) 555		17.2 (13.1-22.4) 45	17.7 (16.1-19.4) 356	
^a To explore differences between group exhibited a non-normal distribution, c consumption in the past week in wave consumption in the last week or in the	os, one-way ANOVA wa data were log-transfo es 1 and 3, no consum e last year; ^e Consume	as used on continuous of rmed prior to testing; ^b I aption in the past year i d alcohol in the past ye	data, and the cl Mean and 95% n waves 1 and ar but not in th	ii ² test on categorical confidence interval; ^c 3, and stated they hac e past week; ^f Among v	data (where continuou Those who reported no I 'always been a non-dri women, this category wa	s data nker'; ^d No s merged
ש שווו נווטא שווש ו דשטו ורש עווטא שווו שוווש ש moderate-intensity or ≥75 minutes of	vigorous-intensity ac	euuluant, suuent, ויטוא tivity per week); inactiv	e (<60 minutes). e	of moderate and <60	minutes of vigorous act	ivity;

below guidelines (anyone not inactive or meeting guidelines).

0.096 0.020 <0.001 <0.001 Difference 50.1 (48.8-51.4)^b 24.2 (23.5-24.9)^b ഹ I c 93 88 79 14 85 ŝ Never 91.4 (83.5-95.7) 3.2 (1.0-9.7) 5.4 (2.2-12.5) 84.9 (76.0-91.0) 15.1 (9.0-24.0) % (95% CI) 25.0 (24.3-25.7)^b c 50.6 (49.4-51.9)^b 88 Q 49 35 ∞ 92 84 2 Non-current 91.3 (83.4-95.6) 6.5 (2.9-14.0) 53.3 (42.9-63.4) 38.0 (28.6-48.5) % (95% CI) 2.2 (0.5-8.5) 8.7 (4.4-16.6) Alcohol consumption category c 50.2 (49.7-50.7)^b 601 24.7 (24.4-24.9)^b 568 528 541 Infrequent ົວ 59 14 32 26 2.3 (1.4-3.9) 90.3 (87.7-92.4) 4.3 (3.0-6.3) 87.9 (85.0-90.2) 9.8 (7.7-12.5) 5.3 (3.8-7.5) % (95% 4,248 4,436 49.7 (49.5-49.9)^b 4,678 25.1 (25.1-25.2)^b 4,480 353 152 Current c 77 82 90.8 (89.9-91.6) 7.5 (6.8-8.3) 1.6 (1.3-2.1) 95.0 (94.3-95.6) 3.3 (2.8-3.8) 1.8 (1.4-2.2) % (95% CI) **Employment status** Variables (wave 3) Mean kg/m² South Asian Mean years Employed Ethnicity Retired Other^c White Other Men Age BM

Table 7.2 Baseline characteristics of participants free of T2DM at wave three and with valid follow-up data, stratified by sex and categories of average weekly volume of alcohol consumption. Observed data.

Family history of T2DM					
No	90.5 (89.6-91.3)	88.9 (86.0-91.2) 	80.0 (70.3-87.1)	89.2 (81.0-94.2)	0.006
Yes	4,175 9.5 (8.7-10.4) 438	526 11.1 (8.8-14.0) 61	72 20.0 (12.9-29.7) 18	83 10.8 (5.8-19.0) 10	
Incident T2DM					
No	89.3 (88.4-90.2)	89.9 (87.2-92.0)	81.5 (72.1-88.3)	86.0 (77.2-91.8)	<0.001
Yes	4,179 10.7 (9.8-11.6)	540 10.1 (8.0-12.8)	75 18.5 (11.7-27.9)	80 14.0 (8.2-22.8)	
	499	61	17	13	
Occupational class					
Admi nis trative (top)	52.2 (50.8-53.7)	31.3 (27.7-35.1)	27.2 (18.9-37.3)	35.5 (26.3-45.9)	<0.001
	2,444	188	25	33	
Professional (middle)	43.0 (41.6-44.4)	56.9 (52.9-60.8)	52.2 (41.8-62.3)	49.5 (39.3-59.7) 46	
Clerical (bottom)	2,012 4.7 (4.2-5.4)	242 11.8 (9.5-14.7)	40 20.7 (13.5-30.3)	40 15.1 (9.0-24.0)	
	222	71	19	14	
Physical activity ^d					
Inactive	12.5 (11.6-13.5)	22.2 (19.0-25.7)	18.5 (11.7-27.9)	22.8 (15.3-32.7)	<0.001
	585	133	17	21	
Below guidelines	36.6 (35.2-38.0)	36.2 (32.4-40.1)	42.4 (32.6-52.9)	35.9 (26.6-46.3)	
	T/17	/ 7		55 10 10 10 10 10 10 10 10 10 10 10 10 10 1	
iviet guidelines	(5.20-0.9) 2.00 2,381	41./ (3/.8-43./) 250	(0.44-0.42) I.45 36	(8.1c-0.15) 5.14 38	
Self-reported general health					
Very good/Excellent	53.1 (51.7-54.5)	46.7 (42.8-50.8)	56.5 (46.1-66.4)	43.5 (33.6-53.9)	<0.001
	2,483	280	52	40	
Good	38.9 (37.5-40.3)	38.4 (34.6-42.4)	31.5 (22.7-41.9)	37.0 (27.6-47.4)	
	1,821	230	29	34	
Fair/Poor	8.0 (7.2-8.8)	14.9 (12.2-17.9)	12.0 (6.7-20.5)	19.6 (12.6-29.1)	
	373	89	11	18	

Smoking					
Never	44.1 (42.6-45.5) 1,952	49.6 (45.5-53.8) 280	77.1 (66.6-85.0) 64	46.7 (36.5-57.2) 42	<0.001
Former	43.4 (41.9-44.8) 1,920	35.3 (31.4-39.3) 199	13.3 (7.4-22.6) 11	38.9 (29.2-49.5) 35	
Current	12.6 (11.6-13.6) 557	15.1 (12.3-18.3) 85	9.6 (4.8-18.3) 8	14.4 (8.5-23.5) 13	
<u>Women</u> Age	-	-	-	-	-
Mean years	50.3 (50.0-50.6) ^b 1,713	52.0 (51.5-52.5) ^b 554	50.8 (49.5-52.0) ^b 101	50.7 (49.2-52.1) ^b 69	<0.001
BMI					
Mean kg/m²	25.2 (25.0-25.4) ^b 1,613	26.7 (26.2-27.1) ^b 523	26.2 (25.1-27.3) ^b 95	27.5 (25.9-28.8) ^b 63	<0.001
Employment status					
Employed	91.2 (89.8-92.5) 1.563	86.3 (83.1-88.9) 478	91.1 (83.6-95.4) 92	84.1 (73.1-91.1) 58	0.021
Retired	5.7 (4.7-6.9) 98	6.0 (4.3-8.3) 33	4.0 (1.5-10.2) 4	5.8 (2.1-14.8) 4	
Other ^c	3.0 (2.3-4.0) 52	7.8 (5.8-10.3) 43	5.0 (2.0-11.5) 5	10.1 (4.8-20.1) 7	
Ethnicity					
White	92.3 (90.9-93.4) 1,573	82.2 (78.7-85.2) 452	42.0 (32.6-52.0) 42	59.4 (47.2-70.6) 41	<0.001
South Asian	2.3 (1.7-3.1) 39	7.8 (5.8-10.4) 43	41.0 (31.7-51.0) 41	20.3 (12.2-31.7) 14	
Other	5.5 (4.5-6.6) 93	10.0 (7.8-12.8) 55	17.0 (10.8-25.8) 17	20.3 (12.2-31.7) 14	

Family history of T2DM					
No	87.5 (85.8-89.0) 1,468	85.9 (82.7-88.6) 468	83.2 (74.4-89.4) 84	83.8 (72.8-90.9) 57	0.425
Yes	12.5 (11.0-14.2) 210	14.1 (11.4-17.3) 77	16.8 (10.6-25.6) 17	16.2 (9.1-27.2) 11	
Incident T2DM					
No	91.4 (89.9-92.6) 1,565	81.2 (77.7-84.3) 450	85.1 (76.6-90.9) 86	81.2 (69.9-88.9) 56	<0.001
Yes	8.6 (7.4-10.1) 148	18.8 (15.7-22.3) 104	14.9 (9.1-23.4) 15	18.8 (11.1-30.1) 13	
Occupational class					
Administrative (top)	21.0 (19.1-23.0) 359	5.2 (3.7-7.4) 29	5.0 (2.0-11.5) 5	5.8 (2.1-14.8) 4	<0.001
Professional (middle)	49.2 (46.8-51.6)	42.1 (38.0-46.2)	24.8 (17.2-34.3)	31.9 (21.8-44.0)	
Clerical (bottom)	843 29.8 (27.7-32.0)	233 52.7 (48.5-56.8)	25 70.3 (60.5-78.5)	22 62.3 (50.1-73.2)	
	511	292	71	43	
Physical activity ^d					
Inactive	29.9 (27.8-32.2)	45.5 (41.4-49.7)	51.5 (41.6-61.2)	46.4 (34.7-58.4)	<0.001
Below guidelines	35.9 (33.7-38.2)	262 26.7 (23.2-30.6)	22 25.7 (18.0-35.3)	32 31.9 (21.8-44.0)	
	615	148	26	22	
Met guidelines	34.2 (31.9-36.4)	27.8 (24.2-31.7)	22.8 (15.5-32.1)	21.7 (13.4-33.3)	
Self-reported general health	COC	4C1	62	CT	
Very good/Excellent	46.1 (43.8-48.5)	35.8 (31.9-39.9)	21.8 (14.7-31.1)	20.3 (12.2-31.7)	100 0V
	06/	198 198	77	14 14	
6000	42.1 (39.7-44.4) 720	(1.40.4-48.1) 246	(0.10-c.14) c.15 55	31.1 (20.8-49.9) 26	
Fair/Poor	11.8(10.4-13.4)	19.7 (16.6-23.2)	23.8 (16.4-33.2)	42.0 (30.8-54.2)	
	202	109	24	29	

Smoking					
Never	48.2 (45.7-50.6)	53.2 (48.9-57.5)	87.9 (79.3-93.3)	67.7 (54.8-78.4)	<0.001
Former	776 34.1 (31.9-36.5)	272 26.8 (23.1-30.8)	80 4.4 (1.6-11.3)	42 21.0 (12.4-33.2)	
	550	137	4	13	
Current	17.7 (15.9-19.6)	20.0 (16.7-23.7)	7.7 (3.7-15.5)	11.3 (5.4-22.2)	
	285	102	7	7	
Table excluded consumption frequen	icy owing to colinear	rity with consumptior	volume.		
^a To explore differences between grou	ps, one-way ANOVA	was used on continuc	ous data, and the chi ²	test on categorical data	(where
continuous data exhibited a non-nor	mal distribution, da	ta were log-transforn	ned prior to testing; ^b N	Aean and 95% confidence	e
interval; ^c Redundant, student, long-te minutes of vigorous-intensity activit, guidelines (anyone not inactive or m	erm sick, long-term c y per week); inactive eeting guidelines).	arer; ^d Meeting guidel (<60 minutes of mod	ines (≥150 minutes of erate and <60 minutes	moderate-intensity or ≥7 . of vigorous activity; belo	75 ow
7.5.2 Age-adjusted models

Although sex-specific differences in dose-response were evident from the meta-analysis reported in Chapter 3, the presence of such a disparity within Whitehall II was formally tested via an interaction was between sex and categories of baseline alcohol consumption. Following a likelihood ratio test of age-adjusted models with and without a sex interaction, the interaction term was found to provide a statistically significant improvement in fit (p=<0.001). This was consistent with results from the earlier meta-analysis and supported the case for the stratification of results by sex.

In age-adjusted analyses, male and female current drinkers showed significantly reduced risks of T2DM, relative to never drinkers (Table 7.3). Among men, risks were of a generally consistent magnitude across categories of current drinkers, at around a 41-50% reduction. Conversely, among women, an inverse dose-response relationship was apparent. Here, risks appeared to decline with each categorical increase in consumption, with a hazard ratio lowest among women who consumed >150.0 g/week (HR 0.28, 95% CI 0.14-0.58).

7.5.3 Multivariable-adjusted models

Adjustment for confounders improved the fit of both models and resulted in substantial attenuations to reductions in risk among current drinkers, with all estimates rendered statistically non-significant (Table 7.3). Among men, the largest reductions in risk were visible among those who reported consuming 0.1-50.0 g/week (HR 0.92, 95% CI 0.53-1.61). Above these volumes, risks were higher than the null except for among the heaviest drinkers (>150.0 g/week), for whom an insubstantial and likely artefactual 1% reduction in risk was observed (HR 0.99, 95% CI 0.56-1.79). For women, the risk of T2DM remained lowest in the highest drinking category, though attenuated from a 73% to a 45% reduction in risk relative to never drinkers (HR 0.55, 95% CI 0.24-1.24).

While not statistically significant, non-current drinkers showed a greater risk than never drinkers among both sexes (men: HR 1.38, 95% CI 0.65-2.94; women: HR 1.07, 95% CI 0.48-2.40). Further, while the risk of T2DM was marginally lower among male infrequent drinkers (HR 0.95, 95% CI 0.52-1.73), female infrequent drinkers showed an elevated risk (HR 1.44, 95% CI 0.79-2.62). Results from analyses that utilised imputed covariate data showed similar findings, though smaller reductions in risk were evident among female current drinkers (Appendix 7.6).

A sizeable change in dose-response was evident according to whether risks were adjusted only for age or all *a priori* risk factors concurrently. To identify the covariates most responsible for

the changes observed, a post-hoc analysis was undertaken for which groups of covariates were added in a mutually exclusive fashion: age only; age plus ethnic and genetic factors (ethnicity and family history of T2DM); age plus adiposity (BMI); age plus other lifestyle factors (physical activity and smoking status); and age plus socio-economic factors (employment status and occupational grade). For consistency between models, the post-hoc analysis was restricted to the multivariable-adjusted sample in Table 7.3. Among both sexes, the greatest improvement in fit is evident following adjustment for BMI and age, relative to age only, while coefficients vary markedly between models (Table 7.4).

	Ag	e adjusted		Mul	tivariable adjusted	
Alcohol consumption (wave 3)	Cases/non-cases	HR (95% CI)	p-value	Cases/non-cases	HR (95% CI)	p-value
Men						
Never drinkers ^a	17/75	(reference)		15/64	(reference)	
Non-current drinkers ^b	13/80	0.77 (0.38-1.55)	0.486	13/72	1.37 (0.64-2.93)	0.412
Infrequent drinkers ^c	60/533	0.53 (0.31-0.95)	0.022	51/467	0.95 (0.52-1.73)	0.870
0.1-50.0g/week	163/1,445	0.50 (0.30-0.85)	0.006	143/1,296	0.92 (0.53-1.61)	0.769
50.1-100.0g/week	115/1,000	0.50 (0.30-0.85)	0.008	105/880	1.02 (0.58-1.81)	0.944
100.1-150.0g/week	75/627	0.52 (0.31-0.85)	0.015	69/566	1.04 (0.57-1.88)	0.907
>150.0g/week ^d	146/1,107	0.59 (0.36-0.95)	0.038	131/997	0.98 (0.56-1.75)	0.958
Log likelihood			-4918			-4166
BIC ^d			9896			8494
Women						
Never drinkers ^a	15/86	(reference)		14/71	(reference)	
Non-current drinkers ^b	13/56	1.13 (0.54-2.38)	0.740	11/45	1.07 (0.48-2.40)	0.869
Infrequent drinkers ^c	103/448	1.16 (0.67-2.00)	0.591	86/386	1.44 (0.79-2.62)	0.240
0.1-50.0g/week	109/920	0.62 (0.36-1.06)	0.082	100/781	1.06 (0.57-1.97)	0.845
50.1-100.0g/week	24/346	0.37 (0.19-0.70)	0.003	22/297	0.81 (0.39-1.70)	0.584
>100.0g/week	15/299	0.28 (0.14-0.58)	0.001	14/267	0.55 (0.24-1.24)	0.151
Log likelihood			-2071			-1700
BICd			4190			3538
Age-adjusted models controlled for employment status, ethnicity, famil	baseline age only. Mu y history of T2DM, occ	ılti vari able-adjustı supational grade, p	ed models (hysical act	controlled for baseli ivity, smoking status	ne age plus baseline B s. Ethnicity was derived	AII, from
responses at waves one and five. ^a F	articipants who repor	ted no consumptio	n in the pa	st week in waves 1 a	nd 3, no consumption	n the past
year in waves 1 and 3, and stated the	iey had 'always been a	a non-drinker' at w	ave 3; ^b Par	ticipants who report	ted no consumption in	thelast
year but had not 'aways been a non	-drinker'; ^c Consumed	alcohol in the past	year but no	ot in the past week; ^d	Bayesian information	criterion.

Table 7.3 Age and multivariable-adjusted dose-response relationship between categories of averageweekly volume of alcohol consumption and T2DM, stratified by sex. Observed data.

	Model 1		Model 2		Model 3		Model 4		Model 5	
Alcohol consumption (wave 3)	HR (95% CI)	o-value	HR (95% CI)	p-value	HR (95% CI)	p-value	HR (95% CI)	p-value	HR (95% CI)	p-value
Men (n=4,869)										
Never drinkers ^a	(reference)		(reference)		(reference)		(reference)		(reference)	
Non-current drinkers ^b	0.86 (0.41, 1.80)	0.684	1.25 (0.59-2.64)	0.554	0.95 (0.45-2.01)	0.902	0.77 (0.36-1.61)	0.483	0.88 (0.42-1.85)	0.731
Infrequent drinkers ^c	0.51 (0.29, 0.91)	0.022	0.90 (0.50-1.62)	0.729	0.50 (0.28-0.90)	0.020	0.46 (0.26-0.81)	0.008	0.53 (0.30-0.95)	0.031
0.1-50.0g/week	0.48 (0.28, 0.82)	0.007	0.88 (0.51-1.52)	0.642	0.48 (0.28-0.81)	0.007	0.46 (0.27-0.78)	0.004	0.54 (0.32-0.93)	0.025
50.1-100.0g/week	0.51 (0.30, 0.88)	0.016	0.97 (0.55-1.70)	0.914	0.49 (0.29-0.85)	0.011	0.48 (0.28-0.83)	0.008	0.60 (0.35-1.03)	0.062
100.1-150.0g/week	0.53 (0.30, 0.92)	0.024	1.04 (0.58-1.86)	0.897	0.50 (0.29-0.87)	0.015	0.48 (0.27-0.85)	0.011	0.63 (0.36-1.11)	0.110
>150.0g/week ^d	0.58 (0.34, 0.99)	0.047	1.13 (0.65-1.97)	0.663	0.50 (0.29-0.86)	0.012	0.49 (0.29-0.85)	0.011	0.68 (0.39-1.16)	0.155
Log likelihood		-4344		-4289		-4242		-4322		-4327
BIC ^d		8747		8663		8552		8738		8747
Women (n=2,094)										
Never dri nkers ^a	(reference)		(reference)		(reference)		(reference)		(reference)	
Non-current drinkers ^b	1.14 (0.52-2.51)	0.749	1.40 (0.63-3.11)	0.408	0.94 (0.43-2.08)	0.886	1.11 (0.50-2.46)	0.791	1.24 (0.56-2.75)	0.587
Infrequent drinkers ^c	1.05 (0.60-1.85)	0.870	1.59 (0.87-2.88)	0.130	0.98 (0.56-1.73)	0.955	1.04 (0.59-1.84)	0.895	1.16 (0.66-2.05)	0.611
0.1-50.0g/week	0.62 (0.36-1.09)	0.099	1.02 (0.56-1.87)	0.949	0.69 (0.39-1.20)	0.188	0.63 (0.36-1.12)	0.118	0.75 (0.42-1.32)	0.314
50.1-100.0g/week	0.37 (0.19-0.71)	0.003	0.67 (0.33-1.37)	0.274	0.45 (0.23-0.88)	0.019	0.36 (0.18-0.72)	0.004	0.51 (0.26-1.01)	0.052
>100.0g/week	0.27 (0.13-0.57)	0.001	0.52 (0.24-1.15)	0.105	0.29 (0.14-0.60)	0.001	0.27 (0.13-0.58)	0.001	0.43 (0.20-0.93)	0.032
Log likelihood		-1803		-1770		-1738		-1800		-1791
BIC ^d		3651		3608		3529		3677		3658
Model 1 adjusted for age; Mod- adiposity (BMI); Model 4 adjus factors (employment status anc	el 2 adjusted for ag ited for age plus oth d occupational gra	ge plus fo ner lifest de). All v	or ethnic and genet yle factors (physic ariables were defi	cic facto al activ ned acc	rs (ethnicity and fa ity and smoking sti ording to informati	imily his atus); an ion avail	tory of T2DM). Moo d Model 5 adjuste able at baseline, a	del 3 adj d for age part froi	usted for age plus e plus socio-econor m ethnicity which v	nic vas
derived from responses at wav	es one and five. ^a Pa	ırticipan	ts who reported no	o consul	mption in the past	week in v	waves 1 and 3, no c	dunsuo	otion in the past yea	ar in
waves 1 and 3, and stated they	had 'always been a	a non-dri	nker' at wave 3; ^b l	Particip	ants who reported	no consi	umption in the last	year bu	t had not 'aways be	en a

non-drinker'; ^cConsumed alcohol in the past year but not in the past week; ^dBayesian information criterion.

Table 7.4 Iteratively-adjusted dose-response relationship between categories of average weeklyvolume of alcohol consumption and T2DM, stratified by sex. Observed data.

7.5.4 Frequency interaction

When multivariable-adjusted survival models included an interaction between the frequency and volume of consumption (Table 7.5), no relationship was observed among men for every 10 g/week increase in the weekly volume of alcohol intake (HR 1.00, 95% CI 0.99-1.01), while women showed a statistically significant 5% reduction in risk for each increment (HR 0.95, 95% CI 0.92-0.99). Among both sexes, the dose-response relationship between the volume of consumption and T2DM risk did not differ according to drinking frequency (men: p=0.361; women: p=0.450), with an interaction term having provided no improvement to the fit of any model. Results from analyses of imputed data are reported in Appendix 7.7, and also show no significant interaction among men (p=0.653) or women (p=0.324).

When non-current and infrequent drinkers were included (Appendix 6.8), the dose-response relationship was unchanged and interaction terms remained statistically insignificant (men: p=0.621; women: p=0.536). Results from analyses that utilised imputed data are reported in Appendix 7.9 and were comparable to those obtained from models applied to the observed data.

Given that models of continuous data assumed a linear dose-response relationship between the volume of alcohol consumption and T2DM, a range of non-linear transformations were tested to explore whether a non-linear association provided a better fit of the underlying data. The dose-response relationship was best described by a linear association.

Table 7.5 Multivariable-adjusted interaction between a continuous measure of average weekly volume of alcohol consumption and drinking frequency, stratified by sex and excluding non-current and infrequent drinkers. Observed data.

	Model 1		Model 2	
Alcohol consumption (wave 3)	HR (95% CI)	p-value	HR (95% CI)	p-value
Men (n=4,266)				
<u>Dose-response by volume</u> Per 10 g/week increase	1.00 (0.99-1.01)	0.522	0.99 (0.98-1.00)	0.164
Difference in risk by frequency Daily Non-daily	-	-	(reference)	0 151
Difference in dose-response by frequency			0.00 (0.00 1.00)	0.131
Daily Non-daily	-	-	(reference) 1.01 (0.99-1.04)	0.361
Log likelihood BIC ^a		-3601 7320		-3600 7334
Women (n=1,566) Dose-response by volume Per 10 g/week increase	0.95 (0.92-0.99)	0.025	0.96 (0.90-1.02)	0.160
Difference in risk by frequency Daily Non-daily	-	-	(reference) 1.01 (0.47-2.18)	0.980
<u>Difference in dose-response</u> <u>by frequency</u> Daily Non-daily	-	-	(reference) 0.96 (0.87-1.06)	0.450
Log likelihood BIC ^a		-988 2078	. ,	-1698 3519

Model 1 reported the linear dose-response relationship between volume alcohol consumption and T2DM. Model 2 included an interaction term between a continuous measure of volume alcohol consumption and whether participants reported drinking alcohol daily or less than daily over the year preceding interview. All models adjusted for baseline covariates: age, BMI, employment status, ethnicity, family history of T2DM, occupational grade, physical activity, smoking status. Ethnicity was derived from responses at waves one and five. All models excluded non-current drinkers from the reference level of exposure (0g/week).

7.5.5 Proportionality assumption

In multivariable-adjusted models, the association between categories of average weekly volume of alcohol consumption and the risk of T2DM did not vary overall as a function of time among men (p=0.820). However, among women, non-proportional hazards were present among very light and never drinkers at baseline (p=0.032). As shown in Appendix 7.10, while the risk of T2DM was higher among infrequent and very light drinkers than never drinkers at baseline, the

magnitude of this difference decreased by around 16-17% with every additional year of followup.

7.5.6 Drink type

As shown in Table 7.6, when consumption from each drink type was included concurrently within a single multivariable-adjusted model, no difference in linear dose-response was evident among male drinkers, as supported by a Wald test of equality between coefficients (p=0.300). Among women, reductions in risk were close to statistical significance across all drink types, but with the magnitude of reduction per 10 g/week increase in consumption being greatest for beer drinkers (HR 0.86, 95% CI 0.74-1.01). Nevertheless, a Wald test indicated no difference in dose-response by drink type (p=0.460). Results from the imputed dataset were comparable (Appendix 7.11). Non-linear dose-response associations were explored, but none provided an improvement in fit over the linear models.

 Table 7.6 Multivariable-adjusted dose-response relationship between categories of average weekly

 volume of alcohol consumption, drink type and T2DM, stratified by sex. Observed data.

	Men (n=4,869))	Women (n=2,0)94)
Alcohol consumption (wave 3)	HR (95% CI)	p-value	HR (95% CI)	p-value
Drink type				
Beer (per 10 g/week)	0.99 (0.98-1.00)	0.206	0.86 (0.74-1.01)	0.060
Spirits (per 10 g/week)	1.01 (0.99-1.03)	0.252	0.93 (0.86-1.01)	0.080
Wine (per 10 g/week)	1.00 (0.97-1.02)	0.642	0.96 (0.91-1.01)	0.095
Log likelihood		-4166		-1699
BIC ^a		8468		3521

Models reported results adjusted for all a priori covariates: age, BMI, employment status, ethnicity, family history of T2DM, occupational grade, physical activity, smoking status. Ethnicity was derived from responses at waves one and five. Models also included variables representing the volume of each drink type consumed during the week prior to interview. ^aBayesian information criterion.

7.6 Summary of results

Differences in metabolic risk were evident at baseline according to T2DM diagnosis, with those that developed the condition being older and more likely to have higher values of BMI, lower physical activity, poorer self-reported health, be of a South Asian ethnic background have a family history of T2DM than those who were not diagnosed (Table 7.1). Risk factors for T2DM were also found to be more common among never and non-current drinkers at baseline, who exhibited worse self-reported health, lower physical activity and a greater proportion of South Asian participants and family history of T2DM than current drinking categories (Table 7.2).

In age-adjusted survival analyses, both male and female current drinkers showed statistically significant reductions in the risk of T2DM, relative to never drinkers (Table 7.3). Reductions were of greatest magnitude among women, for whom a positive dose-response relationship was apparent. Following adjustment for confounding factors, reductions in risk were substantially attenuated, with all estimates rendered statistically non-significant (Table 7.3). While point estimates among men were all very close to the null, sizeable if non-significant reductions in risk were still reported among women at volumes in excess of 50 g/week. Among men, sizeable but statistically non-significant elevations in risk were present among non-current drinkers, while female infrequent drinkers exhibited the highest risk relative to never drinkers.

The dose-response relationship between the volume of alcohol consumption and T2DM risk did not appear to differ according to the frequency of consumption, with an interaction term according to daily or non-daily consumption providing no improvement to the fit of any survival model (Table 7.5). Additionally, no evidence was found in support of differences in doseresponse according to the type of alcoholic drink consumed, with no drink type exhibiting a statistically significant association with T2DM risk after accounting for the consumption of other types of drink (Table 7.11).

7.7 Limitations

Analyses based on both the observed and imputed datasets were restricted to T2DM-free participants at wave three who had valid T2DM diagnosis data. This necessarily omitted individuals who participated at wave three but were absent at all subsequent waves of follow-up such that their T2DM status could not be ascertained. As indicated in Table 6.5, those who did not participate for \geq 4 waves had a worse metabolic risk profile than those that participated across all waves, indicating that the incidence of T2DM was likely to have been underestimated, with analyses applied to a healthier sub-sample of the original cohort.

Additionally, given that participants with unit non-response across \geq 4 waves also reported significantly lower volumes of alcohol consumption at wave one than those with complete-case data (men: p=<0.001; women: p=<0.001), models were likely to have under-sampled lighter and potentially less healthy drinkers. Although attempts were made to impute missing alcohol consumption data, the small number of observed never drinkers was such that an imputation model would not successfully converge. Analyses of both the observed and imputed datasets therefore excluded participants with missing alcohol consumption data. Taken together, there was a possibility that the risk of T2DM associated with lower volumes of alcohol may be underestimated.

Chapter 7: A preliminary conventional survival analysis

By using a small, strictly-defined reference category, it was possible that statistical power was low, impairing the ability to detect differences in risk between exposed and unexposed categories. To explore this further, post-hoc power calculations were undertaken for the detection of differences in hazards between never drinkers and participants in the lowest current drinking category (0.1-50 g/week). Sex-specific survival probabilities and sample sizes across the two categories were used to calculate the power to detect a 20% reduction in hazards at a 0.05 level of significance if such a difference were present. Statistical power measured 0.59 among men and 0.39 among women, or a 41% and 61% probability of failing to detect a 20% difference in the hazard ratio. Contrary to expectation, however, the inclusion of non-current drinkers within the abstention reference category was likely to have done little to remedy the issue. Specifically, statistical power fell to 0.56 among male participants and rose to 0.44 among female participants when non-current drinkers were combined with never drinkers. These oppositional changes in power were attributed to sex-specific differences in the survival probabilities of never and non-current drinkers such that, when combined, the male pooled nondrinking survival probability was drawn closer to that of the current drinking comparison group, while the female abstention survival probability was drawn further away. Despite the limited statistical power of categorical comparisons, dose-response relationships could nonetheless be visually examined by referred to point estimates reported across categories of alcohol consumption.

The breadth of consumption categories was limited by the positive skewness of the volume at which alcohol was consumed within the Whitehall II cohort, with the top categories necessarily defined as any consumption >150.0 g/week among men and >100.0 g/week among women. As such, any increased risk of T2DM as commonly seen among heavier drinkers was likely diluted within the top category by a large number of relatively moderate drinkers. This was especially likely among women, where the median consumption within the highest category was just 134.3 g/week (IQR 110.6, 165.9), or a little over one pint of 4% ABV lager per day.¹¹ Among men, median intake within the top consumption category was 221.2 g/week (IQR 173.8, 292.3). By capturing relatively few heavier drinkers, alcohol consumption data within Whitehall II was likely to have been inadequate for the detection of increases in risk at higher volumes of consumption, as otherwise reported by previous publications. This was further indicated by the lack of improvement to model specification when non-linear associations were tested (Section 7.5.3).

Adding to this issue was the possibility that the volume of alcohol consumption may have been subject to some degree of inaccuracy, with self-reported data limited by various reporting and

recall biases as well as measurement error.^{391,392} The issue of recall bias was posited to have been small, with the reliability of recall over seven-day periods found to be better than over longer retrospective episodes.³⁹³ Reporting biases were expected to have been a larger problem, though the magnitude, direction and determinants of any such bias appeared variable and multifaceted,³⁹¹ negating any real conclusion as to the degree to which reported results may have been under- or overestimated on average. Nevertheless, in one study that investigated the accuracy of participant self-reports, measures were increasingly reliable within older age categories, with no apparent difference between the sexes.³⁹⁴ Accordingly, the accuracy of selfreported alcohol consumption data was posited to have improved over time as the constituent sample grew older.

A further issue concerned the reliability of quantity-frequency questionnaires for the estimation of alcohol consumption. Adopted within Whitehall II, such questionnaires have tended to produce lower drinking estimates than alternative graduated frequency questionnaires, which ask participants to estimate the frequency with which they consumed alcohol at pre-specified volumes.³⁹¹ In this regard, reported volumes of alcohol consumption may have been underestimated.

7.8 Discussion

Age-adjusted analyses reported in Table 7.3 indicate a statistically significant reduction in the risk of T2DM across nearly all categories of current drinking, relative to never drinkers. Among men, the magnitude of such reductions were roughly equivalent at all levels of consumption, at around 50%, while risks among women declined with each categorical increase in consumption. Consistent with results from the updated meta-analysis (Figure 3.8) and other studies,³⁹⁵ less pronounced dose-response associations were evident following adjustment for confounding factors (Table 7.3). Although all coefficients were rendered non-significant, it was possible that attenuated risk estimates were of magnitudes no longer sufficient to be detected as significant owing to low statistical power.

Looking instead to the point estimates reported by the multivariable-adjusted models, male and female non-current drinkers showed a greater risk of T2DM than never drinkers (Table 7.3), potentially supporting the case for having excluded former drinkers from any categorical abstention category. These elevated risks were greatest among men, suggesting that male former drinkers may be less healthy and more likely to have previously been heavy drinkers than their female equivalents. Infrequent drinkers exhibited the highest risk of T2DM among women (Table 7.3), calling into question claims that infrequent drinkers may represent a more

Chapter 7: A preliminary conventional survival analysis

appropriate reference group than never drinkers. In keeping with previous research,^{96,97,239} such an elevated risk might be expected were such women predominantly episodic heavy drinkers. By contrast, male infrequent drinkers showed little difference in risk, relative to never drinkers.

Among male current drinkers, multivariable-adjusted estimates were close or equal to the null. By contrast, female point estimates indicated an inverse relationship between the volume of alcohol consumption and T2DM risk. That reductions in risk may have been specific to women was in accordance with results from the updated meta-analysis (Figures 3.4 and 3.6), with sexspecific effects and a heightened risk among non-current drinkers both supporting hypotheses laid out at the beginning of the chapter.

As reported in Table 7.5, no difference in dose-response was detected according to whether participants consumed alcohol on a daily or non-daily basis, with effect sizes of negligible magnitude. It was possible that the choice of interaction variable represented a poor parameterisation of consumption pattern, failing to truly reflect episodic heavy drinking occasions. However, even if sufficient data were available as to permit the identification of episodic heavy drinkers, it was possible that the number of such individuals may be insufficient to detect an interaction. While the prevalence and odds of excessive episodic heavy drinking appear greatest within younger men and adults of lower socio-economic status or neighbourhood deprivation,^{396,397} Whitehall II predominantly samples middle-aged adults of higher socio-economic status. With an interaction between the volume and frequency of consumption either not present, undetectable or incorrectly specified, analyses in subsequent chapters would forgo any secondary analysis of differences in dose-response according to reported consumption frequency.

The body of current evidence seems to suggest that reductions in risk may be specific to or at least most pronounced among wine drinkers³⁹⁸ – an effect hypothesised to be conferred through the effect of anti-inflammatory compounds common to fruit-based drinks.^{280,281} However, when the average weekly volume of alcohol consumption from each drink type was modelled concurrently, no statistically significant difference in the dose-response relationship was found between drink types for either sex. Of the effect estimates reported, there was a weak indication that beer may be the most strongly associated with a reduction in T2DM risk among women, contrary to other observational studies. While recent publications suggest the beer-specific compound xanthohumol may have an anti-inflammatory effect, preliminary results are currently only available from *in vitro*³⁹⁹ or *in vivo* animal studies.⁴⁰⁰ It was possible that the absence of any significant difference in dose-response by drink type may have been a consequence of having

constrained the models to report linear dose-response associations with T2DM, masking reductions in risk at very low or moderate volumes as a consequence of increases in risk at higher levels of intake. However, as with total volume in Section 7.5.4, transforming consumption from each drink type according to a range of fractional powers provided no improvement in fit relative to a standard linear survival model.

Concordance between the conventional survival analysis and findings from the updated metaanalysis suggest that Whitehall II represents a cohort suitable for further investigating the relationship between alcohol consumption and T2DM. Before constructing survival models capable of accounting for changes to consumption over time, the following chapter first explores the stability of alcohol intake across the life course and whether longitudinal trajectories differ by T2DM diagnosis. Chapter 7: A preliminary conventional survival analysis

Chapter 8

Trajectories of alcohol consumption

8 Trajectories of alcohol consumption

8.1 Introduction

Having established the suitability of Whitehall II for exploring the relationship between the alcohol consumption and T2DM within a conventional survival framework, this chapter begins by describing whether drinking among men and women within the cohort was constant over time. With survival analyses having almost exclusively modelled T2DM risk according to a single cross-sectional measure of consumption, this chapter goes on to report the longitudinal drinking trajectory as stratified according to categories of baseline alcohol consumption. Such analyses will help to highlight the degree of any potential misclassification error that may be inherent to a conventional survival approach, and thereby the validity of results reported by such analyses.

In conflict with findings from Mendelian randomisation studies,⁸⁵ results from the revised metaanalysis of current observational studies (Chapter 3) and a preliminary survival analysis of Whitehall II data (Chapter 7) show a significant and sex-specific dose-response relationship between alcohol consumption and T2DM. Despite observational research indicating reductions in risk among female moderate drinkers and increases in risk at high volumes of consumption among both sexes, little is currently known about how drinking changes over time among those that do and do not develop T2DM. Accordingly, this chapter also reports the longitudinal trajectory of alcohol consumption as stratified by sex and T2DM diagnosis. These results will help shed light upon whether any increases or reductions in risk as reported by current survival analyses were likely to have accrued gradually over the life course as a result of prolonged heavy or moderate consumption, or occurred as a consequence of differences in intake specific to periods of heightened biological sensitivity.⁴⁰¹ For instance, given apparent deteriorations to the alcohol metabolism with increased age,^{196,197,198,199} older age may represent a period of the life course in which alcohol consumption may have particularly manifest effects upon T2DM risk.

8.2 Objectives

The objectives of this chapter are thus to:

- Determine whether sex-specific trajectories of the mean weekly volume of alcohol consumption were constant over the captured adult life course.
- Report the longitudinal stability of alcohol consumption within categories defined according to baseline intake.

• Describe the nature of any differences in the longitudinal trajectory alcohol consumption according to whether or not participants develop T2DM.

8.3 Hypotheses

Based on existing research that combined longitudinal measures of alcohol consumption from multiple cohorts,²⁶⁶ it was expected that the mean volume of alcohol consumption would peak around middle-age before gradually declining thereafter. When stratified according to baseline drinking categories, it was further hypothesised that any decreases in consumption would be most pronounced among heavier baseline drinkers, with non-drinkers representing the most stable baseline category. This supposition was based on results from the Health Professionals Follow-up Study, which reported changes in alcohol consumption over a four-year period.²⁶⁷ Here, study participants were split into three categories according to their alcohol consumption at baseline, representing non-drinkers (0 g/week), moderate drinkers (<105.0 g/week) and heavier drinkers (>105.0 g/week). Stability was greatest among non-drinkers, among whom 93.4% maintained their abstention over the period. By contrast, stability was lowest among heavier drinkers, with 55.8% having reduced their consumption over time.

Regarding differences in the trajectory of alcohol consumption according to the diagnosis of T2DM, a number of plausible differences were hypothesised:

- Firstly, if the risk of T2DM accumulates over time as a result of chronic heavy drinking, the trajectory of alcohol consumption among those who developed T2DM would have been consistently or else predominantly higher on average than among those that did not develop the condition, as illustrated in Figure 8.1a. Similarly, with reductions in risk apparent at more moderate volumes among women (Chapter 3), it was posited that women without T2DM would exhibit a markedly lower volume of consumption over time.
- Secondly, given evidence indicating deteriorations to the alcohol metabolism with increased age,^{196,197,198,199} older age may represent a period during which any deleterious effect of alcohol consumption upon T2DM risk may be particularly pronounced. Under this alternative assumption, and with the incidence of T2DM greatest in older age,⁴⁰² there was a possibility that any disparity in drinking between those that do and do not develop the condition may be greatest during a period immediately preceding diagnosis, such as per the example illustrated in Figure 8.1b. Specifically that, regardless of differences in alcohol consumption earlier in the trajectory, the average volume of

alcohol intake would be greatest among those who developed the condition in the few years prior to diagnosis.

Finally, with a growing number of studies linking the onset of ill-health to a subsequent cessation or attenuation of alcohol consumption,^{144,145,146,141} participants who developed T2DM may exhibit a marked decline in their consumption in line with a gradual deterioration in health status prior to diagnosis (Figure 8.1c). Such a downward trajectory may represent either a proactive attempt by participants to improve their health, or else a response to medical advice or pharmaceutical contraindication following a diagnosis of impaired glucose tolerance or other T2DM risk factors such as obesity and high blood pressure.



Figure 8.1 Hypothesised differences in average alcohol consumption trajectory according to T2DM diagnosis

8.4 Methods

8.4.1 Sample

8.4.1.1 Prospective trajectories of mean weekly volume of alcohol consumption

To model the mean trajectory of alcohol consumption, the sample was defined as any participant who reported their volume of alcohol consumption in the week prior to interview at any point over the course of study since its inception (wave one). Concerned solely with describing changes to the volume of alcohol consumption with increased age, participants were included irrespective of T2DM diagnosis.

8.4.1.2 Trajectories of mean weekly volume of alcohol consumption by T2DM diagnosis

Exploring differences in the trajectory of alcohol consumption according to the development of T2DM, the sample was defined as any participant free of prevalent T2DM at wave three and who reported their volume of alcohol consumption in the week prior to interview at any point over the course of the study since its inception and also participated in at least one clinical examination after wave three such that their diagnosis status could be established.

8.4.2 Variables

8.4.2.1 Alcohol consumption

Alcohol consumption was captured as described in Section 6.3 and treated as a continuous variable. Participants who consumed 0 g/week in the week prior to interview thus comprised an assortment of never, non-current (former) and infrequent drinkers. No distinction between these groups was made in primary analyses that plotted the mean drinking trajectory. If a sick quitter effect were present within the data, evidenced by a decline in consumption over time, then the exclusion of non-current or infrequent drinkers would risk an underestimation of any downward trajectory.

In secondary analyses that stratified the alcohol consumption trajectory according to drinking categories defined at wave one, the stratification variable was coded as per section 7.4.2.1. Never drinkers and non-current drinkers were combined into a single category owing to the absence of questions sufficient for disaggregating the two non-drinking groups at wave one.

Due to the inherent variation of imputed data, between-imputation differences were present when allocating participants to baseline consumption categories when alcohol consumption data were missing. Accordingly, it was inappropriate to stratify longitudinal alcohol consumption trajectories by baseline consumption category when imputed alcohol data were used. Analyses of the imputed dataset were thus restricted to participants with observed measures of baseline consumption, with models then capitalising upon imputed repeated measures of volume of alcohol consumption across waves.

8.4.2.2 T2DM

As documented in Section 6.4, cases of T2DM were defined according to any self-reported doctor diagnosis or self-reported use of hypoglycaemic medication, or a positive FPG result following clinical examination. Wave three was the first period of observation for which both subjective and objective measures were available, and thus represented the wave at which prevalent cases were identified and excluded.

8.4.3 Statistical analysis

8.4.3.1 Prospective trajectories of mean weekly volume of alcohol consumption

Seeking to explore how alcohol consumption changed over the life course, age in years was selected as the timescale of interest for this subset of analyses. Age was defined according to the date on which the self-administered questionnaire was completed. In instances where no such date was documented, the date of clinical examination was used. Where both dates were undocumented, the mean wave-specific date of self-administered questionnaire completion was assumed.

8.4.3.1.1 Linear mixed effects models

Although trajectories of alcohol consumption could have been calculated using a standard regression model, this approach treats repeated measures as a series of unique and independent data points, leading to an overestimation of precision.⁴⁰³ Accordingly, consumption trajectories were calculated using mixed effects models, which nest repeated measures within participants. This inflates standard errors proportionate to the magnitude of correlation between repeated measures, with degrees of freedom calculated according to the number of participants as opposed to the number of data points. The resulting random intercept model is expressed in general terms per Formula 8.1, with the subscript i denoting the participant and the subscript j denoting the repeated measure:

 $y_i(t_{ij}) = (\beta_0 + b_{0i}) + (\beta_1)t_{ij} + \varepsilon_{ij}$

Formula 8.1 Calculation of a linear random intercept model

The true predicted value of weekly alcohol consumption at the age of the jth measurement for the ith participant is therefore calculated as the estimated mean intercept (β_0) plus a random

Chapter 8: Trajectories of alcohol consumption

effect denoting the participant's predicted deviation from the mean intercept value (b_{0i}), with a fixed slope ($\beta_1(t_{ij})$) that equates to the estimated mean rate of change in alcohol consumption per unit of age. Random error was denoted by ϵ_{ij} , representing any remaining variation in alcohol consumption not explained by the model. The resulting random intercept model is illustrated in Figure 8.2a.

As shown in Figure 8.2a, although the nesting of repeated measures allows each participant their own intercept value, slopes are constrained and therefore parallel to the mean trajectory. With existing research indicating a variety of slopes,²⁵⁹ the assumption that any change in alcohol consumption was equal between participants was unlikely to reflect reality.

To capture any underlying differences in slopes between participants, a random slopes model was also constructed, which allowed each participant to exhibit their own rate of change in alcohol consumption per year increase in age. This was achieved by adding a random effects term to the fixed mean slope coefficient, which denotes the predicted deviation of each participant-specific slope from the mean slope ($b_{1i}(t_{ij})$). The resulting random slopes model is expressed per Formula 8.2, and illustrated in Figure 8.2b.

Formula 8.2 Calculation of a linear random slopes model

An unstructured covariance matrix was specified for the random slopes models, which allowed the within-participant covariance between the intercept and repeated measures to take any form.



The red and blue lines refer to the estimated trajectories for participants a and b respectively. The green line refers to the mean trajectory for both participants. The grey interval (t=1) refers to a period of one year.

Figure 8.2 An illustration of random intercept and random slopes models

8.4.3.1.2 Non-linear mixed effects models

Although the random slopes model permitted individual-level variation in the rate of change, shifts in alcohol consumption with increased age were assumed to be linear. Owing to the possibility that changes over the life course may have been non-linear, age was subjected to a series of polynomial transformations as per the methods outlined in Section 3.2.4.1. The fit of each non-linear model was then compared against that of the linear model. Where trajectories were better explained by a non-linear slope, the output of any such models were reported.

8.4.3.1.3 Stability of baseline categories

To explore the longitudinal stability of drinking within baseline categories of alcohol consumption, an interaction term was included between the rate of change and the baseline

category of alcohol consumption. Where a significant interaction was identified, trajectories of were stratified according to categories of baseline intake. These analyses were undertaken using fixed slope models due to a lack of variation between repeated measures when nested within baseline categories. This may have been attributable to a real lack of individual-level variation in the rate of change among participants within the same baseline category, or a consequence of stratified sample sizes being too small to achieve numerical integration.

8.4.3.2 Trajectories of mean weekly volume of alcohol consumption by T2DM diagnosis

Participants were separated into two groups according to whether or not they developed T2DM over the course of the study. The period of observation began at the date of diagnosis for those who developed T2DM, or the final date of participation for those who did not, and in each cases coded as year zero. For each participant the volume of alcohol consumption was then traced backwards to the very first reported measure. A time of -15 years thus represented a measure of alcohol consumption that was taken 15 years prior to diagnosis or censoring from the study.

As the number of participants predicted to have developed T2DM over the course of the study varied between imputations, analyses applied to the imputed dataset were restricted to participants with observed T2DM diagnosis data such that the number of participants remained static between imputations.³⁴²

8.4.3.2.1 Linear mixed effects models

Using methods described in Section 8.4.3.1.1, sex-specific trajectories were first calculated using linear mixed effects models with an interaction between T2DM diagnosis and time. Both fixed and random slopes were estimated, with results reported from the best-fitting model.

8.4.3.2.2 Non-linear mixed effects models

As per section 8.5.1.3, a number of non-linear trajectories were also explored. In this series of comparisons, transformations were restricted to quadratic and cubic exponents due to limitations concerning the range of transformations that can be applied to negative values of time. This was consistent with previous research.^{404,405} When calculating fit statistics for each non-linear mixed effects model, fixed slopes were assumed owing to issues of convergence when random slopes were expressed for some transformations. Results from the best-fitting models were reported allowing for random slopes.

8.4.3.2.3 Sick-quitter effects

Secondary analyses were undertaken to calculate the trajectory of alcohol consumption following diagnosis. These models were constructed in a piecewise manner, with models

constructed separately for consumption data prior to and after the date of diagnosis. Non-linear trajectories were explored in each instance as per the methods in section 8.5.1.3.

8.4.3.2.4 Stability of baseline categories

Finally, the longitudinal stability of alcohol consumption within baseline categories was explored per the methods outlined in Section 8.4.3.1.4.

8.4.3.3 Goodness of fit

Goodness of fit was reported according to the log-likelihood and BIC statistics, as described in Section 7.4.3.2. An improvement in fit was defined as any reduction in the BIC greater than or equal to a value of 10.⁴⁰⁶

8.4.3.4 Statistical package

Mixed models were calculated in Stata 13 using the -mixed- package.⁴⁰⁷ As per the conventional survival analysis reported in Chapter 7, robust Huber-White standard errors were utilised for the calculation of confidence intervals due to the positive skewness of the alcohol consumption variable in its natural form. Although the volume of intake could have been transformed, this would have complicated interpretation and the communication of results to general audiences.

8.5 Results

8.5.1 Prospective trajectories of mean weekly volume of alcohol consumption

8.5.1.1 Descriptive statistics

The weekly volume of alcohol consumption was measured across 31,342 person-observations among men and 13,765 person-observations among women, as reported by 6,882 and 3,402 participants respectively across a mean 4.6 and 4.0 waves. Within the imputed dataset, alcohol consumption was captured across 39,160 person-observations among men and 19,305 person-observations among women, representing 6,895 and 3,413 women across a mean 5.7 waves. The age of participants ranged from 34.1-83.6 years, capturing almost 50 years of the adult life course over the period of study.

8.5.1.2 Linear trajectories

Random slopes models best described the observed (Table 8.1) and imputed (Appendix 8.1) linear trajectory of alcohol consumption across the captured life course for both men and women. Covariance between the random intercepts and random slopes was negative, indicating a convergence of trajectories with increased age.

	Fit statistics	
Linear mixed models	Log-likelihood	BIC ^a
Men		
Intercept only	-184343	368717
Linear mixed model, fixed slopes	-184318	368678
Linear mixed model, random slopes	-184042	368146
Women		
Intercept only	-72552	145133
Linear mixed model, fixed slopes	-72531	145100
Linear mixed model, random slopes	-72500	145057
^a Bayesian information criterion.		

Table 8.1 Crude sex-specific linear trajectory of mean weekly volume of alcohol consumption betweenthe ages of 34-84 years: goodness of fit statistics. Observed data.

To confirm the suitability of stratifying drinking trajectories by sex, linear models were run with and without a sex interaction with age. The interaction term provided a statistically significant improvement in model specification (p=<0.001), supporting the case for sex stratification. As reported in Table 8.2, consumption at baseline was estimated as 110.5 g/week (95% Cl 107.2-113.8) among men and 48.5 g/week (95% Cl 46.0-51.0) among women, falling by an average 2.8 (95% Cl 1.8-3.9) g/week and 2.2 (95% Cl 1.5-3.0) g/week respectively per 10-year increase in age.

Intercept values were slightly higher within the imputed dataset among both sexes, with a steeper rate of decline reported among men than evident within the observed dataset (Appendix 8.2). Among women, the rate of change was so small as to be rendered statistically insignificant (p=0.352). Plotted linear trajectories from analyses of the observed dataset are shown in Figure 8.3, with trajectories based upon the imputed dataset reported in Appendix 8.3.

Mixed models	g/week (95% CI)	p-value
<u>Men (n=6,882)</u>		
Linear model		
Intercept	110.5 (107.2, 113.8)	< 0.001
Age	-2.8 (-3.9, -1.8)	<0.001
Non-linear model		
Intercept	91.8 (90.0, 95.7)	<0.001
Age ¹	12.2 (10.1, 14.4)	<0.001
Age ³	-0.9 (-1.0, -0.8)	<0.001
<u>Women (n=3,402)</u>		
Linear model		
Intercept	48.5 (46.0, 51.0)	<0.001
Age	-2.2 (-3.0, -1.5)	<0.001
Non-linear model		
Intercept	45.4 (43.4, 47.4)	<0.001
InAge	3.5 (1.7, 5.3)	< 0.001
Age ³	-0.2 (-0.3, -0.2)	< 0.001
Age coefficients refer to th	he change in the average volume of v	veekly alcohol

 Table 8.2 Crude sex-specific linear and non-linear trajectories of mean weekly volume of alcohol consumption between the ages of 34-84 years. Observed data.

Age coefficients refer to the change in the average volume of weekly alcohol consumption per 10-year increase in age.



Figure 8.3 Crude sex-specific linear trajectories of mean weekly volume of alcohol consumption between the ages of 34-84 years. Observed data.

8.5.1.3 Non-linear trajectories

A range of non-linear slopes were explored, with fit statistics for each non-linear transformation reported in Appendices 8.4 and 8.5. When applied to both the observed and imputed datasets, non-linear slopes were found to provide a substantial improvement in fit over the linear models. Coefficients from the best-fitting non-linear models are documented in Table 8.2 for the observed dataset and Appendix 8.2 for the imputed dataset, with trajectories plotted in Figure 8.4 and Appendix 8.6 respectively.

Within the observed dataset, the volume of alcohol consumption was consistently highest among men. The mean weekly volume of alcohol consumption increased up to around 56 years of age among men and 51 years of age among women, peaking at 110 g/week (95% Cl 107-112) and 46 g/week (95% Cl 44-48) respectively (Figure 8.4). Beyond 56 and 51 years of age, the mean volume of alcohol consumption steadily declined, falling to 49 g/week (95% Cl 43-55) among men and 23 g/week (95% Cl 19-26) among women by the age of 83.5 years (Figure 8.4). Analyses based upon the imputed dataset reported higher volumes of alcohol consumption, measuring a peak of 114 g/week (95% Cl 111-117) among men aged 55 years and 52 g/week (95% Cl 50-55) among women aged 59 years, before falling to 63 g/week (95% Cl 57-68) among men and 28 g/week (95% Cl 24-33) among women by the age of 83.5 years (Appendix 8.6).



Figure 8.4 Crude sex-specific non-linear trajectories of mean weekly volume of alcohol consumption between the ages of 34-84 years. Observed data.

8.5.1.4 Stability of baseline categories

As reported in Table 8.3 and shown in Figure 8.5, significant differences in the longitudinal slope were identified according to the category of intake at baseline, with similar trends apparent in both sexes. Specifically, slopes between baseline categories tended to converge with increased age, whereby participants in higher categories exhibited downward longitudinal trajectories and those in lower baseline categories exhibited upward or stable slopes over the period, on average.

Among both men and women, changes over the captured life course were most pronounced within the highest baseline category. Relative to the rate of change among non-drinkers, consumption within the highest baseline category fell by an average 29.7 g/week among men and 20.1 g/week among women for every 10 year increase in age. Increases were greatest among infrequent drinkers, whose drinking rose by 8.7 g/week and 3.1 g/week respectively per 10 year increase in age, relative to non-drinkers. However, this rise may have been a consequence of such participants having consumed alcohol more frequently with advancing age such that their drinking was more likely to have been captured by the quantity-frequency questionnaire at each successive wave (Table 6.2). The only stable baseline category was female light drinkers (0.1-50.0 g/week), who showed no statistically significant change in alcohol consumption with increased age, relative to pooled non-drinkers. Although drinking declined

Chapter 8: Trajectories of alcohol consumption

among women defined as more moderate drinkers at baseline (50.1-100.0 g/week), this equated to a mere 2.6 g/week change per decade, relative to non-drinkers. As such, light and moderate baseline consumption categories appeared largely stable over the period of the life course captured by the Whitehall II study.

Results based upon the imputed dataset were similar, with decreasing consumption evident within higher baseline categories and increasing consumption within lower baseline categories (Appendices 8.7 and 8.8). Interestingly, analyses applied to the imputed dataset indicate an increase in alcohol consumption among pooled non-drinkers equal to an average 5.9 g/week (95% CI 4.2-7.6) among men and 5.2 g/week (95% CI 3.8-6.6) among women, suggesting that some former or never drinkers with missing data were predicted to resume or take up drinking alcohol as they aged.

Given that participants varied between 35 and 55 years of age at the time that baseline categories were defined, and with marked differences in alcohol consumption reported by age group in Sections 2.2.1 and 8.5.1.5, there was a possibility that some of the difference in longitudinal trajectories may have been attributable to disparities in the age of participants within each baseline category. A post-hoc sensitivity analyses was thus undertaken which re-ran the mixed effects models with adjustment for differences in age at baseline, parameterised as dates of birth. Adjustment for birth date provided no improvement in the fit of either sexspecific model, and altered intercept and slope coefficients only by fractions of a g/week. The effect of any difference in age at baseline within each category was thus considered negligible.



Figure 8.5 Crude sex-specific linear trajectories of mean weekly volume of alcohol consumption between the ages of 34-84 years, stratified by baseline alcohol consumption category. Observed data.

Linear mixed models	Sample n	g/week (95% CI)	p-value
Men			
Difference in baseline consumption by drinking category			
Non-drinker	220	Reference	
Infrequent drinker	669	0.7 (-2.4, 3.9)	0.651
0.1-50.0 g/week	2,073	30.3 (27.8, 33.0)	<0.001
50.1-100.0 g/week	1,432	76.5 (72.5 <i>,</i> 80.4)	<0.001
100.1-150.0 g/week	881	131.6 (127.3, 125.9)	<0.001
>150.0 g/week	1,563	286.6 (278.5, 294.7)	<0.001
Difference in the rate of change by drinking category			
Non-drinker		Reference	
Infrequent drinker		8.7 (6.8, 10.5)	<0.001
0.1-50.0 g/week		6.2 (4.9 <i>,</i> 7.5)	<0.001
50.1-100.0 g/week		4.5 (2.2, 6.8)	<0.001
100.1-150.0 g/week		-2.7 (-5.0, -0.4)	0.020
>150.0 g/week		-29.7 (-32.8, -26.6)	<0.001
Women			
Difference in baseline consumption by drinking category			
Non-drinker	216	Reference	
Infrequent drinker	764	1.0 (-0.7, 2.6)	0.256
0.1-50.0 g/week	1,428	29.3 (27.5, 31.2)	<0.001
50.1-100.0 g/week	542	76.2 (72.7, 79.6)	<0.001
>100.0 g/week	422	176.5 (167.3, 185.7)	<0.001
Difference in the rate of change by drinking category			
Non-drinker		Reference	
Infrequent drinker		3.1 (1.9, 4.2)	<0.001
0.1-50.0 g/week		-0.0 (-1.2, 1.1)	0.936
50.1-100.0 g/week		-2.6 (-4.8, -0.4)	0.019
>100.0 g/week		-20.1 (23.6, 16.6)	<0.001

 Table 8.3 Crude sex-specific interaction between the trajectory of mean weekly volume of alcohol consumption and baseline alcohol consumption category. Observed data.

8.5.2 Trajectories of mean weekly volume of alcohol consumption by T2DM diagnosis

8.5.2.1 Descriptive statistics

Of the 10,308 individuals originally enlisted at baseline, 8,815 participated at wave three. Among these, 250 prevalent cases were documented and thus excluded. A total 5,723 T2DM-free men and 2,570 T2DM-free women survived to and participated in at least one subsequent wave such that incident diagnosis status could be determined. After excluding repeated measures of alcohol consumption recorded after the date of diagnosis, plus one female participants who did not provide a measure of alcohol consumption for at least one wave prior to their date of

censoring or diagnosis, this left an analytic sample of 5,723 men and 2,569 women with 27,711 male person-observations and 11,734 female person-observations over a mean 4.8 waves. Participants were observed over a median 9.5 years for both sexes, up to a maximum of 28.0 years among men and 27.9 years among women. In total, 620 men and 296 women developed T2DM over the period.

Participants who developed T2DM exhibited a worse metabolic profile at wave one than those who did not develop the condition (Table 8.4), with a greater proportion of such participants being physically inactive, of South Asian ethnicity, a lower occupational grade and having a family history of T2DM as well as a higher BMI and more advanced age. In terms of alcohol consumption behaviour, women who developed T2DM reported a lower volume of weekly alcohol consumption at baseline. Imputed descriptive statistics were comparable (Appendix 8.9).

		Concerned	
Variables (wave one)	% (95% CI) n	% (95% CI) n	Difference ^a
Men			
Age			
Mean years	45.0 (44.6, 45.5) ^b	44.4 (44.2, 44.6) ^b	0.013
	620	5,103	
Alcohol consumption frequency			
None in past year	4.4 (3.0, 6.3)	2.6 (2.2. 3.1)	0.028
	27	132	
<1/week	21.8 (18.8, 25.3)	19.7 (18.6, 20.8)	
	135	1,001	
1-3 times/week	40.3 (36.5, 44.2)	43.7 (42.3, 45.0)	
	249	2,223	
Daily or almost daily	33.5 (29.9, 37.3)	34.1 (32.8, 35.4)	
	207	1,734	
Alcohol consumption volume			
Median g/week	98.8 (89.8, 107.8) ^c	101.5 (98.5, 104.6) ^c	0.570
	617	5,067	
BMI			
Mean kg/m ²	26.1 (25.8, 26.3) ^b	24.3 (24.2, 24.4) ^b	< 0.001
	619	5,094	
Ethnicty			
White	84.8 (81.8, 87.5)	94.6 (93.9, 95.2)	< 0.001
	526	4,815	
South Asian	11.8 (9.5, 14.6)	3.4 (3.0, 4.0)	
	73	175	
Other ^d	3.4 (2.2, 5.1)	2.0 (1.6, 2.4)	
	21	101	
Family history of T2DM			
Yes	81.1 (77.8, 84.1)	91.2 (90.4, 91.9)	<0.001
	495	4,589	
No	18.9 (15.9, 22.2)	8.8 (8.1, 9.6)	
	115	444	
Occupational grade			
Administrative (top)	35.0 (31.3, 38.9)	41.3 (39.9, 42.6)	< 0.001
	217	2,106	
Professional (middle)	54.0 (50.1, 57.9)	52.1 (50.7, 53.4)	
	335	2,657	
Clerical (bottom)	11.0 (8.7, 13.7)	6.7 (6.0, 7.4)	
	68	340	
Physical activity ^e			
Inactive	12.4 (10.0, 15.2)	8.1 (7.4, 8.9)	<0.001
Polow guidelines	76	410	
berow guidennes	40.1 (36.2, 44.0) 216	37.4 (JD.I, 38.8) 1 202	
Met guidelines	240 476 (436 51 5)	1,092 54 5 (53 1 55 8)	
Burnethes		55 (55.1, 55.0)	

Table 8.4 Baseline characteristics of T2DM-free participants, stratified by T2DM diagnosis. Observed data.

Chapter 8: Trajectories of alcohol consumption

Smoking			
Never	42.0 (38.1, 45.9)	49.8 (48.5, 51.2)	<0.001
	258	2,525	
Former	39.3 (35.6, 43.3)	36.4 (35.1, 37.7)	
	242	1,845	
Current	18.7 (15.8, 22.0)	13.8 (12.8, 14.7)	
	115	697	
<u>Women</u>			
Age			
Mean years	467(460474) ^b	45 3 (45 1 45 6) ^b	<0.001
	296	2.274	
Alcohol consumption froquency		_,	
Negotion consumption mequency			-0.001
None in past year	9.5 (6.6, 13.4)	5.7 (4.8, 6.7)	<0.001
at hugek		128 25 4 (22 4 27 4)	
<1/week	51.4 (45.0, 57.0) 153	35.4 (33.4, 37.4) 001	
1-2 times (week	152 28 7 (22 8 24 2)	266 (34 6 38 6)	
1-5 times/week	20.7 (23.0, 34.2)	20.0 (34.0, 38.0) 229	
Daily or almost daily	105 (7 / 1/ 5)	223 22 / (20 7 2/ 1)	
	10.5 (7.4, 14.5)	507	
Alcohol consumption volume	51	507	
Action of consumption volume	<u>,</u>	<u>,</u>	-0.001
Median g/week	29.4 (23.3, 35.5) [°]	46.9 (44.4 <i>,</i> 49.4) ^c	<0.001
	296	2,247	
BMI			
Mean kg/m ²	28.0 (27.4, 28.6) ^b	24.2 (24.0, 24.3) ^b	<0.001
	296	2,273	
Ethnicty			
White	71.2 (65.7, 76.1)	89.1 (87.7, 90.3)	<0.001
	210	2,012	
South Asian	14.6 (11.0, 19.1)	4.6 (3.8, 5.5)	
	43	103	
Other ^d	14.2 (10.7, 18.7)	6.4 (5.4, 7.5)	
	42	144	
Family history of T2DM			
Yes	68.8 (63.1, 73.9)	89.2 (87.8, 90.4)	<0.001
	198	1.989	
No	31.3 (26.1, 36.9)	10.8 (9.6, 12.2)	
	90	242	
Occupational grade			
Administrative (top)	11(2270)	110(127 155)	~0.001
Administrative (top)	4.1 (2.3, 7.0)	14.0 (12.7, 13.3)	<0.001
Professional (middle)	36.8 (31 5 42 5)	43.0 (41 0 45 1)	
	109	978	
Clerical (bottom)	59.1 (53.4. 64.6)	43.0 (40.9. 45.0)	
· · · /	175	977	

Chapter 8: Trajectories of alcohol consumption

Physical activity			
Inactive	31.9 (26.7, 37.6)	22.6 (20.9, 24.4)	0.002
	91	504	
Below guidelines	34.4 (29.1, 40.1)	40.7 (38.7, 42.8)	
	98	907	
Met guidelines	33.7 (28.4, 39.4)	36.6 (34.6, 38.6)	
	96	815	
Smoking			
Never	59.0 (53.3, 64.6)	54.2 (52.2, 56.3)	0.299
	173	1,227	
Former	22.2 (17.8, 27.3)	24.9 (23.2, 26.8)	
	65	564	
Current	18.8 (14.7, 23.7)	20.8 (19.2, 22.5)	
	55	471	

Sample sizes differed according to item non-response at baseline (wave one). Employment status not listed as all participants were employed at wave one.

^aTo explore differences between non-response groups, one-way ANOVA was used on continuous data, and the chi² test on categorical data; ^bMean and 95% confidence interval; ^cMedian and 25th and 75th percentiles; ^de.g. black Caribbean, African and Arabic; ^eMeeting guidelines (≥150 minutes of moderate-intensity or ≥75 minutes of vigorous-intensity activity per week); inactive (<60 minutes of moderate and <60 minutes of vigorous activity; below guidelines (not inactive or meeting guidelines).

8.5.2.2 Linear trajectories

For both sexes, random slopes models best described the observed (Table 8.5) and imputed (Appendix 8.10) linear trajectories of alcohol consumption over the period leading up to the development of T2DM or censoring, suggesting marked variability in drinking between repeated measures.

 Table 8.5 Sex-specific interaction between the linear trajectory of mean weekly volume of alcohol consumption and T2DM diagnosis: goodness of fit statistics. Observed data

	Men		Women	1
Mixed model	Log-likelihood	BIC ^a	Log-likelihood	BIC ^a
Crude linear mixed model, fixed slope	-162536	325134	-61948	123953
Crude linear mixed model, random slopes	-162287	324655	-61907	123889
^a Bayesian information criterion.				

Likelihood ratio tests were undertaken to formally establish the suitability of stratifying linear mixed models by sex. The inclusion of sex interactions between time and diagnosis status each improved the goodness of fit (p=<0.001), supporting the case for sex stratification. Sex stratified results are reported in Table 8.6, with corresponding plots shown in Figure 8.6. Men who did not develop T2DM over the course of study showed a gradual decline in their volume of alcohol consumption up to their time of censoring, while men who did develop the condition exhibited a marked increase in consumption up to their time of their time of diagnosis. Based upon the coefficients

reported in Table 8.6, the mean volume of alcohol consumption thirty years prior to diagnosis or censoring was estimated to have been 108.1 g/week (103.6+[1.5*3]) among male non-cases and 75.6 g/week ([103.6+22.4]+[-16.8*3]) among male cases. By the time of diagnosis or censoring, the weekly volume of alcohol consumption was estimated to be an average 103.6 g/week and 126.0 g/week respectively.

By contrast, women who developed T2DM exhibited consistently lower volumes of alcohol consumption throughout the period of observation than those who were censored. By the end of observation, female consumption measured a mean 46.1 g/week among non-cases and 27.7 g/week (46.1-18.4) among cases, or a difference of 18.4 g/week. Results from analyses based upon the imputed dataset were comparable for both sexes, and reported in Appendices 8.11 and 8.12.

Crude linear mixed models	g/week (95% CI)	p-value
<u>Men (n=5,723)</u>		
Consumption volume		
Intercept	103.6 (100.5, 106.6)	<0.001
Change per 10 years prior to diagnosis or censoring	-1.5 (-2.7, -0.3)	0.011
Difference in consumption at the time of diagnosis or censoring		
Censored	Reference	
T2DM	22.4 (11.2, 33.7)	<0.001
Difference in the rate of change by diagnosis or censoring		
Censored	Reference	
T2DM	16.8 (10.9, 22.7)	<0.001
<u>Women (n=2,569)</u>		
Consumption volume		
Intercept	46.1 (43.5, 48.6)	<0.001
Change per 10 years prior to diagnosis or censoring	-1.5 (-2.5, -0.6)	0.002
Difference in consumption at the time of diagnosis or censoring	P (
Censored	Reference	
T2DM	-18.4 (-24.5, -12.3)	<0.001
Difference in the rate of change by diagnosis or censoring		
Censored	Reference	
T2DM	1 3 (-2 7 5 2)	0 5 2 0
	1.5 (-2.7, 5.2)	0.529

Table 8.6 Crude sex-specific interaction between the linear trajectory of mean weekly volume of alcoholconsumption and T2DM diagnosis. Observed data.


Figure 8.6 Crude sex-specific linear trajectory of mean weekly volume of alcohol consumption, stratified by T2DM diagnosis. Observed data.

8.5.2.3 Non-linear trajectories

A range of non-linear slopes were explored, with fit statistics reported in Appendices 8.13 and 8.14. Consumption trajectories among men and women who developed T2DM were best described as a linear function of time, with a non-linear trajectories provided the best fit of the underlying data among censored participants (Table 8.7 and Figure 8.7). Results from models applied to the imputed dataset were comparable and reported in Appendices 8.15 and 8.16.

At 30 years prior to date of diagnosis or censoring, alcohol consumption was roughly equivalent among men at around 80.0 g/week regardless of whether they later developed T2DM or were censored. However, by the time of diagnosis or censoring, mean alcohol intake among men who didn't develop T2DM was lower than among men who developed the condition, at 92.6 g/week and 126.0 g/week respectively. This equated to a difference of 33.4 g/week, or around 1.8 pints of 4.0% ABV lager.¹¹ Among women, consumption remained consistently higher among those that did not develop T2DM. In contrast to men, disparities were most acute at both the beginning and end of the observation period, at just 13.3 g/week at the time of event or censoring, or around 0.7 pints of 4.0% ABV lager.¹¹

Crude best-fitting mixed models	g/week (95% CI)	p-value
Men		
T2DM (n=620)		
Intercept	126.0 (115.2, 136.9)	<0.001
Time ¹	15.2 (9.4, 21.0)	<0.001
Censored (n=5,103)		
Intercept	92.6 (89.6, 95.6)	<0.001
Time ¹	-33.2 (-36.6, -29.8)	<0.001
Time ²	-1.3 (-1.4, -1.1)	<0.001
<u>Women</u>		
T2DM (n=296)		
Intercept	27.8 (22.2, 33.4)	<0.001
Time ¹	-0.2 (-4.1, 3.7)	0.919
Censored (n=2.273)		
Intercept	41.1 (38.6, 43.6)	<0.001
Time ¹	-15.9 (-19.0, -12.7)	<0.001
Time ²	-0.6 (-0.7, -0.4)	<0.001

Table 8.7 Crude, sex-specific and best-fitting trajectory of mean weekly volume of alcohol consumption,stratified by T2DM diagnosis. Observed data.

Time coefficients refer to the change in the average volume of weekly alcohol consumption per 10 years prior to diagnosis or censoring.



Figure 8.7 Crude sex-specific linear trajectory of mean weekly volume of alcohol consumption, stratified by T2DM diagnosis. Observed data.

8.5.2.4 Sick-quitter effects

Looking specifically at participants who developed T2DM, trajectories were estimated for alcohol consumption beyond the date of diagnosis, with goodness-of-fit statistics for all corresponding models reported in Appendices 8.17 and 8.18. Of the 620 men and 296 women who developed T2DM over the course of the study, alcohol consumption data were observed after the date of diagnosis among 552 and 267 participants respectively. Trajectories of alcohol consumption were estimated based upon a 3,262 person-observations among men and 1,513 person-observations among women.

Table 8.8 Crude trajectories of mean weekly volume of alcohol consumption up to and beyond the date of diagnosis, stratified by sex

	Men		Women	
Piecewise models	g/week (95% CI)	p-value	g/week (95% CI)	p-value
Up to diagnosis				
Intercept	126.0 (115.2, 136.9)	<0.001	27.8 (22.2, 33.4)	<0.001
Time ¹	15.2 (9.4, 21.0)	<0.001	-0.2 (-4.1, 3.7)	0.919
After diagnosis				
Intercept	103.1 (91.4, 114.8)	<0.001	21.2 (16.4, 25.9)	<0.001
Time ¹	-21.2 (-32.2, -10.3)	<0.001	-4.5 (-7.9, -1.2)	0.008
Time coefficients refer to the change in the average volume of weekly alcohol				

consumption per 10 years of follow-up.

As shown in Table 8.8 and Figure 8.8, both sexes showed significant reductions in their consumption following diagnosis, equal to 21.2 g/week per decade among men and 4.5 g/week per decade among women. Trajectories from analyses of the imputed dataset differed little, and are shown in Appendix 8.19.



Figure 8.8 Crude trajectories of mean weekly volume of alcohol consumption up to and beyond the date of T2DM diagnosis, stratified by sex

8.5.2.5 Stability of baseline categories

Significant differences in the longitudinal trajectory of alcohol consumption were identified according to categories of intake at baseline, as reported in Table 8.9 and shown for men and women in Figures 8.9 and 8.10 respectively. The length of each plotted trajectory was dependent upon the time between baseline and diagnosis or censoring. No incident cases were observed in the final wave of observation among participants defined as baseline non-drinkers.

Regardless of diagnosis status, male drinking increased on average within all but the highest baseline consumption categories. Within the highest category, alcohol intake was estimated to fall by an average 20.8 g/week per decade among men that developed T2DM, relative to non-drinkers, or 27.2 g/week per decade among those that were censored. Of the upward trajectories apparent within more moderate categories of baseline consumption, increases were shallower among participants that did not develop T2DM.

Present within both the highest (>100.0 g/week) and second-highest (50.1-100.0 g/week) categories of baseline consumption, downward trajectories were more apparent among women. These slopes were steepest among those that developed T2DM. For instance, of women who drank 50.1-100.0 g/week at baseline, the mean volume of alcohol consumption fell by 17.0 g/week among those who were diagnosed with T2DM and by just 1.9 g/week among those who

were censored, relative to non-drinkers. Although longitudinal increases in consumption were evident among both censored and diagnosed baseline moderate drinkers, elevations appeared too small to be clinically significant. For example, of women who drank just 0.1-50.0 g/week at baseline, consumption rose by an average 3.0 g/week each decade among those that went on to develop T2DM, and by a statistically insignificant 0.4 g/week among women that were censored, relative to non-drinkers. These same longitudinal trends were observed within the imputed dataset, as reported in Appendices 8.20-8.22.

	T2DM		Censored	
Crude linear mixed models	g/week (95% CI)	p-value	g/week (95% CI)	p-value
<u>Men</u>				
Difference in consumption at the time of diagnosis or censoring by baseline consumption category				
Non-drinker	Reference		Reference	
Infrequent drinker	26.2 (6.9, 45.4)	0.008	29.7 (23.5, 36.0)	<0.001
0.1-50.0 g/week	59.1 (40.2, 78.1)	<0.001	51.9 (47.4, 56.4)	<0.001
50.1-100.0 g/week	123.7 (99.5, 148.0)	<0.001	92.4 (85.4, 99.3)	<0.001
100.1-150.0 g/week	162.3 (130.1, 194.5)	<0.001	122.7 (115.3, 130.2)	<0.001
>150.0 g/week	216.1 (182.6, 249.7)	<0.001	188.3 (179.3, 197.2)	<0.001
Difference in the rate of change by				
baseline consumption category ^a				
Non-drinker	Reference		Reference	
Infrequent drinker	11.7 (3.4, 20.0)	0.006	8.3 (6.2, 10.4)	<0.001
0.1-50.0 g/week	14.7 (6.2, 23.2)	0.001	6.1 (4.5, 7.7)	<0.001
50.1-100.0 g/week	25.9 (14.6, 37.3)	<0.001	4.9 (2.1, 7.8)	0.001
100.1-150.0 g/week	26.0 (11.0, 41.1)	0.001	-1.9 (-4.7, 0.8)	0.173
>150.0 g/week	-20.8 (-39.1, -2.5)	0.026	-27.2 (-30.8, -23.7)	<0.001
Women Difference in consumption at the time of diagnosis or censoring by baseline consumption category				
Non-drinker	Reference		Reference	
Infrequent drinker	11.6 (4.8, 18.5)	0.001	14.4 (10.8, 18.0)	<0.001
0.1-50.0 g/week	27.0 (19.7, 34.2)	<0.001	30.9 (27.4, 34.4)	<0.001
50.1-100.0 g/week	39.5 (24.5, 54.6)	<0.001	70.4 (64.3, 76.4)	<0.001
>100.0 g/week	100.7 (68.8, 132.7)	<0.001	109.4 (98.8, 119.9)	<0.001
Difference in the rate of change by				
baseline consumption category ^a				
Non-drinker	Reference		Reference	
Infrequent drinker	6.8 (2.7, 10.9)	0.001	3.6 (2.1, 5.2)	<0.001
0.1-50.0 g/week	3.0 (-1.1, 7.1)	0.155	0.4 (-1.1, 1.9)	0.609
50.1-100.0 g/week	-17.0 (-24.5, -9.4)	<0.001	-1.9 (-4.4, 0.6)	0.139
>100.0 g/week	-25.0 (-52.2, 2.3)	0.072	-18.3 (-22.6, -14.0)	<0.001
^a Rate of change per 10 years prior to d	iagnosis or censoring.			

Table 8.9 Crude sex-specific interaction between the trajectory of mean weekly volume of alcohol consumption and baseline category, stratified by T2DM diagnosis. Observed data.





Figure 8.9 Crude male trajectories of mean weekly volume of alcohol consumption, stratified by baseline alcohol consumption category and T2DM diagnosis. Observed data.



Figure 8.10 Crude female trajectories of mean weekly volume of alcohol consumption, stratified by baseline alcohol consumption category and T2DM diagnosis. Observed data.

8.6 Summary of results

Between the ages of 34 and 84 years, the mean weekly volume of alcohol consumption was markedly higher among men than women. Despite this, the relationship with age was broadly comparable, with volumes increasing up to around 56 years of age among men and 51 years of age among women before decreasing thereafter (Figure 8.4).

When trajectories were stratified by categories of baseline consumption, moderate drinkers were found to remain relatively stable across the adult life course, while heavy drinkers decreased their volume of alcohol consumption and infrequent drinkers increased their intake over the period (Figure 8.5). These changes indicated that, depending upon when in the adult life course a study categorises the alcohol intake of a participant, cross-sectionally defined moderate drinkers may be subject to differential degrees of misclassification error through contamination with heavy and infrequent drinkers.

Stratifying the longitudinal trajectory by T2DM diagnosis revealed that consumption among men and women who did not develop T2DM gradually increased up to around 15 years prior to the time of censoring, before decreasing steadily toward the end of the observation period (Figure 8.7). By contrast, linear increases were evident among men that developed the condition. Thus, by the time of diagnosis or censoring, the mean weekly volume of alcohol consumption among men who didn't develop T2DM was 33.4 g/week lower than among men who developed the condition. A different picture was apparent among women. Here, rather than a similar linear increase among those who developed T2DM, the mean weekly volume of alcohol consumption remained consistently lower than among women who were censored. These differences were most acute at both the beginning and end of the observation period.

Regardless of sex-specific differences in the consumption trajectory prior to developing T2DM, drinking fell among both sexes after the date of diagnosis, at 21.2 g/week among men and 4.5 g/week among women per decade of follow-up (Figure 8.8).

8.7 Limitations

Although Whitehall II captured around 50 years of the adult life course, the dataset was not comprehensive, lacking measures of consumption through early adulthood (18-33 years of age) and advanced old age (\geq 83 years). Although data concerning these periods were missing, it was expected based on existing research that the unobserved period of early adulthood was likely to have been marked by a trend of increasing consumption, with a decrease in drinking during the unobserved period of advanced old age.^{259,266}

As noted in Section 7.6, all analyses of Whitehall II data were dependent upon self-reported measures of alcohol consumption. Consequentially, plotted trajectories were potentially subject to some degree of inaccuracy owing to factors including reporting and recall biases³⁹¹ as well as measurement error.³⁹² In addition, no consideration was given to changes in reported consumption frequency across the life course. This decision was taken after having previously identified no difference in the dose-response relationship according to the frequency of consumption (Table 7.5).

The results reported in this chapter focus on the mean trajectory of sampled participants by age and time. Although some analyses stratified alcohol consumption trajectories according to categories of baseline consumption, providing an indication as to how different drinking groups altered their intake over the life course, all results nonetheless dealt with averages that mask longitudinal changes to the constitution of the mean trajectory. For instance, the declining slopes seen among censored men and women in Figure 8.7 may have been a consequence of either a general decrease in the weekly volume of alcohol consumption to more moderate intakes or the transition of a minority of participants from heavy consumption to non-drinking. To get a rough idea as to the effect of transitions to non-drinking upon plotted trajectories, a post-hoc analysis was undertaken that calculated the best-fitting models in Figure 8.8 when participants with zero consumption (0 g/week) were excluded from each wave. Estimated trajectories are shown in Appendix 8.23. Although little difference was evident among male trajectories beyond an expected elevation in intercept values, both female slopes were shifted upward relative to those reported in Figure 8.7. The degree of change was largest among women who developed T2DM. While results in Table 8.7 indicated a flat trajectory among such participants (-0.2 g/week per decade, p=0.919), the exclusion of non-drinkers resulted in a nearsignificant upward trajectory equal to an average increase of 5.5 g/week per decade (p=0.103). By contrast, among censored women, the difference in slopes between the two models was closer to just 1.5 g/week per decade. Such differences suggest that both female slopes may have been skewed downward by transitions from drinking to non-drinking up to the time of diagnosis or censoring, with transitions to zero consumption being most pronounced among women who developed T2DM.

A further limitation of the longitudinal mean trajectories was the testing of just two polynomial terms within the non-linear mixed effects models. Although two polynomial terms accommodate a broad family of non-monotonic trajectories,¹⁷⁶ the range and complexity of any resulting slopes is limited to a maximum of one turning point. Plotted non-linear trajectories

may therefore be simplistic and fail to reflect multimodal trajectories present within the underlying data. However, while multiple turning points can be accommodated through the introduction of additional polynomial terms,¹⁷⁶ the added flexibility of such an approach comes at the risk of overfitting the data, producing trajectories that provide a better statistical fit of the observed dataset at the cost of poor generalisability to other cohorts.¹⁷⁵

As outlined in Section 6.6.1, mortality was likely to have represented a competing risk, with some study participants potentially censored due to mortality before they would otherwise have developed T2DM. Assuming that participants predominantly died due to chronic non-communicable conditions, those who were censored due to death may have exhibited a pronounced downward trajectory in the period leading up to their final wave of participation. In order to investigate the effect of mortality upon the longitudinal trajectories, the best-fitting models were calculated for each sex excluding data from any participant censored due to death. This involved the exclusion of observations from 594 men and 290 women. As shown in Appendix 8.24, the omission of data from participants censored through death had little impact upon plotted trajectories.

8.8 Discussion

8.8.1 Prospective trajectories of mean weekly volume of alcohol consumption

Analyses confirm that drinking varies with advancing age, increasing up to 56 years of age among men and 51 years of age among women before declining thereafter. This finding is consistent with previous research²⁶⁶ and calls into question the validity of results from analyses that define consumption according to a single baseline measurement.

When consumption trajectories were stratified according to baseline categories of alcohol consumption, mean alcohol intake was found to vary differentially as a function of age (Table 8.3) and in a manner concordant with findings elsewhere.²⁶⁷ Drinking among both sexes was least stable within the highest baseline categories, where volumes of consumption fell markedly with increasing age. Reasons for such a decline are likely to be complex, as indicated by a study of Whitehall II participants aged 61-85 years at wave 11.⁴⁰⁸ Of the 40% of participants who reduced their consumption during the decade preceding wave 11, one of the main reasons given was a decline in opportunities for social drinking (men: 46%; women: 41%), with attenuations also commonly described as a proactive health precaution (men: 45%; women: 34%) or a response to ill-health or pharmaceutical contraindication (men: 20.7%; women: 21.9%). If heavier drinkers were more likely to have reduced their intake owing to ill-health, this may go

some way toward explaining the elevated risks of T2DM evident among non-current drinkers (Chapter 7).

Trajectories stratified according to baseline consumption were found to converge toward moderate volumes with advancing age. There was therefore an indication that the risk of misclassification error may be greatest in studies of older cohorts where cross-sectionally defined moderate drinkers are likely to be contaminated by a greater proportion of infrequent and former heavy drinkers than in younger samples. If infrequent and former heavy drinkers are at greater risk of T2DM than stable moderate drinkers, this longitudinal contamination may help explain why less pronounced reductions in the risk of CHD⁴⁰⁹ and all-cause mortality³⁹⁵ are apparent among older moderate drinkers in conventional survival analyses.

While results from analyses of the imputed dataset showed similar disparities in the longitudinal trajectory by categories baseline consumption, non-drinkers exhibited a more pronounced increase in consumption with advancing age. This suggests either that many former or never drinkers with missing data may have resumed or taken up drinking as they got older, or that the imputation model was poorly specified and did not predict zero values of alcohol consumption effectively.

Whatever the reasons for changes in drinking across the life course, longitudinal variation in the volume of alcohol consumption indicate that the categorisation of study participants according to consumption reported at a single point in time risks introducing misclassification error, with the magnitude of error operating at least in part as a function of participant age.

8.8.2 Trajectories of mean weekly volume of alcohol consumption by T2DM diagnosis

Results from the revised and updated meta-analysis indicate increases in T2DM risk among men from very low volumes of consumption, with reductions in risk specific to moderate female drinkers (Figure 3.4). By examining differences in the trajectory of alcohol consumption by T2DM diagnosis, this chapter aimed to develop a better understanding of whether dose-response relationships reported by conventional survival analyses were the result of harms or benefits having accumulated over time following prolonged exposure to particular volumes of alcohol, or a consequence of acute differences in consumption during periods of the life course in which sensitivity to the effects of alcohol may be especially pronounced.

Among men, alcohol intake was approximately equivalent between groups until just a few years prior to diagnosis, at which point the volume of consumption was greatest among men who developed the condition (Figure 8.7). This finding was most consistent with the second

hypothesis, which contended that sensitivity to the deleterious effects of higher alcohol intakes may be most pronounced later in the life course, with the mean age at diagnosis measuring 61.9 (95% Cl 61.4, 62.3) years among men.

While male consumption increased linearly over the period, those who did not develop the condition exhibited a marked decline in their consumption prior to censoring. As noted in the preceding section, the reasons for such a decline were likely to be a combination of factors, including as a response to deteriorating health or as a proactive health precaution.⁴⁰⁸ Whatever the predominant motivation, the lack of a similar downward trajectory among men who developed T2DM appeared to be in conflict the third hypothesis outlined in Section 8.3, which posited that declining health prior to diagnosis would elicit a sick quitter effect marked by a reduction in consumption. When trajectories of alcohol consumption were calculated beyond the date of diagnosis, reductions were found to be evident among both sexes after the date of diagnosis (Figure 8.8), suggesting that any deterioration in health prior to the diagnosis of T2DM may have been of insufficient magnitude as to elicit a change in drinking behaviour, with consumption only falling following medical advice.

Neither the first or second hypotheses were supported in women. Here, participants who developed T2DM exhibited an average weekly intake consistently below that of women who did not develop T2DM, with disparities by diagnosis status most pronounced during a period 15 years prior to the date of diagnosis or censoring. Regardless of such a disparity, volumes of alcohol consumption within both female groups were well within the range of intake associated with reductions in the risk of T2DM among women in the updated meta-analysis (Figure 3.4). At least two reasons for this conflict with a priori hypotheses were possible. First, the lower volumes consumed by women who developed T2DM may have produced an effect insufficient to offset any increased risk conferred by other factors, with women who developed T2DM having a worse metabolic health profile at wave three than those who were censored (Table 7.1). Secondly, given their worse metabolic health profile, the mean trajectory for women who developed T2DM may have been comprised not primarily of persistent low volume and therefore lower risk drinkers, but of sick guitters who had already attenuated their drinking owing to poor health. Thirdly, it was possible that the absence of distinct female trajectories by diagnosis status may have been attributable to a relatively limited range of alcohol consumption among women within the Whitehall II cohort, with heavier drinkers being few in number. For instance, while estimates from the meta-analysis in Chapter 3 indicate that the risk of T2DM becomes elevated among women at volumes >140 g/week, relative to pooled non-drinkers, just

239 women within the Whitehall II cohort reported consumption anywhere above that level at wave one, falling to just 105 participants by wave 11. A general population cohort with a greater breadth of female consumption similar to that available for men may be better suited to identifying disparities in the mean volume of alcohol consumption among women according to their diagnosis of T2DM.

That the disparity in alcohol intake by diagnosis category was greatest among men in the few years preceding the end of observation suggests that any increased risk conferred by heavy alcohol consumption may occur later in the adult life course and over a relatively acute period of time. Given this indication, survival analyses in Chapter 9 will explore whether the dose-response relationship is better parameterised according to different dimensions of the longitudinal trajectory, providing a clearer picture of differences in dose-response across the life course. Furthermore, while censored participants appeared to reduce their intake during the period leading up to the end of observation, it is unclear whether these downward trajectories represent a proactive attempt to modify negative health behaviours (a 'worried well') or a reaction to deteriorating health ('sick quitters'). Accordingly, survival models reported in Chapter 9 will also be constructed to explore the association between the longitudinal slope and the risk of T2DM independent of the actual volume consumed.

Chapter 9

Longitudinal alcohol consumption and the risk of T2DM

9 Longitudinal alcohol consumption and the risk of T2DM

9.1 Introduction

Having established in Chapter 8 that drinking varies over the life course and differentially within baseline consumption categories, it was possible that estimates derived from conventional survival analyses may fail to adequately capture the complexity of the dose-response relationship between alcohol consumption upon T2DM risk. The primary aim of this chapter was thus to explore the utility of modelling the dose-response relationship according to parameterisations of the longitudinal trajectory other than just the baseline value of consumption. To achieve this, three increasingly complex survival models are considered: an age-varying covariate model, a two-stage model and a shared random effects model.^{410,411} Each approach is subject to its own advantages and limitations, and makes differing assumptions concerning the form of the longitudinal process.^{412,413,414,415}

Notably, the shared random effects approach offers greater flexibility than the other methods, permitting the calculation of differences in dose-response according to whether drinking is defined according to the intercept value of the longitudinal trajectory or consumption at the time of diagnosis. In modelling these two parameters within a single model, the shared random effects approach will help establish whether these different dimensions of the longitudinal process are each independently associated with T2DM risk. Additionally, with alcohol consumption varying as a function of each, the use of an age timescale within such models will permit an exploration of whether the dose-response relationship changes when alcohol is consumed at different points in the adult life course.

With analyses in Chapter 8 having also identified longitudinal decreases in consumption among heavier drinkers and participants who were not diagnosed with T2DM, the shared random effects approach will also be used to estimate the relationship between the rate of change and T2DM risk after adjustment for the volume of past and current consumption. This final analysis will thus explore the sick quitter hypothesis, establishing whether participants who decreased their consumption tended to be at a higher risk of T2DM than those who did not.

9.2 Objectives

To formally explore the utility of using alternative parameterisations of the longitudinal trajectory, and establish whether downward trajectories are consistent with a sick quitter effect, the objectives of this chapter are thus to:

- Compare dose-response relationships reported according to different parameterisations of the longitudinal trajectory.
- Investigate the effect of adjustment for heterogeneous non-drinking groups upon the sex-specific dose-response relationship.
- Establish whether declining trajectories represent a group of drinkers at elevated risk of T2DM.

9.3 Hypotheses

9.3.1 The primary dose-response relationship

Based on results from the preliminary conventional survival model reported in Chapter 7, it is hypothesised that the dose-response relationship estimated according to the intercept of the longitudinal trajectory would indicate a null or increased risk of T2DM among men and a decreased risk among women at higher volumes of consumption (Table 7.5). By contrast, when parameterised according to intake at the time of diagnosis, it is posited that the so-called 'current value' of consumption would be associated with a heightened risk of T2DM, particularly among men. As indicated in Chapter 8, differences in male drinking were most pronounced in the few years prior to diagnosis or censoring, with volumes markedly higher on average among those who developed T2DM (Figure 8.8). With a mean age of diagnosis around 62 years, there was thus some suggestion that an increased risk of T2DM may be most pronounced when alcohol is consumed in higher volumes at older ages, representing a period in the life course during which the metabolism of alcohol is impaired and sensitivity to its deleterious effects potentially intensified.^{196,197,198,199,416,417} Given the possibility that dose-response effects may differ according to a deterioration in the alcohol metabolism with increased age, it was hypothesised that both the intercept and current value parameterisations would have contrasting and independent associations with T2DM risk.

9.3.2 Sick quitter effects

At least as far back as 1988, study participants who reduce their intake over time have been referred to as 'sick quitters' – a group of individuals speculated to attenuate their consumption owing to ill-health.¹³⁸ This is a view maintained within the alcohol literature¹³⁶ and supported by the heightened risk of T2DM among former drinkers (Table 7.3). Elsewhere, T2DM risk factors^{139,140} and poor self-reported health^{141,142,143} have been identified as more prevalent among former drinkers than current drinkers, with the onset of ill-health associated with a cessation from alcohol consumption.^{144,145,146}

Chapter 9: Longitudinal alcohol consumption and the risk of T2DM

Based on such findings, it is possible that downward trajectories of alcohol consumption may be indicative of an increased risk of T2DM owing to a worse metabolic health. For instance, an analysis of participants from the English Longitudinal Study of Ageing showed that declines in the frequency of alcohol consumption were most rapid among participants who reported a deterioration in self-rated health or otherwise exhibited poor self-rated health across the entire period of follow-up.⁴¹⁸ Similarly, although sub-group sample sizes were small, results from the Health Professionals Follow-Up Study indicate that those who decreased their intake over a four-year period tended to show an elevated if non-significant risk of T2DM regardless of baseline consumption category.²⁶⁷

However, a longitudinal decrease in alcohol consumption may not solely occur due to a 'sick quitter' effect. As reported following an analysis of Whitehall II participants aged 61-85 years at wave 11, reasons for a reduction in drinking range from illness, medicinal contraindication, proactive health precaution, a decline in opportunities for social drinking and a history of alcohol misuse.⁴⁰⁸ It was therefore likely that individuals with decreasing consumption may not simply represent a homogenous group of 'sick quitters', but a cohort of 'worried well' or socially isolated adults. It was perhaps because of these disparate motivations that analyses stratified by categories of baseline consumption showed decreasing slopes among heavier baseline drinkers irrespective of T2DM diagnosis (Figures 8.10 and 8.11).

If reductions in alcohol consumption were primarily a response to poor metabolic health, a downward trajectory was expected to be associated with an increased risk of T2DM independent of the volume consumed. Alternatively, if reductions in the volume of alcohol consumption were instead a consequence of proactive attempts to improve health or were a result of fewer opportunities for social drinking due to factors unrelated to health status,⁴⁰⁸ then a downward slopes is expected to be associated with a decreased or null association with T2DM risk respectively. The precise constitution of any downward trajectory will thus determine the direction of any relationship between the rate of change and T2DM.

9.4 Methods

9.4.1 Sample

The survival models chosen to calculate dose-response according to different parametrisations of the drinking trajectory all handle data in a different manner. For instance, while conventional survival models exclude any participant for whom covariate data are missing at baseline, longitudinal methods are able to include such participants by utilising measures obtained at later waves. By contrast, while a conventional survival model can estimate dose-response associations for baseline participants based on all observed cases, longitudinal models drop any case diagnosed at a wave on which covariate data are missing. In order that results from each set of analyses were directly comparable, it was necessary to restrict the sample such that the number of participants and incident cases of T2DM were internally consistent.

Given notable if non-significant differences in the risk of T2DM as reported by infrequent and heterogeneous non-drinkers (Table 7.3), the baseline was set as wave three, the first wave at which drinking data were sufficient for disaggregating and thereby adjusting for differences in risk between such groups. The restriction of the sample thus began by including only those individuals who participated at wave three (men: 6,057; women: 2,758) and who were free of prevalent T2DM at wave three (men: 5,898; women: 2,691). Participants were then excluded if they did not provide valid responses to questions concerning the volume and frequency of alcohol consumption (men: 306; women: 169), and did not participate in at least one other wave such that their diagnosis status could not be determined (men: 136; women: 88). Finally, participants were excluded if incident T2DM was documented on a wave for which covariate data were missing (men: 29; women: 0).

In the case of crude models adjusted only for consumption category, this left an analytical sample of 5,427 men and 2,434 women, or 89.2% of wave three participants. A total incident 560 and 268 cases were identified respectively. Longitudinal alcohol consumption data were available across 21,337 and 8,842 person-observations among men and women, with a median follow-up of 20.2 years and 20.0 years. In multivariable-adjusted models, sample sizes were reduced to 4,793 men and 2,053 women, with 451 and 206 incident cases respectively.

9.4.2 Variables

9.4.2.1 Alcohol consumption

The weekly volume of alcohol consumption was captured and defined as described in Section 6.3 and treated as a continuous variable. As robust standard errors could not be calculated within the shared random effects model, the continuous g/week variable was transformed according to the base 2 logarithm owing to positive skewness. Here, any risk estimate is thus interpreted as the change in risk per two-fold increase in the volume consumed.

All models include adjustment for heterogeneous consumption category, defining participants at each wave according to whether they were never drinkers, non-current drinkers, infrequent drinkers or current drinkers. Using the volume and frequency variables described in Section 6.3,

Chapter 9: Longitudinal alcohol consumption and the risk of T2DM

never drinkers were strictly defined at each wave as those who reported never having consumed alcohol at the current and all prior waves. This definition thus excludes any participant who reported 'always being a non-drinker' at given wave yet also reported non-zero consumption at a prior wave. Such individuals were categorised as non-current drinkers, defined as any individual who reported no alcohol consumption in the year prior to interview but who otherwise reported non-zero consumption at a prior wave. Current drinkers were defined at each wave as anyone who reported non-zero consumption in the week prior to interview, while infrequent drinkers were participants who reported consuming alcohol during the year but did not drink alcohol in the week prior to interview.

As a consequence of seeking to explicitly identify different groups of non-drinkers, continuous alcohol consumption data reported prior to wave three were excluded, as questions concerning never drinking were only available from wave three onwards. Additionally, as never drinking could not be imputed among individuals with missing data due to the small number of such participants, alcohol consumption was restricted to observed data only in all models. Representing the largest category at all waves, current drinkers were selected as the reference group.

9.4.2.2 T2DM

As documented in Section 6.4, cases of T2DM were defined according to any self-reported doctor diagnosis or use of hypoglycaemic medication, or a positive FPG result following clinical examination. Analyses focussed upon incident cases of T2DM and so therefore excluded any prevalent diagnoses at baseline.

As the number of participants predicted to have developed T2DM over the course of the study varied between imputations, analyses applied to the imputed dataset were restricted to participants with observed T2DM diagnosis data such that the number of diagnosed participants remained static between imputations.³⁴²

9.4.2.3 Covariates

Covariates were defined in Section 6.5 and include BMI, ethnicity, employment status, family history of T2DM, occupational grade, physical activity, and smoking status. As described in the statistical analysis section below, age in years was selected as the timescale. As such, it was inappropriate to also adjust for the age of participants at each wave. Instead, adjustment was made for date of birth. Where repeated measures were available, covariates were treated as age-varying within all but the conventional survival analysis where only the baseline values were

used. Given the lack of any indication as to an interaction between the volume and frequency of consumption (Table 7.5), or differences in dose-response according to drink type (Table 7.6), these two factors were not included.

9.4.3 Statistical analysis

9.4.3.1 Time scale

With an interest in testing differences in dose-response according to consumption reported at disparate periods of the adult life course, age in years was selected as the most meaningful timescale. This was contrary to the approach commonly adopted within existing studies, but serves as a conceptual distinction recommended elsewhere over the use of a time-to-event metric due to the incidence of T2DM operating as a function of advancing age as opposed to time since study entry.^{419,420,421}

For the conventional survival analyses, which only considered covariate values at baseline, start times were defined as the age of each participant at baseline, with stop times defined according to the age at diagnosis or censoring. For all other analyses, which utilised repeated measures, start times were equal to the age at participation in each wave, and stop times equal to the age at participation in the next wave of study. In circumstances where the date of participation in a wave was identical to their date of diagnosis or censoring, a value of 0.001 was arbitrarily added to their stop variable such that covariate data for that final wave could be included. The start and stop variables were scaled according to the minimum age at baseline so that intercept values were equal to consumption at 39.6 years of age.

As Stata does not permit the use of imputed survival data, start and stop data were not imputed for waves of unit non-response. Accordingly, results from analyses applied to the imputed dataset only gave account of item non-response present on included covariates during observed waves of participation.

9.4.3.2 Conventional survival model

A proportional hazards model was constructed as per the method described in Chapter 7, with all covariates fixed according to their values at wave three. This approach is consistent with the method conventionally applied for survival analyses of the alcohol-T2DM relationship. Given the highly significant sex interaction reported in Chapter 7, all survival analyses were stratified by sex. As the maximisation procedure of the shared random effects model requires that the functional form of the underlying baseline hazard be explicitly defined,^{415,422} proportional hazards models were run using parametric Weibull, Gompertz and exponential distributions. The parametric distribution that provided the best fit of the unobserved baseline hazard function was then selected for all subsequent models, defined as the model with the smallest BIC.

9.4.3.3 Age-varying covariates survival model

The age-varying covariates model serves as an extension of the conventional survival model and represents the simplest method for estimating the association between a longitudinally varying exposure and a survival process. As per Formula 9.1, alcohol consumption for the ith participant at the jth measurement is represented at each diagnosis by the last *observed* value (d_{ij}). The association parameter (α) linking alcohol consumption and T2DM thus reports the change in risk for a one-unit increase in the last observed or 'current' volume of alcohol consumption. Additional explanatory variables included in the survival model are represented by $x_i(t)$, with the effect of such covariates captured by their associated regression coefficients (β). Notably, although covariate values are allowed to change as a function of age, the regression effect of each covariate value is treated as proportional and thereby independent of age.

$h_i(t)=h_0(t)\exp\{\beta x_i(t)+\alpha d_i(t_{ij})\}$

Formula 9.1 Calculation of a Cox proportional hazard model with time-varying covariates

When obtaining the most probable risk estimates given the observed data, the survival model in Formula 9.1 requires that a complete covariate history is known for all participants.⁴²³ However, in Whitehall II as in other cohorts, longitudinal measures are only captured periodically. The age-varying covariate model thus operates under the assumption that observed values of a covariate remain constant between examinations when drawing on the last known covariate value, as illustrated in Figure 9.1. Although this approach may work well if repeated measures are sufficient in number and frequency as to capture acute changes in consumption, the treatment of the longitudinal trajectory as a step-function is simplistic and can produce a hazard rate unlikely to reflect biological reality.⁴²⁴

A further limitation of the age-varying covariates model is that the longitudinal trajectory is assumed to be measured without error. By giving no account of the inherent variability of the observed values and the degree to which they may differ from the true covariate values, noise-attributable regression dilution can lead to an underestimation of the true longitudinal trajectory and thereby the real effect of alcohol consumption upon T2DM risk.^{412,413,424}



Figure 9.1 Illustration of the longitudinal trajectory and hazard function within age-varying covariate and two-stage survival models

9.4.3.4 Two-stage survival model

The two-stage method provides a better reflection of the inherent uncertainty surrounding the observed values of weekly alcohol consumption. Instead of a survival model that estimates hazards according to the last observed value of alcohol intake, the two-stage approach utilises participant-specific *predictions* of the true volume of alcohol consumption as first estimated via a mixed effects model.⁴¹² By accounting for sampling variability through the inclusion of a random error term, predicted values are closer to the true values of weekly alcohol consumption than the crudely observed data. As such, any underestimation of the association between the weekly volume of alcohol consumption and T2DM tends to be reduced relative to an age-varying covariate model (Figure 9.2).⁴¹²

 $h_i(t) = h_0(t) \exp\{\beta x_i(t_{ij}) + \alpha \hat{y}_i(t_{ij})\}$ $\hat{y}_i(t_{ij}) = (\beta_0 + b_{0i}) + (\beta_1 + b_{1i})t_{ij} + \varepsilon_{ij}$

Formula 9.2 Calculation of a two-stage survival model

As shown in Formula 9.2, predicted values of weekly volume of alcohol consumption for the i^{th} individual at the j^{th} observation ($\hat{y}_i(t_{ij})$) are calculated within the mixed effects model as the estimated mean intercept (β_0) and the predicted deviation of each participant from the mean

Chapter 9: Longitudinal alcohol consumption and the risk of T2DM

intercept (b₀), plus the estimated mean change in alcohol consumption per unit increase in age $(\beta_1(t_{ij}))$ and the predicted participant-specific deviation from the mean slope per year increase in age (b₁(t_{ij})), with ϵ_{ij} representing random error.

Given that the accuracy of a survival model is dependent upon an appropriately specified function of alcohol consumption, linear and non-linear mixed effects models were run on the log₂-transformed volume of alcohol intake according to the restricted range of fractional powers defined in Section 3.2.4.1. Random slopes were assumed in all cases and individual-level predictions from the best-fitting sex-specific models then extracted and added to the parametric survival model.

However, while measurement error is factored in during the calculation of the individual-level predictions, this uncertainty is not carried across into the survival model, which only utilises the fitted values estimated by the mixed model. As a consequence, while a two-stage approach generally provides an improvement in accuracy over age-varying covariate models, producing parameter estimates that are less biased toward the null by random error, resulting coefficients tend to exhibit too great a degree of precision.⁴¹² Additionally, in only using fitted values as predicted for each jth observation, the survival component of the two-stage model still assumes that covariate values are constant between observations (Figure 9.1).

9.4.3.5 Shared random effects survival model

In an attempt to circumvent some of the limitations that carry over into the two-stage method, the dose-response relationship between the volume of alcohol consumption and T2DM risk was also estimated using a shared random effects model. Unlike the two-stage approach, the shared random effects model estimates the longitudinal and survival processes simultaneously within a single model.⁴¹⁵ In doing so, rather than the risk of T2DM being calculated according to the last fitted value available, predictions are instead calculated for all points across the life course ($y_i(t)$), as illustrated in Figure 9.3.⁴²⁵



Figure 9.2 Illustration of the longitudinal trajectory and hazard function within a shared random effects survival model

Though computationally intensive, the result is a longitudinal sub-model that better operationalises the unobserved trajectory of alcohol consumption for each participant and allows for the risk of T2DM among surviving participants at each diagnosis to be calculated according to the predicted participant-specific volume of alcohol consumption at the precise age recorded at diagnosis (Formula 9.3).^{415,422}

 $h_i(t) = h_0(t) \exp\{\beta x_i(t_{ij}) + \alpha y_i(t)\}$

 $y_i(t) = (\beta_0 + b_{0i}) + (\beta_1 + b_{1ij})t_{ij} + \varepsilon_{ij}$

Formula 9.3 Calculation of a shared random effects survival model

In simulation studies, these shared random effects models have been found to produce the least biased estimates of any of the three methods considered,^{411,412} with hazards closest to the true association under study and the most robust to any misspecification within the mixed effects model.^{422,426} Furthermore, by directly accounting for measurement error through the inclusion of random effects within the survival sub-model, parameters estimated by a shared random

effects approach also exhibit a larger and more appropriate degree of precision relative to the two-stage method.^{412,425}

9.4.3.5.1 Intercepts versus current values

To explore how the dose-response relationship differs according to consumption at different points in the adult life course, separate models were run that calculated risk according to consumption as predicted at the intercept $((\beta_0+b_{0i}))$ or the age at diagnosis $((\beta_0+b_{0i})+(\beta_1+b_{1ij})t_{ij})$. Then, to ascertain the independent contribution of the two parameters, each was then adjusted for the other via concurrent inclusion within a single model.

9.4.3.5.2 Sick quitter effects

To test whether participants with downward trajectories represented a group of 'sick quitters' at increased risk of T2DM, shared random effects models were constructed that also included a parameter equal to the association between the rate of change and the risk of T2DM (α_3). By holding constant any differences in consumption at the intercept (α_1) and at the age of diagnosis (α_2), these fully adjusted models report the relationship between the slope of the longitudinal trajectory independent of past and current consumption (Formula 9.4).

 $\begin{aligned} h_i(t) = h_0(t) \exp\{\beta x_i(t_{ij}) + \alpha_1 y^{a_i}(t) + \alpha_2 y^{b_i} + \alpha_3 y^{c_i}\} \\ y^{a_i}(t) = (\beta_0 + b_{0ij})t_{ij} + \varepsilon_{ij} \\ y^{b_i}(t) = (\beta_0 + b_{0ij}) + (\beta_1 + b_{1ij})t_{ij} + \varepsilon_{ij} \\ y^{c_i}(t) = (\beta_1 + b_{1ij})t_{ij} + \varepsilon_{ij} \end{aligned}$

Formula 9.4 Calculation of a shared random effects survival model with an age-dependent slope parameterisation

The resulting slope coefficient is interpreted as the difference in risk per unit increase in the rate of change per year of age (i.e. a slope coefficient equal to one). In order to make this coefficient more meaningful, it was transformed as shown in Formula 9.5 to reflect a 5% increase (HR_{inc}) or decrease (HR_{dec}) in the rate of change per decade increase in age, versus the mean percentage change in weekly alcohol consumption over the period (v).⁴²⁷ The p-values reported for the each HR are equal, referring to the p-value for the untransformed slope coefficient. To simplify the calculation of the mean rate of change per decade, these shared random effects models were calculated with linear slopes.

 $HR_{inc}=exp(\alpha_2^*((log(1+5/100))-log(1+v/100)))$

 $HR_{dec}=exp(\alpha_2^*((log(1-5/100))-log(1+v/100)))$

Formula 9.5 Calculating the risk of T2DM for a 5% increase or decrease in the rate of change versus the mean percentage change over the period

9.4.3.5.3 Reference category

Finally, to formally explore the effect of providing no adjustment for differences in risk between heterogeneous infrequent, non-current and never drinkers, a sensitivity analyses was run that calculated the intercept and current value dose-response associations without the inclusion of the alcohol consumption indicator variable. Doing so was considered equivalent to running a survival analysis relative to pooled non-drinkers.

9.4.3.6 Goodness of fit

As in Chapter 8, goodness of fit was measured according to the log-likelihood and BIC statistics, with an improvement in fit defined as any reduction in the BIC greater than or equal to a value of 10.406

9.4.3.7 Statistical packages

Proportional hazard models were calculated using the -st- package.³⁹⁰ Where fitted values of weekly alcohol consumption were predicted for inclusion within the two-stage survival model, linear and non-linear predictions were created using the -mixed- package.⁴⁰⁷ Shared random effects models were calculated using the -stjm- package.⁴¹⁵ Results from analyses that utilised imputed covariate data were calculated using the -mi- suite of commands.³⁴² The use of imputed data was not supported by the -stjm- package.

9.5 Results

9.5.1 Conventional survival model

Construction of the conventional survival model began with identifying the best-fitting parametric baseline hazard. Fit statistics from the resulting crude models are reported in Table 9.1. For both men and women, the baseline hazard function was best represented by a Weibull distribution, which was thus selected for all subsequent models in this chapter. The observed and predicted baseline hazard functions are shown in Appendix 9.1.

Table 9.1 Relationship between the average weekly volume of alcohol consumption and the risk ofT2DM, stratified by sex. Shared random effects model, goodness of fit statistics, observed data.

Chapter 9: Longitudinal alcohol consumption and the risk of T2DM

Baseline hazard function	Log-likelihood	BIC ^a
Men		
Cox (unspecified)	-4461	8957
Exponential	-1728	3500
Gompertz	-1698	3448
Weibull	-1693	3437
Women		
Cox (unspecified)	-1879	3790
Exponential	-759	1557
Gompertz	-749	1546
Weibull	-747	1542
^a Bayesian information criterion.		

After adjustment for drinking category, a two-fold increase in the weekly volume of alcohol consumption at the intercept was associated with a 4% (HR 1.04, 95% CI 0.98-1.12) increase in the risk of T2DM among men and a statistically significant 26% (HR 0.74, 95% CI 0.65-0.84) decrease in risk among women (Table 9.2).

Multivariable adjustment improved the fit of both male and female conventional survival models. Increases in risk among men were attenuated to the null (HR 0.99, 95% CI 0.92-1.06), and reductions in risk among women reduced to 20% (HR 0.80, 95% CI 0.69-0.93) per two-fold increase in consumption. Results from the imputed dataset, which accounted for missing covariate data, showed comparable results (Appendix 9.2).

Table 9.2 Relationship between the average weekly volume of alcohol consumption and the risk of T2DM, stratified by sex. Conventional survival analysis, observed data.

-	Men		Wome	n
Alcohol consumption (wave	HR (95% CI)	p-value	HR (95% CI)	p-value
Model 1				
Cases/non-cases		560/4,867		268/2,155
Consumption volume				
g/week (log ₂)	1.04 (0.98, 1.12)	0.211	0.74 (0.65, 0.84)	<0.001
Consumption category				
Current drinker	(reference)		(reference)	
Infrequent drinker	1.25 (0.75, 2.08)	0.385	0.48 (0.24, 0.95)	0.036
Non-current drinker	1.74 (0.83, 3.67)	0.142	0.46 (0.19, 1.11)	0.084
Never drinker	2.39 (1.24, 4.60)	0.009	0.42 (0.18, 0.96)	0.040
Log likelihood		-1693		-747
BIC ^a		3437		1542
Model 2				
Cases/non-cases		451/4,342		206/1,847
Consumption volume				
g/week (log ₂)	0.99 (0.92, 1.06)	0.679	0.80 (0.69, 0.93)	0.004
Consumption category				
Current drinker	(reference)		(reference)	
Infrequent drinker	0.84 (0.48, 1.44)	0.519	0.51 (0.23, 1.12)	0.092
Non-current drinker	1.19 (0.56, 2.55)	0.652	0.44 (0.16, 1.20)	0.109
Never drinker	0.81 (0.38, 1.75)	0.597	0.36 (0.14, 0.95)	0.040
Log likelihood		-1253		-521
BIC ^a		2666		1188

Model 1 reported the dose-response relationship between the volume alcohol consumption and T2DM following adjustment for consumption category. Model 2 included additional adjustment for BMI, date of birth, employment status, ethnicity, family history of T2DM, occupational grade, physical activity and smoking status, as defined at baseline. Ethnicity was derived from responses at waves one and five. ^aBayesian information criterion.

9.5.2 Age-varying covariates survival model

Hazard ratios reported according to the current value of alcohol consumption were elevated relative to those associated with consumption at the intercept in Table 9.2. As reported in Table 9.3, doubling of alcohol consumption was associated with an 11% (HR 1.11, 95% Cl 1.03-1.19) increase in the risk of T2DM among men and a 16% (HR 0.84, 95% Cl 0.74-0.95) reduction in risk among women in crude models. Following multivariable adjustment, risks among men were reduced slightly to a 7% increase in hazards (HR 1.07, 95% Cl 0.99-1.15), with an 8% reduction

in hazards reported among women (HR 0.92, 95% CI 0.80-1.06). Hazard ratios estimated using the imputed dataset were comparable (Appendix 9.3).

	Men		Womer	<u>ו</u>
Alcohol consumption	HR (95% CI)	p-value	HR (95% CI)	p-value
Model 1				
Cases/non-cases		560/4,867		268/2,155
Consumption volume				
g/week (log ₂)	1.11 (1.03, 1.19)	0.004	0.84 (0.74, 0.95)	0.005
Consumption category				
Current drinker	(reference)		(reference)	
Infrequent drinker	2.35 (1.37, 4.05)	0.002	0.96 (0.48, 1.91)	0.906
Non-current drinker	2.07 (0.98, 4.37)	0.057	1.15 (0.53, 2.48)	0.722
Never drinker	3.91 (1.95, 7.84)	<0.001	0.67 (0.26, 1.67)	0.387
Log likelihood		-1673		-734
BIC ^a		3406		1523
Model 2				
Cases/non-cases		451/4,342		206/1,847
Consumption volume				
g/week (log ₂)	1.07 (0.99, 1.15)	0.098	0.92 (0.80, 1.06)	0.241
Consumption category				
Current drinker	(reference)		(reference)	
Infrequent drinker	1.61 (0.90, 2.89)	0.110	1.14 (0.53, 2.45)	0.732
Non-current drinker	1.29 (0.57, 2.95)	0.539	1.26 (0.54, 2.96)	0.594
Never drinker	1.30 (0.58, 2.93)	0.528	0.62 (0.23, 1.71)	0.358
Log likelihood		-1157		-478
BIC ^a		2500		1124

Table 9.3 Relationship between the average weekly volume of alcohol consumption and the risk of T2DM, stratified by sex. Age-varying covariate survival analysis, observed data.

Model 1 reported the dose-response relationship between the volume alcohol consumption and T2DM following adjustment for consumption category. Model 2 included additional adjustment for BMI, date of birth, employment status, ethnicity, family history of T2DM, occupational grade, physical activity and smoking status. Ethnicity was derived from responses at waves one and five. ^aBayesian information criterion.

9.5.3 Two-stage survival model

The best-fitting mixed effects model was determined separately for men and women and plotted in Figure 9.3. These trajectories were consistent with those reported in Chapter 8 (Figure 8.3), with the average volume of consumption increasing up to around 55 years of age before declining thereafter.



Figure 9.3 Crude, sex-specific and best-fitting trajectories of the mean weekly volume of alcohol consumption (log₂) between the ages of 40-84 years. Observed data.

Multivariable-adjusted risk estimates were closer to the null and no longer statistically significant when calculated based upon predicted values of alcohol consumption. A 4% (HR 1.04, 95% CI 0.97-1.11) increase in T2DM risk was observed per two-fold increase in alcohol intake among men, and an 8% (HR 0.93, 95% CI 0.85-1.02) reduction in risk among women (Table 9.4). Analyses based on the imputed dataset were comparable and are reported in Appendix 9.4.

	Men		Men Women		ı
Alcohol consumption	HR (95% CI)	p-value	HR (95% CI)	p-value	
Model 1					
Cases/non-cases		560/4,867		268/2,155	
Consumption volume					
g/week (log ₂)	1.05 (0.98, 1.11)	0.151	0.82 (0.76, 0.89)	<0.001	
Consumption category					
Current drinker	(reference)		(reference)		
Infrequent drinker	1.41 (1.01, 1.97)	0.045	1.29 (0.91, 1.84)	0.155	
Non-current drinker	1.31 (0.66, 2.58)	0.440	1.35 (0.81, 2.25)	0.253	
Never drinker	2.52 (1.36, 4.67)	0.003	0.75 (0.36, 1.56)	0.445	
Log likelihood		-1677		-730	
BIC ^a		3413		1514	
Model 2					
Cases/non-cases		451/4,342		206/1,847	
Consumption volume					
g/week (log ₂)	1.04 (0.97, 1.11)	0.262	0.93 (0.85, 1.02)	0.145	
Consumption category					
Current drinker	(reference)				
Infrequent drinker	1.23 (0.85, 1.78)	0.279	1.43 (0.95, 2.13)	0.083	
Non-current drinker	1.04 (0.49, 2.20)	0.923	1.51 (0.85, 2.67)	0.160	
Never drinker	1.06 (0.50, 2.22)	0.880	0.73 (0.33, 1.65)	0.456	
Loa likelihood		-1158		-478	
BIC ^a		2502		1123	

Table 9.4 Relationship between the average weekly volume of alcohol consumption and the risk of T2DM, stratified by sex. Two-stage survival analysis, observed data.

Model 1 reported the dose-response relationship between the volume alcohol consumption and T2DM following adjustment for consumption category. Model 2 included additional adjustment for BMI, date of birth, employment status, ethnicity, family history of T2DM, occupational grade, physical activity and smoking status. Ethnicity was derived from responses at waves one and five. ^aBayesian information criterion.

9.5.4 Shared random effects model

9.5.4.1 Intercept versus the current value

Using the same best-fitting trajectories as shown in Figure 9.3, the multivariable-adjusted risk of T2DM per two-fold increase in the intercept and current values of alcohol consumption were estimated in separate models for each sex (Table 9.5). Intake predicted at the age of 39.6 years was associated with little change in the risk of T2DM among men (HR 0.98, 95% CI 0.92-1.05), while a two-fold increase in the current value of alcohol consumption was associated with a non-significant 5% increase in risk (HR 1.05, 95% CI 0.97-1.14). Among women, both the intercept and current values of alcohol consumption were associated with non-significant reductions in risk, though with an effect strongest according to the intercept parameterisation.

Table 9.5 Multivariable-adjusted relationship between intercept and current value parameterisations of average weekly volume of alcohol consumption and the risk of T2DM, stratified by sex. Shared random effects survival analysis, observed data.

	Men (n=4,793)		Women (n=2,053)	
Parameterisation	HR (95% CI)	p-value	HR (95% CI)	p-value
Intercept value				
Consumption volume				
g/week (log ₂)	0.98 (0.92, 1.05)	0.628	0.89 (0.79, 1.00)	0.054
Consumption category				
Current drinker	(reference)			
Infrequent drinker	1.03 (0.71, 1.50)	0.868	1.29 (0.83, 1.99)	0.259
Non-current drinker	0.84 (0.43, 1.66)	0.619	1.33 (0.71, 2.50)	0.368
Never drinker	0.82 (0.40, 1.68)	0.588	0.63 (0.26, 1.55)	0.315
Log likelihood		-38234		-15907
BIC ^a		76703		32026
Current value				
Consumption volume				
g/week (log ₂)	1.05 (0.97, 1.14)	0.189	0.92 (0.81, 1.06)	0.244
Consumption category				
Current drinker	(reference)		(reference)	
Infrequent drinker	1.32 (0.86, 2.02)	0.209	1.37 (0.83, 2.28)	0.217
Non-current drinker	1.14 (0.55, 2.39)	0.725	1.45 (0.73, 2.90)	0.289
Never drinker	1.14 (0.53, 2.44)	0.741	0.69 (0.27, 1.77)	0.443
Log likelihood		-38233		-15908
BIC ^a		76701		32029

All models included adjustment for consumption category as well as BMI, date of birth, employment status, ethnicity, family history of T2DM, occupational grade, physical activity and smoking status. Ethnicity was derived from responses at waves one and five. ^aBayesian information criterion.

Table 9.6 reports the same two parameterisations but included concurrently within single sexspecific models. After adjusting for the current value of alcohol consumption, a two-fold increase in the intercept was associated with a statistically significant 20% (HR 0.80, 95% CI 0.69-0.92) reduction in risk among men and a 36% (HR 0.64, 95% CI 0.43-0.97) reduction among women. By contrast, when holding intercept values constant, a two-fold increase in the current value of alcohol intake was associated with a statistically significant 35% increase in risk (HR 1.35, 95% CI 1.12-1.62) among men and a 47% increase in risk among women (HR 1.47, 95% CI 0.94-2.31). No differences in goodness of fit were observed between parameterisations.

After adjustment for the intercept and current values of alcohol consumption, risks were elevated at any age among all heterogeneous non-drinking categories except for female never drinkers, for whom a non-significant reduction in risk was apparent relative to pooled current drinkers (HR 0.82, 95% CI 0.31-2.18).

Men (n=4,793)		 Women (n=2,053)		
Parameterisation	HR (95% CI)	p-value	HR (95% CI)	p-value
Intercept value				
Consumption volume				
g/week (log ₂)	0.80 (0.69, 0.92)	0.002	0.64 (0.43, 0.97)	0.034
Current value				
g/week (log ₂)	1.35 (1.12, 1.62)	0.001	1.47 (0.94, 2.31)	0.091
Consumption category				
Current drinker	(reference)		(reference)	
Infrequent drinker	1.56 (1.00, 2.45)	0.052	1.74 (0.99, 3.06)	0.053
Non-current drinker	1.61 (0.73, 3.55)	0.242	1.87 (0.88, 3.99)	0.104
Never drinker	1.43 (0.63, 3.24)	0.397	0.82 (0.31, 2.18)	0.692
Log likelihood		-38228		-15905
BIC ^a		76700		32032

Table 9.6 Multivariable-adjusted associations between conditional intercept and current value parameterisations of average weekly volume of alcohol consumption and the risk of T2DM, stratified by sex. Shared random effects survival analysis, observed data.

All models included adjustment for consumption category as well as BMI, date of birth, employment status, ethnicity, family history of T2DM, occupational grade, physical activity and smoking status. Ethnicity was derived from responses at waves one and five. ^aBayesian information criterion.

9.5.4.2 Reference category

When the separate survival models reported in Table 9.5 were re-run without adjustment for the differences in risk among heterogeneous non-drinking groups, little change in risk estimates was observed among men, while reductions in risk among women were increased. For instance, compared to a non-significant 8% (HR 0.92, 95% CI 0.81-1.06) reduction in risk per two-fold increase in the current value of alcohol intake after adjustment for consumption category, results in Table 9.7 show a significant 13% (HR 0.87, 95% CI 0.81-0.89) reduction in risk.

Table 9.7 Multivariable-adjusted relationship between intercept and current value parameterisations of average weekly volume of alcohol consumption and the risk of T2DM, stratified by sex and without adjustment for alcohol consumption category. Shared random effects survival analysis, observed data.

	Men (n=4,79	3)	Women (n=2,	D53)
Parameterisation	HR (95% CI)	p-value	HR (95% CI)	p-value
Intercept value				
Consumption volume				
g/week (log ₂)	0.99 (0.94, 1.04)	0.569	0.86 (0.79, 0.93)	< 0.001
Log likelihood		-38551		-16088
BIC ^a		77308		32361
Current value				
Consumption volume				
g/week (log ₂)	1.02 (0.97, 1.08)	0.358	0.87 (0.81, 0.95)	0.001
Log likelihood		-38551		-16089
BIC ^a		77307		32364

All models included adjustment for BMI, date of birth, employment status, ethnicity, family history of T2DM, occupational grade, physical activity and smoking status. Ethnicity was derived from responses at waves one and five. ^aBayesian information criterion.

9.5.4.3 Sick quitter effects

The linear trajectory on the \log_2 scale was flat over time among men, falling by 0.2% (p=0.894) per decade. By contrast, female consumption fell by an average 3.1% (p=<0.001) per decade. Table 9.8 reports the multivariable-adjusted association between differences in these rates of change and the risk of T2DM, after adjustment for past and current alcohol consumption.

Among both sexes, a 5% increase in the decennial rate of change was associated with a slight increase in the risk of T2DM (men: HR 1.12, 95% CI 0.59-2.10; women: HR 1.11, 95% CI 1.01-1.62). By contrast, a 5% reduction in the rate of change was associated with a small decrease in
the risk of T2DM among both sexes (men: HR 0.89, 95% CI 0.46-1.73; women: HR 0.97, 95% CI 0.60-0.99).

Table 9.8 Multivariable-adjusted relationship between the rate of change in the average weekly volume of alcohol consumption and the risk of T2DM, stratified by sex. Shared random effects survival analysis, observed data.

	Men (n=4,7	93)	Women (n=2,0	053)
Parameterisation	HR (95% CI)	p-value	HR (95% CI)	p-value
Intercept value				
Consumption volume				
g/week (log ₂)	0.48 (0.24, 0.93)	0.030	0.54 (0.22, 1.33)	0.179
Current value				
Consumption volume				
g/week (log ₂)	2.45 (1.23, 4.88)	0.010	1.80 (0.73, 4.45)	0.202
Slope				
5% increase in the rate of cha	nge			
g/week (log ₂)	1.12 (0.59, 2.10)	0.733	1.11 (1.01, 1.62)	0.897
5% decrease in rate of change	<u>)</u>			
g/week (log ₂)	0.89 (0.46, 1.73)	0.733	0.97 (0.60, 0.99)	0.897
Log likelihood		-38282		-15917
BIC ^a		76808		32056
Log likelihood BIC ^a		-38282 76808		-15917 32056

All models included adjustment for consumption category as well as BMI, date of birth, employment status, ethnicity, family history of T2DM, occupational grade, physical activity and smoking status. Ethnicity was derived from responses at waves one and five. ^aBayesian information criterion.

9.6 Summary of findings

This chapter aimed to investigate the merit of utilising parameterisations of the longitudinal trajectory conventionally overlooked by existing survival analyses, and establish whether participants with downward trajectories of alcohol consumption represent a distinct group of sick quitters at elevated risk of T2DM.

Despite adopting a different timescale, results from the conventional survival models reported in Table 9.2 are consistent with the meta-analysis reported in Chapter 3, whereby consumption predicted at 39.6 years of age was associated with reductions in risk only among women. When the dose-response relationship was instead modelled according to the current value of consumption, female reductions in risk remained present but were of a smaller magnitude regardless of the method used (Tables 9.3 to 9.5).

In order to establish which of these two different dimensions of the longitudinal trajectory might be the most important for estimating T2DM risk, a shared random effects model was constructed that modelled both parameters concurrently (Table 9.6). After adjusting for differences in the current volume of intake, a two-fold increase in alcohol consumption at the intercept was associated with sizeable reductions in T2DM risk. Conversely, when adjusting for differences in intercept values, higher current values of consumption were associated with a greater risk of T2DM. A sensitivity analysis that excluded adjustment for consumption category indicated more pronounced reductions in risk among women for both parameterisations of alcohol consumption (Table 9.7).

Finally, to determine whether participants who decreased their consumption over time represented a group of 'sick quitters' who may have attenuated their intake owing to having manifested symptoms associated with an elevated risk of T2DM, a shared random effects model was constructed that modelled the effect of differences in rate of change independent of past and current consumption (Table 9.8). Although men and women who decreased their consumption at a faster rate appeared to show slightly lower risks of T2DM, associations were not statistically significant.

9.7 Limitations

The dose-response relationship between the weekly volume of alcohol consumption and the risk of T2DM was estimated by linking longitudinal and survival processes via a range of increasingly complex methods. An alternative approach considered was to jointly model both processes with trajectories of alcohol consumption captured via latent classes.^{428,429} Rather than treating participants as a homogenous group by estimating participant-specific slopes constrained according to the same longitudinal function, the joint latent class approach instead treats the population as heterogeneous and thus constituted of multiple disparate trajectory profiles, such as those illustrated in Figure 3.17. Using this approach, participants are divided into a finite number of latent sub-groups, with their membership defined according to a categorical latent variable that can then be included within a survival sub-model.⁴²⁸ As each latent class represents a specific average trajectory, corresponding risk estimates can be used to ascertain the effect of a particular drinking typologies upon the risk of T2DM, such as a pattern of stable moderate consumption. In this sense, a joint latent class model can offer an interpretive benefit over a shared random effects model.

Chapter 9: Longitudinal alcohol consumption and the risk of T2DM

However, the joint latent class approach is subject to its own limitations.⁴²⁸ As described elsewhere, the composition of distinct latent classes is in part dependent upon population heterogeneity or, more specifically, the degree of longitudinal variability in a given variable.²⁵⁹ Thus, if most participants within a cohort exhibit similar consumption trajectories over time, the majority will be categorised into a single latent class with no resulting capacity for exploring differences in risk according to heterogeneous trajectories of clinical interest. This particular shortcoming has been found to apply to the Whitehall II dataset, with heterogeneous nondrinking groups pooled with light drinkers, and heavier drinkers combined with more moderate users of alcohol.^{430,431} Such groupings highlight the arbitrary and potentially non-informative nature of statistically-derived latent classes, and indicate that joint latent class models applied to Whitehall II data for the exploration of T2DM risk were likely to insufficiently disaggregate heterogeneous consumption categories, thereby falling foul of the same misclassification errors present in analytic methods conventionally applied in epidemiologic research. In addition, by modelling heterogeneity in longitudinal consumption solely through the derivation of latent categories, it is not possible to explore the relationship between different dimensions of such heterogeneity and T2DM risk, such as the rate of change or deviation from the mean trajectory.⁴²² It was for these reasons that a shared random effects approach was chosen.

In contrast to the majority of previous studies, which report age-adjusted risk estimates with a time-to-event metric, the analyses reported in this chapter used age as the timescale of interest due to both the incidence of T2DM and volume of alcohol consumption operating as a function of age as opposed to the time and length of participation.^{419,420} Results derived from models in this chapter and those reported elsewhere thus may not be directly comparable. To explore how results may have differed according to the choice of timescale, post-hoc sensitivity analyses were run that reproduced the shared random effects models reported in Tables 9.5 and 9.6 but using a conventional time-to-event timescale. With the intercept and current value parameterisations modelled separately, estimates remain of the same direction and approximate magnitude, with reductions in risk continuing to be most pronounced among women (Appendix 9.5). When both parameters were estimated concurrently, as per Table 9.6, both parameterisations remained statistically significant, with the same differential doseresponse relationships still reported (Appendix 9.6). These sensitivity analyses affirm that similar conclusions are drawn irrespective of the choice of timescale. Specifically, that conventional approaches may oversimplify the relationship between alcohol and T2DM risk through a failure to consider changes in alcohol consumption across the life course.

Utilising an age-based timescale within a shared random effects framework should permits the calculation of dose-response relationships according to intakes at specific ages that may be of utility to policymakers or clinicians. This is achieved by applying a lag effect to the current value parameter, whereby the risk of T2DM at age t is calculated according to the value of the longitudinal trajectory at t-C, where C is the lag of interest.⁴²² Unfortunately, the -stjm- package still in the process of being developed, with such functionality is not yet available. As a result, risks reported Chapter 9 were limited to alcohol consumption at the intercept (40 years of age) and the current value (a mean 62 years of age). Although possible to rescale the age variable such that the intercept intersects age at a point in the life course that may be of greater interested to policymakers, such as the minimum legal drinking age, such a model would be required to extrapolate the longitudinal trajectory far beyond the period observed. The validity of any resulting estimates would thus be questionable given that the functional form of the longitudinal trajectory between middle age and later life was unlikely to hold when extended to a period such as young adulthood, which is marked by pronounced elevations in consumption from very low volumes.^{259,266}

All reported models assumed a linear dose-response relationship between the volume of alcohol consumption and the risk of T2DM. Such an assumption was not consistent with the non-linear dose-response curve reported in Chapter 3. However, as indicated in Chapter 7, the positive skewness and low average volume of alcohol consumption within the Whitehall II cohort were such that it may not have been possible to detect non-linear dose-response relationships, such as increases in risk at higher intakes. To explore the possibility that a non-linear dose-response relationship may be present between an alternative parameterisation of alcohol consumption and T2DM risk, fitted values from the two-stage models reported in Table 9.4 were transformed according to a restricted range of fractional powers, as detailed in Section 3.2.4.1. No improvements in BIC were identified for any non-linear dose-response association, relative to the linear model. As a result of such a limitation, caution should be applied when interpreting the linear dose-response coefficients and seeking to extrapolate them to higher volumes of drinking. Moreover, from a public health perspective, while higher volumes of alcohol consumption at the intercept appear associated with reductions in T2DM risk among both sexes after adjustment for consumption later in the life course (Table 9.6), it should be borne in mind that drinking may still be associated with a heightened risk of other health conditions even at relatively low volumes.432

Although little difference was evident between coefficients reported for models applied to the observed data and those that accounted for item non-response on covariates, all models were restricted to individuals free of missing alcohol consumption and diagnosis data at wave three. As reported in Chapter 6, participants with unit non-response across \geq 4 waves of follow-up (Table 6.5) or item non-response at any point during the study (Table 6.7) exhibited a lower volume of alcohol consumption and worse metabolic risk profile at the beginning of the study than those with observed data. There was thus the possibility that risks associated with lower volumes of alcohol may have been underestimated.

9.8 Discussion

Analyses reported in Chapter 3 and Chapter 7 both indicate a disparity in the dose-response relationship between men and women, with reductions in risk specific to female participants. When the dose-response relationship was parameterised according to drinking at the intercept, two-fold increases in the weekly volume of alcohol consumption were associated with a null effect among men and a borderline significant 11% reduction in risk among women (Table 9.5), consistent with earlier findings.

Capitalising upon the flexibility afforded by shared random effects models, the relationship between intake at the intercept and T2DM risk was recalculated with adjustment for the current value of consumption (Table 9.6). Here, two-fold increases in the intercept were associated with significant reductions in risk in both sexes, but with the magnitude of reduction remaining greatest among women (men: 20%; women: 36%). By contrast, when holding the intercept constant, increases in the current volume of consumption were marked by sizeable increases in risk (men: 35%; women: 45%). With this same disparity in effect found to be apparent when shared random effects models were recalculated based on a time-to-event as opposed to an age-based timescale (Appendix 9.6), there was thus an indication that both parameterisations of the longitudinal trajectory exhibit statistically independent and divergent relationships with T2DM risk. Additionally, that conventional survival analyses which adjust for age and rely upon a single measurement of alcohol consumption fail to report important differences in the dose-response relationship across the adult life course.

Given a mean age of 62 years at the time of diagnosis, and a median follow-up of 20 years from the intercept, the results suggest that increases in risk associated with higher volumes of consumption may be specific to intake at older age. Such an effect would be in accordance with studies that report age-related impairments to the alcohol metabolism^{196,197,198,199} and consequent increases in length of exposure to higher concentrations of pro-inflammatory metabolites such as acetaldehyde and acetate.^{200,201} Unfortunately, although interventional studies have investigated the relationship between alcohol consumption and a range of inflammatory biomarkers, the number of studies is small, methods are heterogeneous and pooled effects statistically insignificant.¹¹⁸ While a J-shaped association was estimated between alcohol consumption and CRP in an analysis based on data from 22 Mendelian randomisation studies,¹³¹ estimates were not disaggregated by age group. As such, it was not possible to state conclusively whether alcohol may differentially affect the concentration of inflammatory biomarkers according to the age of participants. Although evidence concerning the inflammatory pathway is currently tenuous, results in Table 9.6 nonetheless allude to the possibility that any advantageous biological effects of alcohol consumption, such as the linear increases in HDL concentration reported by interventional studies,^{117,118,119} may be outweighed in later life by heightened inflammatory responses following age-related impairments to the alcohol metabolism.

This hypothesis was further substantiated by a post-hoc analysis, which estimated the effect of differences in the longitudinal trajectory after adjusting solely for drinking at the intercept (Appendix 9.7). Relative to the mean slope, a 5% decrease in the rate of change was associated with significant reductions in the risk of T2DM among both sexes (men: HR 0.42, 95% CI 0.29-0.61; women: HR 0.77, 95% CI 0.63-0.94). These reductions in risk may have been a consequence of participants with faster longitudinal declines in alcohol consumption having consumed alcohol in lower volumes during the sensitive period of old age, relative to those who decreased their intake at the average rate.

Adults who reduce their intake over time are typically posited to represent a group of individuals who attenuate their drinking owing to ill-health,^{136,138} with a number of studies linking the onset of poor health to a cessation of drinking^{144,145,146} or a reduction in the frequency of consumption.⁴¹⁸ In an effort to isolate the relationship between differences in the rate of change and T2DM risk, analyses were undertaken that reported a slope coefficient following adjustment for past and current intake (Table 9.8). Among both sexes, participants who reduced their alcohol consumption at a faster rate than the mean experienced a lower risk of T2DM, though risk estimates were non-significant. Accordingly, the fully-adjusted model failed to corroborate the presence of a sick quitter effect, whereby participants with a faster rate of decline would instead be expected to report a higher risk of T2DM, independent of the volume consumed.

It was possible that a sick quitter effect was lost as a consequence of the diverse reasons for which study participants appear to attenuate their alcohol consumption. For instance, while

257

Chapter 9: Longitudinal alcohol consumption and the risk of T2DM

previously published analyses of Whitehall II data indicate that around 21% of men and 22% of women who reduced their intake did so as a direct consequence of illness or pharmaceutical contraindication, a far higher proportion reduced their drinking as a health precaution (men: 45%; women: 34%), with others doing so for reasons as broad as a reduction in opportunities for social drinking, an effort to save money, pressure from family and friends and a history of alcohol misuse.⁴⁰⁸ Elsewhere, an analysis of longitudinal data from the Health and Retirement Study reported differences in longitudinal trajectories according to a range of socio-economic factors, with greater affluence and a higher frequency of socialising all associated with increased probabilities of an upward drinking trajectory.²⁸⁹ Other studies report similar findings, with low incomes^{433,434,435} and low education⁴³⁵ associated with reductions or steeper declines in the volume or frequency of alcohol consumption, suggesting that disposable incomes and social networks may affect drinking behaviours other than just the onset of poor health. Although associations have also been found between drinking trajectories and retirement, effects appear to vary according to differences in occupational drinking cultures.⁴³⁵ While differences in consumption by these socio-economic factors could feasible be a consequence of correlations between socio-economic status and health,⁴³⁶ effects were robust to adjustment for illness and self-reported health.^{289,433,434,435} Thus, despite evidence linking decreases in alcohol consumption to a decline in health, the determinants of downward trajectories are complex, with health protection, family pressures and socio-economic factors all playing a role in determining longitudinal drinking patterns.

As an alternative explanation for the absence of an increased risk among participants with faster rates of decline, it was possible that selection bias may have been a factor. Specifically, participants who were experiencing a decline in health status owing to issues associated with a heightened risk of T2DM, such as such as obesity or hypertension, may have chosen to withdraw themselves from the study. In doing so, sampled participants with downward trajectories would disproportionately represent those who decreased their consumption due to factors not associated with an increased T2DM risk.

In summary, the results reported in this chapter appear to indicate that conventional survival analyses may provide too simplistic a picture of the relationship between alcohol consumption and the risk of T2DM. When intercept and current values of alcohol intake were estimated concurrently, disparate dose-response associations were identified. With age used as the timescale, there was an indication that reductions in risk associated with alcohol consumption in midlife may be countered by a greater sensitivity to alcohol in older age. In the absence of

258

more robust evidence concerning the validity of putative pathways by which alcohol may reduce the risk of T2DM, it would be unwise to recommend that non-drinking adults begin consuming alcohol, especially if such individuals abstained owing to ill-health, pharmaceutical contraindication or a history of alcohol misuse. Furthermore, with higher volumes of alcohol consumption associated with an increased risk of T2DM with advancing age, and relatively moderate volumes in midlife associated with reductions in risk, it would be a reasonable precaution to recommend a general attenuation to the volume of consumption among drinkers.

Chapter 10

Discussion

10 Discussion

10.1 Research questions

Previous meta-analyses suggest that moderate alcohol consumption may reduce the risk of T2DM, with the latest publication at the time of commencing this project having reported a J-shaped association and a peak reduction in risk at around 22-24 g/day among both sexes, relative to quasi-never drinkers.¹⁰ When this meta-analysis was updated through the inclusion of 21 studies that were either newly published or provided updated data with longer periods of follow-up or improved confounder adjustment, significant differences in dose-response were found between men and women. As reported in Chapter 3, the most up-to-date observational data suggest that any reduction in risk may actually be specific to women, peaking among those that consume an average 35 g/day (HR 0.66, 95% CI 0.55-0.78).

However, when constituent studies were investigated in more detail, a series of methodological flaws were identified that may undermine the validity of any inference drawn from this result, including a failure to account for differences in risk among heterogeneous non-drinkers and limited consideration as to the effect of longitudinal changes in alcohol consumption upon the dose-response relationship reported.

As a result of such observations, a number of primary aims were formulated:

- 1. Determine the degree to which categories of alcohol consumption risked being subject to misclassification error as a result of longitudinal changes in alcohol consumption.
- Describe differences in the mean volume of alcohol consumption over the life course according to T2DM diagnosis.
- 3. Formally explore the utility of more advanced survival models in developing a better understanding of the relationship between alcohol consumption and T2DM.

Findings from each resulting analysis are summarised in the following section.

10.2 Summary of findings

10.2.1 Misclassification error

Although analyses of the relationship between alcohol consumption and T2DM have conventionally defined intake according to a single measure as reported at baseline, longitudinal data from multiple cohort studies have shown that alcohol consumption varies across the life course.²⁵⁹ As a consequence of such changes, there was a risk that categories of alcohol

consumption defined in the existing literature may have been subject to misclassification error, complicating interpretations of associated risk estimates. To provide an indication as to how any misclassification error might be distributed across baseline categories, mixed effects models were constructed to plot the mean longitudinal trajectory of alcohol consumption within a series of distinct baseline categories.

When the sex-specific trajectory was stratified in such a manner, marked differences in rates of change were apparent over the adult life course captured by Whitehall II (Figure 8.5). While consumption among moderate baseline drinkers remained little changed over time, male and female participants in the highest categories exhibited a marked decline in drinking with advancing age, with gradual increases in consumption among infrequent drinkers. Accordingly, depending upon the point in the life course at which baseline categories are defined, moderate drinkers may be subject to differential degrees of misclassification error through contamination with heavy and infrequent drinkers. Given the longitudinal convergence of baseline consumption categories toward moderate volumes, misclassification error may be most pronounced in cohorts of older ages. This may go some way toward explaining the less pronounced reductions in the risk of CHD⁴⁰⁹ and all-cause mortality³⁹⁵ reported elsewhere among moderate drinkers in older age groups.

Further stratifying longitudinal trajectories according to the diagnosis of T2DM revealed that the risk of misclassification error appeared greatest among men and women who developed the condition, with greater rates of change as a function of age than those who were censored (Figures 8.10 and 8.11). In particular, consumption among male moderate baseline drinkers increased to such a degree as to be markedly higher than originally classified come the time of diagnosis. As a consequence, conventional survival analyses may risk overestimating the hazardous effects of moderate drinking among men by failing to consider increases to their consumption over time. By contrast, while moderate female drinkers who developed T2DM instead showed little change in consumption over time, they instead risked being increasingly contaminated by infrequent and former heavy drinkers depending upon the point of baseline measurement. As such, studies of women with a shorter time to event may underestimate any advantageous effects of moderate drinking by giving no consideration to the disparate drinking histories of cross-sectionally defined moderate drinkers.

Longitudinal changes to the volume of alcohol consumption as reported in Chapter 8 help draw attention to the risk of misclassification error inherent to conventional survival analyses according to the age of participants at baseline and the time to diagnosis. In so doing, Chapter

8 highlights the possibility that studies may be improved by analyses that give consideration to changes in alcohol consumption across the life course.

10.2.2 Differences in alcohol consumption according to the diagnosis of T2DM

Based on the current literature, a number of hypotheses were proposed concerning how the mean trajectory of alcohol consumption might differ by T2DM status. These included the possibility that the increased risk of T2DM associated with higher volumes of consumption may accumulate over time and therefore be indicated by prolonged or permanently elevated intakes among participants who developed the condition – particularly among men, for whom reductions in risk were not apparent at any level according to the revised and updated meta-analysis reported in Chapter 3. Alternatively, it was possible that increases or decreases in the risk of T2DM might instead be specific to a period of the life course during which sensitivity to the effects of alcohol may be elevated.

While male intake was equivalent around 30 years prior to diagnosis, differences by diagnosis category gradually increased over time. At the point of diagnosis or censoring, consumption among men that developed T2DM was higher than among those who were censored, with a difference equal to around 1.8 pints of 4.0% ABV lager per week on average. With an average age at the time of diagnosis of 62 years, male consumption trajectories were most in keeping with the second hypothesis: that male sensitivity to the deleterious effects of alcohol consumption may be most pronounced later in the life course.

By contrast, neither hypothesis was supported by results from models applied to women. Rather than consumption among those who developed the condition being consistently higher than among censored female participants, or highest specifically during the few years preceding diagnosis, mean consumption was instead consistently below that of women who did not develop T2DM. Such a finding was in conflict with results from the revised meta-analysis in Chapter 3, particularly with mean consumption among both groups of women being permanently within the threshold at which reductions in risk were reported (Chapter 3).

At least two reasons for this incongruity were possible. First, the plotting of mean consumption trajectories may have masked important differences in the constitution of each female trajectory. For instance, the mean trajectory for women who developed T2DM may have been comprised not primarily of persistent low volume and therefore lower risk drinkers, but of both infrequent and former heavy drinkers who may have attenuated their drinking due to poor health prior to study participation. Loosely supporting this, sensitivity analyses that plotted the

same trajectories but excluded zero consumption data indicated that transitions to non-drinking prior to diagnosis or censoring appeared most common among women who developed T2DM, attenuating the difference between the two trajectories (Appendix 8.25). It was therefore plausible that women at risk of T2DM tended to drink alcohol in lower volumes or were more likely to transition to abstention due to adverse health effects from pre-diabetes or its associated risk factors. In this sense, though their volume of drinking was within the range associated with a lower risk of T2DM, this reduction in risk may have been offset by risk factors unaccounted for in the crude analyses. As an alternative explanation, it was also plausible that the absence of trajectories consistent with *a priori* hypotheses may have been attributable to a relatively limited range of alcohol consumption among women, with heavier drinkers being few in number. As such, higher-risk heavier drinkers may have been hidden by a vastly greater proportion of more moderate female drinkers who developed the condition.

10.2.3 Importance of different dimensions of the longitudinal trajectory

10.2.3.1 Intercept versus current consumption

10.2.3.1.1 Primary findings

Despite marked changes to the volume of drinking across the life course, and apparent differences in longitudinal alcohol consumption among men by diagnosis status, survival analyses have conventionally parameterised the dose-response relationship according to just a single cross-sectional measurement obtained at baseline. A final series of analyses was thus undertaken to formally explore the appropriateness of limiting the parameterisation of alcohol consumption in such a fashion.

To achieve this, shared random effects models were constructed that calculated the doseresponse relationship according to whether drinking was defined at the intercept (40 years) or the time of diagnosis (mean 62 years of age), with a further analysis undertaken to model both parameters concurrently so as to establish the relationship of each parameter with T2DM after accounting for future or prior drinking. By using age as the timescale, such analyses also permitted some inference to be made as to differences in the dose-response relationship according to the age at which alcohol was consumed. Although the use of age as a timescale is currently rare within the alcohol literature, with only one study selected as part of the revised and updated meta-analysis having adopted such an approach,²⁴⁵ sensitivity analyses that used the more conventional time-to-event timescale corroborated findings from the primary analyses (Appendices 9.5 and 9.6).

In models that parameterised the longitudinal trajectory according to consumption at the intercept, results were in keeping with those from the revised and updated meta-analyses reported in Chapter 3, whereby reductions in risk were specific to female drinkers (Table 9.5). When the relationship between the weekly volume of alcohol consumption and T2DM risk was instead parameterised according to the current value of alcohol consumption, reductions in risk remained among women but were attenuated in magnitude (Table 9.5). However, when the dose-response relationship was calculated between the intercept and T2DM risk following adjustment for consumption later in the life course, significant reductions in risk were apparent among both sexes, with the magnitude of reduction still greatest among women, at 36% per two-fold increase in consumption versus 20% among men (Table 9.6). By contrast, when the current value of consumption was modelled after holding constant the volume of drinking at the intercept, increased risks of T2DM were evident among both sexes. Such a finding confirms that both parameterisations exhibit statistically independent and divergent relationships with T2DM risk, and thereby suggests that conventional survival analyses fail to capture differences in dose-response across the life course due to their use of a single baseline measure.

That risks associated with higher volumes of consumption appeared more pronounced with advancing age was an effect in accordance with findings from Chapter 8 and consistent with studies that report an age-related deterioration to the alcohol metabolism.^{196,197,198,199} While such deterioration may to individuals of a more advanced age experiencing longer periods of exposure to higher concentrations of pro-inflammatory metabolites such as acetaldehyde and acetate,^{200,201} insufficient evidence is currently available to substantiate a dose-response effect of alcohol consumption upon inflammatory biomarkers, or how any such effect might operate differentially across the life course.

10.2.3.1.2 Possible explanations for sex-specific dose-response effects

Differential biological pathways

Analyses undertaken in Chapters 3, 7 and 9 all report reductions in the risk of T2DM as being either specific to or most pronounced among women when defined at baseline, even following adjustment for consumption at the time of diagnosis. The reproducibility of this difference lends credence to the possibility that putative biological pathways may operate differently between men and women. However, of the studies available, little evidence of a sex interaction is currently apparent. While there is some suggestion from interventional studies that acute alcohol exposure may elicit a greater increase in insulin sensitivity among women than men, such a disparity was no longer statistically significant following the exclusion of a study that represented a primary source of between-study heterogeneity.¹¹² Although studies investigating the effect of alcohol upon HDL concentration are far greater in number, no sex-specific effects have been identified across interventional¹¹⁸ or Mendelian randomisation studies.⁸⁵ Similarly, there is no apparent difference in CRP concentration among men and women sampled across 42 Mendelian randomisation studies,⁸⁵ while a meta-analysis of the few interventional studies investigating alcohol and inflammatory markers reported no sex interaction.¹¹⁸ Though possible that putative biological mechanisms may indeed operate differentially between men and women, robust supporting evidence is lacking.

Differential drink preference

Alternative explanations considered included sex-specific differences in drink preference. While current evidence tends to indicate greater reductions in risk among study participants who consume wine,^{238,246,398} no such effect appeared evident within the Whitehall II cohort, with no difference in dose-response found by drink type within preliminary conventional analyses (Table 7.6). Although the magnitude of reductions in risk appeared greatest for beer consumption, current evidence concerning pathways between beer-specific compounds any anti-inflammatory response was only available from a small number of *in vitro*³⁹⁹ and *in vivo* animal studies.⁴⁰⁰ Given its potency, it seems more likely that the inflammatory effects of ethanol may far outweigh any anti-inflammatory effects of drink-specific compounds.²⁸²

Differential consumption patterns

An alternative hypothesis concerned the possibility that female drinkers within the cohort were more likely to spread their weekly volume of alcohol consumption over a greater number of days, with a larger proportion of men being episodic heavy drinkers. Such an assertion was mooted following research indicating that reductions in the risk of T2DM and other vascular events may be lowest among regular moderate drinkers,^{95,97,240,244} with elevated risks most apparent among binge and episodic heavy drinkers,^{96 244} even if binge drinkers otherwise consume alcohol at moderate volumes the majority of the time.²⁶⁸ When continuous weekly alcohol consumption data were modelled in Chapter 7 with an interaction according whether participants consumed alcohol on a daily or non-daily basis, interaction terms were statistically insignificant for both sexes (Table 7.5). It was likely that the absence of a statistically significant interaction was attributable to Whitehall II volume and frequency data being insufficient to explicitly identify episodic heavy drinking – the specific characteristic most strongly associated with heightened risks relative to regular moderate drinking. However, even were such variables available, it was possible that the number of episodic heavy drinkers may have been insufficient

to detect an interaction, with both excessive consumption and binge drinking most prevalent within lower socio-economic groups^{396,397} and the Whitehall II being skewed toward higher levels of socio-economic status.

Interaction with sex hormones

A final alternative hypothesis pertains to the possibility that sex hormones may play a role in modifying the effect of alcohol upon T2DM risk. To date, two hormonal biomarkers have been implicated as modifiers of T2DM risk: estradiol, a female sex hormone, and SHBG, a protein involved in the transport of sex hormones. Although results pooled from multiple case-control studies have indicated no difference between men and women in the concentration of estradiol by diagnosis status, women who develop T2DM showed lower concentrations of SHBG while men displayed no difference in concentration by diagnosis status.²⁷³ Unfortunately, in addition a lack of any clear pathway by which factors such as SHBG may modify T2DM risk,²⁷³ research concerning a sex-specific effect of alcohol consumption upon hormone transport activity is lacking and almost entirely cross-sectional.

Summary

Current research is insufficient to draw any firm conclusions as to why female drinkers appear to experience the greatest reduction in the risk of T2DM. In the absence of further and more robust analyses concerning sex differences in the effects of alcohol consumption upon putative biological pathways, disparities in consumption pattern are perhaps the most likely determinant, with data from UK-based studies indicating a greater prevalence¹⁰¹ and probability²⁶⁹ of episodic heavy drinking among men.

Whatever the precise reason for differences in dose-response between men and women, it is clear from more advanced survival analyses that the association between alcohol consumption and T2DM risk is likely to be more complex than captured from conventional survival analyses, which fail to consider differences in dose-response across the adult life course.

10.2.3.1.3 Effect of abstention category upon dose-response

Criticism has been levelled at the use of pooled non-drinkers as a reference group, with reductions in risk among current drinkers purported to be artificially increased given its inclusion of former drinkers predisposed to an increased risk of negative health events.^{136,137} In response, the revised and updated meta-analysis reported in Chapter 3 included sex-adjusted dose-response curves stratified according to whether selected studies utilised a pooled non-drinking or strictly-defined never drinking abstention category. The dose-response relationship differed

significantly according to the choice of abstention reference category, with reductions in risk most pronounced among studies that calculated risks relative to pooled non-drinkers (Figure 3.5). Such a finding is in keeping with studies of alcohol consumption and all-cause mortality, which report attenuated reductions in risk among current drinkers when former drinkers are excluded from the abstention reference category.^{136,147,148}

In Chapter 7 (Appendices 7.7 and 7.8) and Chapter 9 (Tables 9.5 and 9.7), sensitivity analyses were undertaken which calculated the dose-response relationship with and without adjustment for or the inclusion of heterogeneous infrequent and non-drinkers. Results from these analyses are in line with the meta-analysis of conventional survival studies, confirming the importance of disaggregating heterogeneous non-drinkers and accounting for any effect they may have upon the perceived harms or benefits of current drinking. In both chapters, effects were most pronounced among women. This nuance may have been attributable in part to the multivariable-adjusted risks among male infrequent and non-drinkers drinkers being closer to the null than among women (e.g. Table 9.5).

10.2.4 Sick quitter effects

Participants who reduce their intake over course of a study have for a long time been referred to as 'sick quitters' who attenuate their consumption owing to ill-health. ^{136,138} Such an assertion is supported by studies that report a higher prevalence of T2DM risk factors^{139,140} and poor self-reported health^{141,142,143} among former drinkers than current drinkers, with the onset of ill-health associated with a subsequent cessation from alcohol consumption.^{144,145,146} However, following a piecewise analysis of consumption trajectories prior to and following diagnosis (Figure 8.9), downward trajectories were only apparent after the date of diagnosis, suggesting that the effect of pre-diabetic symptoms may have been insufficient to elicit a change to drinking behaviour. Furthermore, when trajectories were stratified by baseline consumption, downward slopes were specific to heavier drinkers and evident regardless of sex or diagnosis status (Figures 8.10 and 8.11). Such results indicate that any downward trajectories prior to diagnosis may not have been a consequence of declines to health status of a kind associated with a heightened T2DM risk.

To explore this formally, shared random effects models were constructed that reported the effect of differences in the longitudinal rate of change upon the risk of T2DM after adjustment for past and current volume of consumption. As reported in Table 9.8, men and women who reduced their alcohol consumption at a faster rate than the sample average experienced a lower risk of T2DM, though risk estimates were non-significant. Such a finding appeared contrary to

the sick quitter hypothesis. However, it was possible that the presence of a sick quitter effect was lost as a consequence of the diverse reasons for which study participants attenuate their alcohol consumption, including financial constraints and a decline in opportunities for social drinking⁴⁰⁸ – factors not necessarily associated with T2DM risk. Thus, while some adults who reduce their consumption may indeed do so as a result of a deteriorating health status, a downward trajectory is not necessarily indicative of a heightened risk of T2DM but may instead be part of normal ageing processes.

10.3 Limitations

Despite a systematic attempt to describe and mitigate the impact of methodological shortcomings within the existing literature, a number of weaknesses remain that may potentially have biased or undermined the validity of reported associations.

Perhaps the primary limitation concerns the variable used to denote the average weekly volume of alcohol consumed. Rather than strictly representing each participant's average weekly intake, the variable was derived from questions that referred to the volume consumed during the week prior to each interview. It was possible that the volume of alcohol intake as documented during the week prior to an interview provided a poor surrogate for true average weekly consumption. For instance, heavy drinking during the week prior to interview may not have been indicative of a participant's average weekly consumption throughout a given year, but instead an uncharacteristic period marked by a special occasion such as a family member's birthday or wedding. Such a situation would thus have reflected an episodic heavy drinking occasion as opposed to a high average weekly volume of alcohol consumption. Unfortunately, episodic heavy drinking occasions, as associated elsewhere with a heightened risk of T2DM, ^{96,240,245} could not be captured using the volume and frequency variables available within the Whitehall II cohort. It was perhaps because of this that interactions between these two drinking dimensions were not statistically significant, providing a poor parameterisation of the type of consumption pattern most associated with elevated risks of T2DM. The true effect of unobserved episodic heavy consumption upon the relationship between the weekly volume of alcohol consumption and T2DM risk was thus unknown and dependent upon the distribution of such episodes across the volume distribution.

More useful alcohol consumption data may have been obtained from a graduated frequency questionnaire, which ask participants about the frequency with which they typically consume alcohol according to a series of pre-defined volumes. Such methods have been found to yield higher average volumes of estimated alcohol consumption than quantity-frequency questionnaires,³⁹¹ and would help with the identification of episodic heavy consumption. The identification of such episodes is likely to be critical to providing a more accurate assessment of the benefits and risks associated with alcohol consumption.

Although associations identified within the Whitehall II cohort appear consistent with those reported from studies of general population samples, ²⁹⁶ it should nonetheless be noted that the relatively low volume of alcohol consumption was such that power to detect increases in the risk of T2DM at higher levels of the volume distribution was likely to have been impaired. Thus, although the presence of a non-linear dose-response relationship was tested in Chapter 7 as part of a conventional survival analysis, the null result was unlikely to have been a consequence of a true linear reduction in risk at ever-increasing volumes of alcohol consumption but rather an insufficient number of heavier drinkers with which to detect a turning point in the dose-response relationship. For instance, while point estimates from the meta-analysis in Chapter 3 indicate that the risk of T2DM becomes elevated among women at volumes >140 g/week, relative to pooled non-drinkers, just 152 women within the Whitehall II cohort reported consumption anywhere above that level at wave three, with the figure falling to just 105 by wave 11.

Another notable limitation concerned the application of imputed data. Firstly, regarding imputed values of alcohol consumption, it was not possible to predict never drinking among participants with missing alcohol consumption data due to the small number of strictly defined never drinkers within the Whitehall II cohort. As such, analyses inclusive of a categorical alcohol consumption variable were restricted to participants with observed alcohol consumption data only. Given that the median volume of alcohol consumption was lower and the prevalence of metabolic risk factors greater at the commencement of the study among participants with unit (Table 6.5) or item non-response (Table 6.7) than those with complete-case data, restriction of the dataset may have resulted in an underrepresentation of higher-risk light or non-drinkers in survival analyses undertaken for Chapters 7 and 9.

Aside from limitations concerning the use of imputed categories of alcohol consumption, constraints of the statistical software are such that it was not possible to utilise imputed T2DM and time to event data. As noted above, participants with unit non-response tended to have a worse metabolic risk profile at baseline than those with complete-case data. It was therefore likely that the incidence of T2DM among the original cohort was underestimated, with analyses applied to a healthier sub-sample than originally participated. The precise impact of such attrition upon reported dose-response relationships would be dependent upon how unobserved

271

cases were dispersed across the alcohol consumption distribution. For instance, were attrition more common among heavier drinkers, then any increased risk associated with higher volumes of alcohol intake would have been underestimated. Perhaps as a consequence of the limits concerning the imputed data that could be applied, differences in the primary dose-response relationship between models applied to the observed and imputed datasets were negligible.

10.4 Strengths

A deliberate attempt was made to systematically overcome the limitations inherent to the existing body of epidemiological research in the field, with a view to providing a more accurate and detailed understanding of the dose-response relationship between alcohol consumption and the risk of T2DM.

Firstly, contrary to the vast majority of current studies, participants with zero consumption were disaggregated into heterogeneous infrequent and non-drinking categories, each posited and later confirmed to exhibit disparate and usually elevated risks of T2DM, relative to current drinkers (Table 9.6). Rather than combining these heterogeneous groups into a single reference category, as per convention, risk estimates were instead adjusted for differences in risk between heterogeneous non-drinkers. Such an approach is expected to have helped reduce the extent to which any reductions in risk reported among current drinkers occurred as an artefactual consequence of having compared drinkers against a number of less healthy non-drinkers (Table 7.2). Such an approach is supported by research elsewhere, ^{136,137,148,437} as well as by results from the meta-analysis in Chapter 3, which showed a greater reduction in risk at moderate volumes of consumption when former drinkers were included within the abstention reference category.

In addition, contrary to the minority of current studies to have also distinguished between different non-drinking groups,^{96,219,230,244} never drinkers were not defined according to a single self-report of life-long abstention. With longitudinal research elsewhere having reported that between 52%,¹⁵⁰ 67%³¹⁵ and 87%⁴³⁸ of study participants cross-sectionally defined as never drinkers had previously reported drinking at an earlier wave, never drinkers were instead coded as those who consistently reported being never drinkers and who declared zero consumption at all previous waves, thereby reducing the misclassification of never drinkers.

This project is the first to report differences in the longitudinal trajectory by T2DM diagnosis, highlighting discrepancies in the volume of alcohol consumption and thereby the periods in which increases or decreases in risk may most likely be conferred (Chapter 8). The results reported in Chapter 9 also represent one of the very few instances in which consideration has ever been given to longitudinal changes in alcohol consumption, with just two previous studies known to have documented the risk of T2DM according to temporal variations in consumption.^{153,267} Contrary to the approaches adopted by these two studies, the analyses in reported in Chapter 9 estimated the independent dose-response relationship between different dimensions of the longitudinal trajectory and T2DM, utilising advanced joint modelling techniques that provide more accurate estimates than simpler analytic alternatives.^{411,412}

10.5 Further research

Results reported in Chapter 9 indicate that the risks associated with drinking may be especially pronounced in older age, potentially countering any reductions in risk conferred earlier in the life course. Important now are replication studies designed to establish whether the same dose-response associations are present in different cohorts. Such research should focus upon the analysis of longitudinal datasets with repeated measures of sufficient regularity and covering as broad a duration of the adult life course as to model acute fluctuations in dose-response with advancing age. In selecting studies for testing reproducibility, a number of other important characteristics should be considered given the limitations present within Whitehall II.

Firstly, because of the narrow range of weekly alcohol consumption reported by Whitehall II participants, cohorts should be selected with a greater number of heavier drinkers, providing a better capacity for modelling the non-linear associations indicated in Chapter 3. Secondly, with apparent differences in dose-response according to whether or not individuals participate in binge drinking,^{94,240,245} and an underestimation of average consumption based on quantityfrequency questionnaires,³⁹¹ chosen cohorts should provide more accurate alcohol consumption estimates and sufficient detail for identifying episodic heavy drinking occasions. To achieve this, studies may benefit from the addition of objective transdermal ethanol sensing as a means of monitoring blood alcohol concentrations over extended periods. Such data may prove useful as a means of validating the accuracy of survey-based self-reports, adjusting subjectively reported consumption data as required.⁴³⁹ Thirdly, to reliably test interactions between the volume of alcohol consumption and factors such as episodic heavy drinking, cohorts should be of a size sufficient to permit tests of sound statistical power. One future possibility is the UK Biobank, which sampled close to 500,000 adults aged 40-69 years by 2010 and is estimated to have statistical power sufficient to detect an odds ratio \geq 1.3 after just six years of follow-up for T2DM.440

Although reductions in risk appear either specific to or most prominent among female drinkers, research concerning sex-specific differences in the action of putative biological pathways is weak

and inconclusive. Heterogeneous in design and often conflicting in their conclusions, findings are commonly reported by cross-sectional or small-scale interventional studies. Further research is required to better understand the dose-response effects of alcohol consumption upon hypothesised causal intermediates, including insulin sensitivity, HDL concentration and inflammatory response. Such research may benefit from structural equation models designed to quantify both the effect of alcohol upon intermediate factors and the effect of changes to those intermediate factors upon T2DM risk. At present, no analysis has been published that considers the causal pathway in full.

10.6 Policy implications

Since at least 2004,⁴⁴¹ reviews of the evidence base have reported reductions in the risk of T2DM at moderate volumes of regular consumption, with similar findings reported elsewhere for vascular conditions ranging from ischaemic heart disease and stroke.⁴⁴² Although recent metaanalyses^{10,84} suggest that that peak reductions in the risk of T2DM may be conferred at volumes close to the current recommended limits of 21 units/week among men and 14 units/week among women,⁹⁸ T2DM prevention strategies in the UK give no consideration to the role of alcohol, with public health interventions focussing instead upon the impact of other lifestyle factors such as diet and physical activity.⁹

This omission sits contrary to recommendations by some academics that at-risk patients be encouraged to incorporate moderate drinking into their diet,^{443,444} and the general public advised that moderate drinking may afford an overall health benefit,⁴⁴³ even among abstainers.⁴⁴⁵ That such recommendations have yet to be adopted by the UK government appears attributable to a lack of consensus amongst epidemiologists regarding the health benefits of drinking.¹⁷⁰ This lack of consensus stems from series of methodological shortcomings, including poor confounder adjustment,^{139,446} a lack of established biological mechanisms⁴⁴⁶ and the misclassification of drinking categories.^{446,136,148}

With the benefits of moderate alcohol consumption reported by existing studies likely to have been overestimated, and insufficient evidence available to substantiate a clear biological mechanism by which alcohol may reduce the risk of T2DM, the appropriateness of using the current literature as grounds for recommending a specific volume of moderate drinking either at the population level or among at-risk sub-groups is questionable. This is especially so when considering the net impact of alcohol consumption, with drinking even at low volumes having been associated with increases in the risk of various cancers.⁴³² Indeed there are calls from some

quarters that alcohol guidelines should explicitly discourage the drinking of alcohol for perceived health benefits.⁴⁴⁷

Health policy may need to be more nuanced than these two starkly opposed positions, instead advising a general attenuation to the volume of alcohol consumed. Such a position affords two benefits. First, regardless of whether alcohol consumption exhibits J-shaped or linear associations with assorted non-communicable diseases, the net burden of harms associated with alcohol consumption would be lowered through a general reduction in the volume of consumption. Second, by avoiding any suggestion that moderate alcohol consumption may provide a benefit to health, the danger of inadvertently encouraging abstainers to resume consumption would be mitigated. This second point is important given that adults who abstain appear to do so owing to previous alcohol abuse problems, existing ill-health, pregnancy or contraindication with medication.^{145,146,408}

Evidence from analyses reported in Chapter 9 suggest that initiatives or policies designed to reduce alcohol consumption may need to pay particular attention to older drinkers. Such a focus is supported by the Royal College of Psychiatrists, which argues that alcoholism and alcoholrelated harms within older populations are an under-recognised problem within the UK, with public health policy giving scant consideration to the risks facing drinkers in older age.⁴⁴⁸ In response to a perceived oversight by policymakers, the Royal College of Psychiatrists has advised the introduction of age-specific consumption guidelines as a means of communicating at the population level the heightened risks to health experienced by older drinkers, recommending no more than 10.5 units/week for persons aged ≥ 65 .⁴⁴⁸ However, although the implementation of an age-specific drinking guidelines for older populations may appear appropriate, such an approach should be balanced against the risk of creating a more complex and less unified health message.⁴⁴⁹ With the body of current yet methodologically limited evidence indicating that any cardiovascular benefit from drinking may be limited solely to women aged >55 years who consume volumes below just 5 units/week, and with the risk of non-vascular conditions apparently elevated among both sexes even at low volumes,⁹⁹ health messages for older adults should be no different from those for the general population: that current drinkers should aim to reduce the amount they drink, and that non-drinkers should be encouraged to maintain their abstinence.

One notable recommendation from the 2016 Alcohol Guidelines Review was that the alcohol consumption guideline for men should be reduced from 21 units/week to 14 units/week.⁹⁹ Such a decision was based on new evidence concerning an increased risk of cancer at low volumes of

consumption and the burden of acute alcohol-related harms that appears to disproportionately befall men, such as drink-related road traffic collisions. Nevertheless, such a recommendation is circumstantially in keeping with results reported throughout this project, which indicate that the risk of T2DM may also be greatest among male drinkers, regardless of the age at which alcohol is consumed. To date, precise reasons underlying this difference are unclear. Of the various possibilities considered, perhaps the most likely factor concerns disparities in drinking pattern by sex, with episodic heavy drinking associated with heightened risks for both T2DM^{95,96,97,239,240,244,245} and ischaemic heart disease,²⁶⁸ and such drinking patterns being most prevalent among men.^{101,269} It is therefore reassuring that the 2016 Alcohol Guidelines Review explicitly advises against heavy drinking occasions, with the caveat that adults who continue to binge drink should at the very least aim to reduce the frequency of such behaviour.⁹⁹

While some academics call for the promotion of moderate alcohol consumption among both sexes, observational studies indicative of protective effects at moderate volumes of intake are methodologically weak, with clear causal mechanisms yet to be clearly established. Pending more reliable information concerning the effect of alcohol upon T2DM and a broad range of health conditions, it is inappropriate to recommend that adults take up moderate drinking for a perceived benefit to health, especially given that individuals who abstain from drinking may do so for reasons including a history of addiction, chronic ill-health, pregnancy or pharmaceutical contraindication. Instead, current drinkers should be advised to attenuate both their average weekly consumption and frequency of episodic heavy drinking as a precaution against risks associated with heavy drinking.

Chapter 11: Appendices

Chapter 11

Appendices

Appendix 3.2 The Newcastle-Ottawa quality assessment checklist

NEWCASTLE-OTTAWA QUALITY ASSESSMENT SCALE FOR COHORT STUDIES

<u>Note</u>: A study can be awarded a maximum of one star for each numbered item within the Selection and Outcome categories. A maximum of two stars can be given for Comparability.

Selection

1) Representativeness of the exposed cohort

- a) truly representative of the average current drinker in the community *
- b) somewhat representative of the average current drinker in the community *
- c) selected group of users, e.g. nurses, volunteers
- d) no description of the derivation of the cohort
- 2) Selection of the non-exposed cohort

a) drawn from the same community as the exposed cohort *

- b) drawn from a different source
- c) no description of the derivation of the non-exposed cohort

3) Ascertainment of exposure

- a) secure record, e.g. surgical records *
- b) structured interview *
- c) written self-report, e.g. postal questionnaire
- d) no description
- 4) Demonstration that outcome of interest was not present at start of study
 - a) yes *
 - b) no

Comparability

1) Comparability of cohorts on the basis of the design or analysis

- a) study controls for a measure of adiposity *
- b) study controls for any additional factor *

Outcome

1) Assessment of outcome

- a) independent blind assessment or objective ascertainment *
- b) record linkage *
- c) self report
- d) no description

2) Was follow-up long enough for outcomes to occur

a) yes, at least six years duration *

Chapter 11: Appendices

b) no

3) Adequacy of follow up of cohorts

- a) complete follow up: all subjects accounted for *
- b) subjects lost to follow up unlikely to introduce bias: >5% lost, or description of those lost
- c) follow up rate <95% and no description of those lost
- d) no statement

11.2 Appendices for Chapter 6

Appendix 6.1 Effect of dietary factors upon the alcohol-T2DM relationship

Men	Sample size [®]	Model 1	Model 2		Model 3	Model 4	Model 5	Model 6
Variables of interest	Participants (n) Incident cases (n)	HR (95% CI) p-valu	e HR (95% CI) p-	value	HR (95% CI) p-value	HR (95% CI) p-value	HR (95% CI) p-value	HR (95% CI) p-value
Alcohol consumption at S3								
Never drinkers ^b	90 16	1.00	1.00		1.00	1.00	1.00	1.00
Non-current drinkers ^e	587 63	1.08 (0.61-1.92) 0.78	1 1.11 (0.63-1.97)	0.721	1.10 (0.62-1.95) 0.750	1.11 (0.62-1.97) 0.729	1.08 (0.61-1.92) 0.793	1.10 (0.62-1.95) 0.752
0.1-50.0 g/week	1,429 141	0.96 (0.56-1.66) 0.88	9 0.99 (0.57-1.70)	0.958 (0.97 (0.56-1.68) 0.920	0.98 (0.57-1.70) 0.953	0.96 (0.56-1.65) 0.878	0.98 (0.56-1.69) 0.934
50.1-100.0 g/week	973 103	1.07 (0.61-1.88) 0.80	2 1.10 (0.63-1.92)	0.744	1.08 (0.62-1.89) 0.784	1.10 (0.63-1.93) 0.744	1.07 (0.61-1.87) 0.813	1.10 (0.62-1.92) 0.751
100.1-150.0 g/week	632 69	1.10 (0.62-1.96) 0.74	6 1.11 (0.62-1.98)	0.723	1.10 (0.62-1.97) 0.742	1.12 (0.63-2.01) 0.696	1.10 (0.61-1.96) 0.75	1.11 (0.62-1.99) 0.721
>150.0 g/week	1,124 129	1.03 (0.59-1.80) 0.92	4 1.02 (0.58-1.79)	0.941	1.02 (0.59-1.79) 0.936	1.04 (0.60-1.82) 0.883	1.02 (0.59-1.79) 0.934	1.03 (0.58-1.80) 0.931
BMI								
kg/m ²	4,825 521	1.21 (1.18-1.24) <0.00	1 1.20 (1.17-1.23) <	0.001	1.20 (1.18-1.23) <0.001	1.21 (1.18-1.23) <0.001	1.21 (1.18-1.24) <0.003	1.20 (1.18-1.23) <0.001
Fibre								
g/100g	4825 521		- 0.99 (0.98-1.00)	0.020			I	1.00
LR test versus BMI			,	9.018				0.98 (0.97-1.00) 0.020
Carbohydrate								
log g/100g	4825 521			'	0.88 (0.68-1.15) 0.359			· 1.27 (0.79-2.05) 0.330
LR test versus BMI					0.360			
Polyunsaturated fat								
log g/100g	4825 521	I	1	ī		0.92 (0.76-1.12) 0.405	1	0.99 (0.78-1.27) 0.943
LR test versus BMI						0.405		
Trans fat								
log g/100g	4825 521			ī			1.03 (0.77-1.37) 0.851	1.02 (0.72-1.43) 0.919
LR test versus BMI							0.852	
LR test BMI versus BMI+FFC	~							0.131
^a Defined according to thos	e with valid covariate data in Mode	i 6.						
^b Defined as those who rep	orted consuming no alcohol in the l	ast twelve months and hav	ing always been a non-d	lrinker.				
^c Defined as those who repo	orted consuming no alcohol in the l	ast twelve months and stat	ed that they had not alw	ays been	a non-drinker.			
Model 1: Adjusted for age ((s3), BMI (S3), employment status (S	3), ethnicity, family history	of T2DM (S3), smoking:	status (S3), physical activity (S3), o	occupational grade (S3).		
Model 2: As Model 1, plus	fibre.							
Model 3: As Model 1, plus	carbohydrate.							
Model 4: As Model 1, plus	polyunsaturated fat.							
Model 5: As Model 1, plus	trans fat.							

Model 6: As Model 1, plus fibre, carbohydrate, polyunsaturated fat and trans fat.

				ĺ										
Women	Sample size [®]		Model 1		Model 2		Model 3		Model 4		Model 5		Model 6	
Variables of interest	Participants (n) Incident cases	s (n)	HR (95% CI) p-1	value	HR (95% CI) p-val	ne	HR (95% CI) p-v	alue	HR (95% CI) p	-value	HR (95% CI) p	-value	HR (95% CI)	p-value
Alcohol consumption at S								 						
Never drinkers ^b	104	17	1.00		1.00		1.00		1.00		1.00		1.00	0.467
Non-current drinkers ^c	508	94	1.76 (1.01-3.07) (0.044	1.77 (1.02-3.08) 0.0	44 1	80 (1.04-3.14) 0	037	1.76 (1.01-3.07)	0.044	1.77 (1.02-3.08)	0.043	1.84 (1.06-3.21)	0.031
0.1-50.0g/week	879	100	1.30 (0.73-2.30) 0	0.372	1.30 (0.73-2.30) 0.3	74 1	31 (0.74-2.32) 0	362	1.30 (0.73-2.30)	0.373	1.30 (0.73-2.31)	0.369	1.31 (0.74-2.33)	0.352
50.1-100.0g/week	317	22	1.02 (0.51-2.05) 0	.957	1.02 (0.51-2.05) 0.9	53 1	.03 (0.51-2.08) 0	930	1.02 (0.51-2.05)	0.963	1.03 (0.51-2.07)	0.945	1.02 (0.51-2.05)	0.956
>100.0g/week	277	14	0.73 (0.33-1.58) (0.422	0.72 (0.33-1.58) 0.4	19 0	.72 (0.33-1.57) 0	410	0.73 (0.33-1.58)	0.420	0.73 (0.33-1.59)	0.425	0.71 (0.33-1.55)	0.389
BMI														
kg/m ²	1,838	247	1.13 (1.11-1.16) <0	0.001		1	13 (1.11-1.16) <0	001	1.13 (1.11-1.16)	0.001	1.13 (1.11-1.16)	100.0	1.13 (1.11-1.16)	<0.001
Fibre														
g/100g	1838	247	I	i.	7.0 (10.1-66.0) 00.1	86	I	i.	I	i.	I	1	1.01 (0.99-1.02)	0.456
LR test versus BMI					0.7	35								
Carbohydrate														
log g/100g	1838	247	ı	i.	ı	•	.81 (0.58-1.15) 0	238	I	1	I	1	0.60 (0.34-1.06)	0.078
LR test versus BMI							0	241						
Polyunsaturated fat														
log g/100g	1838	247	'	'	'		'	•	1.03 (0.79-1.34)	0.836	'	ľ	1.20 (0.86-1.69)	0.285
LR test versus BMI										0.836				
Trans fat														
log g/100g	1838	247	I	ł	ı		ı	i.	I	1	0.94 (0.59-1.48)	0.783	1.10 (0.65-1.85)	0.723
LR test versus BMI												0.783		
LR test BMI versus BMI+FF	a													0.503
^a Defined according to tho:	se with valid covariate data in N	Model												
^b Defined as those who rep	oorted consuming no alcohol in	the las	st twelve months and	having	always been a non-drin	ker.								
^c Defined as those who rep	oorted consuming no alcohol in t	the las	t twelve months and :	stated t	hat they had not always	s been a	non-drinker.							
Model 1: Adjusted for age	: (s3), BMI (S3), employment stat	tus (S3)	l, ethnicity, family his	tory of	T2DM (S3), smoking sta	tus (S3)	, physical activity (S3), occ	upational grade (S	3).				
Model 2: As Model 1, plus	s fibre.													

290

Model 4: As Model 1, plus polyunsaturated fat. Model 5: As Model 1, plus trans fat. Model 6: As Model 1, plus fibre, carbohydrate, polyunsaturated fat and trans fat.

Model 3: As Model 1, plus carbohydrate.



Appendix 6.2 Trace plots illustrating convergence on key variables










11.3 Appendices for Chapter 7

Appendix 7.1 Crude Nelson-Aalen cumulative hazard estimate, stratified by sex. Observed data.



1	Men		Womer	E
1	T2DM	Censored	T2DM	Censored
Variables (wave 3)	% (95% CI)	% (95% CI)	% (95% CI)	% (95% CI)
	L	c	ч	ч
Age				
Mean years	50.4 (49.9-50.9) ^a	49.8 (49.6-49.9) ^a	52.1 (51.3-52.8) ^a	50.7 (50.5-51.0) ^a
	620	5.103	296	2.274
Alcohol consumption frequency				
None in past year	4.9 (3.2-6.6)	3.1 (2.6-3.6)	9.8 (6.4-13.3)	6.5 (5.5-7.5)
	30	158	29	148
<1/week	19.2 (16.1-22.3)	16.3 (15.3-17.3)	50.4 (44.6-56.2)	32.3 (30.3-34.3)
	119	832	149	735
1-3 times/week	34.3 (30.5-38.1)	39.7 (38.4-41.1)	26.6 (21.4-31.7)	36.7 (34.6-38.7)
	213	2,028	29	834
Daily or almost daily	41.6 (37.7-27.3)	24.9 (24.8-25.0)	13.2 (9.2-17.2)	24.5 (22.7-26.3)
	258	2,085	39	558
BMI				
Mean kg/m²	27.0 (26.7-27.3) ^a	24.9 (24.8-25.0) ^a	29.5 (28.9-30.2) ^a	25.3 (25.1-25.4) ^a
	620	5,103	296	2,274
Employment status				
Employed	90.8 (88.5-93.1)	90.9 (90.2-91.7)	88.5 (84.9-92.2)	90.7 (89.5-91.9)
	563	4,641	262	2,062
Retired	6.3 (4.4-8.2)	7.5 (6.8-8.2)	5.4 (2.8-8.0)	5.4 (4.5-6.3)
	39	382	16	123
Other ^b	2.9 (1.6-4.2)	1.6 (1.2-1.9)	6.1 (3.3-8.8)	3.9 (3.1-4.7)
	18	80	18	89

Appendix 7.2 Baseline characteristics of participants free of T2DM at wave three and with valid follow-up data, stratified by sex. Imputed data.

Ethnicity				
White	84.8 (82.0-87.7)	94.6 (94.0-95.2)	71.1 (65.9-76.3)	89.1 (87.8-90.3)
	526	4,826	211	2,025
South Asian	11.8 (9.2-14.3)	3.4 (2.9-3.9)	14.5 (10.5-18.6)	4.6 (3.7-5.4)
	73	175	43	104
Other	3.4 (2.0-4.8)	2.0 (1.6-2.4)	14.3 (10.3-18.4)	6.4 (5.4-7.4)
	21	101	42	145
Family history of T2DM	1		!	
No	81.1 (78.0-84.2)	91.2 (90.4-92.0)	68.9 (63.6-74.3)	89.1 (87.8-90.4)
	503	4,653	204	2,026
Yes	18.9 (15.8-22.0)	8.8 (8.0-9.6)	31.1 (25.7-36.3)	10.9 (9.6-12.2)
	117	451	92	248
Occupational class				
Administrative (top)	42.4 (38.5-46.3)	49.9 (48.5-51.3)	5.4 (2.8-8.0)	17.5 (15.9-19.1)
	263	2,546	16	398
Professional (middle)	48.5 (44.6-52.5)	44.3 (43.0-45.7)	42.2 (36.6-47.9)	45.9 (43.8-47.9)
	301	2,263	125	1,043
Clerical (bottom)	9.0 (6.8-11.3)	5.8 (5.1-6.4)	52.4 (46.6-58.1)	36.6 (34.6-38.6)
	56	294	155	833
Physical activity ^c				
Inactive	17.1 (14.1-20.2)	13.6 (12.5-14.5)	41.6 (35.8-47.3)	34.0 (32.0-36.0)
	106	692	123	772
Below guidelines	35.4 (31.5-39.2)	36.6 (35.3-38.0)	34.3 (28.7-39.9)	33.4 (31.4-35.4)
	220	1,869	101	759
Met guidelines	47.5 (43.4-51.5)	49.8 (48.4-51.2)	24.2 (19.1-29.1)	32.7 (30.7-34.6)
	294	2,542	72	743

Very good/Excellent	40.7 (36.7-44.6)	53.5 (52.1-54.9)	30.4 (25.1-35.7)	43.5 (41.2-45.4)
	252	2,731	06	985
Good	45.1 (41.1-49.1)	38.0 (36.6-39.3)	44.7 (38.9-50.6)	42.8 (40.7-44.9)
	280	1,939	132	973
Fair/Poor	14.2 (11.4-17.0)	8.5 (7.7-9.3)	24.9 (19.8-29.9)	13.9 (12.4-15.4)
	88	433	74	316
Smoking				
Never	40.9 (37.0-44.8)	48.4 (47.0-49.8)	55.8 (50.1-61.5)	53.1 (51.0-55.1)
	254	2,470	165	1,207
Former	40.8 (36.9-44.7)	39.7 (38.4-41.1)	27.7 (22.5-32.9)	29.5 (27.6-31.4)
	253	2,028	82	671
Current	18.3 (15.2-21.3)	11.8 (10.9-12.7)	16.5 (12.1-20.8)	17.4 (15.8-19.0)
	113	605	49	396
Alcohol consumption volume not listed as it	was not possible to in	ipute a categorical volum	ie variable in which neve	er and non-current
drinkers were separated.				

Self-reported general health

moderate-intensity or 275 minutes of vigorous-intensity activity per week); inactive (<60 minutes of moderate and <60 minutes of ^aMean and 95% confidence interval; ^bRedundant, student, long-term sick, long-term carer; ^cMeeting guidelines (2150 minutes of vigorous activity; below guidelines (anyone not inactive or meeting guidelines)





Women static constraints of the state of th

50.1-100.0g/week

>100.0g/week

Appendix 7.4 Crude Nelson-Aalen cumulative hazard estimate, stratified by sex and frequency of alcohol consumption. Observed data.



Appendix 7.5 Baseline characteristics of participants free of T2DM at wave three and with valid follow-up data, stratified by sex and categories of average weekly volume of alcohol consumption. Imputed data.

		Alcohol consumptio	n category	
	Current	Infrequent	Non-current	Never
Virishloc (mano 3)	% (95% CI)	% (95% CI)	% (95% CI)	% (95% CI)
variabies (wave 3)	E	c	c	c
Men				
Age				
Mean years	49.7 (49.5-49.9) ^a	50.2 (49.7-50.7) ^a	50.6 (49.4-51.9) ^a	50.1 (48.8-51.4) ^a
	4,678	601	92	93
BMI				
Mean kg/m ²	25.2 (25.1-25.3) ^a	24.7 (24.4-25.0) ^a	24.9 (24.2-25.6) ^a	24.3 (23.5-24.9) ^a
	4,678	601	92	63
Employment status				
Employed	90.8 (90.0-91.6)	87.9 (85.2-90.5)	91.3 (85.4-97.2)	91.4 (85.6-97.2)
	4,248	528	84	85
Reti red	7.5 (6.8-8.3)	9.8 (7.4-12.2)	6.5 (1.4-11.7)	3.2 (-0.4-6.9)
	353	59	9	£
Other ^b	1.6 (1.3-2.0)	2.3 (1.1-3.5)	2.2 (0.0-5.2)	5.4 (0.7-10.0)
	77	14	2	5
Ethnicity				
White	95.0 (94.4-95.6)	90.3 (88.0-92.7)	53.3 (42.9-63.7)	84.9 (77.5-92.4)
	4,444	543	49	79
South Asian	3.3 (2.7-3.8)	5.3 (3.5-7.1)	38.0 (27.9-48.2)	15.1 (7.6-22.5)
	152	32	35	14
Other	1.8 (1.4-2.1)	4.3 (2.7-6.0)	8.7 (2.8-14.6)	I
	82	26	ø	

Family history of T2DM				
No	90.5 (89.7-91.3)	88.8 (86.3-91.4)	80.1 (71.7-88.5)	89.2 (82.8-95.7)
	4,234	534	74	83
Yes	9.5 (8.7-10.3)	11.2 (8.6-13.7)	19.9(11.5-28.3)	10.8 (4.3-17.2)
	445	67	18	10
Incident T2DM				
No	89.3 (88.4-90.2)	89.9 (87.4-92.3)	81.5 (73.4-89.6)	86.0 (78.8-93.2)
	4,179	540	75	80
Yes	10.7 (9.8-11.6)	10.1 (7.7-12.6)	18.5 (10.4-26.6)	14.0 (6.8-21.2)
	499	61	17	13
Occupational class				
Administrative (top)	52.2 (50.8-53.7)	31.3 (27.6-35.0)	27.2 (17.9-36.4)	35.5 (25.6-45.4)
	2,444	188	25	33
Professional (middle)	43.0 (41.6-44.4)	56.9 (52.9-60.9)	52.2 (41.8-62.6)	49.5 (39.1-59.8)
	2,012	342	48	46
Clerical (bottom)	4.7 (4.1-5.4)	11.8 (9.2-14.4)	20.7 (12.2-29.1)	15.1 (7.6-22.5)
	222	71	19	14
Physical activity ^c				
Inactive	12.5 (11.6-13.5)	22.2 (18.8-25.5)	18.5 (10.4-26.6)	23.3 (14.5-32.1)
	585	133	17	22
Below guidelines	36.6 (35.2-38.0)	36.2 (32.3-40.0)	42.4 (32.1-52.7)	35.7 (25.8-45.7)
	1,712	217	39	33
Met guidelines	50.9 (49.5-52.3)	41.7 (37.7-45.6)	39.1 (29.0-49.3)	41.0 (30.8-51.2)
	2,381	251	36	38
Self-reported general health				
Very good/Excellent	53.1 (51.6-54.5)	46.6 (42.6-50.6)	56.5 (46.2-66.8)	43.0 (32.8-53.3)
	2,483	280	52	40
Good	38.9 (37.5-40.3)	38.5 (34.6-42.4)	31.5 (21.8-41.2)	36.8 (26.7-46.8)
	1,821	231	29	34
Fair/Poor	8.0 (7.2-8.8)	14.9 (12.0-17.7)	12.0 (5.2-18.7)	20.2 (11.9-28.6)
	374	89	11	19

Smoking				
Never	46.8 (45.4-48.2)	52.2 (48.2-56.3)	78.8 (70.2-87.4)	48.3 (37.9-58.7)
	2,190	314	72	45
Former	41.3 (39.8-42.7)	33.6 (29.8-37.4)	12.5 (5.5-19.5)	37.7 (27.7-47.8)
	1,930	202	12	35
Current	11.9 (11.0-12.9)	14.2 (11.4-17.0)	8.7 (2.8-14.6)	14.0 (6.8-21.2)
	558	85	8	13
<u>Women</u>				
Age				
Mean years	50.3 (50.0-50.6) ^a	52.0 (51.5-52.5) ^a	$50.8 (49.5-52.0)^{a}$	50.7 (49.2-52.1) ^a
	1,713	554	101	69
BMI				
Mean kg/m²	25.3 (25.0-25.5) ^a	26.7 (26.2-27.1) ^a	26.3 (25.2-27.4) ^a	27.1 (25.7-28.4) ^a
	1,713	554	101	69
Employment status				
Employed	91.2 (89.9-92.6)	86.3 (83.4-89.2)	91.1 (85.4-96.7)	84.1 (75.2-92.9)
	1,563	478	92	58
Retired	5.7 (4.6-6.8)	6.0 (4.0-7.9)	4.0 (0.1-7.8)	5.8 (0.1-11.5)
	98	33	4	4
Other ^b	3.0 (2.2-3.8)	7.8 (5.5-10.0)	5.0 (0.6-9.3)	10.1 (2.8-17.5)
	52	43	5	7
Ethnicity				
White	92.3 (91.0-93.5)	82.2 (78.9-85.4)	42.2 (32.3-52.0)	59.4 (47.5-71.3)
	1,580	455	43	41
South Asian	2.3 (1.6-3.0)	7.8 (5.6-10.1)	40.7 (30.9-50.4)	20.3 (10.6-30.0)
	39	43	41	14
Other	5.5 (4.4-6.5)	10.0 (7.5-12.6)	17.2 (9.6-24.7)	20.3 (10.6-30.0)
	94	56	17	14

Family history of T2DM				
No	87.5 (85.9-89.1)	85.8 (82.8-88.7)	83.2 (75.7-90.6)	83.5 (74.4-92.6)
	1 499	475	84	58
Yes	12.5 (10.9-14.1) 214	14.2 (11.3-17.2) 79	16.8 (9.4-24.3) 17	
Incident T2DM				
No	91.4 (90.0-92.7)	81.2 (78.0-84.5)	85.1 (78.1-92.2)	81.2 (71.7-90.6)
	1 565	450	86	56
Yes	8.6 (7.3-10.0) 148	18.8 (15.5-22.0) 104	14.9 (7.8-21.9) 15	
Occupational class				
Administrative (top)	21.0 (19.0-22.9)	5.2 (3.4-7.1)	5.0 (0.6-9.3)	5.8 (0.1-11.5)
	359	29	5	4
Professional (middle)	49.2 (46.8-51.6)	42.1 (37.9-46.2)	24.8 (16.2-33.3)	31.9 (20.6-43.2)
	843	233	25	22
Clerical (bottom)	29.8 (27.7-32.0)	52.7 (48.5-56.9)	70.3 (61.2-79.4)	62.3 (50.6-74.1)
	511	292	71	43
Physical activity ^c				
Inactive	29.9 (27.8-32.1)	45.5 (41.3-49.6)	51.5 (41.6-61.4)	46.4 (34.3-58.5)
	513	252	52	32
Below guidelines	35.9 (33.6-38.2)	26.7 (23.0-30.4)	25.7 (17.1-34.4)	31.9 (20.6-43.2)
	615	148	26	22
Met guidelines	34.2 (31.9-36.4)	27.8 (24.1-31.5)	22.8 (14.5-31.1)	21.7 (11.8-31.7)
	585	154	23	15

Very good/Excellent	46.1 (43.8-48.5)	35.9 (31.9-39.9)	21.8 (13.6-30.0)	20.3 (10.6-30.0)
	790	199	22	14
Good	42.1 (39.7-44.4)	44.4 (40.3-48.6)	54.5 (44.6-64.3)	37.7 (25.9-49.4)
	721	246	55	26
Fair/Poor	11.8 (10.3-13.3)	19.7 (16.4-23.0)	23.8 (15.3-32.2)	42.0 (30.1-54.0)
	202	109	24	29
Smoking				
Never	51.0 (48.7-53.4)	56.3 (52.1-60.5)	88.9 (82.6-95.2)	70.8 (59.7-81.8)
	874	312	90	49
Former	32.3 (30.1-34.5)	25.2 (21.6-28.9)	4.2 (0.1-8.2)	19.1 (9.5-28.6)
	554	140	4	13
Current	16.6 (14.9-18.4)	18.5 (15.2-21.7)	6.9 (1.9-12.0)	10.2 (2.8-17.5)
	285	102	7	7
Table excluded consumption frequency owing ^a Mean and 95% confidence interval; ^b Redund intensity or >75 minutes of vigorous-intensity	g to colinearity with cons ant, student, long-term s v activity per week): inac	sumption volume. ick, long-term carer; ^c Meet tive (<60 minutes of moder	ing guidelines (≥150 minu ate and <60 minutes of vie	tes of moderate- corous activity:

below guidelines (anyone not inactive or meeting guidelines).

Self-reported general health

Appendix 7.6 Multivariable-adjusted dose response relationship between categories of average weekly volume of alcohol consumption and T2DM, stratified by sex. Imputed data.

	Men (n=5,456	5)	Women (n=2,43	34)
Alcohol consumption (wave 3)	HR (95% CI)	p-value	HR (95% CI)	p-value
Never drinkers ^a	(reference)		(reference)	
Non-current drinkers ^b	1.26 (0.61-2.64)	0.531	1.12 (0.53-2.39)	0.769
Infrequent drinkers ^c	0.99 (0.57-1.74)	0.981	1.62 (0.92-2.88)	0.098
0.1-50.0g/week	0.97 (0.57-1.64)	0.905	1.06 (0.59-1.91)	0.845
50.1-100.0g/week	1.03 (0.60-1.76)	0.925	0.86 (0.42-1.73)	0.666
100.1-150.0g/week	1.05 (0.60-1.84)	0.866	0.58 (0.27-1.25)	0.163
>150.0g/week ^d	1.01 (0.59-1.73)	0.972	-	-
Log likelihood	-4733 (-473	5, -4731)	-1963 (-1964	4, -1962)
BIC ^e	9630 (962	25, 9634)	4066 (406	64, 4068)

Models adjusted for all a priori covariates at baseline: age, BMI, employment status, ethnicity, family history of T2DM, occupational grade, physical activity, smoking status. Ethnicity was derived from responses at waves one and five.

^aParticipants who reported no consumption in the past week in waves 1 and 3, no consumption in the past year in waves 1 and 3, and stated they had 'always been a non-drinker' in wave 3, the first year the always non-drinker variable was available; ^bParticipants who reported no consumption in the last year but had not 'aways been a non-drinker'; ^cConsumed alcohol in the past year but not in the past week; ^dAmong women, this category was merged with those who reported consuming 100.1-150.0g/week; ^eBayesian information criterion. Fit statistics refer to the mean and range of values reported by the first three imputations. Appendix 7.7 Multivariable-adjusted interaction between a continuous measure of average weekly alcohol volume and consumption frequency, stratified by sex and excluding non-current and infrequent drinkers. Imputed data.

	Model 1		Model 2	
Alcohol consumption (wave 3)	HR (95% CI)	p-value	HR (95% CI)	p-value
Men (n=4,770)				
Dose-response by volume				
Per 10 g/week increase	1.00 (0.99-1.01)	0.761	1.00 (0.99-1.01)	0.480
Difference in risk by frequency				
Daily	-	-	(reference)	
Non-daily	-	-	0.89 (0.67-1.20)	0.449
Difference in dose-response by frequency				
Daily	-	-	(reference)	
Non-daily	-	-	1.01 (0.98-1.03)	0.653
Log likelihood	-4074 (-407	6, -4072)	-4074 (-407	5, -4071)
BIC ^a	8267 (826	52, 8270)	8283 (82)	78, 8286)
Women (n=1,814) Dose-response by volume				
Per 10 g/week increase	0.95 (0.92-0.99)	0.021	0.96 (0.91-1.02)	0.156
Difference in risk by frequency				
Daily	-	-	(reference)	
Non-daily	-	-	1.06 (0.52-2.17)	0.867
Difference in dose-response by frequency				
Daily	-	-	(reference)	
Non-daily	-	-	0.95 (0.87-1.05)	0.324
Log likelihood	-1095 (-109	5, -1095)	-1094 (-109	4, -1094)
BIC ^a	2295 (229	95, 2295)	2308 (230	08, 2309)

Model 1 reported the linear dose-response relationship between volume alcohol consumption and T2DM. Model 2 included an interaction term between a continuous measure of volume alcohol consumption and whether participants reported drinking alcohol daily or less than daily over the year preceding interview. All models adjusted for baseline covariates: age, BMI, employment status, ethnicity, family history of T2DM, occupational grade, physical activity, smoking status. Ethnicity was derived from responses at waves one and five. All models

excluded non-current drinkers from the reference level of exposure (Og/week). ^aBayesian information criterion. Fit statistics refer to the mean and range of values reported by the first three imputations.

Appendix 7.8 Multivariable-adjusted interaction between a continuous measure of average weekly volume of alcohol consumption and drinking frequency, stratified by sex and including non-current and infrequent drinkers. Observed data.

-	Model 1		Model 2	
Alcohol consumption (wave 3)	HR (95% CI)	p-value	HR (95% CI)	p-value
Men (n=4,874)				
Dose-response by volume				
Per 10 g/week increase	1.00 (0.99-1.00)	0.504	0.99 (0.98-1.00)	0.240
Difference in risk by frequency				
Daily	-	-	(reference)	
Non-daily	-	-	0.86 (0.65-1.14)	0.301
Difference in dose-response by frequency				
Daily	-	-	(reference)	
Non-daily	-	-	1.00 (0.98-1.03)	0.621
Log likelihood		-4177		-4177
BIC ^a		8473		8489
Women (n=2,094) Dose-response by volume				
Per 10 g/week increase	0.94 (0.90-0.97)	0.001	0.93 (0.86-1.00)	0.055
Difference in risk by frequency				
Daily	-	-	(reference)	
Non-daily	-	-	0.78 (0.35-1.72)	0.536
Difference in dose-response				
by frequency				
Daily	-	-	(reference)	
Non-daily	-	-	0.97 (0.88-1.08)	0.536
Log likelihood		-1700		-1700
BIC ^a		3507		3519

Model 1 reported the linear dose-response relationship between volume alcohol consumption and T2DM. Model 2 included an interaction term between a continuous measure of volume alcohol consumption and whether participants reported drinking alcohol daily or less than daily over the year preceding interview. All models adjusted for baseline covariates: age, BMI, employment status, ethnicity, family history of T2DM, occupational grade, physical activity, smoking status. Ethnicity was derived from responses at waves one and five. All models included non-current drinkers in the reference level of exposure (0g/week).

^aBayesian information criterion.

Appendix 7.9 Multivariable-adjusted interaction between a continuous measure of average weekly volume of alcohol consumption and drinking frequency, stratified by sex and including non-current and infrequent drinkers. Imputed data.

·	Model 1		Model 2	
Alcohol consumption (wave 3)	HR (95% CI)	p-value	HR (95% CI)	p-value
Men (n=5,464)				
Dose-response by volume				
Per 10 g/week increase	1.00 (0.99-1.01)	0.695	1.00 (0.99-1.01)	0.565
Difference in risk by frequency				
Daily	-	-	(reference)	
Non-daily	-	-	0.94 (0.72-1.23)	0.661
Difference in dose-response by frequency				
Daily	-	-	(reference)	
Non-daily	-	-	1.00 (0.98-1.02)	0.910
Log likelihood	-4734 (-473	6, -4731)	-4734 (-473	5, -4731)
BIC ^a	9588 (958	83,9592)	9605 (960	00, 9609)
Women (n=2,437)				
Dose-response by volume	0.00 (0.00 0.07)	-0.004	0.02 (0.06.4.00)	0.020
Per 10 g/week Increase	0.93 (0.90-0.97)	<0.001	0.93 (0.86-1.00)	0.039
Difference in risk by frequency				
Daily	-	-	(reference)	
Non-daily	-	-	0.81 (0.39-1.67)	0.562
Difference in dose-response by				
Daily	-	_	(reference)	
Non-daily	-	_	0.95 (0.87-1.05)	0 347
	_	_	0.00 (0.07 1.00)	0.547
Log likelihood	-1964 (-196	5, -1962)	-1962 (-196	3, -1961)
BIC ^a	4037 (403	35, 4039)	4048 (404	46, 4051)

Model 1 reported the linear dose-response relationship between volume alcohol consumption and T2DM. Model 2 included an interaction term between a continuous measure of volume alcohol consumption and whether participants reported drinking alcohol daily or less than daily over the year preceding interview. All models adjusted for baseline covariates: age, BMI, employment status, ethnicity, family history of T2DM, occupational grade, physical activity, smoking status. Ethnicity was derived from responses at waves one and five. All models included non-current drinkers in the reference level of exposure (0g/week). ^aBayesian information criterion. Fit statistics refer to the mean and range of values reported by the first three imputations. Appendix 7.10 Multivariable-adjusted dose response relationship between categories of average weekly volume of alcohol consumption and T2DM, stratified by sex. Hazards were permitted to vary as a function of linear time. Observed data.

-	Men (n=4,874)		Women (n=2,094)	
Alcohol consumption (wave 3)	HR (95% CI)	p-value	HR (95% CI)	p-value
Risk at baseline				
Never drinkers ^a	(reference)		(reference)	
Non-current drinkers ^b	1.83 (0.30-11.24)	0.515	8.41 (0.79-89.64)	0.078
Infrequent drinkers ^c	0.87 (0.20-3.85)	0.856	11.99 (1.59-90.47)	0.016
0.1-50.0g/week	1.13 (0.29-4.40)	0.864	9.48 (1.23-73.25)	0.031
50.1-100.0g/week	1.44 (0.35-5.86)	0.610	4.28 (0.42-43.71)	0.220
100.1-150.0g/week	0.69 (0.15-3.14)	0.631	2.51 (0.20-30.96)	0.472
>150.0g/week ^d	1.34 (0.33-5.50)	0.686	-	-
Risk per year increase in follow-up				
Never drinkers	(reference)			
Non-current drinkers	0.98 (0.83-1.14)	0.772	0.84 (0.69-1.01)	0.064
Infrequent drinkers	1.01 (0.89-1.14)	0.906	0.83 (0.71-0.97)	0.017
0.1-50.0g/week	0.98 (0.88-1.10)	0.776	0.83 (0.71-0.96)	0.016
50.1-100.0g/week	0.97 (0.86-1.09)	0.636	0.87 (0.73-1.03)	0.113
100.1-150.0g/week	1.03 (0.91-1.17)	0.592	0.88 (0.72-1.06)	0.185
>150.0g/week	0.97 (0.87-1.10)	0.674	-	-
Log likelihood		-4162		-1690
BIC ^e		8647		3657

Models adjusted for all a priori covariates at baseline: age, BMI, employment status, ethnicity, family history of T2DM, occupational grade, physical activity, smoking status. Ethnicity was derived from responses at waves one and five.

^aParticipants who reported no consumption in the past week in waves 1 and 3, no consumption in the past year in waves 1 and 3, and stated they had 'always been a non-drinker' in wave 3, the first year the always non-drinker variable was available; ^bParticipants who reported no consumption in the last year but had not 'aways been a non-drinker'; ^cConsumed alcohol in the past year but not in the past week; ^dAmong women, this category was merged with those who reported consuming 100.1-150.0g/week; ^eBayesian information criterion. Appendix 7.11 Multivariable-adjusted dose-response relationship between categories of average weekly volume of alcohol consumption, drink type and T2DM, stratified by sex. Imputed data.

-	Men (n=5,083)	Women (n=2,22	4)
Alcohol consumption (wave 3)	HR (95% CI)	p-value	HR (95% CI)	p-value
Drink type				
Beer (per 10 g/week)	0.99 (0.98-1.00)	0.236	0.85 (0.72-0.99)	0.042
Spirits (per 10 g/week)	1.01 (0.99-1.03)	0.343	0.94 (0.87-1.01)	0.113
Wine (per 10 g/week)	1.00 (0.98-1.02)	0.773	0.95 (0.90-1.01)	0.081
Log likelihood	-4509 (-457	4, -4378)	-1758 (-175	9, -1758)
BIC ^a	8895 (889	93, 8897)	3640 (364	40, 3641)

Models reported results adjusted for all a priori covariates: age, BMI, employment status, ethnicity, family history of T2DM, occupational grade, physical activity, smoking status. Ethnicity was derived from responses at waves one and five. Models also included variables representing the volume of each drink type consumed during the week prior to interview.

^aBayesian information criterion. Fit statistics refer to the mean and range of values reported by the first three imputations.

11.4 Appendices for Chapter 8

Appendix 8.1 Crude sex-specific linear trajectory of mean weekly volume of alcohol consumption between the ages of 34-84 years: goodness of fit statistics. Imputed data.

	Fit statis	itics
Linear mixed models	Log-likelihood	BIC ^a
Men		
Intercept only	-232502 (-232763,-232191)	465035 (464414-465557)
Linear mixed model, fixed slopes	-232476 (-232731,-232165)	464994 (464371-465504)
Linear mixed model, random slopes	-232361 (-232622,-232023)	464786 (464109-465306)
Women		
Intercept only	-103392 (-103679,-103192)	206814 (206414-207387)
Linear mixed model, fixed slopes	-103390 (-103677,-103191)	206820 (206422-207393)
Linear mixed model, random slopes	-103355 (-103650,-103150)	206756 (206360-207320)
^a Bayesian information criterion. Value	es refer to the mean and range of	fit statistics as reported

from the first three imputations.

Appendix 8.2 Crude sex-specific linear and non-linear trajectories of mean weekly volume of alcohol consumption between the ages of 34-84 years: results. Imputed data.

Mixed models	g/week (95% CI)	p-value
<u>Men (n=6,895)</u>		
Linear model		
Intercept	114.8 (111.3, 118.3)	<0.001
Age	-3.0 (-4.2, -1.9)	<0.001
Non-linear model		
Intercept	88.6 (84.2, 93.0)	<0.001
Age ¹	2.5 (2.1, 2.8)	<0.001
Age ²	-6.1 (-6.8, -6.3)	<0.001
<u>Women (n=3,413)</u>		
Linear model		
Intercept	50.5 (47.6, 53.4)	<0.001
Age	-0.5 (-1.4, 0.5)	0.352
Non-linear model		
Intercept	39.4 (36.1, 42.8)	<0.001
Age ¹	0.8 (0.6, 1.0)	<0.001
Age ³	-0.4 (-0.5, -0.3)	<0.001
Age coefficients refer to the ch	hange in the average volume of w	veekly

Age coefficients refer to the change in the average volume of weekly alcohol consumption per 10-year increase in age.

Appendix 8.3 Crude sex specific trajectories of mean weekly volume of alcohol consumption between the ages of 34-84 years: figure. Imputed data.



Appendix 8.4 Crude sex-specific linear and non-linear trajectories of mean weekly volume of alcohol consumption between the ages of 34-84 years: goodness of fit statistics. Observed data.

	Men		Wome	n
First-order polynomial models	Log-likelihood	BIC ^a	Log-likelihood	BIC ^a
age ⁻²	-184343	368748	-72549	145156
age ⁻¹	-184320	368702	-72538	145133
age ^{-0.5}	-184254	368571	-72520	145098
In(age)	-184130	368322	-72496	145049
age ^{0.5}	-184072	368206	-72494	145045
age	-184042	368146	-72500	145057
age ²	-184056	368174	-72493	145043
age ³	-184098	368259	-72465	144987
Second-order polynomial models	Log-likelihood	BIC ^a	Log-likelihood	BIC ^a
age ⁻² +age ⁻¹	-184343	368717	-72549	145164
age ⁻² +age ^{-0.5}	-184342	368757	-72547	145161
age ⁻² +In(age)	-184337	368747	-72543	145152
age ⁻² +age ^{0.5}	-184329	368730	-72537	145140
age ⁻² +age	-184315	368702	-72529	145125
age ⁻² +age ²	-184284	368641	-72514	145095
age ⁻² +age ³	-184261	368552	-72503	145073
age ⁻¹ +age ⁻²	-184320	368712	-72538	145142
age ⁻¹ +age ^{-0.5}	-184313	368699	-72533	145133
age ⁻¹ +ln(age)	-184304	368680	-72528	145123
age ⁻¹ +age ^{0.5}	-184291	368655	-72522	145110
age ⁻¹ +age	-184276	368624	-72514	145096
age ⁻¹ +age ²	-184246	368564	-72501	145068
age ⁻¹ +age ³	-184225	368523	-72491	145050
age ^{-0.5} +age ²	-184253	368579	-72520	145106
age ^{-0.5} +age ⁻¹	-184247	368565	-72517	145101
age ^{-0.5} +In(age)	-184225	368523	-72508	145083
age ^{-0.5} +age ^{0.5}	-184207	368487	-72500	145067
age ^{-0.5} +age	-184190	368452	-72493	145052
age ^{-0.5} +age ²	-184160	368392	-72480	145027
age ^{-0.5} +age ³	-184143	368358	-72472	145011

In(age)+age ⁻²	-184127	368327	-72495	145058
In(age)+age ⁻¹	-184117	368306	-72492	145051
In(age)+age ^{-0.5}	-184106	368284	-72489	145044
In(age)+age ^{0.5}	-184051	368174	-72469	145005
In(age)+age	-184035	368143	-72463	144993
In(age)+age ²	-184009	368090	-72452	144971
In(age)+age ³	-183997	368067	-72447	144960
age ^{0.5} +age ⁻²	-184068	368208	-72494	145054
age ^{0.5} +age ⁻¹	-184054	368180	-72490	145047
age ^{0.5} +age ^{-0.5}	-184037	368147	-72485	145037
age ^{0.5} +In(age)	-184000	368072	-72473	145013
age ^{0.5} +age	-183968	368009	-72462	144991
age ^{0.5} +age ²	-183941	367954	-72452	144970
age ^{0.5} +age ³	-183933	367938	-72447	144961
age+age ⁻²	-184037	368147	-72500	145066
age+age ⁻¹	-184022	368116	-72497	145060
age+age ^{-0.5}	-184003	368079	-72492	145050
age+In(age)	-183968	368009	-72481	145029
age+age ^{0.5}	-183953	367978	-72476	145019
age+age ²	-183909	367890	-72461	144988
age+age ³	-183905	367883	-72458	144983
age ² +age ⁻²	-184052	368177	-72493	145052
age ² +age ⁻¹	-184037	368147	-72491	145048
age ² +age ^{-0.5}	-184019	368110	-72486	145040
age ² +In(age)	-183990	368052	-72479	145024
age ² +age ^{0.5}	-183975	368023	-72475	145016
age ² +age	-183961	367994	-72470	145008
age ² +age ³	-183953	367978	-72467	145000
age ³ +age ⁻²	-184096	368264	-72465	144996
age ³ +age ⁻¹	-184084	368241	-72464	144995
age ³ +age ^{-0.5}	-184070	368213	-72461	144989
age ³ +In(age)	-184050	368172	-72456	144980
age ³ +age ^{0.5}	-184041	368155	-72454	144975
age ³ +age	-184034	368141	-72452	144971
age ³ +age ²	-184033	368138	-72452	144970
^a Bayesian information criterion.				

	Men (m:	=3/50) ^a
First-order polynomial models	Log-likelihood	BIC ^a
Men		
age ⁻²	-232501 (-232762, -232191)	465034 (464414, 465556)
age ⁻¹	-232501 (-232762, -232191)	465045 (464424, 465567)
age ^{-0.5}	-232501 (-232762, -232191)	465044 (464423, 465567)
In(age)	-232501 (-232761, -232191)	465044 (464423, 465564)
age ^{0.5}	-232492 (-232750, -232181)	465027 (464405, 465542)
age	-232476 (-232731, -232165)	464994 (464371, 465504)
age ²	-232436 (-232686, -232126)	464915 (464295, 465415)
age ³	-232407 (-232655, -232099)	464857 (464241, 465352)
<u>Women</u>		
age ⁻²	-103392 (-103677, -103192)	206823 (206424, 207394)
age ⁻¹	-103389 (-103673, -103190)	206817 (206419, 207385)
age ^{-0.5}	-103389 (-103673, -103189)	206817 (206418, 207386)
In(age)	-103391 (-103677, -103191)	206822 (206421, 207393)
age ^{0.5}	-103392 (-103679, -103192)	206824 (206424, 207397)
age	-103390 (-103677, -103191)	206820 (206422, 207393)
age ²	-103380 (-103666, -103184)	206800 (206408, 207372)
age ³	-103369 (-103655, -103176)	206778 (206391, 207349)

Appendix 8.5 Crude sex-specific linear and non-linear trajectories of mean weekly volume of alcohol consumption between the ages of 34-84 years: goodness of fit statistics. Imputed data.

Second-order polynomial models	Log-likelihood	BIC ^a
Men		
age ⁻² +age ⁻¹	-232501 (-232762, -232191)	465034 (464414, 465556)
age ⁻² +age ^{-0.5}	-232500 (-232762, -232190)	465032 (464412, 465555)
age ⁻² +ln(age)	-232500 (-232761, -232190)	465043 (464423, 465563)
age ⁻² +age ^{0.5}	-232492 (-232750, -232181)	465026 (464405, 465542)
age ⁻² +age	-232475 (-232730, -232164)	464993 (464371, 465503)
age ⁻² +age ²	-232436 (-232686, -232126)	464914 (464294, 465414)
age ⁻² +age ³	-232407 (-232655, -232099)	464856 (464240, 465351)
a co ⁻¹ i a co ⁻²		
age +age	-232501 (-232762, -232191)	465045 (464424, 465567)
age ⁻¹ +age ^{-0.5}	-232500 (-232762, -232190)	465054 (464433, 465576)
age ⁻¹ +ln(age)	-232500 (-232761, -232190)	465054 (464433, 465574)
age ⁻¹ +age ^{0.5}	-232492 (-232750, -232181)	465037 (464415, 465552)
age ⁻¹ +age	-232475 (-232730, -232164)	465004 (464381, 465513)
age ⁻¹ +age ²	-232436 (-232686, -232126)	464925 (464304, 465425)
age ⁻¹ +age ³	-232407 (-232655, -232099)	464867 (464251, 465362)

$age^{-0.5}+age^{2}$	-232500 (-232762, -232190)	465032 (464412, 465555)
age ^{-0.5} +age ⁻¹	-232500 (-232762, -232190)	465054 (464433, 465576)
age ^{-0.5} +In(age)	-232466 (-232722, -232152)	464985 (464357, 465498)
$age^{-0.5}+age^{0.5}$	-232444 (-232699, -232131)	464942 (464314, 465450)
age ^{-0.5} +age	-232422 (-232674, -232109)	464897 (464271, 465401)
$age^{-0.5}+age^{2}$	-232387 (-232635, -232077)	464827 (464206, 465324)
$age^{-0.5}+age^{3}$	-232369 (-232617, -232062)	464791 (464176, 465286)
In(age)+age ⁻²	-232500 (-232761, -232190)	465043 (464423, 465563)
In(age)+age ⁻¹	-232500 (-232761, -232190)	465054 (464433, 465574)
In(age)+age ^{-0.5}	-232466 (-232722, -232152)	464985 (464357, 465498)
In(age)+age ^{0.5}	-232419 (-232671, -232105)	464890 (464264, 465394)
In(age)+age	-232396 (-232646, -232084)	464845 (464222, 465345)
In(age)+age ²	-232367 (-232614, -232058)	464786 (464169, 465281)
In(age)+age ³	-232357 (-232604, -232051)	464766 (464156, 465261)
age ^{0.5} +age ⁻²	-232492 (-232750, -232181)	465026 (464405, 465542)
age ^{0.5} +age ⁻¹	-232492 (-232750, -232181)	465037 (464415, 465552)
age ^{0.5} +age ^{-0.5}	-232444 (-232699, -232131)	464942 (464314, 465450)
age ^{0.5} +In(age)	-232419 (-232671, -232105)	464890 (464264, 465394)
age ^{0.5} +age	-232374 (-232622, -232064)	464800 (464180, 465296)
age ^{0.5} +age ²	-232353 (-232600, -232047)	464759 (464147, 465252)
age ^{0.5} +age ³	-232352 (-232599, -232048)	464756 (464149, 465250)
age+age ⁻²	-232475 (-232730, -232164)	464993 (464371, 465503)
age+age ⁻¹	-232475 (-232730, -232164)	465004 (464381, 465513)
age+age ^{-0.5}	-232422 (-232674, -232109)	464897 (464271, 465401)
age+In(age)	-232396 (-232646, -232084)	464845 (464222, 465345)
age+age ^{0.5}	-232374 (-232622, -232064)	464800 (464180, 465296)
age+age ²	-232348 (-232594, -232044)	464748 (464140, 465240)
age+age ³	-232353 (-232600, -232051)	464758 (464155, 465253)
age ² +age ⁻²	-232436 (-232686, -232126)	464914 (464294, 465414)
age ² +age ⁻¹	-232436 (-232686, -232126)	464925 (464304, 465425)
age ² +age ^{-0.5}	-232387 (-232635, -232077)	464827 (464206, 465324)
age ² +In(age)	-232367 (-232614, -232058)	464786 (464169, 465281)
age ² +age ^{0.5}	-232353 (-232600, -232047)	464759 (464147, 465252)
age ² +age	-232348 (-232594, -232044)	464748 (464140, 465240)
age ² +age ³	-232362 (-232611, -232062)	464778 (464176, 465275)

age ³ +age ⁻²	-232407 (-232655, -232099)	464856 (464240, 465351)
age ³ +age ⁻¹	-232407 (-232655, -232099)	464867 (464251, 465362)
age ³ +age ^{-0.5}	-232369 (-232617, -232062)	464791 (464176, 465286)
age ³ +ln(age)	-232357 (-232604, -232051)	464766 (464156, 465261)
age ³ +age ^{0.5}	-232352 (-232599, -232048)	464756 (464149, 465250)
age ³ +age	-232353 (-232600, -232051)	464758 (464155, 465253)
age ³ +age ²	-232362 (-232611, -232062)	464778 (464176, 465275)
<u>Women</u>		
age ⁻² +age ⁻¹	-103386 (-103670, -103186)	206821 (206421, 207389)
age ⁻² +age ^{-0.5}	-103388 (-103673, -103188)	206826 (206425, 207395)
age ⁻² +ln(age)	-103391 (-103676, -103191)	206831 (206431, 207402)
age ⁻² +age ^{0.5}	-103391 (-103677, -103192)	206832 (206434, 207404)
age ⁻² +age	-103389 (-103675, -103191)	206827 (206431, 207398)
age ⁻² +age ²	-103378 (-103664, -103184)	206806 (206417, 207376)
age ⁻² +age ³	-103368 (-103652, -103176)	206785 (206401, 207353)
age ⁻¹ +age ⁻²	-103386 (-103670, -103186)	206821 (206421, 207389)
age ⁻¹ +age ^{-0.5}	-103388 (-103673, -103189)	206826 (206428, 207395)
age⁻¹+ln(age)	-103388 (-103672, -103190)	206825 (206429, 207394)
age ⁻¹ +age ^{0.5}	-103385 (-103669, -103189)	206819 (206427, 207387)
age ⁻¹ +age	-103380 (-103663, -103186)	206809 (206420, 207376)
age ⁻¹ +age ²	-103368 (-103650, -103177)	206785 (206403, 207350)
age ⁻¹ +age ³	-103358 (-103639, -103168)	206765 (206386, 207328)
age ^{-0.5} +age ²	-103388 (-103673, -103188)	206826 (206425, 207395)
age ^{-0.5} +age ⁻¹	-103388 (-103673, -103189)	206826 (206428, 207395)
age ^{-0.5} +In(age)	-103383 (-103666, -103187)	206815 (206423, 207382)
age ^{-0.5} +age ^{0.5}	-103377 (-103660, -103183)	206804 (206416, 207368)
age ^{-0.5} +age	-103370 (-103652, -103178)	206790 (206405, 207353)
age ^{-0.5} +age ²	-103357 (-103637, -103168)	206763 (206384, 207324)
age ^{-0.5} +age ³	-103348 (-103627, -103160)	206745 (206368, 207304)
In(age)+age ⁻²	-103391 (-103676, -103191)	206831 (206431, 207402)
In(age)+age ⁻¹	-103388 (-103672, -103190)	206825 (206429, 207394)
In(age)+age ^{-0.5}	-103383 (-103666, -103187)	206815 (206423, 207382)
In(age)+age ^{0.5}	-103367 (-103648, -103175)	206784 (206400, 207345)
In(age)+age	-103359 (-103639, -103169)	206767 (206386, 207327)
In(age)+age ²	-103346 (-103624, -103157)	206741 (206364, 207298)
In(age)+age ³	-103338 (-103617, -103151)	206726 (206351, 207283)

age ^{0.5} +age ⁻²	-103391 (-103677, -103192)	206832 (206434, 207404)
age ^{0.5} +age ⁻¹	-103385 (-103669, -103189)	206819 (206427, 207387)
age ^{0.5} +age ^{-0.5}	-103377 (-103660, -103183)	206804 (206416, 207368)
age ^{0.5} +ln(age)	-103367 (-103648, -103175)	206784 (206400, 207345)
age ^{0.5} +age	-103348 (-103627, -103159)	206746 (206368, 207303)
age ^{0.5} +age ²	-103337 (-103615, -103149)	206724 (206348, 207280)
age ^{0.5} +age ³	-103333 (-103611, -103145)	206715 (206339, 207272)
age+age ⁻²	-103389 (-103675, -103191)	206827 (206431, 207398)
age+age ⁻¹	-103380 (-103663, -103186)	206809 (206420, 207376)
age+age ^{-0.5}	-103370 (-103652, -103178)	206790 (206405, 207353)
age+In(age)	-103359 (-103639, -103169)	206767 (206386, 207327)
age+age ^{0.5}	-103348 (-103627, -103159)	206746 (206368, 207303)
age+age ²	-103333 (-103611, -103145)	206715 (206338, 207271)
age+age ³	-103331 (-103610, -103143)	206712 (206335, 207269)
age ² +age ⁻²	-103378 (-103664, -103184)	206806 (206417, 207376)
age ² +age ⁻¹	-103368 (-103650, -103177)	206785 (206403, 207350)
age ² +age ^{-0.5}	-103357 (-103637, -103168)	206763 (206384, 207324)
age ² +In(age)	-103346 (-103624, -103157)	206741 (206364, 207298)
age ² +age ^{0.5}	-103337 (-103615, -103149)	206724 (206348, 207280)
age ² +age	-103333 (-103611, -103145)	206715 (206338, 207271)
age ² +age ³	-103333 (-103613, -103144)	206716 (206337, 207276)
age ³ +age ⁻²	-103368 (-103652, -103176)	206785 (206401, 207353)
age ³ +age ⁻¹	-103358 (-103639, -103168)	206765 (206386, 207328)
age ³ +age ^{-0.5}	-103348 (-103627, -103160)	206745 (206368, 207304)
age ³ +In(age)	-103338 (-103617, -103151)	206726 (206351, 207283)
age ³ +age ^{0.5}	-103333 (-103611, -103145)	206715 (206339, 207272)
age ³ +age	-103331 (-103610, -103143)	206712 (206335, 207269)
age ³ +age ²	-103333 (-103613, -103144)	206716 (206337, 207276)

Fit statistics calculated on fixed effect models due to issues of convergence when some transformations were included as random effects. Values represent the mean and range of fit statistics as reported from the first three imputations. ^aBayesian information criterion.

Appendix 8.6 Crude sex-specific non-linear trajectories of mean weekly volume of alcohol consumption between the ages of 34-84 years: figure. Imputed data.



Appendix 8.7 Crude sex-specific interaction between the trajectory of mean weekly volume of alcohol consumption and baseline alcohol consumption category: results. Imputed data.

Linear mixed models	Sample n	g/week (95% CI)	p-value
Men			
Difference in baseline consumption by drinking category			
Non-drinker	220	Reference	
Infrequent drinker	669	7.8 (3.6, 12.0)	<0.001
0.1-50.0 g/week	2,073	37.6 (33.9, 41.3)	<0.001
50.1-100.0 g/week	1,432	85.6 (81.1, 90.1)	<0.001
100.1-150.0 g/week	881	140.8 (135.1, 146.3)	<0.001
>150.0 g/week	1,563	291.9 (283.1, 301.0)	<0.001
Difference in the rate of change by drinking category			
Non-drinker		Reference	
Infrequent drinker		4.0 (1.9, 6.2)	<0.001
0.1-50.0 g/week		1.7 (-0.3, 3.6)	0.092
50.1-100.0 g/week		-1.3 (-3.8, 1.1)	0.276
100.1-150.0 g/week		-8.9 (-11.6, -6.1)	<0.001
>150.0 g/week		-36.4 (-40.0, -32.8)	<0.001
<u>Women</u>			
Difference in baseline consumption by drinking category			
Non-drinker	216	Reference	
Infrequent drinker	764	4.9 (2.0, 7.8)	0.001
0.1-50.0 g/week	1,428	33.4 (30.4, 36.5)	<0.001
50.1-100.0 g/week	542	82.0 (77.1, 86.8)	<0.001
>100.0 g/week	422	184.1 (173.2, 195.0)	<0.001
Difference in the rate of change by drinking category			
Non-drinker		Reference	
Infrequent drinker		1.2 (-0.4, 2.8)	0.143
0.1-50.0 g/week		-2.6 (-4.3, -1.0)	0.002
50.1-100.0 g/week		-7.1 (-9.5, -4.6)	<0.001
>100.0 g/week		-27.1 (-31.2, -23.0)	<0.001

Appendix 8.8 Crude sex-specific linear trajectories of mean weekly volume of alcohol consumption between the ages of 34-84 years, stratified by baseline alcohol consumption category: figure. Imputed data.



Appendix 8.9 Baseline characteristics of T2DM-free participants, stratified by T2DM diagnosis. Imputed data.

	T2DM	Censored
Variables (wave one)	% (95% CI)	% (95% CI)
	n	n
Men		
Age		
Mean years	45.0 (44.6, 45.5) ^a	44.4 (44.2, 44.6) ^a
	620	5,103
Alcohol consumption frequency		
None in past year	4.4 (2.8, 6.1)	2.6 (2.2, 3.0)
. ,	28	132
<1/week	21.9 (18.6, 25.1)	19.6 (18.5, 20.7)
	136	1,002
1-3 times/week	40.2 (36.4, 44.1)	43.6 (42.3, 45.0)
	249	2,226
Daily or almost daily	33.5 (29.7, 37.2)	34.2 (32.8, 35.5)
	207	1,743
Alcohol consumption volume		
Median g/week	98 8 (89 9, 107 8) ^b	101 8 (98 7 104 9) ^b
-	620	5.103
BMI		,
Mean kg/m^2	26 1 /25 8 26 2) ^a	フル 2 <i>(</i> フル フ フル ル) ^a
	20.1 (25.8, 20.3)	24.3 (24.2, 24.4) 5 103
	020	5,105
Ethnicty		
White	84.8 (82.0, 87.7)	94.6 (94.0, 95.2)
	526	4,826
South Asian	11.8 (9.2, 14.3)	3.4 (2.9, 3.9)
	73	175
Other ^c	3.4 (2.0, 4.8)	2.0 (1.6, 2.4)
	21	101
Family history of T2DM		
Yes	81.1 (78.0, 84.2)	91.2 (90.4, 92.0)
	503	4,652
No	18.9 (15.8, 22.0)	8.8 (8.0, 9.6)
	117	451
Occupational grade		
Administrative (top)	35.0 (31.2, 38.8)	41.3 (39.9, 42.6)
	217	2.106
Professional (middle)	54.0 (50.1, 58.0)	52.1 (50.7, 53.4)
	335	2,657
Clerical (bottom)	11.0 (8.5, 13.4)	6.7 (6.0, 7.3)
	68	340

Physical activity ^d		
Inactive	12.4 (9.8, 15.0) 77	8.3 (7.6, 9.1) 425
Below guidelines	40.0 (36.1.43.9)	37 / (36 1 38 8)
below guidennes	40.0 (30.1, 43.5)	1 Q11
Metguidelines	476 (436 51 5)	54 2 (52 0 55 6)
Met guruennes	47.0 (43.0, 51.5) 295	2,767
Smoking		
Never	42.4 (38.5, 46.3)	49.9 (48.5, 51.2)
	263	2,545
Former	39.1 (35.2, 42.9)	36.4 (35.0, 37.7)
	242	1,855
Current	18.6 (15.5, 21.6)	13.8 (12.8, 14.7)
	115	703
Women		
Age		
Mean years	46.7 (46.0, 47.4) ^a	45.3 (45.1, 45.6) ^a
	296	2,274
Alcohol consumption frequency		
None in past year	9.5 (6.1, 12.8)	5.7 (4.7, 6.6)
	28	129
<1/week	51.4 (45.6, 57.1)	35.3 (33.4, 37.3)
	152	803
1-3 times/week	28.7 (23.5, 33.9)	36.6 (34.6, 38.5)
	85	831
Daily or almost daily	10.5 (7.0, 14.0)	22.5 (20.7, 24.2)
	31	511
Alcohol consumption volume		
Median g/week	294 (23 3 35 5) ^b	470(445 495) ^b
	29.4 (29.5, 59.5)	2,274
BMI		
Mean kg/m ²	28.0 (27.4, 28.6) ^a	24.2 (24.0, 24.3) ^a
-	296	2,274
Ethnicty		
White	71.1 (65.9, 76.3)	89.1 (87.8, 90.3)
	211	2,025
South Asian	14.5 (10.5, 18.6)	4.6 (3.7, 5.4)
	43	103
Other ^c	14.3 (10.3, 18.4)	6.4 (5.4, 7.4)
	42	145
Family history of T2DM		
Yes	68.9 (63.6, 74.3)	89.1 (87.8, 90.4)
	204	2,026
No	31.1 (25.7, 36.4)	10.9 (9.6, 12.2)
	92	248

Occupational grade		
Administrative (top)	4.1 (1.8, 6.3)	14.0 (12.6, 15.5)
	12	319
Professional (middle)	36.8 (31.3, 42.4)	43.0 (41.0, 45.0)
	109	978
Clerical (bottom)	59.1 (53.5, 64.8)	43.0 (40.9, 45.0)
	175	977
Physical activity ^d		
Inactive	32.8 (27.4, 38.3)	23.0 (21.2, 24.7)
	97	522
Below guidelines	34.2 (28.7, 39.7)	40.3 (38.3, 42.3)
	101	917
Met guidelines	33.0 (27.6, 38.4)	36.7 (34.7, 38.7)
	98	835
Smoking		
Never	59.1 (53.5, 64.7)	54.3 (52.2, 56.3)
	175	1,234
Former	22.2 (17.4, 27.0)	24.9 (23.1, 26.7)
	66	567
Current	18.7 (14.2, 23.1)	20.8 (19.1, 22.5)
	55	473

^aMean and 95% confidence interval; ^bMedian and 25th and 75th percentiles; ^ce.g. black Caribbean, African and Arabic; ^dMeeting guidelines (≥150 minutes of moderate-intensity or ≥75 minutes of vigorous-intensity activity per week); inactive (<60 minutes of moderate and <60 minutes of vigorous activity; below guidelines (not inactive or meeting guidelines).

Appendix 8.10 Sex-specific interaction between the linear trajectory of mean weekly volume

of alcohol consumption and T2DM diagnosis: goodness of fit statistics. Imputed data

Mixed model	Log-likelihood	BIC ^a
Men		
Crude linear mixed model, fixed slope	-173035 (-173255, -172857)	346132 (345776, 346571)
Crude linear mixed model, random slopes	-172951 (-173171, -172765)	345984 (345613, 346424)
Women		
Crude linear mixed model, fixed slope	-67229 (-67566, -67016)	134514 (134089, 135188)
Crude linear mixed model, random slopes	-67196 (-67546, -66969)	134467 (134014, 135167)
^a Bayesian information criterion. Values refer to	o the mean and range of fit sta	atistics as reported by
the first three imputations.		

Appendix 8.11 Crude sex-specific interaction between the linear trajectory of mean weekly volume of alcohol consumption and T2DM diagnosis: results. Imputed data.

Crude linear mixed models g/week (95	% CI)	p-value
<u>Men (n=5,723)</u>		
Consumption volume		
Intercept 104.9 (101.9, 10)7.9)	<0.001
Change per 10 years prior to diagnosis or censoring -1.5 (-2.7,	-0.4)	0.009
Difference in consumption at the time of diagnosis or consoring		
Consored Pofer	onco	
		<0.001
	\$5.5)	<0.001
Difference in the rate of change by diagnosis or censoring		
Censored Refer	ence	
T2DM 17.0 (11.1, 2	23.0)	<0.001
Women (n=2.570)		
Consumption volume		
Intercept 48.3 (45.7, 5	50.8)	<0.001
Change per 10 years prior to diagnosis or censoring -1.0 (-2.0,	-0.0)	0.045
Difference in consumption at the time of diagnosis or censoring		
Censored Refer	ence	
T2DM -18.2 (-24.4, -1	12.0)	<0.001
Difference in the rate of change by diagnosis or censoring		
Censored Refer	ence	
T2DM 2.1 (-1.7,	6.0)	0.279

Appendix 8.12 Crude sex-specific linear trajectory of mean weekly volume of alcohol consumption, stratified by T2DM diagnosis: figure. Imputed data.





331

	T2DM	·	No T2DM	
First-order models	Log-likelihood	^a BIC	Log-likelihood	^a BIC
Men				
time ⁻²	-12665	25361	-149882	299804
time ⁻¹	-12665	25361	-149866	299772
time ¹	-12648	25327	-149877	299794
time ²	-12647	25324	-149882	299805
time ³	-12649	25329	-149876	299792
Women				
time ⁻²	-4758	9543	-57154	114345
time ⁻¹	-4758	9543	-57153	114343
time ¹	-4759	9545	-57152	114342
time ²	-4759	9545	-57159	114355
time ³	-4759	9545	-57159	114355
Second-order models	Log-likelihood	^a BIC	Log-likelihood	^a BIC
Men				
time ⁻² +time ⁻¹	-12665	25368	-149813	299678
time ⁻² +time	-12648	25334	-149875	299801
time ⁻² +time ²	-12647	25332	-149880	299812
time ⁻² +time ³	-12649	25336	-149874	299799
time ⁻¹ +time ⁻²	-12665	25368	-149813	299678
time ⁻¹ +time	-12648	25334	-149862	299774
time ⁻¹ +time ²	-12647	25332	-149864	299778
time ⁻¹ +time ³	-12649	25336	-149857	299764
time+time ⁻²	-12648	25334	-149875	299801
time+time ⁻¹	-12648	25334	-149862	299774
time+time ²	-12647	25332	-149709	299468
time+time ³	-12647	25333	-149730	299510
time ² +time ⁻²	-12647	25332	-149880	299812
time ² +time ⁻¹	-12647	25332	-149864	299778
time ² +time	-12647	25332	-149709	299468
time ² +time ³	-12646	25331	-149779	299609
time ³ +time ⁻²	-12649	25336	-149874	299799
time ³ +time ⁻¹	-12649	25336	-149857	299764
time ³ +time	-12647	25333	-149730	299510
time ³ +time ²	-12647	25331	-149779	299609

Appendix 8.13 Crude sex-specific linear and non-linear trajectories of mean weekly alcohol consumption, stratified by T2DM diagnosis: goodness of fit statistics. Observed data.
Chapter 11: Appendices

<u>Women</u>				
time ⁻² +time ⁻¹	-4758	9549	-57152	114351
time ⁻² +time	-4758	9550	-57147	114340
time ⁻² +time ²	-4758	9550	-57154	114354
time ⁻² +time ³	-4758	9550	-57154	114354
1 2				
time ⁻¹ +time ⁻²	-4758	9549	-57152	114351
time ⁻¹ +time	-4758	9549	-57146	114339
time ⁻¹ +time ²	-4758	9549	-57152	114351
time ⁻¹ +time ³	-4758	9549	-57153	114352
time+time ⁻²	-4758	9550	-57147	114340
time+time ⁻¹	-4758	9549	-57146	114339
time+time ²	-4759	9552	-57103	114252
time+time ³	-4759	9552	-57111	114268
time ² +time ⁻²	-4758	9550	-57154	114354
time ² +time ⁻¹	-4758	9549	-57153	114351
time ² +time	-4759	9552	-57103	114252
time ² +time ³	-4759	9552	-57129	114305
time ³ +time ⁻²	-4758	9550	-57154	114354
time ³ +time ⁻¹	-4758	9549	-57153	114352
time ³ +time	-4759	9552	-57111	114268
time ³ +time ²	-4759	9552	-57129	114305

Fit statistics calculated on models with fixed slopes and without robust standard errors due to issues of convergence for some transformations when random slopes were expressed. ^aBayesian information criterion.

Appendix 8.14 Crude sex-specific linear and non-linear trajectories of mean weekly alcohol
consumption, stratified by T2DM diagnosis: goodness of fit statistics. Imputed data.

	T2DM	No T2DM
First-order		
models	Log-likelihood ^a BIC	Log-likelihood ^a BIC
Men		
time ⁻²	-13499 (-13501, -13497) 27029 (27024-27032)	-159549 (-159770, -159370) 319139 (318782-319581)
time ⁻¹	-13499 (-13501, -13497) 27029 (27024-27032)	-159532 (-159754, -159353) 319105 (318747-319549)
time ¹	-13482 (-13483, -13481) 26995 (26992-26997)	-159546 (-159767, -159368) 319133 (318776-319576)
time ²	-13480 (-13482, -13479) 26992 (26988-26994)	-159549 (-159769, -159370) 319138 (318780-319579)
time ³	-13482 (-13484, -13480) 26996 (26992-26998)	-159540 (-159760, -159361) 319121 (318763-319562)
Women		
time ⁻²	-5140 (-5152, -5132) 10307 (10292-10331)	-62040 (-62367, -61827) 124118 (123691-124772)
time ⁻¹		(2000) (2000) (2000) (2000) (2000) (2000) (2000) (2000)
י וווע		-02039 (-02305, -01825) 124115 (12308/-124/08)
time	-5140 (-5152, -5132) 10307 (10291-10331)	-62042 (-62369, -61828) 124121 (123694-124775)
time ²	-5140 (-5152, -5132) 10307 (10291-10331)	-62045 (-62372, -61832) 124128 (123701-124781)
time ³	-5140 (-5152, -5132) 10307 (10292-10331)	-62044 (-62370, -61830) 124126 (123698-124778)
Second-order		
models	Log-likelihood ^a BIC	Log-likelihood ^a BIC
Men		
time ⁻² +time ⁻¹	-13499 (-13501, -13497) 27037 (27032-27040)	-159486 (-159710, -159306) 319023 (318662-319472)
time ⁻² +time	-13482 (-13483, -13481) 27002 (27000-27004)	-159544 (-159766, -159366) 319139 (318782-319582)
time ⁻² +time ²	-13480 (-13481, -13479) 26999 (26996-27002)	-159547 (-159767, -159368) 319145 (318787-319586)
time ⁻² +time ³	-13482 (-13484, -13480) 27003 (26999-27006)	-159538 (-159759, -159359) 319128 (318770-319568)
time ⁻¹ +time ⁻²	-13499 (-13501, -13497) 27037 (27032-27040)	-159486 (-159710, -159306) 319023 (318662-319472)
time ⁻¹ +time	-13482 (-13483, -13480) 27002 (27000-27005)	-159529 (-159751, -159350) 319109 (318752-319554)
time ⁻¹ +time ²	-13480 (-13481, -13479) 26999 (26996-27002)	-159529 (-159750, -159349) 319108 (318749-319551)
time ⁻¹ +time ³	-13482 (-13484, -13480) 27003 (26999-27006)	-159519 (-159741, -159340) 319090 (318731-319532)

ti me+ti me ⁻² ti me+ti me ⁻¹	-13482 (-13483, -13481) 27002 (27000-27004) -13487 (-13483 -13480) 27007 (27000-27005)	-159544 (-159766, -159366) 319139 (318782-319582) -159529 (-159751 -159350) 319109 (318757-319554)
ume-time ² time+time ²	-13480 (-13482, -13479) 26999 (26996-27002)	-159366 (-159592, -159193) 318783 (318437-319234)
ti me+ti me ³	-13481 (-13482, -13479) 27001 (26997-27003)	-159388 (-159615, -159215) 318827 (318482-319280)
time ² +time ⁻²	-13480 (-13481 -13479) 26999 (26996-27002)	-159547 (-159767 -159368) 319145 (318787-319586)
time ² +time ⁻¹	-13480 (-13481, -13479) 26999 (26996-27002)	-159529 (-159750, -159349) 319108 (318749-319551)
time ² +time	-13480 (-13482, -13479) 26999 (26996-27002)	-159366 (-159592, -159193) 318783 (318437-319234)
ti me ² +ti me ³	-13480 (-13481, -13478) 26999 (26995-27001)	-159439 (-159666, -159265) 318929 (318581-319382)
time ³ +time ⁻²	(90025-99997) 50075 (08851- 88851-) 58851-	-159538 (-159759 -159359) 319128 (218770-319568)
time ³ +time ⁻¹	-13482 (-13484 -13480) 27003 (26999-27006)	-159519 (-159741 -159340) 319090 (318731-319532)
time ³ +time	-13481 (-13482, -13479) 27001 (26997-27003)	
ti me ³ +ti me ²	-13480 (-13481, -13478) 26999 (26995-27001)	-159439 (-159666, -159265) 318929 (318581-319382)
Women		
ti me ⁻² +ti me ⁻¹	-5138 (-5149, -5131) 10311 (10297-10333)	-62038 (-62364, -61824) 124122 (123694-124775)
time ⁻² +time	-5140 (-5152, -5132) 10314 (10298-10338)	-62036 (-62364, -61823) 124120 (123692-124775)
ti me ⁻² +ti me ²	-5140 (-5152, -5132) 10314 (10298-10338)	-62040 (-62367, -61826) 124127 (123700-124781)
time ⁻² +time ³	-5140 (-5152, -5132) 10314 (10298-10338)	-62039 (-62365, -61825) 124125 (123697-124778)
time ⁻¹ +time ⁻²	-5138 (-5149, -5131) 10311 (10297-10333)	-62038 (-62364, -61824) 124122 (123694-124775)
time ⁻¹ +time	-5139 (-5150, -5131) 10312 (10296-10335)	-62035 (-62363, -61821) 124117 (123690-124772)
time ⁻¹ +time ²	-5139 (-5150, -5131) 10312 (10296-10335)	-62039 (-62365, -61825) 124124 (123696-124777)

-62037 (-62363, -61824) 124122 (123694-124774)

-5139 (-5151, -5131) 10313 (10296-10336)

time⁻¹+time³

Chapter 11: Appendices

ti me+ti me ⁻²	-5140 (-5152, -5132) 10314 (10298-10338)	-62036 (-62364, -61823) 124120 (123692-124775)
ti me+ti me ⁻¹	-5139 (-5150, -5131) 10312 (10296-10335)	-62035 (-62363, -61821) 124117 (123690-124772)
ti me+ti me ²	-5140 (-5152, -5132) 10314 (10298-10338)	-61987 (-62313, -61776) 124021 (123598-124673)
ti me+ti me ³	-5140 (-5152, -5132) 10314 (10298-10338)	-61997 (-62324, -61785) 124040 (123617-124694)
ti me ² +ti me ⁻²	-5140 (-5152, -5132) 10314 (10298-10338)	-62040 (-62367, -61826) 124127 (123700-124781)
ti me ² +ti me ⁻¹	-5139 (-5150, -5131) 10312 (10296-10335)	-62039 (-62365, -61825) 124124 (123696-124777)
ti me ² +ti me	-5140 (-5152, -5132) 10314 (10298-10338)	-61987 (-62313, -61776) 124021 (123598-124673)
ti me ² +ti me ³	-5140 (-5151, -5132) 10314 (10298-10337)	-62017 (-62344, -61804) 124080 (123654-124735)
ti me ³ +ti me ⁻²	-5140 (-5152, -5132) 10314 (10298-10338)	-62039 (-62365, -61825) 124125 (123697-124778)
ti me ³ +ti me ⁻¹	-5139 (-5151, -5131) 10313 (10296-10336)	-62037 (-62363, -61824) 124122 (123694-124774)
ti me ³ +ti me	-5140 (-5152, -5132) 10314 (10298-10338)	-61997 (-62324, -61785) 124040 (123617-124694)
ti me ³ +ti me ²	-5140 (-5151, -5132) 10314 (10298-10337)	-62017 (-62344, -61804) 124080 (123654-124735)
Fit statistics cal for some transf reported by the	Iculated on models with fixed slopes and without ormations when random slopes were expressed. \ first three imputations.	robust standard errors due to issues of convergence 'alues refer to the mean and range of fit statistics as

Chapter 11: Appendices

Appendix 8.15 Crude, sex-specific and best-fitting trajectories of mean weekly volume of alcohol consumption, stratified by T2DM diagnosis: results. Imputed data.

Crude best-fitting mixed models	g/week (95% CI)	p-value
Men		
T2DM (n=620)		
Intercept	127.1 (116.0, 138.2)	<0.001
Time ¹	15.4 (9.4, 21.3)	<0.001
Censored (n=5,103)		
Intercept	93.4 (90.4, 96.3)	<0.001
Time ¹	-34.9 (-38.6, -31.1)	<0.001
Time ²	-1.3 (-1.5, -1.2)	<0.001
<u>Women</u>		
T2DM (n=296)		
Intercept	30.1 (24.5, 35.7)	<0.001
Time ¹	1.2 (-2.5, 5.0)	0.514
Censored (n=2,274)		
Intercept	42.7 (40.1, 45.3)	<0.001
Time ¹	-17.0 (-20.4, -13.6)	<0.001
Time ²	-0.6 (-0.8, -0.5)	<0.001

Time coefficients refer to the change in the average volume of weekly alcohol consumption per 10 years prior to diagnosis or censoring.

Appendix 8.16 Crude sex-specific best-fitting trajectories of mean weekly volume of alcohol consumption, stratified by T2DM diagnosis: figure. Imputed data.



	Men		Women	
Function	Log-likelihood	^a BIC	Log-likelihood	^a BIC
Up to diagnosis				
time ⁻²	-12665	25361	-4758	9543
time ⁻¹	-12665	25361	-4758	9543
time ¹	-12648	25327	-4759	9545
time ²	-12647	25324	-4759	9545
time ³	-12649	25329	-4759	9545
time ⁻² +time ⁻¹	-12665	25368	-4758	9549
time ⁻² +time	-12648	25334	-4758	9550
time ⁻² +time ²	-12647	25332	-4758	9550
time ⁻² +time ³	-12649	25336	-4758	9550
time⁻¹+time	-12648	25334	-4758	9549
time ⁻¹ +time ²	-12647	25332	-4758	9549
time ⁻¹ +time ³	-12649	25336	-4758	9549
time+time ²	-12647	25332	-4759	9552
time+time ³	-12647	25333	-4759	9552
time ² +time ³	-12646	25331	-4759	9552
After diagnosis				
time ⁻²	-6777	13582	-2721	5468
time ⁻¹	-6775	13578	-2721	5467
time ¹	-6767	13562	-2719	5463
time ²	-6766	13561	-2718	5461
time ³	-6768	13565	-2718	5462
time ⁻² +time ⁻¹	-6769	13573	-2720	5472
time ⁻² +time	-6766	13566	-2718	5469
time ⁻² +time ²	-6766	13568	-2718	5468
time ⁻² +time ³	-6768	13571	-2718	5467
time⁻¹+time	-6766	13566	-2718	5469
time ⁻¹ +time ²	-6766	13568	-2718	5468
time ⁻¹ +time ³	-6768	13571	-2718	5468
time+time ²	-6766	13567	-2718	5468
time+time ³	-6766	13568	-2718	5468
time ² +time ³	-6766	13567	-2718	5468

Appendix 8.17 Crude linear and non-linear trajectories of alcohol consumption up to and beyond the date of diagnosis, stratified by sex: goodness of fit statistics. Observed data.

Fit statistics calculated on models with fixed slopes and without robust standard errors due to issues of convergence for some transformations when random slopes were expressed. ^aBayesian information criterion.

Appendix 8.18 Crude linear and non-linear trajectories of alcohol consumption up to and beyond the date of diagnosis, stratified by sex: goodness of fit statistics. Imputed data.

Cha	apte	er 1:	1: A	ppe	endi	ices					
	aBIC		10331)	10329)	10331)	10331)	10331)	10333)	10338)	10338)	

	Mer		Wor	men
Function	Log-likelihood	aBIC	Log-likelihood	^a BIC
Up to diagnosis				
ti me ⁻²	-13499 (-13501, -13497)	27029 (27024, 27032)	-5140 (-5152, -5132)	10307 (10292, 10331)
ti me ⁻¹	-13499 (-13501, -13497)	27029 (27024, 27032)	-5139 (-5151, -5131)	10306 (10290, 10329)
ti me ¹	-13482 (-13483, -13481)	26995 (26992, 26997)	-5140 (-5152, -5132)	10307 (10291, 10331)
ti me ²	-13480 (-13482, -13479)	26992 (26988, 26994)	-5140 (-5152, -5132)	10307 (10291, 10331)
ti me ³	-13482 (-13484, -13480)	26996 (26992, 26998)	-5140 (-5152, -5132)	10307 (10292, 10331)
ti me ⁻² +ti me ⁻¹	-13499 (-13501, -13497)	27037 (27032, 27040)	-5138 (-5149, -5131)	10311 (10297, 10333)
ti me ⁻² +ti me	-13482 (-13483, -13481)	27002 (27000, 27004)	-5140 (-5152, -5132)	10314 (10298, 10338)
ti me ⁻² +ti me ²	-13480 (-13481, -13479)	26999 (26996, 27002)	-5140 (-5152, -5132)	10314 (10298, 10338)
ti me ⁻² +ti me ³	-13482 (-13484, -13480)	27003 (26999, 27006)	-5140 (-5152, -5132)	10314 (10298, 10338)
ti me ⁻¹ +ti me	-13482 (-13483, -13480)	27002 (27000, 27005)	-5139 (-5150, -5131)	10312 (10296, 10335)
ti me ⁻¹ +ti me ²	-13480 (-13481, -13479)	26999 (26996, 27002)	-5139 (-5150, -5131)	10312 (10296, 10335)
ti me ⁻¹ +ti me ³	-13482 (-13484, -13480)	27003 (26999, 27006)	-5139 (-5151, -5131)	10313 (10296, 10336)
ti me+ti me ²	-13480 (-13482, -13479)	26999 (26996, 27002)	-5140 (-5152, -5132)	10314 (10298, 10338)
ti me+ti me ³	-13481 (-13482, -13479)	27001 (26997, 27003)	-5140 (-5152, -5132)	10314 (10298, 10338)
ti me ² +ti me ³	-13480 (-13481, -13478)	26999 (26995, 27001)	-5140 (-5151, -5132)	10314 (10298, 10337)

After diagnosis				
time ⁻²	-8746 (-8862, -8664)	17521 (17357, 17753)	-3806 (-3880, -3724)	7638 (7474, 7786)
time ⁻¹	-8744 (-8860, -8662)	17517 (17354, 17749)	-3808 (-3888, -3723)	7643 (7473, 7804)
time ¹	-8736 (-8853, -8657)	17501 (17342, 17735)	-3809 (-3894, -3722)	7645 (7470, 7814)
time ²	-8736 (-8853, -8657)	17501 (17343, 17735)	-3809 (-3894, -3722)	7645 (7470, 7815)
time ³	-8738 (-8854, -8659)	17505 (17347, 17738)	-3810 (-3895, -3722)	7646 (7471, 7816)
time ⁻² +time ⁻¹	-8738 (-8856, -8658)	17513 (17352, 17748)	-3799 (-3862, -3723)	7631 (7479, 7757)
time ⁻² +time	-8735 (-8853, -8656)	17507 (17348, 17742)	-3804 (-3879, -3722)	7641 (7477, 7790)
time ⁻² +time ²	-8736 (-8853, -8657)	17508 (17351, 17742)	-3804 (-3880, -3721)	7642 (7476, 7793)
time ⁻² +time ³	-8737 (-8854, -8659)	17511 (17353, 17744)	-3804 (-3880, -3721)	7642 (7476, 7793)
time ⁻¹ +time	-8735 (-8852, -8656)	17507 (17348, 17741)	-3807 (-3888, -3722)	7647 (7477, 7809)
time ⁻¹ +time ²	-8736 (-8853, -8657)	17508 (17350, 17742)	-3807 (-3888, -3721)	7648 (7476, 7810)
time ⁻¹ +time ³	-8737 (-8854, -8658)	17510 (17353, 17744)	-3807 (-3888, -3721)	7648 (7476, 7810)
time+time ²	-8735 (-8853, -8656)	17507 (17349, 17742)	-3809 (-3894, -3722)	7651 (7477, 7820)
time+time ³	-8735 (-8853, -8657)	17507 (17350, 17742)	-3809 (-3894, -3722)	7651 (7477, 7821)
time ² +time ³	-8735 (-8852, -8656)	17506 (17348, 17741)	-3809 (-3894, -3722)	7652 (7477, 7822)
Fit statistics calcul	ated on models with fixed	d slopes and without robu	st standard errors due to i	ssues of
convergence for sol	ne transformations wher	random slopes were expr	essed. Values refer to the	mean and range of fit
statistics as report	ed by the first three impu	tations. ^a Bayesian inform	ation criterion.	

36) 24) 14) 14) 16) 30) 33) 33)

Chapter 11: Appendices

Appendix 8.19 Crude trajectories of mean weekly volume of alcohol consumption up to and beyond the date of diagnosis, stratified by sex. Imputed data.



Appendix 8.20 Crude sex-specific interaction between the trajectory of mean weekly volume of alcohol consumption and baseline alcohol consumption category, stratified by T2DM diagnosis: results. Imputed data.

	T2DM		Censored	
Crude linear mixed models	g/week (95% CI)	p-value	g/week (95% Cl)	p-value
Men				
Difference in consumption at the time of diagnosis or censoring by baseline consumption category				
Non-drinker	Reference		Reference	
Infrequent drinker	24.9 (5.3, 44.5)	0.013	27.7 (21.1, 34.3)	<0.001
0.1-50.0 g/week	59.4 (40.8, 78.1)	<0.001	50.0 (44.9, 55.0)	<0.001
50.1-100.0 g/week	122.3 (97.9, 146.7)	<0.001	89.6 (83.1, 96.2)	<0.001
100.1-150.0 g/week	160.6 (126.2, 195.1)	<0.001	120.9 (112.9, 129.0	<0.001
>150.0 g/week	209.1 (176.0, 242.2)	<0.001	184.2 (175.0, 193.5)	<0.001
Difference in the rate of change by				
baseline consumption category ^a				
Non-drinker	Reference		Reference	
Infrequent drinker	10.8 (1.4, 20.1)	0.024	7.4 (5.0, 9.8)	<0.001
0.1-50.0 g/week	14.4 (5.1, 23.7)	0.002	5.1 (3.1, 7.0)	<0.001
50.1-100.0 g/week	25.1 (12.9, 37.3)	<0.001	3.3 (0.7, 6.0)	0.012
100.1-150.0 g/week	24.1 (6.0, 42.2)	0.009	-3.3 (-6.4, -0.3)	0.034
>150.0 g/week	-27.5 (-45.8, -9.3)	0.003	-30.0 (-33.9, -26.4)	<0.001
Women Difference in consumption at the time of diagnosis or censoring by baseline consumption category				
Non-drinker	Reference		Reference	
Infrequent drinker	14.4 (7.4, 21.4)	<0.001	14.0 (10.0, 18.0)	<0.001
0.1-50.0 g/week	28.5 (21.1, 35.9)	<0.001	29.5 (25.6, 33.4)	<0.001
50.1-100.0 g/week	39.6 (24.4, 54.6)	<0.001	68.8 (62.3, 75.4)	<0.001
>100.0 g/week	102.0 (65.6, 138.5)	<0.001	106.2 (95.5, 116.9)	<0.001
Difference in the rate of change by				
baseline consumption category ^a				
Non-drinker	Reference		Reference	
Infrequent drinker	8.3 (4.0, 12.7)	<0.001	3.6 (1.8, 5.4)	<0.001
0.1-50.0 g/week	4.1 (-0.2, 8.4)	0.064	-0.2 (-1.9, 1.5)	0.849
50.1-100.0 g/week	-17.4 (-25.0, -9.7)	<0.001	-2.5 (-5.3, 0.2)	0.071
>100.0 g/week	-25.0 (-52.4, 2.4)	0.074	-20.4 (-24.7, 16.1)	<0.001

^aRate of change per 10 years prior to diagnosis or censoring.

Appendix 8.21 Crude male trajectories of mean weekly volume of alcohol consumption, stratified by baseline alcohol consumption category and T2DM diagnosis: figure. Imputed data.









Appendix 8.23 Crude sex-specific best-fitting trajectory of mean weekly volume of alcohol consumption, stratified by T2DM diagnosis and excluding non-drinkers. Observed data.





Appendix 8.24 Crude sex-specific and best-fitting trajectory of mean weekly volume of alcohol consumption, stratified by T2DM diagnosis: a competing risk sensitivity analysis. Observed data.





Chapter 11: Appendices

11.5 Appendices for Chapter 9

Appendix 9.1 Alternative distributional functions of the cumulative baseline hazard



Men 40 Cumulative hazard (%) 30 20 10 0 70 50 60 40 80 Age (years) Cox (unspecified) ----- Exponential ---- Gompertz ----- Weibull

	Men		Wome	n
Alcohol consumption (wave 3)	HR (95% CI)	p-value	HR (95% CI)	p-value
Model 1				
Cases/non-cases		560/4,867		268/2,155
Consumption volume				
g/week (log ₂)	1.04 (0.98, 1.12)	0.211	0.74 (0.65, 0.84)	<0.001
6) Week (1062)	1.0 (0.50) 1.12)	0.211		.0.001
Consumption category				
Current drinker	(reference)		(reference)	
Infrequent drinker	1.25 (0.75, 2.08)	0.385	0.48 (0.24, 0.95)	0.036
Non-current drinker	1.74 (0.83, 3.67)	0.142	0.46 (0.19, 1.11)	0.084
Never drinker	2.39 (1.24, 4.60)	0.009	0.42 (0.18, 0.96)	0.040
Log likelihood	-1693 (-	-1693, -1693)	-742	7, (-747, -747)
BIC ^a	3437	(3437, 3437)	1542	(1542, 1532)
Model 2				
Cases/non-cases		560/4,867		268/2,155
Consumption volume				
g/week (log ₂)	1.00 (0.93, 1.06)	0.888	0.81 (0.71, 0.93)	0.003
Consumption category				
Current drinker	(reference)		(reference)	
Infrequent drinker	0.94 (0.57, 1.54)	0.805	0.61 (0.30, 1.22)	0.160
Non-current drinker	1.16 (0.57, 2.37)	0.686	0.42 (0.18, 1.00)	0.050
Never drinker	0.96 (0.49, 1.88)	0.911	0.38 (0.16, 0.90)	0.028
Log likelihood	-1507 (-	1508, -1505)	-63	9 (-640, -638)
BIC ^a	3177	(3173, 3180)	1427	(1425, 1428)

Appendix 9.2 Relationship between the average weekly volume of alcohol consumption and the risk of T2DM, stratified by sex. Conventional survival analysis, imputed data.

Model 1 reported the dose-response relationship between the volume alcohol consumption and T2DM following adjustment for consumption category. Model 2 included additional adjustment for BMI, date of birth, employment status, ethnicity, family history of T2DM, occupational grade, physical activity and smoking status, as defined at baseline. Ethnicity was derived from responses at waves one and five. ^aBayesian information criterion. Fit statistics refer to the mean and range of values reported from the first three imputations.

Appendix 9.3 Relationship between the average weekly volume of alcohol consumption and the risk of T2DM, stratified by sex. Age-varying covariate survival analysis, imputed data.

	Men		Women	
Alcohol consumption	HR (95% CI)	p-value	HR (95% CI)	p-value
Model 1				
Cases/non-cases		560/4,867		268/2,155
Consumption volume				
g/week (log ₂)	1.11 (1.03, 1.19)	0.004	0.84 (0.74, 0.95)	0.005
Consumption category				
Current drinker	(reference)		(reference)	
Infrequent drinker	2.35 (1.37, 4.05)	0.002	0.96 (0.48, 1.91)	0.906
Non-current drinker	2.07 (0.98, 4.37)	0.057	1.15 (0.53, 2.48)	0.722
Never drinker	3.91 (1.95, 7.84)	<0.001	0.67 (0.26, 1.67)	0.387
Log likelihood	-1673 (-1673, -1673)		-734 (-734, -734)	
BIC ^a	3406 (3406, 3406)		1523 (1523, 1523)	
Model 2				
Cases/non-cases		560/4,867		268/2,155
Consumption volume				
g/week (log ₂)	1.07 (1.00, 1.15)	0.053	0.93 (0.82, 1.05)	0.227
Consumption category				
Current drinker	(reference)		(reference)	
Infrequent drinker	1 68 (0 99 2 86)	0.054	1 19 (0 60 2 34)	0.623
Non-current drinker	1.19 (0.56, 2.54)	0.654	1.27 (0.59, 2.72)	0.536
Never drinker	1.67 (0.84, 3.33)	0.146	0.67 (0.28, 1.64)	0.386
		0.2.0	(0.000
Log likelihood	-1456 (-	1456, -1455)	-620 (-621, -620)	
BIC ^a	3101	(3099, 3102)	1414	(1412, 1415)

Model 1 reported the dose-response relationship between the volume alcohol consumption and T2DM following adjustment for consumption category. Model 2 included additional adjustment for BMI, date of birth, employment status, ethnicity, family history of T2DM, occupational grade, physical activity and smoking status. Ethnicity was derived from responses at waves one and five. ^aBayesian information criterion. Fit statistics refer to the mean and range of values reported from the first three imputations.

Appendix 9.4 Relationship between the average weekly volume of alcohol consumption and the risk of T2DM, stratified by sex. Two-stage survival analysis, imputed data.

	Men		Women	
Alcohol consumption	HR (95% CI)	p-value	HR (95% CI)	p-value
Model 1				
Cases/non-cases		560/4,867		268/2,155
Consumption volume				
g/week (log ₂)	1.05 (0.98, 1.11)	0.151	0.82 (0.76, 0.89)	<0.001
Consumption category				
Current drinker	(reference)		(reference)	
Infrequent drinker	1.41 (1.01, 1.97)	0.045	1.29 (0.91, 1.84)	0.155
Non-current drinker	1.31 (0.66, 2.58)	0.440	1.35 (0.81, 2.25)	0.253
Never drinker	2.52 (1.36, 4.67)	0.003	0.75 (0.36, 1.56)	0.445
Log likelihood	-1677 (-1677, -1677)		-730 (-730730)	
BIC ^a	3413 (3413, 3413)		1514 (1514, 1514)	
Model 2				
Cases/non-cases		560/4,867		268/2,155
Consumption volume				
g/week (log ₂)	1.04 (0.98, 1.10)	0.209	0.92 (0.85, 1.00)	0.056
Consumption category				
Current drinker	(reference)		(reference)	
Infrequent drinker	1.25 (0.90, 1.74)	0.189	1.38 (0.97, 1.96)	0.073
Non-current drinker	0.93 (0.46, 1.85)	0.831	1.41 (0.84, 2.35)	0.190
Never drinker	1.33 (0.72, 2.45)	0.370	0.74 (0.37, 1.50)	0.404
Log likelihood	-1457 (-:	1458, -1456)	-620 (-620619)	
BIC ^a	3103 ((3101, 3104)	1412	(1411, 1413)

Model 1 reported the dose-response relationship between the volume alcohol consumption and T2DM following adjustment for consumption category. Model 2 included additional adjustment for BMI, date of birth, employment status, ethnicity, family history of T2DM, occupational grade, physical activity and smoking status. Ethnicity was derived from responses at waves one and five. ^aBayesian information criterion. Fit statistics refer to the mean and range of values reported from the first three imputations.

Appendix 9.5 Multivariable-adjusted relationship between intercept and current value parameterisations of average weekly volume of alcohol consumption and the risk of T2DM, stratified by sex. Shared random effects survival analysis, time-to-event timescale, observed data.

	Men (n=4,793)		Women (n=2,053)	
Parameterisation	HR (95% CI)	p-value	HR (95% CI)	p-value
Intercept value				
Consumption volume				
g/week (log ₂)	1.02 (0.95, 1.09)		0.90 (0.80, 1.02)	0.111
Consumption category				
Current drinker	(reference)			
Infrequent drinker	1.14 (0.77, 1.69)		1.31 (0.80, 1.02)	0.250
Non-current drinker	0.96 (0.47, 1.93)		1.35 (0.70, 2.59)	0.369
Never drinker	1.03 (0.49, 2.16)		0.67 (0.27, 1.66)	0.385
Log likelihood		-38075		-15914
BIC ^a		76384		32039
Current value				
Consumption volume				
g/week (log ₂)	1.05 (0.97, 1.13)	0.258	0.95 (0.83, 1.08)	0.411
Consumption category				
Current drinker	(reference)		(reference)	
Infrequent drinker	1.28 (0.83, 1.97)	0.268	1.47 (0.89, 2.43)	0.135
Non-current drinker	1.10 (0.53, 2.30)	0.796	1.54 (0.77, 3.10)	0.221
Never drinker	1.19 (0.55, 2.55)	0.664	0.77 (0.30, 1.97)	0.584
Log likelihood		-38074		-15915
BIC ^a		76383		32041

All models included adjustment for consumption category as well as BMI, date of birth, employment status, ethnicity, family history of T2DM, occupational grade, physical activity and smoking status. Ethnicity was derived from responses at waves one and five. ^aBayesian information criterion.

Appendix 9.6 Multivariable-adjusted associations between conditional intercept and current value parameterisations of average weekly volume of alcohol consumption and the risk of T2DM, stratified by sex. Shared random effects survival analysis, time-to-event timescale, observed data.

	Men (n=4,793)		Women (n=2.053)	
Demonstration				
Parameterisation	HR (95% CI)	p-value	HR (95% CI)	p-value
Later and so the				
<u>Intercept value</u>				
Consumption volume				
g/week (log ₂)	0.83 (0.67, 1.02)	0.075	0.32 (0.15, 0.72)	0.005
Current value				
g/week (log ₂)	1.28 (1.01, 1.63)	0.042	3.05 (1.33, 6.96)	0.008
Consumption category				
Current drinker	(reference)		(reference)	
Infrequent drinker	1.41 (0.90, 2.20)	0.134	2.06 (1.16, 3.64)	0.013
Non-current drinker	1.26 (0.59, 2.68)	0.553	2.28 (1.06, 4.89)	0.035
Never drinker	3.52 (1.80, 14.64)	0.547	0.97 (0.36, 2.62)	0.957
Log likelihood		-38073		-15909
BIC ^a		76389		32038

All models included adjustment for consumption category as well as BMI, date of birth, employment status, ethnicity, family history of T2DM, occupational grade, physical activity and smoking status. Ethnicity was derived from responses at waves one and five. ^aBayesian information criterion.

Appendix 9.7 Multivariable-adjusted relationship between the rate of change in the average weekly volume of alcohol consumption and the risk of T2DM, stratified by sex and adjusted for intake at the intercept. Shared random effects survival analysis, observed data.

	Men (n=4,793)		Women (n=2,053)	
Parameterisation	HR (95% CI)	p-value	HR (95% CI)	p-value
Intercept value				
Consumption volume				
g/week (log ₂)	1.14 (1.03, 1.26)	0.010	0.96 (0.83, 1.11)	0.581
Consumption category				
Current drinker	(reference)		(reference)	
Infrequent drinker	1.69 (1.08, 2.64)	0.023	1.80 (1.05, 3.10)	0.034
Non-current drinker	1.97 (0.88, 4.42)	0.100	2.00 (0.93, 4.29)	0.077
Never drinker	1.83 (0.76, 4.40)	0.175	0.84 (0.31, 2.29)	0.730
Slope				
5% increase in the rate of ch	<u>nange</u>			
g/week (log ₂)	2.42 (1.65, 3.56)	<0.001	2.89 (1.28, 6.54)	0.011
5% decrease in rate of chan	ge			
g/week (log ₂)	0.42 (0.29, 0.61)	<0.001	0.77 (0.63, 0.94)	0.011
Log likelihood		-38285		-15918
BIC ^a		76804		32048

All models included adjustment for consumption category as well as BMI, date of birth, employment status, ethnicity, family history of T2DM, occupational grade, physical activity and smoking status. Ethnicity was derived from responses at waves one and five. ^aBayesian information criterion.

Chapter 12: Reference list

Chapter 12 Reference list

12 Reference list

1 Sarwar N, Gao P, Seshasai SR, Gobin R, Kaptoge S, Di Angelantonio E, et al. Diabetes mellitus, fasting blood glucose concentration, and risk of vascular disease: a collaborative meta-analysis of 102 prospective studies. Lancet. 2010; 375(9733): 2215–22.

2 Shah AD, Langenberg C, Rapsomaniki E, Denaxas S, Pujades-Rodriguez M, Gale CP, et al. Type 2 diabetes and incidence of cardiovascular diseases: a cohort study in 1.9 million people. Lancet Diabetes Endocrinol. 2015; 3(2): 105–113.

3 Vinik AI, Maser RE, Mitchell BD, Freeman R. Diabetic autonomic neuropathy. Diabetes Care. 2003; 26(5): 1553–79.

4 Kanavos P, van den Aardweg S, Schurer W. Diabetes expenditure, burden of disease and management in 5 EU countries. [Internet]. LSE Health, London School of Economics; 2012. Available from:

http://www.lse.ac.uk/LSEHealthAndSocialCare/research/LSEHealth/MTRG/LSEDiabetesReport 26Jan2012.pdf

5 The Health and Social Care Information Centre. Health Survey for England – 2014: Trend Tables. Leeds: The Health and Social Care Information Centre; 2015.

6 The Health and Social Care Information Centre. QOF 2014-15: Prevalence, achievements and exceptions at region and nation level V1.1. Leeds: The Health and Social Care Information Centre; 2015.

7 Sharma M, Nazareth I, Petersen I. Trends in incidence, prevalence and prescribing in type 2 diabetes mellitus between 2000 and 2013 in primary care: a retrospective cohort study. BMJ Open. 2016; 6(1): e010210.

8 Department of Health. National Service Framework for Diabetes. London: Department of Health; 2001.

9 Public Health England. A systematic review and meta-analysis assessing the effectiveness of pragmatic lifestyle interventions for the prevention of type 2 diabetes mellitus in routine practice. London: Public Health England; 2015.

10 Baliunas DO, Taylor BJ, Irving H, Roerecke M, Patra J, Mohapatra S, et al. Alcohol as a Risk Factor for Type 2 Diabetes A systematic review and meta-analysis. Dia Care. 2009; 32(11): 2123–32.

11 Drinkaware. Unit and Calorie Calculator [Internet]; 2014. Available from: https://www.drinkaware.co.uk/understand-your-drinking/unit-calculator

12 Maahs DM, West NA, Lawrence JM, Mayer-Davis EJ. Chapter 1: Epidemiology of Type 1 Diabetes. Endocrinol Metab Clin North Am. 2010; 39(3): 481–97.

13 Feener EP, King GL. Vascular dysfunction in diabetes mellitus. Lancet. 1997; 350 Suppl 1: SI9–13.

14 Nathan DM, Meigs J, Singer DE. The epidemiology of cardiovascular disease in type 2 diabetes mellitus: how sweet it is ... or is it? Lancet. 1997; 350 Suppl 1: SI4–9.

15 Mazzone T, Chait A, Plutzky J. Cardiovascular disease risk in type 2 diabetes mellitus: insights from mechanistic studies. Lancet. 2008; 371(9626): 1800–9.

16 Morrish NJ, Wang SL, Stevens LK, Fuller JH, Keen H. Mortality and causes of death in the WHO Multinational Study of Vascular Disease in Diabetes. Diabetologia. 2001; 44 Suppl 2: S14-21.

17 Kannel WB, McGee DL. Diabetes and glucose tolerance as risk factors for cardiovascular disease: the Framingham study. Diabetes Care. 1979; 2(2): 120–6.

18 Hillen T, Coshall C, Tilling K, Rudd AG, McGovern R, Wolfe CD; South London Stroke Register. Cause of stroke recurrence is multifactorial: patterns, risk factors, and outcomes of stroke recurrence in the South London Stroke Register. Stroke. 2003; 34(6): 1457–63.

19 Hier DB, Foulkes MA, Swiontoniowski M, Sacco RL, Gorelick PB, Mohr JP, et al. Stroke recurrence within 2 years after ischemic infarction. Stroke. 1991; 22(2): 155–61.

20 Sacco RL, Shi T, Zamanillo MC, Kargman DE. Predictors of mortality and recurrence after hospitalized cerebral infarction in an urban community: the Northern Manhattan Stroke Study. Neurology. 1994; 44(4): 26–34.

21 Elneihoum AM, Göransson M, Falke P, Janzon L. Three-year survival and recurrence after stroke in Malmö, Sweden: an analysis of stroke registry data. Stroke. 1998; 29(10): 2114–7.

22 Megherbi SE, Milan C, Minier D, Couvreur G, Osseby GV, Tilling K, et al. Association between diabetes and stroke subtype on survival and functional outcome 3 months after stroke: data from the European BIOMED Stroke Project. Stroke. 2003; 34(3): 688–94.

23 Luchsinger JA, Tang MX, Stern Y, Shea S, Mayeux R. Diabetes mellitus and risk of Alzheimer's disease and dementia with stroke in a multiethnic cohort. Am J Epidemiol. 2001; 154(7): 635–41.

24 Pendlebury ST, Rothwell PM. Prevalence, incidence, and factors associated with pre-stroke and post-stroke dementia: a systematic review and meta-analysis. Lancet Neurol. 2009; 8(11): 1006–18.

25 Hex N, Bartlett C, Wright D, Taylor M, Varley D. Estimating the current and future costs of Type 1 and Type 2 diabetes in the UK, including direct health costs and indirect societal and productivity costs. Diabet Med. 2012; 29(7): 855–62.

26 Harker R. NHS funding and expenditure SN/SG/724. London: House of Commons Library; 2012. Available at: http://www.parliament.uk/briefing-papers/SN00724.pdf

27 The Economist Intelligence Unit. The Silent Epidemic. An Economic Study of Diabetes in Developed and Developing Countries. London: The Economist; 2007.

28 Crawford R, Emmerson C. NHS and social care funding: the outlook to 2021/22. London: Nuffield Trust; 2012.

29 Roberts A, Marshall L, Charlesworth A. A decade of austerity? The funding pressures facing the NHS from 2010/11 to 2021/22. London: Nuffield Trust; 2012.

30 Appleby J. UK's health and social care spending plans: more of the same? BMJ. 2015; 351: h6458.

31 The Health and Social Care Information Centre. Health Survey for England, 2014: Trend tables - Population number estimates tables. Leeds: The Health and Social Care Information Centre; 2015.

32 Green A, Støvring H, Andersen M, Beck-Nielsen H. The epidemic of type 2 diabetes is a statistical artefact. Diabetologia. 2005; 48(8): 1456–8.

33 Blak BT, Thompson M, Dattani H, Bourke A. Generalisability of The Health Improvement Network (THIN) database: demographics, chronic disease prevalence and mortality rates. Inform Prim Care. 2011; 19(4): 251–5.

34 Lewis JD, Schinnar R, Bilker WB, Wang X, Strom BL. Validation studies of the health improvement network (THIN) database for pharmacoepidemiology research. Pharmacoepidemiol Drug Saf. 2007; 16(4): 393–401.

35 González ELM, Johansson S, Wallander M-A, Rodríguez LAG. Trends in the prevalence and incidence of diabetes in the UK: 1996–2005. J Epidemiol Community Health 2009; 63(4): 332–6.

36 Ardisson Korat AV, Willett WC, Hu FB. Diet, lifestyle, and genetic risk factors for type 2 diabetes: a review from the Nurses' Health Study, Nurses' Health Study 2, and Health Professionals' Follow-up Study. Curr Nutr Rep. 2014; 3(4): 345–354.

37 Stringhini S, Tabak AG, Akbaraly TN, Sabia S, Shipley MJ, Marmot MG, et al. Contribution of modifiable risk factors to social inequalities in type 2 diabetes: prospective Whitehall II cohort study. BMJ. 2012; 345: e5452.

38 Hardoon SL, Morris RW, Thomas MC, Wannamethee SG, Lennon LT, Whincup PH. Is the Recent Rise in Type 2 Diabetes Incidence From 1984 to 2007 Explained by the Trend in Increasing BMI? Evidence from a prospective study of British men. Dia Care. 2010; 33(7): 1494–6.

39 Kodama S, Horikawa C, Fujihara K, Heianza Y, Hirasawa R, Yachi Y, et al. Comparisons of the strength of associations with future type 2 diabetes risk among anthropometric obesity indicators, including waist-to-height ratio: a meta-analysis. Am J Epidemiol. 2012; 176(11): 959–69.

40 Kahn BB, Flier JS. Obesity and insulin resistance. J Clin Invest. 2000; 106(4): 473–481.

41 Slavin JL, Martini MC, Jacobs DR Jr, Marquart L. Plausible mechanisms for the protectiveness of whole grains. Am J Clin Nutr. 1999; 70(3 Suppl): 459S-463S.

42 Anderson JW, Bryant CA. Dietary fiber: diabetes and obesity. Am J Gastroenterol. 1986; 81(10): 898–906.

43 Greenwood DC, Threapleton DE, Evans CEL, Cleghorn CL, Njkjaer C, Woodhead C, et al. Glycemic Index, Glycemic Load, Carbohydrates, and Type 2 Diabetes: Systematic review and dose–response meta-analysis of prospective studies. Diabetes Care. 2013; 36(12): 4166–4171.

44 de Souza RJ, Mente A, Maroleanu A, Cozma AI, Ha V, Kishibe T, et al. Intake of saturated and trans unsaturated fatty acids and risk of all cause mortality, cardiovascular disease, and type 2 diabetes: systematic review and meta-analysis of observational studies. BMJ. 2015; 351: h3978.

45 Hamman RF. Genetic and environmental determinants of non-insulin-dependent diabetes mellitus (NIDDM). Diabetes Metab Rev. 1992; 8(4): 287–338.

46 Wallin A, Di Giuseppe D, Orsini N, Patel PS, Forouhi NG, Wolk A. Fish consumption, dietary long-chain n-3 fatty acids, and risk of type 2 diabetes: systematic review and meta-analysis of prospective studies. Diabetes Care. 2012; 35(4): 918–29.

47 Schisterman EF, Cole SR, Platt RW. Overadjustment bias and unnecessary adjustment in epidemiologic studies. Epidemiology. 2009; 20(4): 488–95.

48 Scarborough P, Rayner M, van Dis I, Norum K. Meta-analysis of effect of saturated fat intake on cardiovascular disease: overadjustment obscures true associations. Am J Clin Nutr. 2010; 92(2): 458–9.

49 Weijnen CF, Rich SS, Meigs JB, Krolewski AS, Warram JH. Risk of diabetes in siblings of index cases with Type 2 diabetes: implications for genetic studies. Diabet Med. 2002; 19(1): 41–50.

50 Kaprio J, Tuomilehto J, Koskenvuo M, Romanov K, Reunanen A, Eriksson J, et al. Concordance for type 1 (insulin-dependent) and type 2 (non-insulin-dependent) diabetes mellitus in a population-based cohort of twins in Finland. Diabetologia. 1992; 35(11): 1060–7.

51 Committee on Diabetic Twins, Japan Diabetes Society. Diabetes mellitus in twins: a cooperative study in Japan. Diabetes Res Clin Pract. 1988; 5(4): 271–80.

52 Medici F, Hawa M, Ianari A, Pyke DA, Leslie RD. Concordance rate for type II diabetes mellitus in monozygotic twins: actuarial analysis. Diabetologia. 1999; 42(2):146–50.

53 Billings LK, Florez JC. The genetics of type 2 diabetes: what have we learned from GWAS? Ann N Y Acad Sci. 2010; 1212: 59-77.

54 Gaulton KJ, Ferreira T, Lee Y, Raimondo A, Mägi R, Reschen ME, et al. Genetic fine mapping and genomic annotation defines causal mechanisms at type 2 diabetes susceptibility loci. Nat Genet. 2015; 47(12): 1415–25.

55 Hara K, Shojima N, Hosoe J, Kadowaki T. Genetic architecture of type 2 diabetes. Biochem Biophys Res Commun 2014; 452(2): 213–20.

56 Fuchsberger C, Flannick J, Teslovich TM, Mahajan A, Agarwala V, Gaulton KJ, et al. The genetic architecture of type 2 diabetes. Nature. 2016; 536(7614): 41–7.

57 Kwak SH, Park KS. Genetic Studies on Diabetic Microvascular Complications: Focusing on Genome-Wide Association Studies. Endocrinol Metab (Seoul) 2015; 30(2): 147–158. 58 Dudbridge F. Polygenic Epidemiology. Genet Epidemiol. 2016; 40(4): 268–272.

59 van Dongen J, Slagboom PE, Draisma HH, Martin NG, Boomsma DI. The continuing value of twin studies in the omics era. Nat Rev Genet. 2012; 13(9): 640–53.

60 Jeon CY, Lokken RP, Hu FB, van Dam RM. Physical activity of moderate intensity and risk of type 2 diabetes: a systematic review. Diabetes Care 2007; 30(3): 744–52.

61 Aune D, Norat T, Leitzmann M, Tonstad S, Vatten LJ. Physical activity and the risk of type 2 diabetes: a systematic review and dose-response meta-analysis. Eur J Epidemiol. 2015; 30(7): 529-42.

62 Mozaffarian D, Hao T, Rimm EB, Willett WC, Hu FB. Changes in diet and lifestyle and longterm weight gain in women and men. N Engl J Med. 2011; 364(25): 2392–404.

63 Rana JS, Li TY, Manson JE, Hu FB. Adiposity compared with physical inactivity and risk of type 2 diabetes in women. Diabetes Care. 2007; 30(1): 53–8.

64 Mandroukas K, Krotkiewski M, Hedberg M, Wroblewski Z, Björntorp P, Grimby G. Physical training in obese women. Effects of muscle morphology, biochemistry and function. Eur J Appl Physiol Occup Physiol. 1984; 52(4): 355–61.

65 Prior SJ, McKenzie MJ, Joseph LJ, Ivey FM, Macko RF, Hafer-Macko CE, et al. Reduced skeletal muscle capillarization and glucose intolerance. Microcirculation. 2009; 16(3): 203–12.

66 Dela F, Handberg A, Mikines KJ, Vinten J, Galbo H. GLUT 4 and insulin receptor binding and kinase activity in trained human muscle. J Physiol. 1993; 469: 615–24.

67 Vinten J, Petersen N, Sonne B, Galbo H. Effect of physical training on glucose transporters in fat cell fractions. Biochim Biophys Acta. 1985; 841(2): 223–7.

68 Stallknecht B, Andersen PH, Vinten J, Bendtsen LL, Sibbersen J, Pedersen O, et al. Effect of physical training on glucose transporter protein and mRNA levels in rat adipocytes. Am J Physiol. 1993; 265(1 Pt 1): E128-34.

69 Goodyear LJ, Kahn BB. Exercise, glucose transport, and insulin sensitivity. Annu Rev Med. 1998; 49: 235–61.

70 Mikines KJ, Sonne B, Farrell PA, Tronier B, Galbo H. Effect of physical exercise on sensitivity and responsiveness to insulin in humans. Am J Physiol. 1988; 254(3 Pt 1): E248–59.

71 Richter EA, Mikines KJ, Galbo H, Kiens B. Effect of exercise on insulin action in human skeletal muscle. J Appl Physiol (1985). 1989; 66(2): 876–85.

72 Mayer-Davis EJ, D'Agostino R Jr, Karter AJ, Haffner SM, Rewers MJ, Saad M, et al. Intensity and amount of physical activity in relation to insulin sensitivity: the Insulin Resistance Atherosclerosis Study. JAMA. 1998; 279(9): 669–74. 73 Pan A, Wang Y, Talaei M, Hu FB, Wu T. Relation of active, passive, and quitting smoking with incident type 2 diabetes: a systematic review and meta-analysis. Lancet Diabetes Endocrinol. 2015; 3(12): 958–67.

74 Pham NM, Nguyen CT, Binns CW, Lee AH. Non-linear association between smoking cessation and incident type 2 diabetes. Lancet Diabetes Endocrinol. 2015; 3(12): 932.

75 Audrain-McGovern J, Benowitz N. Cigarette Smoking, Nicotine, and Body Weight. Clin Pharmacol Ther. 2011; 90(1): 164–168.

76 Aubin HJ, Farley A, Lycett D, Lahmek P, Aveyard P. Weight gain in smokers after quitting cigarettes: meta-analysis. BMJ. 2012; 345: e4439.

77 Freathy RM, Kazeem GR, Morris RW, Johnson PC, Paternoster L, Ebrahim S, et al. Genetic variation at CHRNA5-CHRNA3-CHRNB4 interacts with smoking status to influence body mass index. Int J Epidemiol. 2011; 40(6): 1617–28.

78 Chiolero A, Faeh D, Paccaud F, Cornuz J. Consequences of smoking for body weight, body fat distribution, and insulin resistance. Am J Clin Nutr. 2008; 87(4): 801–9.

79 Attvall S, Fowelin J, Lager I, Von Schenck H, Smith U. Smoking induces insulin resistance--a potential link with the insulin resistance syndrome. J Intern Med. 1993; 233(4): 327–32.

80 Facchini FS, Hollenbeck CB, Jeppesen J, Chen YD, Reaven GM. Insulin resistance and cigarette smoking. Lancet. 1992; 339(8802): 1128–30.

81 Rönnemaa T, Rönnemaa EM, Puukka P, Pyörälä K, Laakso M. Smoking is independently associated with high plasma insulin levels in nondiabetic men. Diabetes Care. 1996; 19(11): 1229–32.

82 Eliasson B, Attvall S, Taskinen MR, Smith U. Smoking cessation improves insulin sensitivity in healthy middle-aged men. Eur J Clin Invest. 1997; 27(5): 450–56

83 Bergman BC, Perreault L, Hunerdosse D, Kerege A, Playdon M, Samek AM, et al. Novel and reversible mechanisms of smoking-induced insulin resistance in humans. Diabetes. 2012; 61(12): 3156–66.

84 Li XH, Yu FF, Zhou YH, He J. Association between alcohol consumption and the risk of incident type 2 diabetes: a systematic review and dose-response meta-analysis. Am J Clin Nutr. 2016; 103(3): 818–29.

85 Holmes MV, Dale CE, Zuccolo L, Silverwood RJ, Guo Y, Ye Z, et al. Association between alcohol and cardiovascular disease: Mendelian randomisation analysis based on individual participant data. BMJ. 2014; 349: g4164.

86 Smith GD, Ebrahim S. Mendelian randomization: prospects, potentials, and limitations. Int J Epidemiol. 2004; 33(1): 30–42.

87 Edenberg HJ. The genetics of alcohol metabolism: role of alcohol dehydrogenase and aldehyde dehydrogenase variants. Alcohol Res Health. 2007; 30(1): 5–13.

88 Macgregor S, Lind PA, Bucholz KK, Hansell NK, Madden PA, Richter MM, et al. Associations of ADH and ALDH2 gene variation with self report alcohol reactions, consumption and dependence: an integrated analysis. Hum Mol Genet. 2009; 18(3): 580–93.

89 Bierut LJ, Goate AM, Breslau N, Johnson EO, Bertelsen S, Fox L, et al. ADH1B is associated with alcohol dependence and alcohol consumption in populations of European and African ancestry. Mol Psychiatry. 2012; 17(4): 445–50.

90 Beulens JWJ, Rimm EB, Hendriks HFJ, Hu FB, Manson JE, Hunter DJ, et al. Alcohol consumption and type 2 diabetes: influence of genetic variation in alcohol dehydrogenase. Diabetes. 2007; 56(9): 2388–94.

91 Tuma DJ, Casey CA. Dangerous byproducts of alcohol breakdown--focus on adducts. Alcohol Res Health. 2003; 27(4): 285–90.

92 Hsiang CY, Wu SL, Cheng SE, Ho TY. Acetaldehyde-induced interleukin-1beta and tumor necrosis factor-alpha production is inhibited by berberine through nuclear factor-kappaB signaling pathway in HepG2 cells. J Biomed Sci. 2005; 12(5): 791–801.

93 Garcia C, Feve B, Ferré P, Halimi S, Baizri H, Bordier L, et al. Diabetes and inflammation: fundamental aspects and clinical implications. Diabetes Metab. 2010; 36(5): 327–38.

94 Rehm J, Shield KD, Roerecke M, Gmel G. Modelling the impact of alcohol consumption on cardiovascular disease mortality for comparative risk assessments: an overview. BMC Public Health. 2016; 16: 363.

95 Conigrave KM, Hu BF, Camargo CA Jr, Stampfer MJ, Willett WC, Rimm EB. A prospective study of drinking patterns in relation to risk of type 2 diabetes among men. Diabetes. 2001; 50(10): 2390–5.

96 Carlsson S, Hammar N, Grill V, Kaprio J. Alcohol consumption and the incidence of type 2 diabetes: a 20-year follow-up of the Finnish twin cohort study. Diabetes Care. 2003; 26(10): 2785–90.

97 Hodge AM, English DR, O'Dea K, Giles GG. Alcohol intake, consumption pattern and beverage type, and the risk of Type 2 diabetes. Diabet Med. 2006; 23(6): 690–7.

98 Department of Health. Sensible drinking: Report of an inter-departmental working group. London: Department of Health; 1995.

99 Department of Health. Alcohol Guidelines Review: report from the guidelines development group to the UK Chief Medical Officers. London: Department of Health; 2016. Available from: https://www.gov.uk/government/consultations/health-risks-from-alcohol-new-guidelines

100 Gray L, Leyland AH. Chapter 2: Alcohol. The Scottish Health Survey 2014: Volume 1: Main Report. Edinburgh: The Scottish Government Health Directorate; 2015.

101 Office for National Statistics. Adult Drinking Habits in Great Britain, 2014. London: Office for National Statistics; 2014. Available from: http://www.ons.gov.uk/ons/rel/ghs/opinions-and-lifestyle-survey/adult-drinking-habits-in-great-britain--2013/index.html

102 Lebovitz HE. Insulin resistance: definition and consequences. Exp Clin Endocrinol Diabetes. 2001; 109 Suppl 2: S135-48.

103 Cefalu WT. Insulin resistance: cellular and clinical concepts. Exp Biol Med (Maywood). 2001; 226(1): 13–26.

104 Hulthe J, Fagerberg B. Alcohol Consumption and Insulin Sensitivity: A Review. Metab Syndr Relat Disord. 2005; 3(1): 45–50.

105 Bell RA, Mayer-Davis EJ, Martin MA, D'Agostino RB Jr, Haffner SM. Associations between alcohol consumption and insulin sensitivity and cardiovascular disease risk factors: the Insulin Resistance and Atherosclerosis Study. Diabetes Care. 2000; 23(11): 1630–6.

106 Lazarus R, Sparrow D, Weiss ST. Alcohol intake and insulin levels. The Normative Aging Study. Am J Epidemiol. 1997; 145(10): 909-16.

107 Moriya S, Yokoyama H, Hirose H, Ishii H, Saito I. Correlation between insulin resistance and gamma-glutamyl transpeptidase sensitivity in light drinkers. Alcohol Clin Exp Res. 2003; 27(8 Suppl): 52S–57S.

108 Flanagan DE, Moore VM, Godsland IF, Cockington RA, Robinson JS, Phillips DI. Alcohol consumption and insulin resistance in young adults. Eur J Clin Invest. 2000; 30(4): 297–301.

109 Kiechl S, Willeit J, Poewe W, Egger G, Oberhollenzer F, Muggeo M, et al. Insulin sensitivity and regular alcohol consumption: large, prospective, cross sectional population study (Bruneck study). BMJ. 1996; 313(7064): 1040–4.

110 Konrat C, Mennen LI, Cacès E, Lepinay P, Rakotozafy F, Forhan A, et al. Alcohol intake and fasting insulin in French men and women. The D.E.S.I.R. Study. Diabetes Metab. 2002; 28(2):116 23.

111 Vernay M, Balkau B, Moreau JG, Sigalas J, Chesnier MC, Ducimetiere P, et al. Alcohol consumption and insulin resistance syndrome parameters: associations and evolutions in a longitudinal analysis of the French DESIR cohort. Ann Epidemiol. 2004; 14(3): 209–14.

112 Schrieks IC, Heil AL, Hendriks HF, Mukamal KJ, Beulens JW. The effect of alcohol consumption on insulin sensitivity and glycemic status: a systematic review and meta-analysis of intervention studies. Diabetes Care. 2015; 38(4): 72332.

113 Toth PP. Reverse cholesterol transport: high-density lipoprotein's magnificent mile. Curr Atheroscler Rep. 2003; 5(5): 386–93. 114 Lund-Katz S, Phillips MC. High density lipoprotein structure-function and role in reverse cholesterol transport. Subcell Biochem. 2010; 51: 183–227.

115 Zhang Q, Zhang Y, Feng H, Guo R, Jin L, Wan R, et al. High Density Lipoprotein (HDL) Promotes Glucose Uptake in Adipocytes and Glycogen Synthesis in Muscle Cells. PLoS One. 2011; 6(8): e23556.

116 Mineo C, Shaul PW. Novel biological functions of high-density lipoprotein cholesterol. Circ Res. 2012; 111(8): 1079–90.

117 Rimm EB, Williams P, Fosher K, Criqui M, Stampfer MJ. Moderate alcohol intake and lower risk of coronary heart disease: meta-analysis of effects on lipids and haemostatic factors. BMJ. 1999; 319(7224): 1523–8.

118 Brien SE, Ronksley PE, Turner BJ, Mukamal KL, Ghali WA. Effect of alcohol consumption on biological markers associated with risk of coronary heart disease: systematic review and metaanalysis of interventional studies. BMJ. 2011; 342: d636.

119 Gepner Y, Golan R, Harman-Boehm I, Henkin Y, Schwarzfuchs D, Shelef I, et al. Effects of Initiating Moderate Alcohol Intake on Cardiometabolic Risk in Adults With Type 2 Diabetes: A 2-Year Randomized, Controlled Trial. Ann Intern Med. 2015; 163(8): 569–79.

120Kim JY, Song EH, Lee HJ, Oh YK, Park YS, Park JW, et al. Chronic Ethanol Consumptioninduced Pancreatic β -Cell Dysfunction and Apoptosis through Glucokinase Nitration and Its Down-regulation. J Biol Chem. 2010; 285(48): 37251–37262.

121 Haase CL, Tybjærg-Hansen A, Nordestgaard BG, Frikke-Schmidt R. HDL Cholesterol and Risk of Type 2 Diabetes: A Mendelian Randomization Study. Diabetes. 2015; 64(9): 3328–33.

122 Stewart SH. Alcohol and inflammation: a possible mechanism for protection against ischemic heart disease. Nutr Metab Cardiovasc Dis. 2002; 12(3):148–51.

123 De Lorgeril M, Salen P. Is alcohol anti-inflammatory in the context of coronary heart disease? Heart. 2004; 90(4): 355–357.
124 Akash MS, Rehman K, Chen S. Role of inflammatory mechanisms in pathogenesis of type 2 diabetes mellitus. J Cell Biochem. 2013; 114(3): 525–31.

125 Donath MY, Böni-Schnetzler M, Ellingsgaard H, Ehses JA. Islet inflammation impairs the pancreatic beta-cell in type 2 diabetes. Physiology (Bethesda). 2009; 24: 325–31.

126 Bell S, Mehta G, Moore K, Britton A. Ten-year alcohol consumption typologies and trajectories of C-reactive protein, interleukin-6 and interleukin-1 receptor antagonist over the following 12 years: a prospective cohort study. J Intern Med. 2016. doi: 10.1111/joim.12544. [Epub ahead of print]

127 Kraus VB, Jordan JM. Serum C-Reactive Protein (CRP), Target for Therapy or Trouble? Biomark Insights. 2006; 1:77–80.

128 Moller DE. Potential role of TNF-alpha in the pathogenesis of insulin resistance and type 2 diabetes. Trends Endocrinol Metab. 2000; 11(6): 212–7.

129 Hotamisligil GS. Mechanisms of TNF-alpha-induced insulin resistance. Exp Clin Endocrinol Diabetes. 1999; 107(2): 119–25.

130 Lorenzo M, Fernández-Veledo S, Vila-Bedmar R, Garcia-Guerra L, De Alvaro C, Nieto-Vazquez I. Insulin resistance induced by tumor necrosis factor-alpha in myocytes and brown adipocytes. J Anim Sci. 2008; 86(14 Suppl): E94–104.

131 Silverwood RJ, Holmes MV, Dale CE, Lawlor DA, Whittaker JC, Smith GD, et al. Testing for non-linear causal effects using a binary genotype in a Mendelian randomization study: application to alcohol and cardiovascular traits. Int J Epidemiol. 2014; 43(6): 1781–1790.

132 Brunner EJ, Kivimäki M, Witte DR, Lawlor DA, Davey Smith G, Cooper JA. Inflammation, Insulin Resistance, and Diabetes—Mendelian Randomization Using CRP Haplotypes Points Upstream. PLoS Med. 2008; 5(8): e155.

133 The Interleukin 1 Genetics Consortium. Cardiometabolic effects of genetic upregulation of the interleukin 1 receptor antagonist: a Mendelian randomisation analysis. Lancet Diabetes Endocrinol. 2015; 3(4): 243–253.

134 IL6R Genetics Consortium Emerging Risk Factors Collaboration. Interleukin-6 receptor pathways in coronary heart disease: a collaborative meta-analysis of 82 studies. Lancet. 2012; 379(9822): 1205–1213.

135 The Interleukin-6 Receptor Mendelian Randomisation Analysis (IL6R MR) Consortium. The interleukin-6 receptor as a target for prevention of coronary heart disease: a mendelian randomisation analysis. Lancet. 2012; 379(9822): 1214–1224.

136 Fillmore KM, Stockwell T, Chikritzhs T, Bostrom A, Kerr W. Moderate alcohol use and reduced mortality risk: systematic error in prospective studies and new hypotheses. Ann Epidemiol. 2007; 17(5 Suppl): S16-23.

137 Shaper AG. Alcohol and mortality: A review of prospective studies. Br J Addict. 1990; 85: 837–847.

138 Shaper AG, Wannamethee SG, Walter M. Alcohol and mortality in British men: Explaining the U-shaped curve. Lancet 1988; 2(8623): 1267–73.

139 Naimi TS, Brown DW, Brewer RD, Giles WH, Mensah G, Serdula MK, et al. Cardiovascular risk factors and confounders among nondrinking and moderate-drinking U.S. adults. Am J Prev Med. 2005; 28(4): 369–73.

140 Hansel B, Thomas F, Pannier B, Bean K, Kontush A, Chapman MJ, et al. Relationship between alcohol intake, health and social status and cardiovascular risk factors in the Urban Paris-Ile-de-France Cohort: is the cardioprotective action of alcohol a myth? Eur J Clin Nutr. 2010; 64(6): 561–8.

141 Ng Fat L, Shelton N. Associations between self-reported illness and non-drinking in young adults. Addiction. 2012; 107(9): 1612–20.

142 Wannamethee G, Shaper AG. Men Who Do Not Drink: A Report from the British Regional Heart Study. Int J Epidemiol. 1988; 17(2): 307–316.

143 Power C, Rodgers B, Hope S. U-shaped relation for alcohol consumption and health in early adulthood and implications for mortality. Lancet. 1998; 352(9131): 877.

144 Wannamethee G, Shaper AG. Changes in drinking habits in middle-aged British men. J R Coll Gen Pract. 1988; 38(315): 440–2.

145 Liang W, Chikritzhs T. Reduction in alcohol consumption and health status. Addiction. 2011; 106(1): 75–81.

146 Ng Fat L, Cable N, Shelton N. Worsening of Health and a Cessation or Reduction in Alcohol Consumption to Special Occasion Drinking Across Three Decades of the Life Course. Alcohol Clin Exp Res. 2015; 39(1): 166–174.

147 Di Castelnuovo A, Costanzo S, Bagnardi V, Donati MB, Iacoviello L, de Gaetano G. Alcohol dosing and total mortality in men and women: an updated meta-analysis of 34 prospective studies. Arch Intern Med. 2006; 166(22): 2437–45.

148 Stockwell T, Zhao J, Panwar S, Roemer A, Naimi T, Chikritzhs T. Do "Moderate" Drinkers Have Reduced Mortality Risk? A Systematic Review and Meta-Analysis of Alcohol Consumption and All-Cause Mortality. J Stud Alcohol Drugs. 2016; 77(2): 185–198.

149 Fagrell B, De Faire U, Bondy S, Criqui M, Gaziano M, Gronbaek M, et al. The effects of light to moderate drinking on cardiovascular diseases. J Intern Med. 1999; 246(4): 331–40.

150 Rehm J, Irving H, Ye Y, Kerr WC, Bond J, Greenfield TK. Are Lifetime Abstainers the Best Control Group in Alcohol Epidemiology? On the Stability and Validity of Reported Lifetime Abstention. Am J Epidemiol. 2008; 168(8): 866–871.

151 Ng Fat L, Cable N, Marmot MG, Shelton N. Persistent long-standing illness and nondrinking over time, implications for the use of lifetime abstainers as a control group. J Epidemiol Community Health. 2014; 68(1): 71–7.

152 Naimi TS, Stockwell T, Zhao J, Xuan Z, Dangardt F, Saitz R, et al. Selection biases in observational studies affect associations between 'moderate' alcohol consumption and mortality. Addiction. 2016. doi: 10.1111/add.13451. [Epub ahead of print]

153 Djoussé L, Biggs ML, Mukamal KJ, Siscovick DS. Alcohol Consumption and Type 2 Diabetes among Older Adults: The Cardiovascular Health Study. Obesity. 2007; 15(7): 1758–65.

154 Dias P, Oliveira A, Lopes C. Social and behavioural determinants of alcohol consumption. Ann Hum Biol. 2011; 38(3): 337–44.

155 World Health Organisation. Global status report on alcohol and health 2014. Geneva, Switzerland: World Health Organisation. Available from: http://apps.who.int/iris/bitstream/10665/112736/1/9789240692763_eng.pdf

156 Knott C, Bell S, Britton A. Alcohol Consumption and the Risk of Type 2 Diabetes: A Systematic Review and Dose-Response Meta-analysis of More Than 1.9 Million Individuals From 38 Observational Studies. Diabetes Care. 2015; 38(9): 1804–12.

157 Lee AJ, Crombie IK, Smith WC, Tunstall-Pedoe H. Alcohol consumption and unemployment among men: the Scottish Heart Health Study. Br J Addict. 1990; 85(9): 1165–70.

158 Tomkins S, Saburova L, Kiryanov N, Andreev E, McKee M, Shkolnikov V, Leon DA. Prevalence and socio-economic distribution of hazardous patterns of alcohol drinking: study of alcohol consumption in men aged 25-54 years in Izhevsk, Russia. Addiction. 2007; 102(4): 544– 53.

159 World Health Organization. Diabetes Mellitus: Report of a WHO Expert Committee. Technical Report Series 310. Geneva, Switzerland: WHO; 1965.

160 World Health Organization. WHO Expert Committee on Diabetes Mellitus: Second Report. Technical Report Series 646. Geneva, Switzerland: WHO; 1980.

161 World Health Organization. Diabetes Mellitus: Report of a WHO Study Group. Technical Report Series 727. Geneva, Switzerland: WHO; 1985.

162 World Health Organization. Definition, Diagnosis and Classification of Diabetes Mellitus and its Complications. Part 1: Diagnosis and Classification of Diabetes Mellitus. Geneva, Switzerland: WHO; 1999.

163 World Health Organization. Use of Glycated Haemoglobin (HbA1c) in the Diagnosis ofDiabetes Mellitus. Abbreviated Report of a WHO Consultation. Geneva, Switzerland: WHO;2011.

164 Tillett S, Newbold E. Grey literature at The British Library: revealing a hidden resource. Interlending & Document Supply. 2006; 34(2): 70–73.

165 Fleiss JL. Statistical Methods for Rates and Proportions. New York, NY: John Wiley & Sons Inc; 1981.

166 McHugh ML. Interrater reliability: the kappa statistic. Biochem Med (Zagreb). 2012; 22(3): 276–82.

167 Mao Q, Lin Y, Zheng X, Qin J, Yang K, Xie L. A meta-analysis of alcohol intake and risk of bladder cancer. Cancer Causes Control. 2010; 21(11): 1843–50.

168 Norat T, Lukanova A, Ferrari P, Riboli E. Meat consumption and colorectal cancer risk: dose-response meta-analysis of epidemiological studies. Int J Cancer. 2002; 98(2): 241–56.

169 Fedirko V, Tramacere I, Bagnardi V, Rota M, Scotti L, Islami F, et al. Alcohol drinking and colorectal cancer risk: an overall and dose-response meta-analysis of published studies. Ann Oncol. 2011; 22(9): 1958–72.

170 House of Commons Science and Technology Committee. Alcohol guidelines. Eleventh Report of Session 2010–12, [Vol. 1]: Report, Together with Formal Minutes, Oral and Written Evidence. London: The Stationery Office; 2012. Available from:

http://www.publications.parliament.uk/pa/cm201012/cmselect/cmsctech/1536/1536.pdf

171 Zhang J, Yu KF. What's the relative risk? A method of correcting the odds ratio in cohort studies of common outcomes. JAMA. 1998; 280(19): 1690–1.

172 Hamling J, Lee P, Weitkunat R, Ambühl M. Facilitating meta-analyses by deriving relative effect and precision estimates for alternative comparisons from a set of estimates presented by exposure level or disease category. Stat Med. 2008; 27(7): 954–70.

173 Zeisser C, Stockwell TR, Chikritzhs T. Methodological biases in estimating the relationship between alcohol consumption and breast cancer: The role of drinker misclassification errors in meta-analytic results. Alcohol Clin Exp Res. 2014; 38(8): 2297–2306. 174 Royston P, Ambler G, Sauerbrei W. The use of fractional polynomials to model continuous risk variables in epidemiology. Int J Epidemiol. 1999; 28(5): 964–74.

175 Greenland S. Dose-response and trend analysis in epidemiology: alternatives to categorical analysis. Epidemiology. 1995; 6(4): 356-65.

176 Royston P, Altman DG. Regression using fractional polynomials of continuous covariates: parsimonious parametric modelling (with discussion). Appl Stat. 1994; 43: 425–467.

177 StataCorp. Stata user's guide: release 13. College Station, TX: StataCorp LP; 2013.

178 Greenland S, Longnecker MP. Methods for trend estimation from summarized doseresponse data, with applications to meta-analysis. Am J Epidemiol. 1992; 135(11): 1301–9.

179 Orsini N, Bellocco R, Greenland S. Generalized least squares for trend estimation of summarized dose-response data. Stata Journal. 2006; 6(1): 40–57.

180 Borenstein M, Hedges LV, Higgins JPT, Rothstein HR, eds. Introduction to Meta-Analysis. Chicheser, UK: John Wiley & Sons, Ltd; 2009.

181 Higgins JP, Thompson SG, Deeks JJ, Altman DG. Measuring inconsistency in meta-analyses. BMJ. 2003; 327(7414): 557–60.

182 Higgins JPT, Green S, eds. Cochrane Handbook for Systematic Reviews of Interventions Version 5.1.0 [updated March 2011]. The Cochrane Collaboration; 2011. Available at: http://handbook.cochrane.org

183 Scherer RW, Langenberg P, von Elm E. Full publication of results initially presented in abstracts. Cochrane Database Syst Rev. 2007; (2): MR000005.

184 Hopewell S, Loudon K, Clarke MJ, Oxman AD, Dickersin K. Publication bias in clinical trials due to statistical significance or direction of trial results. Cochrane Database Syst Rev. 2009; (1): MR000006.

185 Egger M, Davey Smith G, Schneider M, Minder C. Bias in meta-analysis detected by a simple, graphical test. BMJ. 1997; 315(7109): 629–34.

186 Begg CB, Mazumdar M. Operating characteristics of a rank correlation test for publication bias. Biometrics. 1994; 50(4): 1088–101.

187 Egger M, Davey Smith G, Schneider M, Minder C. Bias in meta-analysis detected by a simple, graphical test. BMJ. 1997; 315(7109): 629–34.

188 Sterne JA, Harbord RM. Funnel plots in meta–analysis. The Stata Journal. 2004;4(2):127-141.

189 Wells GA, Shea B, O'Connell D, Peterson J, Welch V, Losos M, et al. The Newcastle-Ottawa Scale (NOS) for assessing the quality of nonrandomized studies in meta-analyses. Available from: http://www.ohri.ca/programs/clinical_epidemiology/oxford.asp

190 Hu FB, Manson JE, Stampfer MJ, Colditz G, Liu S, Solomon CG, et al. Diet, lifestyle, and the risk of type 2 diabetes mellitus in women. N Engl J Med. 2001; 345(11): 790–7.

191 Tabák AG, Jokela M, Akbaraly TN, Brunner EJ, Kivimäki M, Witte DR. Trajectories of Glycemia, Insulin Sensitivity and Insulin Secretion Preceding the Diagnosis of Type 2 Diabetes: The Whitehall II Study. Lancet. 2009; 373(9682): 2215–2221.

192 Cole-Harding S, Wilson JR. Ethanol metabolism in men and women. J Stud Alcohol. 1987; 48(4): 380-7.

193 Mumenthaler MS, Taylor JL, O'Hara R, Yesavage JA. Gender differences in moderate drinking effects. Alcohol Res Health. 1999; 23(1): 55–64.

194 Kwo PY, Ramchandani VA, O'Connor S, Amann D, Carr LG, Sandrasegaran K, et al. Gender differences in alcohol metabolism: relationship to liver volume and effect of adjusting for body mass. Gastroenterology. 1998; 115(6): 1552–7.

195 Oldroyd J, Banerjee M, Heald A, Cruickshank K. Diabetes and ethnic minorities. Postgrad Med J. 2005; 81(958): 486-90.

196 Wynne HA, Cope LH, Mutch E, Rawlins MD, Woodhouse KW, James OF. The effect of age upon liver volume and apparent liver blood flow in healthy man. Hepatology. 1989; 9(2): 297– 301. 197 Zoli M, Iervese T, Abbati S, Bianchi GP, Marchesini G, Pisi E. Portal blood velocity and flow in aging man. Geronology. 1989; 35(2-3): 61–65.

198 Meier P, Seitz HK. Age, alcohol metabolism and liver disease. Curr Opin Clin Nutr Metab Care. 2008; 11(1): 21–26.

199 Dufour MC, Archer L, Gordis E. Alcohol and the elderly. Clin Geriatr Med. 1992; 8(1): 127-41.

200 Wang HJ, Gao B, Zakhari S, Nagy LE. Inflammation in Alcoholic Liver Disease. Annu Rev Nutr. 2012; 32: 343–368.

201 Guo R, Ren J. Alcohol and Acetaldehyde in Public Health: From Marvel to Menace. Int J Environ Res Public Health. 2010; 7(4): 1285–1301.

202 Schneider AL, Pankow JS, Heiss G, Selvin E. Validity and reliability of self-reported diabetes in the atherosclerosis risk in communities study. Am J Epidemiol. 2012; 176(8): 738–43.

203 Edenberg HJ. The Genetics of Alcohol Metabolism: Role of Alcohol Dehydrogenase and Aldehyde Dehydrogenase Variants. Alcohol Res Health. 2007; 30(1): 5–13.

204 Cho YS, Lee JY, Park KS, Nho CW. Genetics of type 2 diabetes in East Asian populations. Curr Diab Rep. 2012; 12(6): 686–96.

205 Eisen EA, Robins JM, Picciotto S. Healthy worker effect in occupational studies. In: El-Shaarawi AH, Piegorsch W eds. Encyclopedia of environmetrics. Chichester, UK: John Wiley & Sons; 2012: 1269–72.

206 Berlin JA, Santanna J, Schmid CH, Szczech LA, Feldman HI. Individual patient- versus grouplevel data meta-regression for the investigation of treatment effect modifiers: ecological bias rears its ugly head. Stat Med. 2002; 21: 371–87.

207 Stampfer MJ, Colditz GA, Willett WC, Manson JE, Arky RA, Hennekens CH, et al. A prospective study of moderate alcohol drinking and risk of diabetes in women. Am J Epidemiol. 1988; 128(3): 549–58.

208 Colditz GA. A prospective assessment of moderate alcohol intake and major chronic diseases. Ann Epidemiol. 1990; 1(2): 167–77.

209 Rimm EB, Chan J, Stampfer MJ, Colditz GA, Willett WC. Prospective study of cigarette smoking, alcohol use, and the risk of diabetes in men. BMJ. 1995; 310(6979): 555–9.

210 Beulens JWJ, Stolk RP, Schouw YT van der, Grobbee DE, Hendriks HFJ, Bots ML. Alcohol Consumption and Risk of Type 2 Diabetes Among Older Women. Dia Care. 2005;28(12): 2933– 8.

211 Joosten MM, Grobbee DE, van der A DL, Verschuren WMM, Hendriks HFJ, Beulens JWJ. Combined effect of alcohol consumption and lifestyle behaviors on risk of type 2 diabetes. Am J Clin Nutr. 2010; 91(6): 1777–83.

212 Nagaya T, Yoshida H, Takahashi H, Kawai M. Heart rate-corrected QT interval in resting ECG predicts the risk for development of type-2 diabetes mellitus. Eur J Epidemiol. 2010; 25(3): 195–202.

213 Sawada SS, Lee I-M, Naito H, Noguchi J, Tsukamoto K, Muto T, et al. Long-term trends in cardiorespiratory fitness and the incidence of type 2 diabetes. Diabetes Care. 2010; 33(6): 1353–7.

214 Holbrook TL, Barrett-Connor E, Wingard DL. A prospective population-based study of alcohol use and non-insulin-dependent diabetes mellitus. Am J Epidemiol. 1990; 132(5): 902–9.

215 Kawakami N, Takatsuka N, Shimizu H, Ishibashi H. Effects of Smoking on the Incidence of Non-Insulin-dependent Diabetes Mellitus Replication and Extension in a Japanese Cohort of Male Employees. Am J Epidemiol. 1997; 145(2): 103–9.

216 Tsumura K, Hayashi T, Suematsu C, Endo G, Fujii S, Okada K. Daily alcohol consumption and the risk of type 2 diabetes in Japanese men: the Osaka Health Survey. Diabetes Care. 1999; 22(9): 1432–7.

217 Ajani UA, Hennekens CH, Spelsberg A, Manson JE. Alcohol consumption and risk of type 2 diabetes mellitus among US male physicians. Arch Intern Med. 2000; 160(7): 1025-30.

218 Wei M, Gibbons LW, Mitchell TL, Kampert JB, Blair SN. Alcohol intake and incidence of type 2 diabetes in men. Diabetes Care. 2000; 23(1): 18–22.

219 Kao WH, Puddey IB, Boland LL, Watson RL, Brancati FL. Alcohol consumption and the risk of type 2 diabetes mellitus: atherosclerosis risk in communities study. Am J Epidemiol. 2001; 154(8): 748–57.

220 Meisinger C, Thorand B, Schneider A, Stieber J, Döring A, Löwel H. Sex differences in risk factors for incident type 2 diabetes mellitus: the MONICA Augsburg cohort study. Arch Intern Med. 2002; 162(1): 82–9.

221 Wannamethee S, Shaper A, Perry I, Alberti K. Alcohol consumption and the incidence of type II diabetes. J Epidemiol Community Health. 2002; 56(7): 542–548.

222 Lee DH, Ha MH, Kim JH, Christiani DC, Gross MD, Steffes M, et al. Gammaglutamyltransferase and diabetes--a 4 year follow-up study. Diabetologia. 2003; 46(3): 359–64.

223 Nakanishi N, Suzuki K, Tatara K. Alcohol Consumption and Risk for Development of Impaired Fasting Glucose or Type 2 Diabetes in Middle-Aged Japanese Men. Dia Care. 2003; 26(1): 48–54.

224 Sawada SS, Lee I-M, Muto T, Matuszaki K, Blair SN. Cardiorespiratory fitness and the incidence of type 2 diabetes: prospective study of Japanese men. Diabetes Care. 2003; 26(10): 2918–22.

225 Wannamethee SG, Camargo CA Jr, Manson JE, Willett WC, Rimm EB. Alcohol drinking patterns and risk of type 2 diabetes mellitus among younger women. Arch Intern Med. 2003; 163(11): 1329–36.

226 Lee DH, Folsom AR, Jacobs DR Jr. Dietary iron intake and Type 2 diabetes incidence in postmenopausal women: the Iowa Women's Health Study. Diabetologia. 2004; 47(2): 185–94.

227 Waki K, Noda M, Sasaki S, Matsumura Y, Takahashi Y, Isogawa A, et al. Alcohol consumption and other risk factors for self-reported diabetes among middle-aged Japanese: a population-based prospective study in the JPHC study cohort I. Diabetic Medicine. 2005; 22(3): 323–31.

228 Hu G, Jousilahti P, Peltonen M, Bidel S, Tuomilehto J. Joint association of coffee consumption and other factors to the risk of type 2 diabetes: a prospective study in Finland. Int J Obes. 2006; 30(12): 1742–9.

229 Strodl E, Kenardy J. Psychosocial and non-psychosocial risk factors for the new diagnosis of diabetes in elderly women. Diabetes Res Clin Pract. 2006; 74(1): 57–65.

230 Burke V, Zhao Y, Lee AH, Hunter E, Spargo RM, Gracey M, et al. Predictors of type 2 diabetes and diabetes-related hospitalisation in an Australian Aboriginal cohort. Diabetes Res Clin Pract. 2007; 78(3): 360–8.

231 Maty SC, Lynch JW, Raghunathan TE, Kaplan GA. Childhood socioeconomic position, gender, adult body mass index, and incidence of type 2 diabetes mellitus over 34 years in the Alameda County Study. Am J Public Health. 2008; 98(8): 1486–94.

232 Onat A, Hergenç G, Küçükdurmaz Z, Uğur M, Kaya Z, Can G, et al. Moderate and heavy alcohol consumption among Turks: long-term impact on mortality and cardiometabolic risk. Turk Kardiyol Dern Ars. 2009; 37(2): 83–90.

233 Roh WG, Shin HC, Choi JH, Lee YJ, Kim K. Alcohol consumption and higher incidence of impaired fasting glucose or type 2 diabetes in obese Korean men. Alcohol. 2009; 43(8): 643–8.

234 Boggs DA, Rosenberg L, Ruiz-Narvaez EA, Palmer JR. Coffee, tea, and alcohol intake in relation to risk of type 2 diabetes in African American women. Am J Clin Nutr. 2010; 92(4): 960–6.

235 Jee SH, Foong AW, Hur NW, Samet JM. Smoking and Risk for Diabetes Incidence and Mortality in Korean Men and Women. Dia Care. 2010; 33(12): 2567–72.

236 Nagaya T, Yoshida H, Takahashi H, Kawai M. Resting heart rate and blood pressure, independent of each other, proportionally raise the risk for type-2 diabetes mellitus. Int J Epidemiol. 2010; 39(1): 215–22.

237 Balkau B, Soulimane S, Lange C, Gautier A, Tichet J, Vol S, et al. Are the same clinical risk factors relevant for incident diabetes defined by treatment, fasting plasma glucose, and HbA1c? Diabetes Care. 2011; 34(4): 957–9.

238 Beulens JWJ, van der Schouw YT, Bergmann MM, Rohrmann S, Schulze MB, Buijsse B, et al. Alcohol consumption and risk of type 2 diabetes in European men and women: influence of beverage type and body size. Journal of Internal Medicine. 2012; 272(4): 358–70.

239 Cullmann M, Hilding A, Östenson C-G. Alcohol consumption and risk of pre-diabetes and type 2 diabetes development in a Swedish population. Diabet Med. 2012; 29(4): 441–52.

240 Sato KK, Hayashi T, Harita N, Koh H, Maeda I, Endo G, et al. Relationship between drinking patterns and the risk of type 2 diabetes: the Kansai Healthcare Study. J Epidemiol Community Health. 2012; 66(6): 507–11.

241 Stringhini S, Tabak AG, Akbaraly TN, Sabia S, Shipley MJ, Marmot MG, et al. Contribution of modifiable risk factors to social inequalities in type 2 diabetes: prospective Whitehall II cohort study. BMJ. 2012; 345: e5452.

242 Teratani T, Morimoto H, Sakata K, Oishi M, Tanaka K, Nakada S, et al. Dose-response relationship between tobacco or alcohol consumption and the development of diabetes mellitus in Japanese male workers. Drug Alcohol Depend. 2012; 125(3): 276–82.

243 Abbasi A, Corpeleijn E, Gansevoort RT, Gans ROB, Hillege HL, Stolk RP, et al. Role of HDL cholesterol and estimates of HDL particle composition in future development of type 2 diabetes in the general population: the PREVEND study. J Clin Endocrinol Metab. 2013; 98(8): E1352–1359.

244 Heianza Y, Arase Y, Saito K, Tsuji H, Fujihara K, Hsieh SD, et al. Role of alcohol drinking pattern in type 2 diabetes in Japanese men: the Toranomon Hospital Health Management Center Study (TOPICS 11). Am J Clin Nutr. 2013; 97(3): 561–8.

245 Rasouli B, Ahlbom A, Andersson T, Grill V, Midthjell K, Olsson L, et al. Alcohol consumption is associated with reduced risk of Type 2 diabetes and autoimmune diabetes in adults: results from the Nord-Trøndelag health study. Diabetic Medicine. 2013;3 0(1): 56–64.

246 Shi L, Shu X-O, Li H, Cai H, Liu Q, Zheng W, et al. Physical Activity, Smoking, and Alcohol Consumption in Association with Incidence of Type 2 Diabetes among Middle-Aged and Elderly Chinese Men. PLoS ONE. 2013; 8(11): e77919. 247 de Vegt F, Dekker JM, Groeneveld WJ, Nijpels G, Stehouwer CD, Bouter LM, et al. Moderate alcohol consumption is associated with lower risk for incident diabetes and mortality: the Hoorn Study. Diabetes Res Clin Pract. 2002; 57(1): 53–60.

248 Lambert PC, Sutton AJ, Abrams KR, Jones DR. A comparison of summary patient-level covariates in meta-regression with individual patient data meta-analysis. J Clin Epidemiol. 2002; 55(1): 86–94.

249 Berlin JA, Santanna J, Schmid CH, Szczech LA, Feldman HI; Anti-Lymphocyte Antibody Induction Therapy Study Group. Individual patient- versus group-level data meta-regressions for the investigation of treatment effect modifiers: ecological bias rears its ugly head. Stat Med. 2002; 21(3): 371–87.

250 Schmid CH, Stark PC, Berlin JA, Landais P, Lau J. Meta-regression detected associations between heterogeneous treatment effects and study-level, but not patient-level, factors. J Clin Epidemiol. 2004; 57(7): 683–97.

251 Baker WL, White CM, Cappelleri JC, Kluger J, Coleman CI; Health Outcomes, Policy, and Economics (HOPE) Collaborative Group. Understanding heterogeneity in meta-analysis: the role of meta-regression. Int J Clin Pract. 2009; 63(10): 1426–34.

252 Higgins JP, Thompson SG. Controlling the risk of spurious findings from meta-regression. Stat Med. 2004; 23(11): 1663–82.

253 Sanderson S. Tatt ID, Higgins JPT. Tools for assessing quality and susceptibility to bias in observational studies in epidemiology: a systematic review and annotated bibliography. Int J Epidemiol. 2007; 36(3): 666–676.

254 Jüni P, Witschi A, Bloch R, Egger M. The hazards of scoring the quality of clinical trials for meta-analysis. JAMA. 1999; 282(11): 1054–60.

255 Greenland S, O'Rourke K. On the bias produced by quality scores in meta-analysis, and a hierarchical view of proposed solutions. Biostatistics. 2001; 2(4): 463–71.

256 Stang A. Critical evaluation of the Newcastle-Ottawa scale for the assessment of the quality of nonrandomized studies in meta-analyses. Eur J Epidemiol. 2010; 25(9): 603–5.

257 Hartling L, Milne A, Hamm MP, Vandermeer B, Ansari M, Tsertsvadze A, Dryden DM: Testing the Newcastle Ottawa Scale showed low reliability between individual reviewers. J Clin Epidemiol. 2013; 66: 982–993

258 Oremus M, Oremus C, Hall GB, McKinnon MC; ECT & Cognition Systematic Review Team. Inter-rater and test-retest reliability of quality assessments by novice student raters using the Jadad and Newcastle-Ottawa Scales. BMJ Open. 2012; 2(4): e001368.

259 Sher KJ, Jackson KM, Steinley D. Alcohol Use Trajectories and the Ubiquitous Cat's Cradle: Cause for Concern? J Abnorm Psychol. 2011; 120(2): 322–35.

260 Wiesner M, Weichold K, Silbereisen RK. Trajectories of alcohol use among adolescent boys and girls: identification, validation, and sociodemographic characteristics. Psychol Addict Behav. 2007; 21(1): 62–75.

261 Van Der Vorst H, Vermulst AA, Meeus WH, Deković M, Engels RC. Identification and prediction of drinking trajectories in early and mid-adolescence. J Clin Child Adolesc Psychol. 2009; 38(3): 329-41.

262 Toumbourou JW, Williams IR, Snow PC, White VM. Adolescent alcohol-use trajectories in the transition from high school. Drug Alcohol Rev. 2003; 22(2): 111–6.

263 Tran NT, Williams GM, Alati R, Najman JM. Trajectories and predictors of alcohol consumption over 21 years of mothers' reproductive life course. SSM Population Health. 2015; 1: 40–47.

264 Berg N, Kiviruusu O, Karvonen S, Kestilä L, Lintonen T, Rahkonen O, et al. A 26-year followup study of heavy drinking trajectories from adolescence to mid-adulthood and adult disadvantage. Alcohol Alcohol. 2013; 48(4): 452–7.

265 Kaplan MS, Huguet N, Feeny D, McFarland BH, Caetano R, Bernier J, et al. Alcohol use patterns and trajectories of health-related quality of life in middle-aged and older adults: a 14-year population-based study. J Stud Alcohol Drugs. 2012; 73(4): 581–90.

266 Britton A, Ben-Shlomo Y, Benzeval M, Kuh D, Bell S. Life course trajectories of alcohol consumption in the United Kingdom using longitudinal data from nine cohort studies. BMC Med. 2015; 13(47): 1–9.

267 Joosten MM, Chiuve SE, Mukamal KJ, Hu FB, Hendriks HF, Rimm EB. Changes in alcohol consumption and subsequent risk of type 2 diabetes in men. Diabetes. 2011; 60(1): 74–9.

268 Roerecke M, Rehm J. Alcohol consumption, drinking patterns, and ischemic heart disease: a narrative review of meta-analyses and a systematic review and meta-analysis of the impact of heavy drinking occasions on risk for moderate drinkers. BMC Medicine. 2014; 12: 182.

269 Nazareth I, Walker C, Ridolfi A, Aluoja A, Bellon J, Geerlings M, et al. Episodic heavy drinking in Europe: a cross section study in primary care in six European countries. Alcohol Alcohol. Alcohol. 2011; 46(5): 600–6.

270 Khaw KT, Barrett-Connor E. Fasting plasma glucose levels and endogenous androgens in non-diabetic postmenopausal women. Clin Sci (Lond). 1991; 80(3): 199–203.

271 Kalish GM, Barrett-Connor E, Laughlin GA, Gulanski BI. Association of endogenous sex hormones and insulin resistance among postmenopausal women: results from the Postmenopausal Estrogen/Progestin Intervention Trial. J Clin Endocrinol Metab. 2003; 88(4): 1646–52.

272 Phillips GB. Relationship between serum sex hormones and the glucose-insulin-lipid defect in men with obesity. Metabolism. 1993; 42(1): 116–20.

273 Ding EL, Song Y, Malik VS, Liu S. Sex differences of endogenous sex hormones and risk of type 2 diabetes: a systematic review and meta-analysis. JAMA. 2006; 295(11): 1288–99.

274 Rohwer RD, Liu S, You NC, Buring JE, Manson JE, Song Y. Interrelationship Between Alcohol Intake and Endogenous Sex-Steroid Hormones on Diabetes Risk in Postmenopausal Women. J Am Coll Nutr. 2015; 34(4): 273–80.

275 Frydenberg H, Flote VG, Larsson IM, Barrett ES, Furberg AS, Ursin G, et al. Alcohol consumption, endogenous estrogen and mammographic density among premenopausal women. Breast Cancer Res. 2015; 17: 103.

276 Muti P, Trevisan M, Micheli A, Krogh V, Bolelli G, Sciajno R, et al. Alcohol consumption and total estradiol in premenopausal women. Cancer Epidemiol Biomarkers Prev. 1998; 7(3): 189– 93.

277 Gavaler JS, Deal SR, Van Thiel DH, Arria A, Allan MJ. Alcohol and estrogen levels in postmenopausal women: the spectrum of effect. Alcohol Clin Exp Res. 1993; 17(4): 786–90.

278 Sarkola T, Mäkisalo H, Fukunaga T, Eriksson CJ. Acute effect of alcohol on estradiol, estrone, progesterone, prolactin, cortisol, and luteinizing hormone in premenopausal women. Alcohol Clin Exp Res. 1999; 23(6): 976–82.

279 Mendelson JH, Lukas SE, Mello NK, Amass L, Ellingboe J, Skupny A. Acute alcohol effects on plasma estradiol levels in women. Psychopharmacology (Berl). 1988; 94(4): 464–7.

280 Goldberg DM, Soleas GJ, Levesque M. Moderate alcohol consumption: the gentle face of Janus. Clin Biochem. 1999; 32(7): 505–18.

281 Boban M, Stockley C, Teissedre PL, Restani P, Fradera U, Stein-Hammer C, et al. Drinking pattern of wine and effects on human health: why should we drink moderately and with meals? Food Funct. 2016; 7(7): 2937–42.

282 Lachenmeier DW, Godelmann R, Witt B, Riedel K, Rehm J. Can resveratrol in wine protect against the carcinogenicity of ethanol? A probabilistic dose-response assessment. Int J Cancer. 2014; 134(1): 144–53.

283 Mortensen EL, Jensen HH, Sanders SA, Reinisch JM. Better psychological functioning and higher social status may largely explain the apparent health benefits of wine: a study of wine and beer drinking in young Danish adults. Arch Intern Med. 2001; 161(15): 1844–8.

284 Paschall M, Lipton RI. Wine preference and related health determinants in a U.S. national sample of young adults. Drug Alcohol Depend. 2005; 78(3): 339–44.

285 Greenfield T. K., Kerr W. C. Commentary on Liang & Chikritzhs (2011): Quantifying the impacts of health problems on drinking and subsequent morbidity and mortality—life-course measures are essential. Addiction. 2011; 106(1): 82–3

286 Britton A, Marmot MG, Shipley MJ. How does variability in alcohol consumption over time affect the relationship with mortality and coronary heart disease? Addiction. 2010; 105(4): 639–45.

287 Fillmore KM, Chikritzhs T. Commentary on Britton et al. The dangers of declining drink. Addiction. 2010; 105(4): 646–7.

288 Maggs JL, Patrick ME, Feinstein L. Childhood and adolescent predictors of alcohol use and problems in adolescence and adulthood in the National Child Development Study. Addiction. 2008; 103 Suppl 1: 7–22.

289 Platt A, Sloann FA, Costanzo P. Alcohol-Consumption Trajectories and Associated Characteristics Among Adults Older Than Age 50. J Stud Alcohol Drugs. 2010; 71(2): 169–179.

290 Steptoe A, Breeze E, Banks J, Nazroo J. Cohort profile: the English longitudinal study of ageing. Int J Epidemiol. 2013; 42(6): 1640–8.

291 Wadsworth M, Kuh D, Richards M, Hardy R. Cohort Profile: The 1946 National Birth Cohort (MRC National Survey of Health and Development). Int J Epidemiol. 2006; 35(1): 49–54.

292 Benzeval M, Der G, Ellaway A, Hunt K, Sweeting H, West P, et al. Cohort profile: west of

Scotland twenty-07 study: health in the community. Int J Epidemiol. 2009; 38(5): 1215–23.

293 Creavin ST, Gallacher J, Bayer A, Fish M, Ebrahim S, Ben-Shlomo Y. Metabolic syndrome, diabetes, poor cognition, and dementia in the Caerphilly prospective study. J Alzheimers Dis. 2012; 28(4): 931–9.

294 Marmot M, Brunner E. Cohort Profile: the Whitehall II study. Int J Epidemiol. 2005; 34(2): 251–6.

295 Sabia S, Dugravot A, Kivimaki M, Brunner E, Shipley MJ, Singh-Manoux A. Effect of Intensity and Type of Physical Activity on Mortality: Results From the Whitehall II Cohort Study. American Journal of Public Health. 2012; 102(4): 698–704.

296 Batty GD, Shipley M, Tabák A, Singh-Manoux A, Brunner E, Britton A, Kivimäki M. Generalizability of Occupational Cohort Study Findings. Epidemiology. 2014; 25(6): 932–933. 297 Goddard E. Estimating alcohol consumption from survey data: Updated method of converting volumes to units, National Statistics Methodological Series No. 37. Newport: Office for National Statistics; 2007.

298 Britton A, O'Neill D, Bell S. Underestimating the Alcohol Content of a Glass of Wine: The Implications for Estimates of Mortality Risk. Alcohol Alcohol 2016. doi: http://dx.doi.org/10.1093/alcalc/agw027. [Epub ahead of print]

299 Boniface S, Kneale J, Shelton N. Actual and perceived units of alcohol in a self-defined "usual glass" of alcoholic drinks in England. Alcohol Clin Exp Res. 2013; 37(6): 978–83.

300 Gill JS, Donaghy M. Variation in the alcohol content of a 'drink' of wine and spirit poured by a sample of the Scottish population. Health Educ Res. 2004; 19(5): 485–91.

301 Moody A. Chapter 4: Diabetes and hyperglycaemia. In. Craig R, Mindell J, editors. Health Survey for England 2011. Volume 1: Health, social care and lifestyles. Leeds: Health and Social Care Information Centre; 2012.

302 Alberti KG, Zimmet PZ. Definition, diagnosis and classification of diabetes mellitus and its complications. Part 1: diagnosis and classification of diabetes mellitus provisional report of a WHO consultation. Diabet Med. 1998; 15(7): 539–53.

303 Yeomans MR. Alcohol, appetite and energy balance: is alcohol intake a risk factor for obesity? Physiol Behav. 2010; 100(1): 82–9.

304 Traversy G, Chaput J-P. Alcohol Consumption and Obesity: An Update. Curr Obes Rep. 2015; 4(1): 122–130.

305 French MT, Norton EC, Fang H, Maclean JC. Alcohol consumption and body weight. Health Econ. 2010; 19(7): 814–32.

306 Mozaffarian D, Hao T, Rimm EB, Willett WC, Hu FB. Changes in Diet and Lifestyle and Long-Term Weight Gain in Women and Men N Engl J Med. 2011; 364(25): 2392–2404.

307 Wannamethee SG, Field AE, Colditz GA, Rimm EB. Alcohol intake and 8-year weight gain in women: A prospective study. Obes Res. 2004; 12(9): 1386–96.

308 Willett WC, Sampson L, Stampfer MJ, Rosner B, Bain C, Witschi J, et al. Reproducibility and validity of a semiquantitative food frequency questionnaire. Am J Epidemiol. 1985; 122(1):51–65.

309 Bingham SA, Gill C, Welch A, Cassidy A, Runswick SA, Oakes S, et al. Validation of dietary assessment methods in the UK arm of EPIC using weighed records, and 24-hour urinary nitrogen and potassium and serum vitamin C and carotenoids as biomarkers. Int J Epidemiol. 1997; 26 Suppl 1: S137–51.

310 Brunner E, Stallone D, Juneja M, Bingham S, Marmot M. Dietary assessment in Whitehall II: comparison of 7 d diet diary and food-frequency questionnaire and validity against biomarkers. Br J Nutr. 2001; 86(3): 405–14.

311 Ainsworth BE, Haskell WL, Whitt MC, Irwin ML, Swartz AM, Strath SJ, et al. Compendium of physical activities: an update of activity codes and MET intensities. Med Sci Sports Exerc. 2000; 32(9 Suppl): S498–504.

312 Stringhini S, Sabia S, Shipley M, Brunner E, Nabi H, Kivimaki M, et al. Association of socioeconomic position with health behaviors and mortality. The Whitehall II study. JAMA. 2010; 303(12): 1159–1166.

313 World Health Organization. Global Recommendations on Physical Activity for Health. Geneva: World Health Organization; 2010.

314 Jefferis BJ, Manor O, Power C. Social gradients in binge drinking and abstaining: trends in a cohort of British adults. J Epidemiol Community Health. 2007; 61(2): 150–3.

315 Caldwell TM, Rodgers B, Clark C, Jefferis BJ, Stansfeld SA, Power C. Life course socioeconomic predictors of midlife drinking patterns, problems and abstention: findings from the 1958 British Birth Cohort Study. Drug Alcohol Depend. 2008; 95(3): 269–78.

316 Droomers M, Schrijvers CT, Stronks K, van de Mheen D, Mackenbach JP. Educational differences in excessive alcohol consumption: the role of psychosocial and material stressors. Prev Med. 1999; 29(1): 1–10.

317 Harhay MO, Bor J, Basu S, McKee M, Mindell JS, Shelton NJ et al. Differential impact of the economic recession on alcohol use among white British adults, 2004-2010. Eur J Public Health. 2014; 24(3): 410–5.

318 Fone DL, Farewell DM, White J, Lyons RA, Dunstan FD. Socioeconomic patterning of excess alcohol consumption and binge drinking: a cross-sectional study of multilevel associations with neighbourhood deprivation. BMJ Open. 2013; 3(4).

319 Sims M, Diez Roux AV, Boykin S, Sarpong D, Gebreab SY, Wyatt SB, et al. The socioeconomic gradient of diabetes prevalence, awareness, treatment, and control among African Americans in the Jackson Heart Study. Ann Epidemiol. 2011; 21(12): 892–8.

320 Rabi DM, Edwards AL, Southern DA, Svenson LW, Sargious PM, Norton P, et al. Association of socio-economic status with diabetes prevalence and utilization of diabetes care services. BMC Health Serv Res. 2006; 6: 124.

321 Agardh EE, Ahlbom A, Andersson T, Efendic S, Grill V, Hallqvist J, et al. Explanations of socioeconomic differences in excess risk of type 2 diabetes in Swedish men and women. Diabetes Care. 2004; 27(3): 716–21.

322 Connolly V, Unwin N, Sherriff P, Bilous R, Kelly W. Diabetes prevalence and socioeconomic status: a population based study showing increased prevalence of type 2 diabetes mellitus in deprived areas. J Epidemiol Community Health. 2000; 54(3): 173–7.

323 Elkind MS. Inflammatory mechanisms of stroke. Stroke. 2010; 41(10 Suppl): S3-8.

324 Wannamethee SG, Shaper AG, Ebrahim S. HDL-Cholesterol, total cholesterol, and the risk of stroke in middle-aged British men. Stroke. 2000; 31(8): 1882–8.

325 Zhang Y, Tuomilehto J, Jousilahti P, Wang Y, Antikainen R, Hu G. Total and high-density lipoprotein cholesterol and stroke risk. Stroke. 2012; 43(7): 1768–74.

326 Gast KB, Tjeerdema N, Stijnen T, Smit JW, Dekkers OM. Insulin resistance and risk of incident cardiovascular events in adults without diabetes: meta-analysis. PLoS One. 2012; 7(12): e52036.

327 Shahid M, Sun RL, Liu Y, Bao JL, Huang CX, Liao Y, et al. Is high high-density lipoprotein cholesterol beneficial for premature coronary heart disease? A meta-analysis. Eur J Prev Cardiol. 2015. pii: 2047487315610662.

328 Kaptoge S, Seshasai SR, Gao P, Freitag DF, Butterworth AS, Borglykke A, et al. Inflammatory cytokines and risk of coronary heart disease: new prospective study and updated meta-analysis. Eur Heart J. 2014; 35(9): 578–89.

329 MacKinnon DP, Krull JL, Lockwood CM. Equivalence of the Mediation, Confounding and Suppression Effect. Prev Sci. 2000; 1(4): 173.

330 Rubin DB. Inference and missing data. Biometrika. 1976; 63(3): 581–92.

331 Muthén B, Kaplan D, Hollis M. On structural equation modeling with data that are not missing completely at random. Psychometrika. 1987; 52(3):431–462.

332 Biering K, Hjollund NH, Frydenberg M. Using multiple imputation to deal with missing data and attrition in longitudinal studies with repeated measures of patient-reported outcomes. Clinical Epidemiology. 2015; 7: 91–106.

333 Diehr P, Patrick DL. Trajectories of health for older adults over time: accounting fully for death. Ann Intern Med. 2003; 139(5 Pt 2):416–20.

334 Satagopan JM, Ben-Porat L, Berwick M, Robson M, Kutler D, Auerbach AD. A note on competing risks in survival data analysis. Br J Cancer. 2004; 91(7): 1229–35.

335 Ferrucci L, Guralnik JM, Studenski S, Fried LP, Cutler GB, Walston JD, et al. Designing randomized, controlled trials aimed at preventing or delaying functional decline and disability in frail, older persons: a consensus report. J Am Geriatr Soc. 2004; 52(4): 625–634.

336 Diehr P, Patrick DL, Spertus J, Kiefe CI, McDonell M, Fihn SD. Transforming self-rated health and the SF-36 scales to include death and improve interpretability. Med Care. 2001; 39(7): 670–80. 337 Berry SD, Ngo L, Samelson EJ, Kiel DP. Competing Risk of Death: An Important Consideration in Studies of Older Adults. Journal of the American Geriatrics Society. 2010; 58(4): 783–787.

338 Hernán MA, Hernández-Díaz S, Robins JM. A structural approach to selection bias. Epidemiology. 2004; 15(5): 615–25.

339 Kang H. The prevention and handling of the missing data. Korean J Anesthesiol. 2013; 64(5): 402–406.

340 Sterne JA, White IR, Carlin JB, Spratt M, Royston P, Kenward MG, et al. Multiple imputation for missing data in epidemiological and clinical research: potential and pitfalls. BMJ. 2009; 338: b2393

341 Little RJA, Rubin DB. Statistical analysis with missing data, Second Edition. New York: John Wiley and Sons; 2002

342 StataCorp LP. Stata multiple-imputation reference manual: release 13. College Station, TX: StataCorp LP; 2013.

343 Lee KJ, Carlin JB. Multiple imputation for missing data: fully conditional specification versus multivariate normal imputation. Am J Epidemiol. 2010; 171(5): 624–32.

344 Royston P, White IR. Multiple imputation by chained equations (MICE): implementation in Stata. J Stat Softw. 2011; 45(4): 1–20.

345 Schafer JL. Analysis of incomplete multivariate data. London: Chapman & Hall; 1997

346 van Buuren S, Boshuizen HC, Knook DL. Multiple imputation of missing blood pressure covariates in survival analysis. Stat Med. 1999; 18(6): 681–94.

347 White IR, Royston P. Imputing missing covariate values for the Cox model. Stat Med. 2009; 28(15): 1982–98.

348 Graham JW. Missing Data Analysis: Making it Work in the Real World. Annu Rev Psychol. 2009; 60: 549–576.

349 Rubin DB. Inference and missing data. Biometrika. 1976; 63(3): 581–92.

350 Collins LM, Schafer JL, Kam CM. A comparison of inclusive and restrictive strategies in modern missing data procedures. Psychol Methods. 2001; 6(4): 330–51.

351 Goldberg DP. Detecting Psychiatric Illness by Questionnaire. London: Oxford University Press; 1972.

352 Stansfeld SA, Marmot MG. Social class and minor psychiatric disorder in British civil servants: a validated screening survey using the General Health Questionnaire. Psychol Med. 1992; 22(3): 739–749.

353 Folstein MF, Folstein SE, McHugh PR. "Mini-mental state". A practical method for grading the cognitive state of patients for the clinician. J Psychiatr Res. 1975; 12(3): 189–98.

354 Tombaugh TN, McIntyre NJ. The mini-mental state examination: a comprehensive review. J Am Geriatr Soc. 1992; 40(9): 922–35.

355 Ahacic K, Kåreholt I, Helgason AR, Allebeck P. Non-response bias and hazardous alcohol use in relation to previous alcohol-related hospitalization: comparing survey responses with population data. Subst Abuse Treat Prev Policy. 2013; 8: 10.

356 Ewing JA. Detecting alcoholism. The CAGE questionnaire. JAMA. 1984; 252(14): 1905–7.

357 Mayfield D, McLeod G, Hall P. The CAGE questionnaire: validation of a new alcoholism screening instrument. Am J Psychiatry. 1974; 131(10): 1121–3.

358 Powers JR, Young AF. Longitudinal analysis of alcohol consumption and health of middleaged women in Australia. Addiction. 2008; 103(3): 424–32.

359 Hurcombe H, Bayley M, Goodman A. Ethnicity and alcohol: a review of the literature. York: Joseph Rowntree Foundation; 2010

360 Yaogo A, Fombonne E, Kouanda S, Lert F, Melchior M. Life course socioeconomic position and alcohol use in young adulthood: results from the French TEMPO cohort study. Alcohol Alcohol. 2014; 49(1): 109–16. 361 Sundquist J, Johansson SE. The influence of socioeconomic status, ethnicity and lifestyle on body mass index in a longitudinal study. Int J Epidemiol. 1998; 27(1): 57–63.

362 Ball K, Crawford D. Socioeconomic status and weight change in adults: a review. Soc Sci Med. 2005; 60(9): 1987–2010.

363 McLaren L. Socioeconomic status and obesity. Epidemiol Rev. 2007; 29: 29-48.

364 Kivimäki M, Batty GD, Singh-Manoux A, Nabi, H, Tabak AG, Akbaraly TN, et al. Association between common mental disorder and obesity over the adult life course. Br J Psychiatry. 2009; 195(2): 149–155.

365 Drøyvold WB, Midthjell K, Nilsen TI, Holmen J. Change in body mass index and its impact on blood pressure: a prospective population study. Int J Obes (Lond). 2005; 29(6): 650–5.

366 Denke MA, Sempos CT, Grundy SM. Excess body weight. An underrecognized contributor to high blood cholesterol levels in white American men. Arch Intern Med. 1993; 153(9): 1093–103.

367 Shamai L, Lurix E, Shen M, Novaro GM, Szomstein S, Rosenthal R, et al. Association of body mass index and lipid profiles: evaluation of a broad spectrum of body mass index patients including the morbidly obese. Obes Surg. 2011; 21(1): 42–7.

368 Ferrie JE, Shipley MJ, Davey Smith G, Stansfeld SA, Marmot MG. Change in health inequalities among British civil servants: the Whitehall II study. J Epidemiol Community Health. 2002; 56(12): 922–6.

369 Singh-Manoux A, Gimeno D, Kivimaki M, Brunner E, Marmot MG. Low HDL cholesterol is a risk factor for deficit and decline in memory in midlife: the Whitehall II study. Arterioscler Thromb Vasc Biol. 2008; 28(8): 1556–1562.

370 Wagenaar AF, Kompier MA, Houtman IL, van den Bossche SN, Taris TW. Employment contracts and health selection: unhealthy employees out and healthy employees in? J Occup Environ Med. 2012; 54(10): 1192–200.

371 Nazroo J, Kapadia D. Ethnic inequalities in labour market participation? Manchester: ESRC Centre on Dynamics of Ethnicity; 2013.

372 Catney G, Sabater A. Disadvantage in labour market: participation, skills and geographical inequalities. York: Joseph Rowntree Foundation; 2015.

373 Durstine JL, Grandjean PW, Cox CA, Thompson PD. Lipids, lipoproteins, and exercise. J Cardiopulm Rehabil. 2002; 22(6): 385–98.

374 Brown H, Roberts J. Exercising choice: The economic determinants of physical activity behaviour of an employed population. Soc Sci Med. 2011; 73(3): 383–390.

375 Tucker-Seeley RD, Subramanian SV, Li Y, Sorensen G. Neighborhood safety, socioeconomic status, and physical activity in older adults. Am J Prev Med. 2009; 37(3): 207–13.

376 Gidlow C, Johnston LH, crone D, Ellis NJ, James D. A systematic review of the relationship between socio-economic position and physical activity. Health Educ .J 2006; 65(4): 338–67.

377 Chen M, Wu Y, Narimatsu H, Li X, Wang C, Luo J, et al. Socioeconomic Status and Physical Activity in Chinese Adults: A Report from a Community-Based Survey in Jiaxing, China. PLoS One. 2015; 10(7): e0132918.

378 Buck D, Frosini F. Clustering of unhealthy behaviours over time: Implications for policy and practice. London: The Kings Fund; 2012.

379 Primatesta P, Falaschetti E, Gupta S, Marmot MG, Poulter NR. Association between smoking and blood pressure: evidence from the health survey for England. Hypertension. 2001; 37(2): 187–93.

380 He BM, Zhao SP, Peng ZY. Effects of cigarette smoking on HDL quantity and function: implications for atherosclerosis. J Cell Biochem. 2013; 114(11): 2431–6.

381 StataCorp LP. Stata power and sample size reference manual: release 13. College Station,TX: StataCorp LP; 2013.

382 Cox DR. Regression models and life-tables (with discussion). JR Stat Soc B. 1972; 34(2): 187–220.

383 Curran PJ, West SG, Finch JF. The Robustness of Test Statistics to Nonnormality and Specification Error in Confirmatory Factor Analysis. Psychological Methods. 1996; 1(1): 16–29.

384 Chou CP, Bentler PM, Satorra A. Scaled test statistics and robust standard errors for nonnormal data in covariance structure analysis: a Monte Carlo study. Br J Math Stat Psychol. 1991; 44(Pt 2): 347–57.

385 Hox JJ, Maas CJ, Brinkhuis MJ. The effect of estimation method and sample size in multilevel structural equation modeling. Stat Neerl. 2009; 64(2): 157–170.

386 Jones RH. Bayesian information criterion for longitudinal and clustered data. Stat Med. 2011; 30(25): 3050–6.

387 Williams R. Imputation of missing data in an unbalanced panel using ICE [Online forum comment]; 25 Oct 2013. Message posted to: http://www.stata.com/statalist/archive/2013-10/msg00916.html

388 Meng X, Rubin DB. Performing likelihood ratio tests with multiply-imputed data sets. Biometrika. 1992; 79(1): 103–111.

389 Statistical Consulting Group. milrtest [Internet]. UCLA, Institute for Digital Research and Education; 2008. Available from:

http://www.ats.ucla.edu/stat/stata/ado/analysis/milrtest.sthlp.htm

390 StataCorp LP. Stata survival analysis and epidemiological tables reference manual: release13. College Station, TX: StataCorp LP. Available from: www.stata.com/manuals13/st.pdf

391 Del Boca FK, Darkes J. The validity of self-reports of alcohol consumption: state of the science and challenges for research. Addiction. 2003; 98 Suppl 2: 1–12.

392 Stockwell T, Donath S, Cooper-Stanbury M, Chikritzhs T, Catalano P, Mateo C. Underreporting of alcohol consumption in household surveys: a comparison of quantity-frequency, graduated-frequency and recent recall. Addiction. 2004; 99(8): 1024–33.

393 Lemmens P, Tan ES, Knibbe RA. Measuring quantity and frequency of drinking in a general population survey: a comparison of five indices. J Stud Alcohol. 1992; 53(5): 476–86.

394 Stockwell T, Zhao J, Macdonald S. Who under-reports their alcohol consumption in telephone surveys and by how much? An application of the 'yesterday method' in a national Canadian substance use survey. Addiction 2014; 109(10): 1657–66.

395 Knott CS, Coombs N, Stamatakis E, Biddulph JP. All cause mortality and the case for age specific alcohol consumption guidelines: pooled analyses of up to 10 population based cohorts. BMJ. 2015; 350: h384.

396 Fone DL, Farewell DM, White J, Lyons RA, Dunstan FD. Socioeconomic patterning of excess alcohol consumption and binge drinking: a cross-sectional study of multilevel associations with neighbourhood deprivation. BMJ Open. 2013; 3(4): e002337.

397 Lewer D, Meier P, Beard E, Boniface S, Kaner E. Unravelling the alcohol harm paradox: a population-based study of social gradients across very heavy drinking thresholds. BMC Public Health. 2016; 16: 599.

398 Huang J, Wang X, Zhang Y. Specific Types of Alcoholic Beverage Consumption and Risk of Type 2 Diabetes: A Systematic Review and Meta-analysis. J Diabetes Investig. 2016. doi: 10.1111/jdi.12537. [Epub ahead of print]

399 Peluso MR, Miranda CL, Hobbs DJ, Proteau RR, Stevens JF. Xanthohumol and related prenylated flavonoids inhibit inflammatory cytokine production in LPS-activated THP-1 monocytes: structure-activity relationships and in silico binding to myeloid differentiation protein-2 (MD-2). Planta Med. 2010; 76(14): 1536–43.

400 Lupinacci E, Meijerink J, Vincken JP, Gabriele B, Gruppen H, Witkamp RF. Xanthohumol from hop (Humulus lupulus L.) is an efficient inhibitor of monocyte chemoattractant protein-1 and tumor necrosis factor-alpha release in LPS-stimulated RAW 264.7 mouse macrophages and U937 human monocytes. J Agric Food Chem. 2009; 57(16): 7274–81.

401 Kuh D, Ben-Shlomo Y, Lynch J, Hallqvist J, Power C. Life course epidemiology. J Epidemiol Community Health. 2003; 57(10): 778–83.

402 Holden SH, Barnett AH, Peters JR, Jenkins-Jones S, Poole CD, Morgan CL, et al. The incidence of type 2 diabetes in the United Kingdom from 1991 to 2010. Diabetes Obes Metab. 2013; 15(9): 844–52.

403 Park S, Lake ET. Multilevel Modeling of a Clustered Continuous Outcome: Nurses' Work Hours and Burnout. Nurs Res. 2005; 54(6): 406–413.

404 Færch K, Witte DR, Tabák AG, Perreault L, Herder C, Brunner EJ, et al. Trajectories of cardiometabolic risk factors before diagnosis of three subtypes of type 2 diabetes: a post-hoc analysis of the longitudinal Whitehall II cohort study. Lancet Diabetes Endocrinol. 2013; 1(1): 43–51.

405 Vistisen D, Witte DR, Tabák AG, Herder C, Brunner EJ, Kivimäki M, et al. Patterns of obesity development before the diagnosis of type 2 diabetes: the Whitehall II cohort study. PLoS Med. 2014; 11(2): e1001602.

406 Raftery AE. Bayesian Model Selection in Social Research. Sociological Methodology. 1995; 25(1): 111–163.

407 StataCorp LP. Stata multilevel mixed-effects reference manual: release 13. College Station, TX: StataCorp LP; 2013.

408 Britton A, Bell S. Reasons why people change their alcohol consumption in later life: findings from the Whitehall II Cohort Study. PLoS One. 2015; 10(3): e0119421. 409 Hvidtfeldt UA, Tolstrup JS, Jakobsen MU, Heitmann BL, Grønbaek M, O'Reilly E, et al. Alcohol intake and risk of coronary heart disease in younger, middle-aged, and older adults. Circulation. 2010; 121(14): 1589–97.

410 Gould AL, Boye ME, Crowther MJ, Ibrahim JG, Quatey G, Micallef S, et al. Joint modeling of survival and longitudinal non-survival data: current methods and issues. Report of the DIA Bayesian joint modeling working group. Stat Med. 2015; 34(14): 2181–2195.

411 Ibrahim JG, Chu H, Chen LM. Basic Concepts and Methods for Joint Models of Longitudinal and Survival Data. J Clin Oncol. 2010; 28(16): 2796–2801.

412 Sweeting MJ, Thompson SG. Joint modelling of longitudinal and time-to-event data with application to predicting abdominal aortic aneurysm growth and rupture. Biom J. 2011; 53(5): 750–763.

413 Prentice RL. Covariate Measurement Errors and Parameter Estimation in a Failure Time Regression Model. Biometrika. 1982; 69(2): 331–342.

414 Hutcheon JA, Chiolero A, Hanley JA. Random measurement error and regression dilution bias. BMJ. 2010; 340: c2289.

415 Crowther MJ, Abrams KR, Lambert PC. Joint modeling of longitudinal and survival data. The Stata Journal 2013; 13(1): 165–184.

416 Gärtner U, Schmier M, Bogusz M, Seitz HK. Blood alcohol concentrations after oral alcohol administration—effect of age and sex. Z Gastroenterol. 1996; 34(10): 675–9.

417 Lucey MR, Hill EM, Young JP, Demo-Dananberg L, Beresford TP. The influences of age and gender on blood ethanol concentrations in healthy humans. J Stud Alcohol. 1999; 60(1): 103–10.

418 Holdsworth C, Mendonça M, Pikhart H, Frisher M, de Oliveira C, Shelton N. Is regular drinking in later life an indicator of good health? Evidence from the English Longitudinal Study of Ageing. J Epidemiol Community Health. 2016. doi: 10.1136/jech-2015-206949. [Epub ahead of print]

419 Thiébaut AC, Bénichou J. Choice of time-scale in Cox's model analysis of epidemiologic cohort data: a simulation study. Stat Med. 2004 ;23(24): 3803–20

420 Korn EL, Graubard BI, Midthune D. Time-to-event analysis of longitudinal follow-up of a survey: choice of the time-scale. Am J Epidemiol. 1997; 145(1): 72–80.

421 Commenges D, Letenneur L, Joly P. Re: "Serum transferrin saturation, stroke incidence, and mortality in women and men. The NHANES I Epidemiologic Followup Study". Am J Epidemiol. 1997; 146(8): 683–4.

422 Rizopoulos D, Verbeke G, Molenberghs G. Shared parameter models under random effects misspecification. Biometrika. 2008; 95: 63–74

423 Cox DR. Partial likelihood. Biometrika. 1975; 62(2): 269–276.

424 Tsiatis AA, DeGruttola V, Wulfsohn MS. Modeling the Relationship of Survival to Longitudinal Data Measured with Error. Applications to Survival and CD4 Counts in Patients with AIDS. J Am Stat Assoc.1995; 90(429): 27–37

425 Asar Ö, Ritchie J, Kalra PA, Diggle PJ. Joint modelling of repeated measurement and timeto-event data: an introductory tutorial. Int J Epidemiol. 2015; 44(1): 334–44.

426 Huang X, Stefanski LA, Davidian M. Latent-model robustness in joint models for a primary endpoint and a longitudinal process. Biometrics. 2009; 65(3): 719–27.

427 Fournier MC, Foucher Y, Blanche P, Buron F, Giral M, Dantan E. A joint model for longitudinal and time-to-event data to better assess the specific role of donor and recipient factors on long-term kidney transplantation outcomes. Eur J Epidemiol. 2016; 31(5): 469–79.

428 Proust-Lima C, Séne M, Taylor JM, Jacqmin-Gadda H. Joint latent class models for longitudinal and time-to-event data: a review. Stat Methods Med Res. 2014; 23(1): 74–90.

429 Muthén B, Muthén LK. Integrating person-centered and variable-centered analyses: growth mixture modeling with latent trajectory classes. Alcohol Clin Exp Res. 2000; 24(6): 882– 91.

430 Bell S. The association between longitudinal trajectories of alcohol intake and risk of coronary heart disease, cardiovascular disease and all-cause mortality in Great Britain: evidence from 8 prospective cohort studies. Abstract presented at 41st KBS Meeting; Munich; 2015 Jun 1-5.

431 Bell S. Joint latent class models [Internet]. Message to: Knott C. 2015 Apr 23 [cited 2016 Jun 10].

432 Connor J. Alcohol consumption as a cause of cancer. Addiction 2016. doi: 10.1111/add.13477. [Epub ahead of print]

433 Brennan PL, Schutte KK, Moos RH. Retired status and older adults' 10-year drinking trajectories. J Stud Alcohol Drugs. 2010; 71(2): 165–8.

434 Iparraguirre J. Socioeconomic determinants of risk of harmful alcohol drinking among people aged 50 or over in England. BMJ Open. 2015; 5(7): e007684.

435 Holdsworth C, Frisher M, Mendonça M, de Oliveiria C, Pikhart H, Shelton N. Lifecourse transitions, gender and drinking in later life. Ageing and Society. doi: 10.1017/S0144686X15001178. [Epub ahead of print]

436 Frisher M, Mendonça M, Shelton N, Pikhart H, de Oliveira C, Holdsworth C. Is alcohol consumption in older adults associated with poor self-rated health? Cross-sectional and longitudinal analyses from the English Longitudinal Study of Ageing. BMC Public Health. 2015; 15: 703.

437 Bell S, Britton A. Protective effects of moderate alcohol consumption on fatty liver: a spurious association? J Hepatol. 2015; 62(5): 1209-11.

438 Bell S, Britton A. Reliability of a retrospective decade-based life-course alcohol consumption questionnaire administered in later life. Addiction. 2015; 110(10): 1563–73.

439 Leffingwell TR, Cooney NJ, Murphy JG, Luczak S, Rosen G, Dougherty DM, et al. Continuous Objective Monitoring of Alcohol Use: 21st Century Measurement using Transdermal Sensors. Alcohol Clin Exp Res. 2013; 37(1): 16–22.

440 Palmer LJ. UK Biobank: bank on it. Lancet. 2007; 369(9578): 1980-2.

441 Howard AA, Arnsten JH, Gourevitch MN. Effect of alcohol consumption on diabetes mellitus: a systematic review. Ann Intern Med. 2004; 140(3): 2119.

442 Rehm J, Baliunas D, Borges GLG, Graham K, Irving H, Kehoe T, et al. The relation between different dimensions of alcohol consumption and burden of disease: an overview. Addiction. 2010; 105(5): 817–43.

443 Ronksley PE, Brien SE, Turner BJ, Mukamal KJ, Ghali WA. Association of alcohol consumption with selected cardiovascular disease outcomes: a systematic review and meta-analysis. BMJ. 2011; 342: d671.

444 Movva R, Figueredo VM. Alcohol and the heart: to abstain or not to abstain? Int J Cardiol. 2013; 164(3): 267–76.

445 Rubin E. To drink or not to drink: that is the question. Alcohol Clin Exp Res. 2014; 38(12): 2889–92.

446 Fekjaer HO. Alcohol-a universal preventive agent? A critical analysis. Addiction. 2013; 108(12): 2051–7.

447 Gilmore W, Chikritzhs T, Stockwell T, Jernigan D, Naimi T, Gilmore I. Alcohol: taking a population perspective. Nat Rev Gastroenterol Hepatol. 2016; 13(7): 426–34.

448 Crome I, Dar K, Janikiewicz S, Rao T, Tarbuck A. Our invisible addicts: first report of the Older Persons' Substance Misuse Working Group of the Royal College of Psychiatrists. London: RCPsych; 2011.

449 Britton A. Do We Need Age-Specific Alcohol Consumption Guidelines? Alcohol Alcohol. 2012; 47(3): 203.